# NOTICE

THIS DOCUMENT HAS BEEN REPRODUCED FROM MICROFICHE. ALTHOUGH IT IS RECOGNIZED THAT CERTAIN PORTIONS ARE ILLEGIBLE, IT IS BEING RELEASED IN THE INTEREST OF MAKING AVAILABLE AS MUCH INFORMATION AS POSSIBLE

# NASA Contractor Report 159381

A FEASIBILITY STUDY FOR ADVANCED TECHNOLOGY INTEGRATION FOR GENERAL AVIATION

(NASA-CR-159381)A FEASIBILITY STUDY FORN81-15974ADVANCED TECHNOLOGY INTEGRATION FOR GENERALAVIATION Final Report, 10 Apr. 1979 - 9UnclasApr. 1980 (Kansas Univ.)533 pUnclasHC A24/MF A01CSCL 01B G3/01 41219

David L. Kohlman, Garey T. Matsuyama, Kevin E. Hawley, and Paul T. Meredith

UNIVERSITY OF KANSAS CENTER FOR RESEARCH, INC. Lawrence, Kansas 66045

1.

3

NASA Contract NAS1-15770 November 1980



Î

24

Langley Res. arch Center Hampton Virginia 23665

### PREFACE

This volume constitutes the final report for NASA Contract NAS-1-15770. This contract was funded by NASA Langley Research Center for the period of 10 April 1979 through 9 April 1980, although the project actually began in January 1979.

This project was used to fulfill part of the requirements for the Master of Engineering (ME) and Doctor of Engineering (DE) degrees for three graduate students at the University of Kansas. Major Garey T. Matsuyama, a DE candidate on leave from the United States Air Force Academy, served as project director. Assisting him were Kevin Hawley and Paul Meredith, ME candidates, who both made significant contributions to the project.

The objective of the ME-DE program at the University of Kansas is to provide students with an educational experience which includes graduate level vechnical and management courses, practical engineering experience through an internship program in industry or government, and a significant engineering project which provides direct experience with management problems, interpersonal relations, communication, and challenging technical problems.

We are grateful to the NASA Langley Research Center for supporting this project which not only provided the unique experiences required for the success of this educational program

i

tut also contribured to advancing the level of aeronautical technology.

١.

David L. Kohlman Principal Investigator Professor of Aerospace Engineering

# CONTENTS

.

.

																			page
Preface		•••	•	••	• •	•	•	•	•	•	•	•	•	•	•	•	•	•	i
List of H	figures .	• • •	•	••	• •	•	•	•	•	•	•	•	•	•	•	•	•	٠	xv
List of 7	Tables.	•••	•	•••	• •	•	•	•	•	•	•	•	•	•	•	•	•	•	xix
List of S	Symbols .	•••	•	••	• •	•	•	•	•		•	•	•	•	•	•		•	xxiii
List of A	Acronyma.		•			•	•	•	•	•	•	•	•	•	•	•	•	•	xxvii
1. SUMM	ARY	•••	•			•	•	•	•	•	•	•		•	•		•	•	1
2. INTRO	DUCTION					•	•		•	•	•	•	•	•	•	•		•	3
2.1	Backgrou	ınd .	•	•••		•	•	•		•	•		•			•	•	•	4
2 ?	Purpose		•			•	•	•							•	•			5
2.3	Scope.		•			•	•	•	•		•				•		•		6
2.4	Approach	1				•		•		•			•	•	•	•		•	7
2.5	Report I	Format																	7
3. VIST	IS TO MAR	WIFAC'	TIR:	ERS	ANT	RI	ESF	LAR	СН	ा	- 'AC	TT	.T.1	די	S		-	_	9
J								- •••	.~~							•	•	•	-
3.1	Introduc	tion	•	•••	• •	•	•	•	•	•	•	•	•	•	•	•	•	•	9
3.2	Manuiact and Meet	turers tings	3, Ì •	Rese • •	earo	h I	Fac	:11 •	it ,	ie	s,	•	•	•	•	•	•		9
	3.2.1	Manuf	fac	tur	ing	Fac	2 <b>i</b> ]	it	ie	s	•	•	•	•	•	•	•	•	10
	2 2 2	Chang		**	e <b>h</b> e	. м.		. f a	<b>~</b> +		- <b>-</b>								
	3.2.2	Visit	at	ion	Sch	ned	ule	11a ?.	•	•	•	•	•	•	•	•	•	•	10
	3.2.3	Profe	ess	iona	al M	lee	tir	ngs	•	•	•	•		•	•	•	•	•	10
	3.2.4	Resea	arc	h Fa	acil	lit:	ies	5	•	•	•	•	•	•	•	•		•	14
	3.2.5	Facil	lit	y Co	onta	acts	5.		•		•	•		•	•		•	•	16
3.3	Meetings	s.,			•		•	•	•	•	•		•	•		•	•	•	16
		D4		1			14		<b>c</b> .				~						
	1.2.1	Faci]	iss lit	ies	5 W1		ма •	.nu	a	•	.ur	•				•	•		16

			3.3.1.1	Airframe	16
			3.3.1.2	Propulsion	20
			3.3.1.3	Avionics	20
			3.3.1.4	Systems	21
		3.3.2	NASA Rese	arch Centers	22
			3.3.2.1	Ames Research Center Topics	22
			3.3.2.2	Langley Research Center Topics	22
			3.3.2.3	Lewis Research Center Topics .	23
	3.4	Results	of the Fa	cility Visits	23
		3.4.1	Technolog	y and the Industry	23
		3.4.2	Technolog	y and Cost	24
			3.4.2.1	Developmental Costs	24
			3.4.2.2	Certification Costs	25
			3.4.2.3	Product Liability Costs	26
	3.5	Recomme	ndations f	or NASA	27
		3.5.1	Communica	tions	27
		3.5.2	Basic Res	earch and Product Development .	28
		3.5.3	Research	Contracts	28
4.	TECH	NOLOGY E	VALUATION		31
	4.1	Method	of Technol	ogy Evaluation	31
		4.1.1	Evaluation	n Techniqu <b>e</b> .	32
		4.1.2	Category	Selection	33
		4.1.3	Category	Weightings	38
	4.2	Candida	te Technolo	ogy Identification	41
		4.2.1	Data Base		41
			4.2.1.1	Literature Search	41

		4.2.1.2	Visits and Meetings	44
	4.2.2	Prelimina	ry Candidate Technologies	45
	4.2.3	Final Can	didate Technologies	52
4.3	Applica	tion of the	e Evaluation Technique	57
	4.3.1	Mechaniza Technique	tion of the Evaluation	57
	4.3.2	Determina	tion of Relative Benefits	58
		4.3.2.1	Relative Benefit Scale	59
		4.3.2.2	Technology Baselines	59
		4.3.2.3	Technology Application and Integration	59
	4.3.3	Sensitivi	ty to Relative Benefits	61
		4.3.3.1	Consistency of Relative Benefits	62
		4.3.3.2	Uncertainty of Relative Benefits	64
4.4	Observa Evaluat	ations on t ion Techni	he Utilization of the que	64
	4.4.1	Qualitati	ve Values of Raters	65
	4.4.2	Stability	of Figures of Merit	65
	4.4.3	Orthogona	lity of Categories	66
	4.4.4	Effect of	Empty Weight	67
	4.4.5	Significa	nce of Figure of Merit Scores .	69
4.5	Evaluat	tion Techni	que Results	70
	4.5.1	Overall T	echnology Ranking	70
	4.5.2	Aerodynam	nics	70
		4.5.2.1	Winglets	75
		4.5.2.2	Spoilers	76

	4.5.2.3	Fowler Flaps	77
	4.5.2.4	Low-Drag Surface Coatings	78
	4.5.2.5	Active Laminar Flow Control	80
	4.5.2.6	Leading Edge Devices	81
	4.5.2.7	Advanced Low and Medium Speed Airfoils	82
	4.5.2.8	Advanced Natural Lamınar Flow Airfoils	87
	4.5.2.9	Improved Stall/Spin Character- istics Through Aerodynamic Tailoring	88
4.5.3	Aircraft	Systems	91
	4.5.3.1	Icephobic Surface Coatings	92
	4.5.3.2	AC Electrical Systems	93
4.5.4	Crashwort	chiness	93
	4.5.4.1	Load Limiting Seats	94
	4.5.4.2	Energy Absorbing Floor	96
	4.5.4.3	Improved Restraints	98
	4.5.4.4	Burst/Tear Resistant Fuel Tanks	9 <b>9</b>
4.5.5	Flight Co	ontrol Systems	103
	4.5.5.1	Fly-by-Wire	105
	4.5.5.2	Fly-by-Light	106
	4.5.5.3	Active Gust Alleviation	107
	4.5.5.4	Active Ride Smoothing	108
	4.5.5.5	Active Flutter Control	109
	4.5.5.6	Active Controls for Relaxed Inherent Stability	10 <b>9</b>
	4.5.5.7	Integrated Yaw Damper	110

ć

.

	4.5.5.8	Integrated Low-Cost Wing Leveler	111
	4.5.5.9	Separate Surface Technology	112
	4.5.5.10	Direct Side Force Control	113
	4.5.5.11	Direct Lift Control	114
	4.5.5.12	Single Lever Thrust/Drag Control	114
	4.5.5.13	Fluidic Automatic Flight Control System	115
	4.5.5.14	Active Stall Prevention	116
4.5.6	Informati	on Systems	117
	4.5.6.1	Dígital Data Links	117
	4.5.6.2	Cathode Ray Tube (CRT) Displays	120
	4.5.6.3	Heads Up Displays (HUD)	124
	4.5.6.4	Micro & UD	125
	4.5.6.5	Systems Status Displays	126
	4.5.6.6	Integrated Avionics and Displays	130
	4.5.6.7	Fiber Optics for Data Transmission	135
	4.5.6.8	Laser Gyros	135
4.5.7	Navigatio	n Concepts	136
	4.5.7.1	NAVSTAR/GPS	137
	4.5.7.2	Inertial Navigation Systems (INS)	138
	4.5.7.3	Doppler Navigation	139
	4.5.7.4	Microwave Landing Systems (MLS)	140
	4.5.7.5	Loran C	144
	4.5.7.6	Omega	145

 $\mu_{1},\dots,\mu_{n}\in (1,2)(2^n)$ 

	4.5.8	Noise		146
		4.5.8.1	Quiet, Efficient Propeller Technology	146
		4.5.8.2	Low-Level Pressurization	149
		4.5.8.3	Ducted Propulsors	149
	4.5.9	Propulsion	n	150
		4.5.9.1	GATE Engine	150
		4.5.9.2	Stratified Charge Rotary Combustion Engine	153
		4.5.9.3	High Compression Ratio Lean Burn (HCRLB) Engine	154
		4.5.9.4	Stratified Charge Reciprocating Engine	155
		4.5.9.5	Advanced Diesel Engine	158
		4.5.9.6	Liquid Cooling	159
		4.5.9.7	Improved Turbocharging	162
	4.5.10	Structura	l Materials	163
		4.5.10.1	Fiberglass Composites	163
		4.5.10.2	Kevlar Composites	164
		4.5.10.3	Graphite Composites	165
. 6	Other Pr	romising Te	echnologies	165
	4.6.1	Prop-Fan		166
	4.6.2	Quiet, Cle Turbofan	ean General Aviation (QCGAT)	167
. 7	Summary	of the Teo	chnology Evaluation	168
	4.7.1	Features o	of the Evaluation Technique	168
	4.7.2	Attractive	e Technologies	169
	4.7.3	Unattract:	ive Technologies	170

4

4

•••

5.	DESI	GN STUDI	ES	••••••	171
	5.1	Design	Specificat	ions	171
		5.1.1	6-Passeng	er Design Specifications	172
		5.1.2	Commuter	Design Specifications	172
	5.2	Technol	logy Integr	ation	173
		5.2.1	6-Passeng	er Technology Integration	175
		5.2.2	Commuter	Technology Integration	178
	5.3	Paramet	ric Trade	Studies	130
		5.3.1	6-Passeng	er Trade Studies	180
		5.3.2	Commuter	Trade Studies	184
	5.4	Concept	ual Design	15	192
		5.4.1	6-Passeng	er Designs	193
			5.4.1.1	Current Technology Baseline	195
			5.4.1.2	Advanced Technology Conventional Configuration	196
			5.4.1.3	Advanced Technology Canard Configuration	199
		5.4.2	Commuter	Designs	202
			5.4.2.1	Current Technology Baseline	206
			5.4.2.2	Conventional Configuration Advanced Technology Commuter .	209
			5.4.2.3	Canard Configuration Advanced Technology Commuter .	205
		5.4.3	Canard An	alysis Problems	214
	5.5	Design	Study Summ	ary	217
		5.5.1	GASP Comp	arison5	217
			5.5.1.1	6-Passenger Designs	217
			5.5.1.2	Commuter Designs	219

		5.5.2	Parametric Comparisons	222
			5.5.2.1 6-Passenger Designs	222
			5.5.2.2 Commuter Designs	226
6.	GENE	RAL AVIA	TION SYNTHESIS PROGRAM	231
	6.1	Program	Familiarity	231
		6.1.1	Canard Configurations	232
		6.1.2	Propulsion System Sizing	233
		6.1.3	Drag and Weight Calculations	233
	6.2	Program	Modifications	234
		6.2.1	Twin Engine Centerline Thrust	234
		6.2.2	Brake Specific Fuel Consumption (BSFC) .	234
		6.2.3	Commuter Mission Profile	235
	6.3	Program	Benchmarks	236
	6.4	Use of	GASP	238
	6.5	Results	of GASP Implementation	238
7.	RESU	LTS		241
	7.1	Overvie	w of Tasks Performed	241
	7.2	Results	of Tasks Performed	243
		7.2.1	Manufacturer and Research Center Visits.	243
		7.2.2	Technology Evaluation	244
			7.2.2.1 Attractive Technologies	244
			7.2.2.2 Unattractive Technologies	245
			7.2.2.3 Technologies Not Examined	245
		7.2.3	Design Studies	245
		7.2.4	GASP	246
		7.2.5	QVLM	246

8.	CONCI	LUSTONS A	AND RECON	MEN	ND.	TI	ON	13	•	•	•	•	•	•	•	•	•	•	•	247
	8.1	Conclusi	lons	•	•	•	•	•	•	•	•	•	•	•	•	٠	•	•	•	247
	8.2	Recommen	ndations	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	248
BIB	LIOGRA	PHY AND	REFERENC	CE I	LIS	ST	•	•	•	•	•	•	•	•	•	•	•	•	•	251
APP	ENDIX	<b>A</b>		•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	279
	A.1	AIRFRAM!	E MANUFAC	TUE	REF	۱S	•	•	•	•	•	•	•	•	•	•	٠	•	•	280
		A.1.1	Company	Α.	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	280
		A.1.2	Company	в.	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	284
		A.1.3	Company	c.	•	•	•	•	•	•	•	,	•	•	•	•	•	•	•	287
		A.1.4	Company	D.	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	290
		A.1.5	Company	E.	•	•	•	•	٠	•	•	•	•	•	•	•	٠	•	•	292
		A.1.6	Company	F.	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	297
		A.1./	Company	G.	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	299
		A.1.8	Company	н.	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	302
		A.1.9	Company	Ι.	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	303
		A.1.10	Company	J.	٠	•	•	•	•	•	•	•	•	•	•	•	•	٠	•	308
		A.1.11	Company	к.	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	308
	<b>A.</b> 2	PROPULS	ION MANUI	TAC 1	ruf	REF	s	•	•	•	•	•	•	•	•	•	•	•	•	309
		A.2.1	Company	L.		•	•	٠	•	٠	•	•	•	•	•	•	•	•	•	30 <b>9</b>
		A.2.2	Company	Μ.	•	•	•	•	•	•	•	•	•	•	•	•	•	•		311
		A.2.3	Company	N.	•	•	•	٠	•	•	•	•	•	•	•	•	•	•	•	313
		A.2.4	Company	0.	•	•	•	•	•	•	•	•	•	•	٠	•	•	٠	•	315
		A.2.5	Company	Ρ.	•	•	•	•	•	•	•	•	•	•	•	•	•	•	•	317
		A.2.6	Company	Q.	•	•	•	•	•	•	•		•	•	•	•	•	•	•	326
		A.2.7	Company	R.		•	•	•	•	•	•	•	•	•	•	•	•	•	s	329
		A.2.8	Company	s.							•						•			331

いたかと見ている。

	A.2.9	Compar	уT.	•	• •	•	•	•		•	•	•	٠	•	•	•	332
A.3	AVIONICS	S AND A	UTOP	ILO	ΤM	ANU	FA	сπ	JRE	RS	•	•	•	•	•	•	335
	A.3.1	Compar	y U.	•	••	•	•	•	• •	•	•	•	•	•	•	•	335
	A.3.2	Compar	yν.	•	•••	٠	•	•	••	•	•	•	•	•	•	•	337
	A.3.3	Compar	yW.	•	. •	•	•	•	•••	•	•	•	•	•	•	•	338
	A.3.4	Compar	yX.	•	•••	•	•	•	••	•	•	•	•	•	•	•	3 <b>39</b>
	A.3.5	Compar	yΥ.	•	•••	•	•	•	••	•	•	•	•	•	•	•	340
	A.3.6	Compar	yZ.	•		•	•	•	••	•	•	•	•	•	•	•	342
	A.3.7	Compar	y AA	••	••	•	•	•	• •	•	•	•	•	•	•	•	346
A.4	SYSTEMS	MANUFA	CTUR	ERS	•••	•	•	•		•	•	•	•	•	•	•	348
	A.4.1	Compar	y BB	•	•••	•	•	•		•	•	•	•	•	•	•	348
	A.4.2	Compar	y CC	•	•••	•	•	-	• •	•	•	•	•	•	•	•	350
	A.4.3	Compar	y DD	•	•••	•	•	•	••	•	•	•	•	•	•	•	352
	A.4.4	Compar	y EE	•	• •	•	•	•		•	•	•	•	•	•	•	352
A. 5	NASA RES	SEARCH	CENT	ERS	•••	•	•	•	••	•	•	•	•	•	•	•	354
	A.5.1	Ames H	lesca	rch	Ce	nte	r	•	• •	•	•	•	•	•	•	•	354
	A.5.2	Langle	y Re	sea	rch	Ce	ent	er	• •	•	•	•	•	•	•	•	362
	A.5.3	Lewis	Rese	arc	h C	ent	er	•		•	•	•	•	•	•	•	369
APPENDIX	B	• • •			• •	•	•	•	• •	•	•	•	•	•	•	•	393
<b>B.1</b>	Category	/ Selec	tion	Su	rve	у.	•	•	• •	•	•	•	•	•	•	•	394
в.2	Category	7 Ratin	ig Su	rve	y.	•	•	•	•••	•	•	•	•	•	•	•	401
B.3	Candidat	te Tech	nolo	89	Ide	nti	if <b>i</b>	cat	t i o	n.	•	•	•	•	•	•	423
В.4	Pessimi: (PLOE)	stic, l Figure	ikel of N	y, I Ieri	Opt t S	imi tud	ist lie	ic. s	, a 	nd	Ex •	Pe	ect •	ed •	•		432
B.5	Tabular With An Weight	Result Examin	s Of atio	Th n O	eT fT	ect he	nno Ef	log fec	gy ct	Eva Of	ilu Em	iat ipt	io y	n			458

	B.5.1	Average Category Weights	459
	B.5.2	Maximum Category Weight for Empty Weight	459
	B.5.3	Minimum Category Weight for Empty Weight	460
APPENDIX	c		483
<b>C.</b> 1	6-Passe	nger Designɛ	483
	C.1.1	Trade Studies	483
	C.1.2	Current Technology Design - 6PAXBL	485
	C.1.3	Advanced Technology Conventional Design - 6PAXAD	488
	C.1.4	Advanced Technology Canard Design - 6PAXC	489
	C.1.5	Tail Sizing Computer Program	489
C.2	Commute	r Design	492
	C.2.1	Trade Studies	492
	C.2.2	Current Technology Configuration - COMBL	494
	C.2.3	Advanced Technology Conventional Configuration - ADCOM	494
	C.2.4	Advanced Technology Canard Configuration - ADCOMCN	494
<b>C.</b> 3	Range F	actor Parametric Analysis	498

xiii

### LIST OF FIGURES

Figure		page
3.1	Locations of Facilities and Meetings	11
4.1	Illustration of the Evaluation Technique	34
4.2	Sample of the Pessimistic, Likely, Optimistic, and Expected Figures of Merit	63
4.3	c <sub>l</sub> vs t/c of NASA Low-Speed and NACA Airfoils	83
4.4	Effect of t/c on $c_{l_{max}}$ and $l/d$ for NASA Low-Speed	
	Airfoils	84
4.5	Design Family of NASA Low-Speed Airfoils	85
4.6	Modification to NASA 63 <sub>2</sub> -215 Airfoil	85
4.7	Experimental Results of a Modification to an NACA 63 <sub>2</sub> -215 Airfoil	86
4.8	Experimental Results of Low and Medium Speed Airfoils	87
4.9	Leading Edge Modification for Stall/Spin Alleviation	<b>9</b> 0
4.10	Flat-Topped Lift Curve From Leading Edge Modification for Stall/Spin Alleviation	91
4.11	Load-Limiting Seat Concepts	95
4.12	Energy Absorbing Floor Concepts	97
4.13	Five-Point Improved Restraint Vs Conventional Restraint	99
4.14	Modified Fuel Tank Installed in Wing	100
4.15	Airframe On Catapult With Earthen Hill, Steel Tubes, and Rock Piles Visible	101

.

ï١

4.16	Typical Impact of Airframe Following a Catapult Acceleration	102
4.17	Wing Damage Where a 1-Ply Tank Did Not Leak	103
4.18	Preliminary Fiber Optic Symbol Generator (Operational)	127
4.19	Micro HUD With Fiber Optic Link and Mirror	128
4.20	Micro Hud Image Display	129
4.21	Comparison of Pitch, Heading, and Roll Computations	132
4.22	Typical ILS Implementation	141
4.23	Typical MLS Implementation	142
5.1	6-Passenger Cruise Trade Studies (at 20000 ft)	181
5.2	6-Passenger Cruise Trade Studies (at 30000 ft)	182
5.3	6-Passenger Takeoff Trade Studies	183
5.4	6-Passenger Landing Trade Studies	185
5.5	Commuter Cruise Trade Studies	187
5.6	Commuter Balanced Field Length Trade Studies	188
5.7	Commuter Balanced Field Length Trade Studies	189
5.8	Commuter Balanced Field Length Trade Studies	190
5.9	6-Passenger Cabin Layout	194
5.10	Current Technology 6-Passenger Design	197
5.11	Advanced Technology Conventional Configuration 6-Passenger Design	200
5.12	Advanced Technology Canard Configuration 6-Passenger Design	203
5.13	Commuter Cabin Layout	205
5.14	Commuter Baseline Configuration	207
5.15	Advanced Technology Commuter	210

5.16	Advanced Technology Commuter Canard Configuration	212
5.17	QVLN Validation Configuration	215
5.18	Comparison of Experimental and QVIM Results for the Validation Configuration	216
5.19	6-Passenger Range Factor Comparisons	225
5.20	Commuter Range Factor Comparisons	229
B.1	Results of Round 3 of the Category Rating Survey.	403

.

Ĵ

-

- **- -**

	would be would be the buckbory watting burkbyt t	
<b>B.2</b>	Mean Category Rating Trends	411
<b>B.</b> 3	Airplane A PLOE Figure of Merit Comparisons	448
<b>B.4</b>	Airplane B PLOE Figure of Merit Comparisons	453

# LIST OF TABLES

.

N.4. 1.1

•

.

Table		page
3.1	General Aviation Manufacturing Facilities Visited	12
3.2	Professional Meetings	15
3.3	NASA Research Centers	15
3.4	Facility Contacts	17
4.1	Technology Evaluation Categories	36
4.2	Category Weighting Survey Participants	39
4.3	Evaluation Technique Category Weightings	42
4.4	Evaluation Technique Category Rankings	43
4.5	Computerized Literature Search Results	45
4.6	Preliminary Candidate Technologies	46
4.7	Final Candidate Technologies	54
4.8	Technology Baselines	60
4.9	Six Factors for Airplanes A and B	63
4.10	Overall Technology Ranking for Airplane A	71
4.11	Overall Technology Ranking for Airplane B	73
4.12	Aerodynamics	75
4.13	Aircraft Systems	92
4.14	Crashworthiness	94
4.15	Flight Control Systems	104
4.16	Information Systems	118

xix

PRECEDING PAGE BLANK NOT FILMED

4.17	1975 Characteristics of Some Electronic Displays.	121
4.18	Electronic Display Technology Perspective	122
4.19	Six Major Display Problem Areas and Their Impact on New Technologies	123
4.20	Navigation Concepts	136
4.21	Noise	146
4.22	Propulsion	151
4.23	Structural Materials	163
5.1	6-Passenger Design Specifications	173
5.2	Commuter Design Specifications	174
5.3	6-Passenger Advanced Technologies	176
5.4	Commuter Advanced Technologies	179
5.5	6-Passenger Trade Study Summary	186
5.6	Commuter Trade Study Summary	191
5.7	6-Passenger Current Technology Baseline Characteristics	198
5.8	6-Advanced Technology Conventional Configuration Characteristics	201
5.9	Commuter Current Technology Baseline Characteristics	208
5.10	Advanced Technology Conventional Configuration Commuter Characteristics	211
5.11	6-Passenger Comparison by GASP	218
5.12	Commuter Comparison by GASP	220
5.13	Comparison of Baseline and Advanced Technology Commuters for the Multi-Mission	221
5.14	Data Used for 6-Passenger Range Factor Comparisons	224

5.15	Data Used for Commuter Range Factor Comparisons	228
B.1	Round 1 Average Category List	395
B.2	Round 2 Average Category List	396
B.3	Round 3 Average Category List	397
B.4	Round 4 Average Category List	398
B.5	Final Category List Used in the Evaluation Technique	399
B.6	Category Rating Survey Stability	421
B.7	Search of Aircraft Design, Aircraft Propulsion, and Aircraft Structures	424
B.8	Search of Flight Controls, Navigation Aids, Avionics, Canard Configurations, and NAVSTAR/GPS.	425
B.9	Pessimistic Technology Evaluation	433
<b>B.10</b>	Likely Technology Evaluation	438
B.11	Optimistic Technology Evaluation	443
B.12	Technology Ranking By Technology Group for Airplane A	461
B.13	Overall Technology Ranking for Airplane A	464
<b>B.1</b> 4	Technology Ranking by Technology Group for Airplane B	466
B.15	Overall Technology Ranking for Airplane B	469
<b>B.1</b> 6	Overall Technology Ranking With Empty Weight = 9.5 for Airplane A	471
B.17	Overall Technology Ranking With Empty Weight = 9.5 for Airplane B	474
B.18	Overall Technology Ranking With Empty Weight = 0 for Airplane A	476
<b>B.</b> 19	Overall Technology Ranking With Empty Weight = 0 for Airplane B	479

C.1	6PAXBL Weight Breakdown	•	•	487
C.2	6PAXBL Drag Breakdown	•	•	488
C.3	6PAXAD Weight Breakdown	•	•	<b>49</b> 0
C.4	6PAXAD Drag Breakdown	•	•	491
C.5	COMEL Weight Breakdown	•	•	495
C.6	COMBL Drag Breakdown	•	•	496
C.7	ADCOM Weight Breakdown	•	•	497
C.8	ADCOM Drag Breakdown	•		498

# LIST OF SYMBOLS

Symbol	Definition	Dimensions <u>SI (Britis</u> h)
A or AR	wing aspect ratio	
Ae	effective aspect ratio (A)(e)	
b or b <sub>w</sub>	wing span	m (ft)
₽ <sub>f</sub> /₽	flap span to wing span ratio	
BFL	balanced field length	m (ft)
<sup>b</sup> ij	relative benefit of technology i in category j	
BSFC	brake specific fuel consumption	kg/kW/hr (1b/hp/hr)
c	wing chord	m (ft)
c <sub>d</sub>	2D drag coefficient	
с <sub>р</sub>	3D drag coefficient	
с <sub>р</sub>	zero lift drag coefficient	
c <sub>f</sub> /c	flap chord to wing chord ratio	
cl	2D lift coefficient	<b></b>
$c_{\ell_{\beta}}$	variation of rolling moment coefficient due to sideslip angle	rad <sup>-1</sup> (deg <sup>-1</sup> )
с <sub>г</sub>	3D lift coefficient	
CL <sub>i</sub>	confidence level of survey participant i	
c <sub>m</sub> c/4	section pitching moment coefficient about the quarter chord	
с <sub>то</sub>	section pitching moment coefficient at zero angle of attack	

с <sub>щ</sub>	variation of pitching moment coefficient with angle of attack	rad <sup>-1</sup> (deg <sup>-1</sup> )
°n <sub>r</sub>	variation of yawing moment coefficient with yaw rate	rad <sup>-1</sup> (deg <sup>-1</sup> )
CR	compression ratio	
dC <sub>m</sub> /dC <sub>L</sub>	change in pitching moment coefficient with respect to lift coefficient	
Dn	propeller diameter x rotation speed	m/s (ft/sec)
e	Airplane efficiency factor	****
f/a or F/A	fuel to air ratio	
FM <sub>1</sub>	figure of merit of technology i	
h	altitude	m (ft)
L/D	lift-to-drag ratio	
М	Mach number	
n	number of survey participants	
P/W	power to weight ratio (usually P /W max gross)	kW/N (hp/1b)
- P	dynamic pressure	$\frac{N/m^2}{(1b/ft^2)}$
R	range	km (nm)
R <sub>j</sub>	mean rating of category j	
R <sub>c</sub> j	corrected mean rating of category j	
Re	Reynold's number	
RF	range factor	kg/km/1 (1b/nm/gal)
R <sub>ij</sub>	rating given by participant i to category j	

t

S	wing area	$m^2$ (ft <sup>2</sup> )
sfc or SFC	specific fuel consumption	kg/kW/hr (lb/hp/hr)
t/c	airfoil section thickness-to- chord ratio	
<sup>T</sup> ijk	voting indicator of participant i in category j for rating k	
TNC	number of participants who changed their rating of category i between two successive survey rounds	
TPC <sub>1</sub>	total percent change in the number of participancs changing their ratings of category i between two successive survey rounds	
TSFC	thrust specific fuel consumption	N/kg/hr (1bf/1bm/hr)
V	velocity	m/s (ft/sec)
V <sub>jk</sub>	number of votes in category j for rating k	
w_j	weighting of category j	
W	weight	kg (1b)
₩ <sub>E</sub> /₩ <sub>G</sub>	empty to gross weight ratio	
W/S	wing loading	$\frac{N/m^2}{(lb/ft^2)}$

# Greek Symbols:

t.

i 1

ł

α	angle of attack	rad (deg)
β	sideslip angle	rad (deg)
Δ	increment (used as a prefix)	
δ <sub>e</sub>	elevator deflection	rad (deg)
η p	propeller efficiency	

ካ <b>₽/₩</b> ₽	effective power loading (usually that required at a specific flight condition, as opposed to the maximum)	kW/N (hp/1b)
μ	runway friction coefficient	<b></b>
ρ	air density	kg/m <sup>3</sup> (slug/ft <sup>3</sup> )

# LIST OF ACRONYMS

-

.

Acronym	Definition
ACARS	Automatic Communication and Reporting System
ACEE	Aircraft Energy Efficiency program
ADCOM	Advanced Technology Commuter, Conventional Configuration
ADCOMCN	Advanced Technology Commuter, Canard Configuration
ADF	Automatic Direction Finder
ADI	Attitude Direction Indicator
AFCS	Automatic Flight Control System
AGL	Above Ground Level
AHRS	Attitude Heading Reference System
ALT	Altitude
ARA	Advanced Research Airplane (Princeton)
ARAD or ARA-D	Advanced propeller sections developed by Aircraft Research Association, Ltd. (ARA) and funded by Dowty Rotol, Ltd.
ARINC	Aeronautical Radio, Inc.
ASEE	American Society for Engineering Education
ATC	Air Traffic Control
ATIS	Automatic Terminal Information Service
ATLIT	Advanced Technology Light Twin research airplane (addified Piper Seneca) tested by NASA Langley, Piper, and the University of Kansas, Flight Research Laboratory.

xxvii

A/C	Aircraft
BDC	Bottom Dead Center
BFL	Balanced Field Length
BHP	Brake Horsepower
BMDP	Biomedical Computer Programs
BMEP	Brake Mean Effective Pressure
BSFC	Brake Specific Fuel Consumption
CADCAM	Computer Aided Design/Computer Aided Manufacturing
CAMI	Civil Aeromedical Institute (FAA-Oklahoma City)
CAS	Collision Avcidance System
CCV	Control Configured Vehicle
CEP	Circular Error Probability
СНТ	Cylinder Head Temperature
CL	Confidence Level
COMBL	Current technology Commuter Baseline configuration
CONUS	Continental United States
СРМ	Critical Path Method
CR	Compression Ratio
CRT	Cathode Ray Tube
СТ	Candidate Technology
CVCC	Compound Vortex Controlled Combustion system (Honda stratified charge system)
DAAS	Demonstrator Advanced Avionics System
DABS	Discrete Address Beacon System
DARE	Digital Avionics Research system (NASA~Langley)
DDBS	Digital Data Broadcast System

.

٠

•

DLC	Direct	Lift	Control	

DME Distance Measuring Equipment

DOC Direct Operating Cost

DOT Department of Transportation

EET Energy Efficient Transport (element of the ACEE program)

EGT Exhaust Gas Temperature

EHSI Electronic Horizontal Situation Indicator

EMI Electromagnetic Interference

EPA Environmental Protection Agency

ESHP Equivalent Shaft Horsepower

ET Evaluation Technique

FAA Federal Aviation Administration

FAR Federal Aviation Regulations

FBL Fly-by-Light

•,

FBW Fly-by-Wire

FCT Final Candidate Technologies

FM Figure of Merit

FOD Foreign Object Damage

GA or G.A. General Aviation

GALIA General Aviation Manufacturer's Association

GAP General Aviation Propeller program

GASP General Aviation Synthesis Program

GATE General Aviation Turbine Engine

GA(W) General Aviation Whitcomb airfoil series

GPS	Global Positioning System
HCRLB	High Compression Ratio/Lean Burn reciprocating engine
HMD	Head-Mounted Display
HSI	Horizontal Situation Indicator
HUD	Heads-Up Display
IDCC	Integrated Data Control Center
IDRF	Impact Dynamics Research Facility (NASA-Langley)
IEEE	Institute of Electrical and Electronics Engineers
IFR	Instrument Flight Rules
ILS	Instrument Landing System
INS	Inertial Navigation System
IPS	Intermittent Positive Control
ISA	International Civil Aeronautical Organization (ICAO) Standard Atmosphere
LaRC	Langley Research Center (NASA)
LC	Liquid Crystal
LCC	Life Cycle Cost
LED	Light Emitting Diode
LFC	Laminar Flow Control
LS	Low-Speed airfoil series (NASA)
MLS	Microwave Landing System
MSL	Mean Sea Level
MSOMLA	Modified Seat-Occupant Model for Light Aircraft
MTBF	Mean Time Between Failures
NACA	National Advisory Committee for Aeronautics

xxx

NASA	National Aeronautics and Space Administration
NASTRAN	NASA Structural Analysis computer code
NAVSTAR	Navigation System using Time And Ranging
Nicad	Nickel/Cadmium battery
NLF	Natural Laminar Flow
NLFC	Natural Laminar Flow Control
NTIS	National Technical Information Service
OEM	Original Equipment Manufacturer
osu	Ohio State University
PCAAS	Preliminary Candidate Advanced Avionics System
PCT	Preliminary Candidate Technology
PERT	Program Evaluation and Review Technique
PLOE	Pessimistic-Likely-Optimistic-Expected (scenarios to examine the Evaluation Technique)
POL	Problem-Oriented Language computer codes
PROCO	Programmed Combustion (Ford stratified charge system)
QCGAT	Quiet, Clean, General Aviation Turbofan
QVLM	Quasi-Vortex Lattice Method
RC	Rotary Combustion
RF	Range Factor
Redhawk	Langley Research Center/University of Kansas, Flight Research Laboratory research airplane (modified Cessna Cardinal)
RSR	Rapid Solidification Rate
R/C	Radio Controlled ,
R & D	Research and Development

SAS	Stability Augmentation System
SCAN	Problem-oriented language computer code developed at the University of Illinois.
SCAN (NASA)	NASA Selected Current Aerospace Notices. Index published twice each month.
SCAS	Stability and Control Augmentation System
SFC or sfc	Specific Fuel Consumption
SHP or shp	Shaft Horsepower
SIU	Southern 111inois University
SSSA	Separate Surface Stability Augmentation system
STAR (NASA)	NASA Scientific and Technical Aerospace Reports. Index published twice each month.
STAT	Small Transport Aircraft Technology
STC	Supplemental Type Certificate
STI	Systems Technology, Inc.
тво	Time Between Overhaul
TCCS	Texaco Controlled Combustion System (Texaco stratified charge system)
TCHLST	Computer program written at the University of Kansas, Flight Research Center which incorporates the Evaluation Technique.
TDC	Top Dead Center
TIT	Turbine Inlet Temperature
TSFC	Thrust Specific Fuel Consumption
TSO	Technical Standard Order
UG3RD	Upgraded 3rd Generation Air Traffic Control system
VFR	Visual Flight Rules
VHF	Very High Frequency
VLF	Very Low Frequency

xxxii

VLM	Vortex Lattice Method
VOR	Very High Frequency Omni-Directional Range
VORTAC	VOR and Tactical Air Navigation System
V/STOL	Vertical/Short Takeoff and Landing

•

### CHAPTER 1

#### SUMMARY

A study directed toward the identification and evaluation of applicable advanced technologies for general aviation was performed. An extensive data base was generated through visits to 31 general aviation manufacturers and 3 NASA research centers as well as through an exhaustive literature search. An evaluation technique was developed which allowed candidate technologies to be ranked according to potential benefit. Finally, design studies were performed for a 6-passenger personal/business airplane and a 19-passenger commuter airplane. The General Aviation Synthesis Program (GASP) was utilized during the design studies for propulsion system and vehicle sizing as well as mission performance analysis.

In assembling the data base, extensive notes which were acquired from the visits were edited and are included in the report as an appendix. 56 of the 137 technologies initially identified were evaluated and are discussed separately in the report.

The results of the technology evaluation indicated that propulsion, aerodynamic, and composite technologies are extremely attractive to general aviation. When these technologies were incorporated into the design synthesis of the two airplanes, higher wing loadings and smaller airplanes resulted. Fuel savings of 50% for the 6-passenger airplane and 40% for the commuter were realized.

### CHAPTER 2

### INTRODUCTION

General aviation represents 96% of the civilian pilot force flying 99% of the civil aircraft and 84% of the total flight hours in the United States. During 1979 alone, factory billings for aircraft shipments amounted to \$2.2 billion. However, propeller technology is largely of World War II vintage, reciprocating propulsion systems are cooled by excessively rich mixtures during climb, autopilot functions are fed back to the pilot as distracting control movements, and construction is typified by conventional aluminum sheet-stringer structure with protruding rivet heads in a large number of current aircraft.

This apparent conflict, where the fleet size is large and heavily utilized, the industry is enjoying record sales, while the equipment appears outmoded, points to some unique characteristics of the general aviation environment. Specifically, (1) the users (particularly single-engine and commuter users) are much more sensitive to purchase price than their heavy-jet, commercial airline counterparts, (2) manufacturers are reluctant to incur increased production costs through product improvement for an apparently already satisfied market, and (3) the capital intensive nature of the industry may easily spell financial disaster for the manufacturer who misjudges the product improvement expectations of the user.

However, the fuel price and availability problems which began in 1973, together with strong public reaction to noise and emissions pollution, have resulted in the acceleration in development of several new and promising technologies.

### 2.1 BACKGROUND

The recent development of many different technologies with potential application to the general aviation fleet of aircraft has reached a point where radically improved airplanes can now be foreseen. However, to realize these improvements applicable technologies must be identified and actively pursued in an orderly and timely manner. Also, synergistic effects resulting from the integration of appropriate technologies require identification in order to better define research.

To illustrate this latter point, consider the development of an advanced natural laminar flow airfoil. Thorough evaluation of this technology would appear to dictate eventual full scale tunnel tests and flight tests. However, since surface roughness can impose severe penalties on the performance of this airfoil, a composite or bonded wing may be called for in order to eliminate the problem of rivet lines and butted skin joints. Here, a decision to pursue natural laminar airfoils through conventional manufacturing techniques may lead to an erroneous evaluation, while a decision to investigate a composite wing may be delayed if its potential benefit is seen only as an improvement in empty weight (when a conventional airfoil is utilized).
The key issues facing both research institutions (NASA Research Centers) and a capital intensive industry when confronted with the question of which technologies to pursue appear to lie in how to identify those technologies which offer great potential for improving safety, performance, and cost as well as how to identify those with questionable benefits. An attendant result of such an evaluation appears to be the identification of those technologies with noteworthy (as opposed to highly significant) benefits which might be attained for a rather low level of development effort.

Significant pioneering work done in this area of technology evaluation was recently completed by Bergey (Ref. 18). The present research represents an effort to continue with this type of work on a much broader scale and in much greater depth.

2.2 PURPOSE

In light of the above discussion, the purpose of this research is threefold:

- Identify candidate technologies which appear to offer improvements in safety, fuel efficiency, performance, and utility of general aviation airplanes.
- (2) Quantify the magnitude of these improvements.
- (3) Investigate the synergistic effects of advanced technology integration on general aviation airplanes.

2.3 SCOPE

Recognizing the different requirements of different types of airplanes within general aviation, this research effort was directed to an investigation of the impact of new technologies on a small airplane with a 6-passenger cabin (including pilots) and a larger airplane with at least a 12-passenger cabin (excluding pilots). Performance guidelines for these two airplanes were very broad. General guidelines adopted for this study included high maximum lift-to-drag ratios on the order of 18, cruise speeds on the order of 250 knots, and landing speeds below 60 knots.

Those tasks which were specified as a part of this study included the following:

- (1) Representative manufacturers within the general aviation industry and certain NASA Research Centers were visited in order to integrate the views of government agencies and industry concerning new technologies.
- (2) Promising new technologies were identified, and their impacts on two different airplanes were evaluated.
- (3) Trade studies for the two specified airplanes configured conventionally with aft tails and also as canards were performed (although the canard studies met with difficulties).
- (4) The General Aviation Synthesis Program (GASP) was utilized in the evaluation of those technologies which affect vehicle weight and performance.

## 2.4 APPROACH

The approach utilized to accomplish the required tasks leading to successful attainment of the goal of this research may be broadly categorized as follows:

(1) Identify and develop a data base.

- (a) Conduct a literature search.
- (b) Acquire, review, and tabulate pertinent documents.
- (c) Visit manufacturers and NASA research centers.

(2) Identify, develop, and test an evaluation technique.

(3) Identify and evaluate technologies.

- (4) GASP
  - (a) Gain familiarity with the program.
  - (b) Modify as required to evaluate technologies.
  - (c) Benchmark against current technologies.
  - (d) Use to size configurations.

(5) Design and evaluate two advanced technology airplanes.

## 2.5 REPORT FORMAT

Each of the four major tasks listed previously in Section 2.3 is inscussed in a separate chapter within this report, and an associated appendix is included where supporting documentation for three chapters is provided. Here Appendix A is provided for Chapter 3 (visits), Appendix B is provided for Chapter 4 (technoevaluation), and Appendix C is provided for Chapter 5 (designs). GASP modifications are discussed in Chapter 6.

Finally, Chapter 7 summarizes the results of the research effort and Chapter 8 closes with conclusions and recommendations.

## CHAPTER 3

## VISITS TO MANUFACTURERS

## AND RESEARCH FACILITIES

## 3.1 INTRODUCTION

Visits to manufacturers and NASA research facilities formed one of three major thrusts of the present research. The purpose of this particular effort was to interview representative industry and research facilities with the goal of developing the information base and contacts required to support the project. The 34 facilities which were visited provided a wealth of information and, as a spinoff, directed attention to many other sources of pertinent information.

This chapter presents an overview of the planning, initiation, proceedings, and results of the visitation phase. Appendix A is included as a supplement, where notes assembled by the research team during each visit have been compiled, edited, and are presented in abbreviated form.

## 3.2 MANUFACTURERS, RESEARCH FACILITIES, AND MEETINCS

The work statement for this research effort listed 28 manufacturers and research facilities to be visited. As the project progressed, the list was modified to accomodate changes in order to provide the required data base for the ensuing technology evaluation. In order to support the final visitation list without exceeding budget constraints, different modes of transportation

were utilized. Figure 3.1 illustrates the scope of the visitation phase by presenting the geographical location of the facilities involved in this research project.

## 3.2.1 Manufacturing Facilities

Table 3.1 provides a listing of those manufacturers visited.

١.

## 3.2.2 Changes to the Manufacturer Visitation Schedule

These changes represent the addition of five facilities to those already planned for. The following is a summary of the changes and the rationale for including them.

- (1) Bellanca Aircraft Engineering. This firm was responsible for the development of the Bellanca Skyrocket, a high performance single engine aircraft. The inclusion of this firm was believed to be important because of their achievements in low drag airframes and composite construction techniques.
- (2) Hartzell, Hamilton Standard, and McCauley. An examination of the initial list of facilities detected deficiencies in the propulsion area. While engine manufacturers were included, representatives of the propeller manufacturers were not. These three companies were added to the facility list to broaden the propulsion technology base.
- (3) Curtiss-Wright. This facility was added because of their work with rotary combustion engines.

## 3.2.3 Professional Meetings

These meetings proved valuable to the research efforts of the project and provided significant insight to technology



Fig. 3.1. Locations of Facilities and Meetings

.

Type Facility	Company	Location
Airframe	Beech Aircraft Corporation	Wichitz, Kansas
	Bellanca Aircraft Corporation	Alexandria, Minnesota
	Bellanca Aircraft Engineering, Inc.	Middletown, Delaware
	Gessna Aircraft Corporation	
	Pawnee Division	Wichita, Kansas
	Wallace Division	Wichita, Kansas
	Gates Learjet Corporation	Wichita, Kansas
	Gulfstream American Corporation	Savannah, Georgia
	Mooney Aircraft Corporation	Kerrville, Texas
	Piper Aircraft Corporation	Lakeland, Florida
	Rockwell International	
	General Aviation Division	Bethany, Oklahoma
	Rutan Aircraft Factory	Mojave, California
Avionics and Autonilots	Brittain Industries, Inc.	Tulsa, Oklahoma
	Cessna Aircraft Corporation	
	Aircraft Radio & Control Division (ARC)	Boonton, New Jersey

# Table 3.1. General Aviation Manufacturing Facilities Visited

Avionics	Edo Aire/Mitchell	Mineral Wells, Texas
Autopilots (cont)	King Radio Corporation	Olathe, Kansas
	Narco Avionics	Fort Washington, Pennsylvania
	Rockwell International	
	Avionics and Missile Group (Collins Avionics)	Cedar Rapids, Iowa
	Sperry Flight Systems	
	Avionics Division	Phoenix, Arizona
Systems	Bell Helicopter Textron	Fort Worth, Texas
	Bertea Corporation (Division, Parker Hannifin Corp)	Irvine, California
	Garrett AiResearch Industrial Division	Los Angeles, California
	Systems Technology Inc. (STI)	Hawthorne, California
Propulsion	Avco Lycoming Stratford Division	Stratford, Connecticut
	Cessna Aircraft Corporation	
	McCauley Accessory Division	Dayton, Ohio
	Curtiss-Wright Corporation	Wood-Ridge, New Jersey
	Garrett AiResearch Manufacturing Co.	Phoenix, Arizona
	Hartzell Propeller Inc.	Piqua, Ohio

# Table 3.1. General Aviation Manufacturing Facilities Visited (Continued)

Propulsion (cont)	Teledyne Continental Motors	
	Aircraft Products Division	Mobile, Alabama
	United Technologies Corporation	
	Hamilton Standard Division	Windsor Locks, Connecticut
	<ul> <li>Pratt and Whitney of Canada</li> </ul>	Montreal, Canada
	Williams Research Corporation	Walled Lake, Michigan

## Table 3.1. General Aviation Manufacturing Facilities Visited (Concluded)

implementation, current state-of-the-art, and advanced technology development efforts. An attendant benefit was the acquisition of reference material. The presentations provided exposure to many subject areas of interest and the acquired papers provided documented support for the opinions expressed by the speakers. The NEAA meeting differed significantly from the others in that it provided product information as well as an exposure to the marketing strategies of manufacturers. Table 3.2 provides a list of the meetings attended.

## 3.2.4 Research Facilities

Three NASA research facilities were visited as shown in Table 3.3. These centers provided the necessary technical background for emerging technologies which would otherwise be

Meeting	Organization	Month (1979	) Location
lst Central Okiahoma AlAA Mini Symposium	American Institute of Aeronautics and Astronautics	Feb	Norman, Oklahoma
Business Aircraft Meeting	Society of Automotive Engineers	Apr	Wichita, Kansas
Display Systems Engineering Short Course	University of California at Los Angeles	Jun	Los Angeles, California
AIAA Systems and Technology Meeting	American Institute of Aeronautics and Astronautics	Aug	New York, New York
NBAA National Convention	National Business Aircraft Association	Aug	Atlanta, Georgia
General Aviation Propulsion Conference	Lewis Research Center (NASA)	Nov	Cleveland, Ohio

Table 3.2. Professional Meetings

unavailable (unpublished). Also, they willingly discussed both strengths and weaknesses of candidate technologies within their fields of expertise.

Table 3.3. NASA Research Centers

.

Center	Date	Location
Ames Research Center	June, 1979	Moffett Field, Calif.
Langley Research Center	August, 1979	Hampton, Virginia
Lewis Research Center	June, 1979	Cleveland, Ohio

## 3.2.5 Facility Contacts

Successful implementation and completion of the visitation phase relied heavily upon the contacts established with the facilities. Table 3.4 lists the final contacts at each facility.

#### 3.3 MEETINGS

All meetings within the visitation phase proved quite fruitful. In every case, the reception of the oroject staff by the facility representatives was exceptional, and an open and candid atmosphere prevailed throughout all discussions. Areas of mutual interest regarding current work on advanced technologies as well as the impressions and attitudes of the general aviation industry were discussed. To cover each of these meetings individually is impractical in this text. Therefore, an abbreviated compilation of notes generated by the project staff at each of the facilities visited is included as Appendix A.

The following discussion is intended to serve as an introduction to that appendix. As such, it provides a brief description of the subject areas discussed during the meetings. Manufacturer visits are first discussed, followed by NASA visits.

## 3.3.1 Discussions With Manufacturing Facilities

3.3.1.1 <u>Airframe</u>. The airframe manufacturers are, in every sense, the generalists of the industry. Their final product, an operational airplane, must include products representing many technologies and disciplines. Thus, these manufacturers must have an awareness of all aspects of the industry.

Facility	Contact
Ames Research Center	Seth Anderson Research Assistant Intra-Agency Programs
Avco Lycoming	Walt Schraeder Director of Advanced Technology Engines
Beech Aircraft Corporation	Bill Wise Vice President Advanced Technology
Bell Helicopter Textron	Hugh Upton Group Engineer Research Electronics
Bellanca Aircraft Corporation	Andrew Vano Chief Engineer
Bellanca Aircraft Engineering, Inc.	August Bellanca President & Chief Engineer
Bertea Corporation	John C. Hall Group Vice President
Brittain Industries, Inc.	Charles Walters President
Cessna Aircraft Corporation	
ARC Division	Virgil Davis Chief Engineer
McCauley Accessory Division	W.B. Voisard Chief Engineer
Pawnee Division	Harvey Nay Chief Engineer
Wallace Division	Emmett Kraus Supervisor of Advanced Design
Curtiss-Wright Corporation	Bill Silvestri RC Engine Program Manager

# Table 3.4. Facility Contacts

5

۰,

Facility	Contact
Edo Aire/Mitchell	John Nixon Chief Engineer
Garrett AiResearch	
Industrial Division	Dick Barcus Project Engineer Aircraft Systems & Control
Manufacturing Company	M.C. Steele Director of Engineering
Gates Learjet Corporation	Richard Etherington Director of Technical Engineering
Gulfstream American Corporation	Bob Stewart Assistant to the Vice President of Engineering
Hartzell Propeller Inc.	Ben Harlamert Vice President, Engineering & Chief Engineer
King Radio Corporation	Dan Rodgers Group Leader Special Programs
Langley Research Center	Bruce Holmes Aerospace Technologist Flight Mechanics Division
Lewis Research Center	William Strack Supervisory Aerospace Engineer Propulsion Section
Mooney Aircraft Corporation	Fen Taylor Chief of Aerodynamics and Performance
Narco Avionics	Norman Messinger Manager Advanced Development

t.

# Table 3.4. Facility Contacts (Continued)

Facility	Contact
Piper Aircraft Corporation	Grahame Gates Director Advanced Engineering
Rockwell International	
Collins Avionics	G.L. Benning Vice President of Advanced Technology and Engineering
General Aviation Division	Larry McHughes Director, Engineering
Rutan Aircraft Factory	Bert Rutan
Sperry Flight Systems	W.T. Robertson Manager, Engineering
Systems Technology Inc.	Irving L. Ashkenas Vice President
Teledyne Continental Motors	L. Waters Vice President Aeronautical Engineering
United Technologies Corp.	
Hamilton Standard Division	D.F. Phillips Head of Technical Planning
Pratt & Whitney of Canada	Sid Monaghan Chief, Research and Development Support
Williams Research Corporation	Edward J. Lays Senior Applications Engineer

Table 3.4. Facility Contacts (Concluded)

and a start of the second start

ŧ

Key areas included in the discussions were:

- (1) Propulsion and powerplants. New generation engines, engine controls, propellers, and their integration into an airframe.
- (2) Structures. Composites, metal-metal bonding.
- (3) Systems. Flight control systems, micro-computer based systems, and avionics.
- (4) Aerodynamics. New generation NASA airfoil sections, and computational aerodynamics.
- (5) Configuration. The advantages and disadvantages of canards and tandem wings.

3.3.1.2 <u>Propulsion</u>. Much developmental work is occuring within this industry, and many promising concepts are emerging. The "new generation" of general aviation powerplants and propellers was a prime topic throughout the industrial and research community. Topics discussed with the propulsion representatives were:

- (1) General Aviation Turbine Engine GATE.
- (2) General Aviation Propeller Study GAP.
- (3) Advanced propellers configuration aspects, advanced airfoil sections, composite materials.
- (4) Propulsion integration airframe-powerplant integration.
- (5) High speed propellers propfan.
- (6) Positive displacement engines diesel, rotary combustion engine, advanced reciprocating engine concepts.
- (7) Powerplant control integrated controls, microprocessorbased controls.

3.3.1.3 Avionics. Advanced technology prevails within this manufacturing group due to the competitive nature of this particular market. Survivability demands continued research and development in new technology areas, resulting in a level of sophistication that often surpasses military and commercial markets.

Areas covered during the discussions included:

- (1) Digital technology analog versus digital avionics.
- (2) Integration multi-function avionics packages and standardization, multiplexing.
- (3) Displays electronic displays and instrumentation.
- (4) Flight controls digital and electronic flight control systems.

3.3.1.4 <u>Systems</u>. Systems manufacturers offer areas of technology transfer to general aviation. For the most part, their main contributions lie in areas other than general aviation. Broad foundations in other fields allow "spinoff" technologies to filter in under circumstances and costs that are acceptable to the community of manufacturers and users.

The areas discussed included:

- Flight control actuators advanced hydraulics applied to general aviation.
- (2) Turbocharging advancements for general aviation.
- (3) Displays Heads up and head mounted displays.
- (4) Fiber optics signal transmission, airframe structural monitoring.
- (5) Flight control systems microprocessor based systems, fluidics.

It should be noted that comments by industry representatives were not limited solely to the topics listed. Due to the nature of the interviews, the personal interests of the participants, and the degree of integration required for production of general aviation hardware, much overlap between subject areas existed. Appendix A should be consulted for further information.

## 3.3.2 NASA Research Centers

Appendix A also includes the summarized notes from the visits to the Ames, Langley, and Lewis Research Centers.

## 3.3.2.1 Ames Research Center Topics.

(1) Avionics

- (a) Preliminary Candidate Advanced Avionics Systems (PCAAS).
- (b) Demonstrator for Advanced Avionics Systems (DAAS).
- (2) Stall/spin aerodynamic tailoring.
- (3) Cooling drag.
- (4) Small Transport Aircraft Technology (STAT).
- (5) Aerodynamics.
- 3.3.2.2 Langley Research Center Topica.
- Crash dynamics seats, restraints, structures, fire prevention.
- (2) Composites types, testing, characteristics.
- (3) Aerodynamics
  - (a) 3-dimensional
  - (b) 2-dimensional
    - natural laminar flow sections
    - low speed sections

high speed sections

- (4) Avionics and controls pilot-ATC interface, fluidics, pilot workload.
- (5) Stall/spin research.
- 3.3.2.3 Lewis Research Center Topics.
- (1) GATE
- (2) Positive displacement engines
  - (a) Advanced reciprocating engines.
  - (b) Alternative engine systems
    - · Diesel engines
    - Rotary combustion engines
- (3) Propeller Technology ~ turboprop, prop fan, variable pitch fan, turbofan, GAP program.

## 3.4 RESULTS OF THE FACILITY VISITS

The visitation phase of the project proved to be quite successful. A strong data base was developed through the interviews as well as from those sources of information identified during the interviews. Also, the insight and opinions of the various representatives provided indications of the practicality and/or feasibility for advanced technology development.

## 3.4.1 <u>Technology</u> and the Industry

To general aviation, like other industries, a technology is of no use unless:

- (1) it satisfies a need,
- (2) the developer can afford it,
- (3) it is profitable to the user, and
- (4) it is introduced at the right time.

The most important factor affecting technology implementation is user acceptance. Manufacturers cannot afford to pursue a technology merely to improve a product unless the user requires it. A market for the technology must exist, and the associated developmental costs must be acceptable to the manufacturer.

## 3.4.2 Technology and Cost

Three major cost constraints exist for the general aviation manufacturer when incorporating new technologies. In general, these may be grouped into the broad categories of developmental costs, certification costs, and product liability costs. 3.4.2.1 <u>Developmental Costs</u>. Developmental costs can be extreme, particularly in high technology areas. Many companies cannot absorb these and must rely instead on developments within NASA research centers or industries external to general aviation. Often, the production base or technology requirements of the automotive industry or the spinoffs realized from other technologies serve to reduce general aviation developmental costs. For example, the cost of the 8,000 aircraft turbochargers produced annually benefits significantly from the 1 million units demanded by the automotive and tru king industry.

3.4.2.2 <u>Certification Costs</u>. While the project was concerned primarily with an assessment of advanced technologies, the feasibility of promising technologies being integrated into future airplanes appeared to be jeopardized as much by certification costs as by technological risks. Some manufacturers, for example, pointed out that these costs can amount to 10 to 100 times the cost of developing the technology itself.

Since this difficulty presented serious implications for the research at hand, an effort was made to solicit the opinions of manufacturers regarding certification procedures. This was done in order to evaluate manufacturer perceptions of the process and is not intended to be an evaluation of the process itself. In every case discussed, members of the present research team noted that sufficient information to allow an unbiased evaluation of the problems identified was lacking. However, the following points appear to merit further discussion and consequently are incorporated into this report.

- (1) Interpretations of the same regulation by different FAA regional offices may sometimes result in markedly different certification requirements for the same technology in different regions.
- (2) Delays resulting from "excessive red tape" are expensive to manufacturers. One frustrated manufacturer produced documentation which indicated that an application filed more than 6 months earlier had yet to generate a response.
- (3) Personnel qualifications were addressed by one manufacturer.

In this particular case, the manufacturer was irritated because an inexperienced individual with a civil engineering background was given the responsibility for establishing compliance with propulsion/airframe certification requirements.

(4) New technologies are often evaluated by old or outdated techniques. In one case, an attempt to utilize a finite element computer code (NASTRAN) to validate structural integrity computations was perceived by the manufacturer to be hampered by a lack of familiarity with the code on the part of the certifying officials.

The significant and common denominator which appeared whenever a manufacturer chose to identify "difficulties with certification" was that the issue was usually an emotional one. Better communications between manufacturers and FAA regional offices such that both parties recognize the difficulties faced by the other appears to be a fundamental requirement if the issues are to be resolved. The inescapable fact is that manufacturers pay a heavy premium in order to certify a new technology.

3.4.2.3 <u>Product Liability Costs</u>. These costs are also high. One airframe manufacturer pointed out that 15% of a single engine airplane's price represented product liability costs. Another general aviation manufacturer pointed to insurance premiums of \$3 million per year to illustrate these costs.

Nuisance suits seem to prevail within this industry, and several examples were pointed out by some of the manufacturers.

Whether justified or not, such suits represent a very real cost to the manufacturer.

The disturbing aspect of high product liability costs is that they point to a history of large court settlements. This would appear to deter the incorporation of new technologies and, instead, promote an atmosphere of conservatism within the industry.

Two facets of product liability deserve mention. In one case, manufacturers may avoid new technologies for fear that the incorporation of improvements to systems may be interpreted to mean that deficiencies exist in previous models of the same system. This otherwise unnecessary exposure to suits can be a very real deterrent to the incorporation of new technologies.

The second point to be made is that the incorporation of a new technology always has attendant risks. Here, the experience gained with a mature system will no longer assist the manufacturer in avoiding the many unforeseen problems associated with new technologies. Product liability takes on added significance under these conditions.

## 3.5 RECOMMENDATIONS FOR NASA

The following suggestions and recommendations are the result of observations made during the visits to manufacturers and research facilities.

#### 3.5.1 Communications

Better communications need to be maintained between NASA and

the industry with regard to forthcoming research topics and the results of completed research projects. For example, several companies indicated that they were not aware of the final results of the Redhawk and ATLIT programs. While this points to a disturbing lick of awareness of technical publications on the part of industry, NASA will promote its image immensely if industry officials are advised of the availability of these reports. Interestingly, NASA publications designed to fulfill this task (SCAN's, STAR's, Tech Briefs, etc.) apparently are not reaching the industry or are not being used by them.

## 3.5.2 Basic Research and Product Development

Industry encourages NASA involvement with basic recearch and discourages any efforts aimed toward product development. Since industry involvement with basic research is small (less than 5% of the total engineering budget for one of the industry's leaders), this research team emphasizes that NASA should continue its efforts in basic research. High risk technologies such as those associated with propulsion, and sophisticated aerodynamic analytical tools such as those associated with natural laminar flow airfoils and numerical optimization techniques deserve special attention. Avionics manufacturers, on the other hand, appear wary of any NASA efforts which may be perceived to lead to standardization similar to those embodied in ARINC specifications.

## 3.5.3 Research Contracts

NASA should not fund programs which diminish a competitive

advantage which may already be enjoyed by a manufacturer through its own research efforts. In those cases where the problem is more perceived than real, NASA should clearly and publicly define the limits of the research contract being awarded.

H

## CHAPTER 4

## TECHNOLOGY EVALUATION

The goal of this phase of the study was to establish a rank ordering of technologies according to their benefits to general aviation airplanes. This involved three major tasks: (1) selecting the method of evaluation, (2) identifying the candidate technologies, and (3) evaluating the candidate technologies.

## 4.1 METHOD OF TECHNOLOGY EVALUATION

In selecting the evaluation method, the following criteria were applied to the available techniques in an effort to determine the technique best suited to the requirements of the present research effort.

- (1) The method must allow the evaluation of specific technologies.
- (2) The method must allow dissimilar technologies to be evaluated in a consistent manner.
- (3) The method must be within the means (cost and time) of the project.
- (4) The method must be as objective as possible subject to the three previous criteria.

Many of the techniques used in technological forecasting did not fulfill the requirements of this research effort. For example, regression and trend analyses assume one has suitable data to construct extrapolations. Such data does not exist for many of the technologies considered in this study because they have had only limited use in \_eneral aviation airplanes. A direct technology

31

# PRECEDING PAGE BLANK NOT FILMED

evaluation using the Delphi Method (Ref. 125) was rejected as beyond the means of the project due to the large number of technologies to be investigated. The evaluation method finally adopted is based on a technology figure of merit concept and has already met with some success (Ref. 18). This method, here termed "evaluation technique," is a simple linear compensatory model and is discussed below.

## 4.1.1 Evaluation Technique

In the evaluation technique (ET), candidate technologies (CT) are evaluated relative to current technologies by assessing their impacts on a group of categories. The categories model the major factors involved in the operation of an airplane and are assigned weightings (w) according to their relative importance. The impact of a CT in a given category is quantified by a relative benefit (b), where b > 0 indicates that the CT offers an improvement in the category (relative to current technology) and b < 0 indicates that the CT causes a degradation in the category (relative to current technology). Summing the products of the relative benefits and category weightings yields a figure of merit (FM) for the CT which is a measure of how much improvement the candidate technology offers in the overall operation of the airplane. The figure of merit is defined by Equation 4.1 while Fig. 4.1 illustrates the general concept of the evaluation technique.

$$FM_{i} = \sum_{j=1}^{n} b_{ij} w_{j}$$
(4.1)

where : n is the number of categories,

b<sub>ij</sub> is the relative benefit of technology i in category j,

 $w_i$  is the weighting of category j, and

FM, is the figure of merit of technology i.

Referring to Fig. 4.1, one notes that the adoption of this type of technology evaluation method is not without difficulties. First, the categories must be selected, and second, their weightings must be determined. So, while the ET inherently satisfied the first three criteria, the fourt criteria, that of objectivity, demanded that special attention be given to the category selection and weighting.

## 4.1.2 Category Selection

The first step in defining the category group was the establishment of the following selection criteria:

- The group of categories must be broad enough to model the major factors involved in the operation of an airplane.
- (2) There should be a minimum of overlap between individual categories.
- (3) The number of categories should be as small as possible while satisfying the first two criteria.

Various selection schemes were studied, and it soon became obvious that all of them involved generating a consensus from a set of opinions. The Delphi Method, although too involved for the overall technology evaluation, seemed well suited to the category selection problem and was therefore used. The survey group for the cateogry selection consisted of the project staff and the

		Categories —	Cost	Safety	Weight	Noise		Figure of Merit
Candidate ◀ ☵ Technologies	Weights → CT's →	9	10	8	5		FM	
	Turboprops	-4 x 9	+3 x 10	+5 x 10	+2 x 5		+44	
	Wankel	0 x 9	+1 x 10	+2 x 8	0 x 5		+26	
	Winglets	-1 x 9	0 x 10	-1 x 8	0 x 5		-17	
	L.E. Slats	-2 x 9	+2 x 10	-1 x 8	0 x 5		- 6	
	NAVSTAR/GPS	-1 x 9	+1 x 10	0 x 8	0 x 5		+ 1	
	Ļ	<i>\</i> →	$\downarrow$	ļ	Ļ	ţ	↓	
		······································	(Relative	Benefit) x	(Category	Weight)		

Figure 4.1. Illustration of the Evaluation Technique

the faculty members of the Department of Aerospace Engineering at the University of Kansas.

The category selection surveys involved the following procedures:

- Obtain a category list, confidence level, and comments from each of the participants. The confidence level is a measure of a participant's confidence in his response.
- (2) Analyze the group response and generate an "average" category list.
- (3) Feed back the average category list along with participant comments.
- (4) Repeat (2) through (3) until a viable "average" category list is obtained.

The Delphi Method worked quite well although the analysis of the group response was complicated by the fact that the survey dealt with symbols (in the form of category names) instead of numbers. The survey converged to an acceptable category list (17 categories) in four survey rounds. The final category list is given in Table 4.1.

Referring to Table 4.1, one notes that some categories appear to overlap. Examples are the categories of Fuel Efficiency, Reliability, and Direct Operating Cost. The problem is minimized when one considers that Fuel Efficiency is concerned with fuel availability as well as with fuel costs. Likewise, Reliability is concerned with operational readiness as well as maintenance costs.

With these factors in mind, the category list was considered acceptable. Further discussion of the category selection survey may be found in Appendix B.1.

Category	Definition
Ceiling	Altitude at which the maximum rate of climb is 0.51 m/s (service ceiling).
Crashworthiness	The characteristics of an air- plane which determine the level of occupant protection in the event of a crash.
Cruise Speed	Maximum continuous cruise speed.
Direct Operating Cost	All costs directly attributable to flying and keeping an air- plane operational. All scheduled and unscheduled main- tenance costs and fuel costs are included.
Emissions	Pollutants produced during the operation of an airplane; does not include noise.
Empty Weight	Airplane weight without fuel, crew, and payload.
Exterior Noise	Noise perceived at ground level due to the operation of an airplane.
Fuel Efficiency	Airplane cruise efficiency measured in air-miles per pound-fuel.
Interior Noise	Noise perceived by the occu- pants of an airplane.

Table 4.1. Technology Evaluation Categories

TADIE 4.1. TECHNOIDX' EVALUATION CALENOLIAS (CONCINN	Table 4.1.	Technology	Evaluation	Categories	(concluded
--	------------	------------	------------	------------	------------

Category	Definition
Pilot Workload	The amount of time, concentra- tion, and effort a pilot must devote to the safe operation of an airplane. This includes the effects of airplane handl- ing qualities.
Purchase Price	The price paid for a new air- plane by the user, including avionics and equipment costs.
Range	The distance an airplane can fly without refueling, allow- ing for appropriate fuel reserves.
Reliability	A measure of the probability of failure of an airplane com- ponent or system.
Ride Qualities	A measure of the effects of aircraft motion on the smooth- ness and comfort of the ride experienced by the occupants.
Safety	A measure of an airplane's inherent characteristics which reduce the probability of an accident.
Static Comfort	A measure of an airplane's inherent comfort. This in- cludes roominess, seat com- fort, ventilation, decor, ease of entry, etc.
Takeoff/Landing Performance	This parameter includes take- off and landing speeds, field length requirements, and rates of climb and descent.

## 4.1.3 Category Weightings

Having completed the category selection survey, the Delphi Method was again used in the determination of the category weights. This survey, termed the "category rating survey", was conducted on a much larger scale than the category selection survey with 42 participants representing general aviation manufacturers and user groups, university faculty, NASA centers, and project staff. A list of the participants is given in Table 4.2.

Upon reviewing the 17 categories to be weighted, it was noted that in general, different types of airplanes would have different sets of category weights because of differing operational priorities. Because the statement of work (for this study) specified that two types of airplanes were to be investigated, two sets of category weightings were generated. In the rest of the text, the two airplanes are referred to as Airplane A, which is a six passenger (including pilots) airplane for business and/or personal transportation, and Airplane B, which is a 19 passenger (excluding pilots) commuter airliner.

The category rating survey was conducted as outlined below: (1) Participants gave each category (for each airplane) a rating (R) constrained by  $0 \le R \le 10$  with a minimum scale increment of .5. R = 10 was assigned to the category considered most important and the other categories were rated in a relative manner with R = 0 meaning the category is of no importance relative to the most important one. Duplications in ratings were allowed because two or more categories could be of equal

Table 4.2. Category Rating Survey Participants

Aircraft Owners and Pilots Association AVCO Lycoming Beech Aircraft Corporation **Bell Helicopter Textron** Bellanca Aircraft Corporation Bellanca Aircraft Engineering, Inc. Brittain Industries, Inc. Cessna Aircraft Company - Aircraft Radio and Comprol Division Cessna Aircraft Company - Pawnee Division Cessna Aircraft Company - Wallace Division Commuter Airlines Association of America Curtiss-Wright Corporation Edo-Aire Mitchell Division Garrett-AiResearch - Industrial Division Gates Learjet Corporation Gulfstream American Corporation Hartzell Propeller, Inc. King Radio Corporation Mooney Aircraft Corporation Narco Avionics Division NASA Ames Research Center NASA Langley Research Center NASA Lewis Research Center National Business Aircraft Association

Table 4.2. Category Rating Survey Participants (concluded)

Parker Hammifin Corporation - Control System Division (formerly Bertea Corporation)
Pf.pes: Aircraft Corporation
Pratt & Whitney Aircraft of Canada, Ltd.
Rockwell International - Collins Division
Rockwell International - General Aviation Division
Sperry Flight Systems - Avionics Division
Systems Technology Incorporated
Teledyne Continental
Williams Research Corporation
University of Kansas, Aerospace Engineering Faculty<sup>\*</sup>
Project Staff<sup>\*</sup>

These groups each contributed more than one participant

importance. Participants also rated their degree of confidence in their response with a confidence level (CL) constrained by  $0 \ge CL \le 1$  for each airplane. Comments on the category ratings were encouraged.

- (2) Feedback to the participants consisted of information regarding the mean category ratings, the category rating distributions (in the form of histograms), and participant comments for each of the airplanes.
- (3) The survey continued until the participant response reached a predetermined level of stability (less than 15% change in participant voting).

The category rating survey required three rounds to achieve satisfactory convergence. The category weightings used in the evaluation technique are simply the mean category ratings from the final survey rounds.

Table 4.3 presents the final category weightings for Airplanes A and B and Table 4.4 shows the category rankings in order of importance. Details of the category rating survey may be found in Appendix B.2.

## 4.2 CANDIDATE TECHNOLOGY IDENTIFICATION

This section deals with the selection of the technologies to be analyzed with the evaluation technique and discusses the data base used. Since a large list of advanced technologies is easily created, several constraints were placed on the selection process. An early list, defined as "Preliminary Candidate Technologies," and a final list, defined as "final Candidate Technologies," are both discussed here.

## 4.2.1 Data Base

In selecting candidate technologies, one must first collect information regarding technologies in general. The project relied on three major sources of information as detailed below.

4.2.1.1 <u>Literature Search</u>. A computerized literature search was conducted using the Lockheed DIALOG system. In preparing for the computer search, a manual search was made of the NASA STAR index to identify key words, technology areas, and sample titles. This

41

Ś

Category	Weightings	
	Airplane A <sup>*</sup>	Airplane B**
Ceiling	5.429	5.447
Crashworthiness	6.816	7.333
Cruise Speed	7.922	7.480
Direct Operating Cost	7.906	9.451
Emissions	1.606	2.156
Empty Weight	4.340	5.657
Exterior Noise	4.287	5.413
Fuel Efficiency	8.157	8.606
Interior Noise	7.037	7.452
Pilot Workload	7.349	6.998
Purchase Price	8.515	7.761
Range	7.290	7.181
Reliability	8.854	9.547
Ride Qualities	6.110	7.238
Safety	9.422	9.544
Static Comfere	6.244	6.797
Takeoff/Landing Performence	6.874	7.407

# Table 4.3. Evaluation Technique Category Weightings

\* Airplane A - 6 Passenger Business/Personal Airplane

\*\* Airplane B - 19 Passenger Commuter Airliner

,
Rank	Cat	egories:		
Order	Airplane A <sup>*</sup>	Airplane B		
1	Safety	Reliability		
2	Reliability	Safety		
3	Purchase Price	Direct Operating Cost		
4	Fuel Efficiency	Fuel Efficiency		
5	Cruise Speed	Purchase Price		
6	Direct Operating Cost	Cruise Speed		
7	Pilot Workload	Interior Noise		
8	Range	Takeoff/Landing Performance		
9	Interior Noise	Crashworthiness		
10	Takeoff/Landing Performance	Ride Qu <b>alities</b>		
11	Crashworthiness	Range		
12	Static Comfort	Pilot Workload		
13	Ride Qualities	Static Comfort		
14	Ceiling	Empty Weight		
15	Empty Weight	Ceiling		
16	Exterior Noise	Exterior Noise		
17	Emissions	Emissions		

# Table 4.4. Evaluation Technique Category Rankings

100 H

transfert deligner in Statistic Barris

١

k

\* Airplane A - 6 Passenger Business/Personal Airplane

\*\* Airplane B - 19 Passenger Commuter Airline

resulted in six primary search topics (a-f) to which three specific topics (g-i) were later added.

- (a) Aircraft Design
- (b) Aircraft Propulsion
- (c) Aircraft Structures
- (d) Flight Controls
- (e) Navigational Aids
- (f) Avionics
- (g) Canard Configurations
- (h) NAVSTAR/GPS

#### (i) GASP (General Aviation Synthesis Program)

Table 4.5 presents the initial results of the computer search in terms of the number c2 abstracts printed for each search topic; note that some topics were combined to eliminate duplications. The 1655 abstracts were reviewed individually to identify pertinent articles. This resulted in 107 articles which were obtained from NTIS. The specific search procedures used for each topic may be found in Appendix B.3.

4.2.1.2 <u>Visits and Meetings</u>. Quite a few articles were found through recommendations of various people and from the references given in other articles. Also, aviation related magazines and journals were helpful in finding information.

The bibliography lists the most important books, articles, and papers obtained while the discussions of Chapter 3 and Appendix A summarize the information gained from the visits and meetings.

# Table 4.5. Computerized Literature Search Results

	Topic	Abstracts Printed
(a)	Aircraft Design	_33
<b>(</b> b)	Aircraft Propulsion	206
(c)	Aircraft Structures	275
(d)	Flight Controls	780
(e)	Navigational Aids	780
(f)	Avionics	780
(g)	Canard Configurations	22
(h)	NAVSTAR/GPS	89
(i)	GASP	0

# 4.2.2 Preliminary Candidate Technologies

The preliminary candidate technologies (PCT) were the result of the first formal attempt to identify technologies felt to have some application to General Aviation airplanes. Selecting the PCT proved more difficult than expected for several reasons, not the least of which was how to define a "technology." To select technologies in a consistent manner, three criteria were formulated.

 Operational results of a technology are not themselves technologies: Examples: "stall/spin prevention" and "increased TBO (time between overhaul)" are the results of the application of various technologies.

- (2) Design variables (parameters) are not technologies. Examples: "high wing loading," "high aspect ratio."
- (3) Non-specific "wish list" technologies are to be avoided. Examples: "low-cost sensors," "low-cost weather radar." These criteria were followed as closely as possible but marginal technologies were given the benefit of the doubt to insure that all applicable technologies were considered. The PCT selection process resulted in 137 technologies which are given in Table 4.6, grouped according to technology area.

Table 4.6. Preliminary Candidate Technologies

- I. AERODYNAMICS (11)
  - Winglets
  - Variable geometry winglets
  - Spoilers
  - Fowler flaps
  - Low drag surface coatings
  - Leading edge devices
  - Advanced low and medium speed airfoils (turbulent)
  - Supercritical airfoils
  - Advanced natural laminar flow airfoils
  - Improved stall/spin through aerodynamic tailoring
  - Active laminar flow control

# II. AIRCRAFT SYSTEMS (12)

- Microwave anti-icing
- Sonic/pulsating anti-icing

II. AIRCRAFT SYSTEMS (concluded) • Lithium hydroxide/hydrogen peroxide batteries • Air cycle environmental systems • Improved lead-acid batteries • AC electrical systems • Passive anti-icing through icephobic coatings • Variable cycle environmental systems · Accurate fuel monitoring and management · High speed brushless alternator • Single unit starter-generator • Air bearings III. COMPUTATIONAL METHODS (4) • Computational aerodynamics • Computational structural design/analysis • Aeroacoustic modeling/analysis CADCAM IV. CRASHWORTHINESS (7) • Load limiting seats Improved restraints · Foam filled fuel tanks • Burst/tear resistant fuel tanks • Energy absorbing floor • Anti-misting fuel treatment • Trangible fuel fittings

.

•

-1

v.	FLIGHT CONTROL SYSTEMS (21)
	• Fly-by-wire
	• Fly-by-light
	• Active gust alleviation
	• Active ride smoothing
	• Active flutter supression
	• Active controls for relaxed inherent stability (CCV)
	• Integrated low-cost wing leveler
	• Integrated yaw damper
	• Separate surface technology
	• Winglets for lateral-directional control
	• Direct side-force control
	• Direct lift control
	• Single level thrust/drag control
	• Force-stick controllers
	• Digital automatic flight controls
	• Fluidic automatic flight controls
	<ul> <li>Stick shaker/pusher for stall prevention</li> </ul>
	• Stabilizer/elevator spoilers for stall prevention
	• Pneumatic actuators
	• Hydraulic actuators
	• Electro-mechanical actuators
VI.	INFORMATION SYSTEMS (28)
	• Flush antennas
	• Digital data links

VI. INFORMATION SYSTEMS (continued) • Single function CRT displays • Time-shared CRT displays • HUD • Micro HUD • Warning annuciators • Total panel-mounted avionics • Active outside imaging • Fluidic shed-vortex airspeed sensor • Airplane health/diagnostic systems • Fluidic rate sensors • Multiplexing ARINC-type broadcast hierarchy • Integrated avionics and displays • Piezo-resistive pressure transducers • Fiber optics for data transmission • Liquid crystal displays • Flat CRT displays • Touch sensitive CRT • Weather radar • Alternate weather detection • Radar altimeter • Onboard computing capability • 3-axis magnetometer acceleration sensor

ļ

VI. INFORMATION SYSTEMS (concluded)
• Improved stall warning
• Laser gyros
VII. <u>MATERIALS/PROCESSES</u> (12)
• Metal/metal bonding
• Fiberglass composites
• Kevlar composites
• Graphite composites
• Honeycomb core composite skin panels
• Single crystal metal
• Powdered metal
• Isothermal forging
• Diffusion bonding
• Friction welding
• Corrosion resistant coatings
• Matched-die fiber reinforced plastic (FRP)
VIII. <u>NAVIGATION CONCEPTS</u> (11)
• VOR/DME RNAV
• Scanning VOR/DME RNAV
• Omega
• Differential Omega
• VLF NAVCOM
• Loran C
• NAVSTAR/GPS

VIII.	NAVIGATION CONCEPTS (concluded)
	• Inertial navigation
	• Doppler navigation
	• MLS
	• Inertial smoothing
IX.	NOISE (6)
	<ul> <li>Noise absorbing materials</li> </ul>
	• Improved mufflers
	<ul> <li>Variable engine/prop gearing</li> </ul>
	• Quiet propeller technology
	• Low level pressurization
	• Ducted propulsors
X.	PROPULSION (25)
	• Advanced diesel engine
	<ul> <li>Advanced rotary combustion engine</li> </ul>
	<ul> <li>Advanced reciprocating engine</li> </ul>
	• GATE engine
	• Auto engine conversions
	• QCCAT engine
	• Liquid cooling
	• Stratified charge
	<ul> <li>Improved turbocharging</li> </ul>
	• Variable timing
	<ul> <li>Electronic ignition</li> </ul>

......

X. <u>PROPULSION</u> (concluded)

Automatic mixture control
Lean burn combustion
Density compensating fuel injection
Total microprocessor engine control
Variable bypass turbofan
Variable pitch fan
Efficient propeller technology
Prop fan
Cooled turbine blades
Ceramic turbines
Composite propellers
Torsionally (aeroelastically) tailored propeller blades

# • Single lever throttle/mixture control

# 4.2.3 Final Candidate Technologies

As a group, the 137 PCT of Table 4.7 were not equally suited to the evaluation technique. First, the PCT exhibit different technology levels. In the propulsion area for example, complete advanced technology engines are compared against component technologies. One cannot really compare the two although both may be important. Second, the PCT exhibit different technology orders, meaning that while most technologies are first order (i.e., <u>on</u> the airplane) some are second order (i.e., <u>not on</u> the airplane). Examples are "advanced natural laminar flow airfoils" (first order) and "computational aerodynamics" (second order). The first requires the second. Hence, the two technologies are not really comparable. In addition to these fundamental problems, four other constraints are noted: (1) the ET category weightings are valid only for six passenger and commuter airplanes, (2) the project scope requires 1990 technology implementation, (3) technologies for which data are lacking cannot be evaluated objectively, and (4) some of the technologies are already gaining acceptance in general aviation.

These factors gave rise to a second set of criteria which were used to select the final candidate technologies (FCT).

- (1) Try to reduce the differences in technology levels.
- (2) Delete second order technologies.
- (3) Delete technologies not applicable to the six passenger or commuter airplanes.
- (4) Delete technologies which will not be ready for implementation by 1990.
- (5) Delete technologies for which sufficient data are lacking.
- (6) Delete technologies already gaining acceptance in general aviation.

Application of the above criteria resulted in 56 final candidate technologies which are presented in Table 4.7 and are subsequently discussed in Section 4.4. Appendix B.3 contains a complete discussion of the candidate technology selection process.

1.	AERODYNAMICS (9)
	• Winglets
	• Spoilers
	• Fowler flaps
	<ul> <li>Low drag surface coatings</li> </ul>
	• Active laminar flow control
	• Leading edge devices
	• Advanced low/medium speed airfoils
	• Improved stall/spin through aerodynamic tailoring
II.	AIRCRAFT SYSTEMS (2)
	<ul> <li>Anti-icing surface coatings</li> </ul>
	• AC electrical systems
III.	CRASHWORTHINESS (4)
	• Load limiting seats
	• Energy absorbing floor
	• Improved restraints
	• Burst/tear resistant fu <b>el tanks</b>
IV.	FLIGHT CONTROL SYSTEMS (14)
	• Fly-by-wire
	• Fly-by-light
	• Active gust alleviation
	• Active ride smoothing
	• Active flutter suppression

# Table 4.7. Final Candidate Technologies

# Table 4.7. Final Candidate Technologies (continued)

CALL BOT DRA CALL

IV. FLIGHT CONTROL SYSTEMS (concluded) • Active controls for relaxed inherent stability (CCV) • Integrated yaw damper • Integrated low-cost wing leveler • Separate surface technology (SSSA) • Direct side-force control • Direct lift control • Single lever thrust/drag control • Fluidic automatic flight control system • Active stall prevention V. INFORMATION SYSTEMS (8) • Digital data links • CRT displays • HUD • Micro HUD • Systems status display • Integrated avionics and displays • Fiber optics for data transmission · Laser gyros VI. <u>NAVIGATION</u> CONCEPTS (6) • NAVSTAR/GPS Inertial navigation • Doppler navigation

# Table 4.7. Final Candidate Technologies (concluded)

VI. NAVIGATION CONCEPTS (concluded) • Microwave Landing System (MLS) • Loran C • Omega VII. NOISE (3) • Quiet, efficient propeller technology Low-level pressurization Ducted propulsors VIII. PROPULSION (7) • GATE engine Stratified charge rotary combustion engine HCRLB reciprocating engine \* Stratified charge reciprocating engine Advanced diesel engine Liquid cooling Improved turbocharging IX. STRUCTURAL MATERIALS (3) Fiberglass composites • Kevlar composites \* Graphite composites

# 4.3 APPLICATION OF THE EVALUATION TECHNIQUE

The application of the ET involved three main tasks: the mechanization of the ET, the determination of the relative benefits, and the analysis of the ET-generated figures of merit. The reader may find Figure 4.1 useful in the following discussion.

#### Mechanization of the Evaluation Technique

Because of the sheer number of calculations required to evaluate 56 technologies in 17 categories for two airplane types, a FORTRAN computer program known as TCHLST was written to compute the figures of merit. Since the considerations of paramount interest to any user (input, output, and efficiency) are particularly significant for the case at hand, input and output were simplified by employing a problem oriented language (POL) known as SCAN, while efficiency was maintained by using arrays, array pointers, and packed arrays. While SCAN is available from the University of Illinois for several mainframes, its limited use together with other peculiarities built into TCHLST which are system-dependent (word size and system software) have resulted in a decision not to include the program listing with this report. The program capabilities, however, are briefly discussed here in order to support the resulting data.

Input data, if not managed properly, can quickly invalidate results. Hence SCAN was employed to allow data inputs in the form:

#### AIRPLANE A

AERODYNAMICS

'WINGLETS' .5 0 0 .5 0 -2 0 .5 0 0 -1 .5 0 0 0 0 0 0 'SPOILERS'-.5 0 2 1 0 1 0 1 0 .5 0 1 0 2 0 0 0

and incorporate commands such as

PRINT INPUT

PRINT TABULAR CATEGORIES OMIT CRASHWORTHINESS PRINT BARCHART ALL RESET OMIT RESET PRINT, etc.

in order to (1) allow easy data entry, (2) allow accurate input verification, (3) allow quick, accurate data analysis, and (4) provide for sensitivity analyses under different category weightings. Efficiency was maintained by array management where, for example, different addresses were packed into the left and right half of a word size (36 bits). Hence, although not portable, the program was very efficient, with costs on the order of \$3 to compute, sort, format, and output original data as well as perturbed data (which could be demanded in real time).

#### 4.3.2 Determination of Relative Benefits

In assigning relative benefits, three factors were found to be of great importance. They are the relative benefit scale, the technology baseline, and the technology application and integration.

4.3.2.1 <u>Relative Benefit Scale</u>. The scale of relative benefits  $(-b_{max} \leq b_{ij} \leq b_{max})$  must be broad enough to allow adequate differentiation between technology impacts but narrow enough that guessing is avoided. This is a strong function of the amount of information available for a given technology so in general, an optimum scale for one technology will not be optimum for another technology. A workable compromise was found to be  $-3 \leq b_{ij} \leq 3$  where increments of  $\pm 1$  are normally used but, if justified, increments of  $\pm .5$  are allowed.

4.3.2.2 <u>Technology Baselines</u>. A technology baseline must be established before any relative benefits are obtained because by definition, a relative benefit compares a candidate technology to a current technology baseline. Since different classes of airplanes utilize different technologies, separate technology baselines were established for the six passenger and commuter airplanes. These baselines are presented in Table 4.8.

4.3.2.3 <u>Technology Application and Integration</u>. In evaluating a given technology, a decision must be made as to how the technology will be used and to what extent other airplane parameters will change. For example, spoilers do not offer much (if any) advantage over ailerons unless the freed trailing edge is used for additional flap span. Even then, one must decide if

Airplaue A*	Airplane B <sup>**</sup>
• Reciprocating engine	• Turboprop
• Aluminum structure	• Aluminum structure
• Limited flush riveting and/or bonding	<ul> <li>Extensive flush riveting and/or bonding</li> </ul>
<ul> <li>Conventional avionics with VOR/DME RNAV but no auto- pilot</li> </ul>	• Conventional avionics with autopilot but no RNAV
• Plain or single-slotted flaps	• Single-slotted Fowler flaps
<ul> <li>Conventional controls and control surfaces</li> </ul>	<ul> <li>Conventional controls and control surfaces</li> </ul>
• No anti-icing	• Anti-icing in the form of hot air or boots

# Table 4.8. Technology Baselines

\*Airplane A - 6 Passenger Business/Personal Airplane \*\* Airplane B - 19 Passenger Commuter Airplane

takeoff and landing performance will be improved with a fixed wing loading, or if an improvement in cruise efficiency will be sought by increasing wing loading. Similar technology application and integration effects are exhibited by many technologies, especially in the areas of composite structures and advanced propulsion. After reviewing the performance characteristics of current six passenger and commuter airplanes, it was decided that technologies offering either cruise or takeoff and landing performance improvements would be used to benefit cruise since the current airplanes seemed to have adequate takeoff and landing performance.

# 4.3.3 Sensitivity to Relative Benefits

Uncertainty in generating relative benefits is unavoidable and therefore must be accounted for. Using the guidelines of Section 4.3.2, three sets of relative benefits were generated for each technology and are termed "pessimistic," "likely," and "optimistic" (PLO) relative benefits. Their use and definition are analogous to the pessimistic, most likely, and optimistic time estimates frequently used in CPM and PERT analyses. Likewise, a Beta distribution is assumed such that an "expected" relative benefit is obtained as given by Equation 4.2 (Ref. 219).

$$b_{E} = \frac{b_{p} + 4b_{L} + b_{o}}{6}$$
(4.2)

To avoid unnecessary computations, only the PLO relative benefits are found and the expected figures of merit  $(FM_E)$  are obtained directly from the PLO figures of merit. It is easy to show that the expected figure of merit is given by Equation 4.3.

$$FM_{E} = \frac{FM_{P} + 4FM_{L} + FM_{O}}{6}$$
(4.3)

The PLO relative benefits were generated by an iterative process. First, information on each of the technologies was obtained from the project data base: Second, the PLO relative benefits for each technology were estimated. Finally, the PLO relative benefits were reviewed and updated as required. The PLO relative benefits are preliminary in that they are used only to test the evaluation procedure.

I

Figure 4.2 presents a sample of the pessimistic, likely, optimistic, and expected (PLOE) figures of merit where the expected figures of merit were computed with equation 4.3. (Complete results may be found in Appendix B.4.) This figure illustrates the characteristics of consistency and uncertainty which are discussed below.

4.3.3.1 <u>Consistency of Relative Benefits</u>. To obtain a meaningful ranking of the technologies, each must be evaluated in a similar manner with the same level of objectivity. Doing so results in a consistent set of likely relative benefits. While it is impossible to insure absolute consistency, the PLOE figures of merit provide information which allows one to check for reasonable consistency. This is done by comparing the likely and expected figures of merit for each technology. If a technology exhibits a large difference in these figures of merit, it implies that the likely relative benefit was itself either optimistic  $(FM_L > FM_E)$  or pessimistic  $(FM_L < FM_E)$  and needs to be refined. The data of Figure 4.5 is typical and shows that the preliminary likely relative benefits



Figure 4.2. Sample of the Pessimistic, Likely, Optimistic, and Expected Figure of Merit

are quite consistent in that the likely and expected figures of merit are nearly identical for most technologies. The small inconsistencies exhibited by a few technologies were kept in mind when the final relative benefits were determined.

4.3.3.2 <u>Uncertainty of Relative Benefits</u>. It was hoped that the PLOE approach, in addition to providing a check on consistency, would give a measure of the uncertainty involved in the evaluation of each technology.

For example, examining Fig. 4.2 one might conclude that the degree of uncertainty in the evaluation of a technology is directly related to the difference in its optimistic and pessimistic figures of merit. This has not proved to be a legitimate assumption because technologies which impact nearly all of the categories exhibit large changes in figure of merit for small changes in the relative benefits which results in an exaggerated perception of uncertainty.

The initial testing of the evaluation technique (PLOE studies) proved that the procedures established for evaluating the technologies worked well and resulted in consistent analyses. The following sections address the final evaluation of the candidate technologies in detail.

#### 4.4 OBSERVATIONS ON THE UTILIZATION OF THE EVALUATION TECHNIQUE

This section briefly discusses some of the more salient points involving the use and interpretation of the evaluation technique. Specific items addressed include:

- the effect of qualitative values of different raters in assigning relative benefits to technologies,
- (2) the stability of figures of merit under different conditions,
- (3) the orthogonality of categories,
- (4) the effect of varying the weighting for Empty Weight, and
- (5) the significance of figure of merit scores.

# 4.4.1 Qualitative Values of Raters

As the evaluation technique was applied, it became evident that different raters had a different range of figures of merit resulting from different relative benefits applied to each of the technologies. However, an examination of the several results indicated remarkable consistency between the relative rankings of technologies. It was therefore decided to use one rater and emphasize the refinement of relative benefits generated by the rater rather than to standardize the results of several raters at the expense of exhaustive refinement.

#### 4.4.2 Stability of Figures of Merit

After several data sets were analyzed, it became evident that the figures of merit:

- exhibited consistency in ranking technologies as was predicted by the PLOE studies discussed previously in Section 4.3.3.
- (2) could change significantly if its anticipated application and integration changed (Section 4.3.2.3) regardless of whether

this change resulted from a technology breakthrough or a more innovative (less conservative) application. In light of (2) above, every effort has been made to define the baseline and the perceived application of each technology as

# 4.4.3 Orthogonality of Categories

shown in Section 4.5.

The orthogonality of categories and their weightings as developed by Surveys 1 and 2 and discussed in Sections 4.1.2 and 4.1.3 was investigated by performing a factor analysis of the relative benefit matrix (56 x 17) for both Airplanes A and B. The Biomedical Computer Programs (BMDP) statistical computer package as described in Reference 218 and available on-line at the University of Kansas was used to generate the factor data.

The intent of this study was to determine the possibility of grouping the 17 categories into a smaller, orthogonal set of components. If this could be done (and it was strongly suspected that such would be the case), then the possibility of reducing the number of categories into a smaller and more manageable set needed to be investigated (see Section 4.1.2).

BMDP results for both Airplane A and Airplane B indicated that six factors would explain a significant amount of the variance in both relative benefit data sets (74% for Airplane A and 76% for Airplane B). Furthermore, the first three factors explained 50% and 51% of the data variance for Airplane A and B respectively. These three factors included 11 of the 17 categories for both

6ť

airplanes where ten of the 11 categories were common to both. The six factors are shown in Table 4.9.

An examination of this table, however, quickly identifies extreme difficulties in implementing an evaluation technique consisting of only six orthogonal components. Specifically, the loss of differentiation between vehicle performance measurements would defeat the purpose of the technology evaluation. For example, benefits in cruise speed could not be differentiated from those accorded to empty weight. Likewise, improvements in noise control could not be differentiated from takeoff and landing performance. Although purchase price is grouped with DOC and reliability for Airplane B, it is statistically tied to static comfort and crashworthiness for Airplane A. Yet, purchase price received a score other than zero in 51 of 56 technologies for Airplane A and 44 of 51 technologies for Airplane B.

Armed with this information, it was decided that the 17 categories developed by Survey 1 provided a better measurement of a technology's impact than the six factors resulting from the factor analysis.

# 4.4.4 Effect of Empty Weight

A concern expressed early in the development of the evaluation technique dealt with the quantification of category weightings through the Delphi Method. Since a major benefit of the method is to allow respondents to preserve their views without coercion from others, a decision to halt further rounds of the surveys is often

# Table 4.9. Six Factors for Airplanes A and B

Factor	Airplane A Categories	Factor	Airplane B Categories
1	Range Cruise Speed Ceiling Fuel Efficiency Empty Weight	1	Fuel Efficiency Ceiling Cruise Speed Range Empty Weight
2	Exterior Noise Interior Noise Takeoff/Landing Performance	2	Exterior Noise Interior Noise Takeoff/Landing Performance
3	Reliability DOC Emissions	3	DOC Reliability Purch <b>ase</b> Price
4	Static Comfort Crashworthiness Purchase Price	4	Safety Pilot Workload
5	Safety Pilot Workload	5	Crashworthiness Static Comfort
14	Ride Quality	6	Emissions Ride Quality

made based on the stability of responses. This implies that the use of an average rating may be misleading if a large dispersion of responses is noted. (Note that "stability" is implied by a large number of unchanged responses from one round to the next, and does not imply that ratings have converged about a mean.) Such was the case for the Empty Weight category for Airplane B. As shown in the histograms in Appendix B.2., these ratings ranged almost uniformly from 0 to 9.5. Hence, results reflecting Empty Weights of 0, 5.657, and 9.5 (low, mean, hi) for Airplane B are included in Appendix B.5.

# 4.4.5 Significance of Figure of Merit Scores

Before proceeding to the actual results of the evaluation technique, one should be aware of the following aspects of the figures of menit:

- (1) Values near zero imply no significant improvement in effect on the user. For example, an improvement in range may be offset by high cost or maintenance.
- (2) Positive numbers reflect net benefits to the user while negative numbers reflect net penalites to the user.
- (3) Macroscopic figures of merit (FM) are fairly stable, where  $FM_i = 50$  signals greater benefit from technology i than  $FM_j = 30$  does for technology j. However,  $FM_i = 50$  is difficult to differentiate from say,  $FM_k = 45$ . A major exception to this rule is noted in (4) below.
- (4) Those technologies impacting many categories have higher figures of merit than those which affect only a few. Hence, the four technologies listed under the crashworthiness group have scores which do not truly measure their benefit to the user. This is due to the fact that although they all received maximum scores (+3) for Crashworthiness, they did not impact DOC, Safety, Reliability, etc. (Also, Crashworthiness received relatively low scores as shown in Tables 4.3 and 4.4.)
- (5) The figures of merit reflect relative benefits as opposed to benefit/risk. Hence, although the propulsion group reflects particularly high benefits to the user, the extremely high risk of development to the manufacturer is not accounted for.

69

C - Z

#### 4.5 EVALUATION TECHNIQUE RESULTS

This section discusses the final figures of merit obtained when Equation 4.1 was applied to the 56 final candidate technologies (FCT) identified in Table 4.7. The 17 category weights used here are those shown in Table 4.3 and are average weights. Technology groupings were established for convenience and have no bearing on the relative benefits or the figures of merit.

Each technology group is first discussed in general terms followed by a tabular ranking of those technologies which were considered to be in the particular group. A brief discussion of each technology then follows.

#### 4.5.1 Overall Technology Ranking

Tables 4.10 and 4.11 show overall rankings for the 56 and 51 technologies considered for Airplanes A and B respectively. Both lists are identical except for five propulsion technologies which were not considered appropriate for Airplane B.

#### 4.5.2 Aerodynamics

Nine technologies were evaluated and are shown in Table 4.12. As shown in Table 4.10, the first three technologies under Airplane A were among the top seven of 56 technologies. Winglets, when considered for a new wing, offer little benefit to the user of either airplane. Spoilers (implying full-span Fowler Flaps) appear rather attractive for Airplane B as shown in Table 4.11. Table 4.10. Overall Technology Ranking For Airplane A

а Ж AIRPLANE A

Pic or 17.326 14.191 12.677 57.924 54.217 31.767 31.140 29.0% 25.027 23.90 14.020 13.154 FL . 205 52.375 51.20 11.785 30-454 33.283 24..08 22.219 10.21 ÷2.453 24.203 25.534 -] CEIL WINY SPED DOC SION NGMT MOIS EFFC NOIS WAKL PRCE AGE BLTY GLY STAT TOLD CATGEORY MEIGHTS 5.43 5.82 7.92 7.91 1.61 4.34 4.29 8.16 7.04 7.35 8.51 7.29 8.85 6.11 9.42 6.24 6.87 ď 3 å 3.0 ... : ... ô. 5 <u>ہ</u> 2:0 å • ċ 5 °. 2.0 2 0.2 2.0 å 9 . å å ď å ď 1.0 2.0 -1.0 -0.5 -1.0 ŝ j ð .. ••• 2 1.0 0.5 ? ... 2 .0 2.0 0.5 0.5 • • 5 5 • -1.0 -1.0 -1.0 -2.0 -0.5 1.0 -2.0 -0.5 -1.0 -0.5 -0.5 1.0 -3.0 -1-0 -2.0 -1.0 -0.5 5.0 -2.0 1.0 -0.5 5.0 -2.0 • • **.** • 0.5 . . . 5 . 0 • • ... 1.0 2 0.S 2.0 • . 2.0 ... 2.0 2 20 2 ?: . 2 ... . å • • **.** å å • 0.1 ... 3 2.0 • • • • • . • . : . • . . . • 1.0 3.0 1.0 ••• ű.t 3.0 -1.0 2.0 3.0 -0.5 -0.5 0.5 <u>.</u> 5.0 -0.5 . . . • • 6 6 ••• 2.0 • . • . • • . • ځ . 5 5 • \*\*\*TABULAR RESULTS FOR ALL TECHNOLOGIES FOLLOW: 1.0 2.0 2.0 1.0 .. 0.5 0.5 2.0 **°** 0.5 5 5.0 0.5 0.5 2.0 • • • • . . 6 • 1.0 •• 2.0 **5.0** <u>م</u> • • ? •••• 0.5 . . • • . • • • 3.0 3.0 3.0 • • • • 6 . 5 6 2.0 ••• ... -0.5 2.0 **2.**0 **~~** 1.0 • . • • 6 5 ċ ď • ő 6 6 6 . INTEG AVIONICS AND BSPTS IMPR STLL/SPN--AERO TLRG STRT CHG ROT COMB ENGINE LOW/HEDM SPEED AIRFOLLS 2 STR ABV BIESEL ENGINE ACTIVE STALL PREVENTION INTEG LOW-COST NG LVLER ITTOVED TURBCCHARGING SEPARATE SFC TECHNOLOGY ENERGY ABSORBING FLOOR MAT LAM FLOW AIRFOILS QUIET EFFICIENT PROPS FIBERGLASS COMPOSITES LEADING EDGE DEVICES **BYS STATUS DISPLAYS** LCAD LIMITING SEATS IMPROVED RESTRAINTS GRAPHITE COMPOSITES DIGITAL DATA LINKS KEVLAR CONPOSITES LIBUID COOLING FOULER FLAPS GATE ENGINE 5P01LERS

Table 4.10. Overall Technology Ranking For Airplane A (concluded)

~

STRAT CHG RECIP ENGINE	0.	0.	0.	1.0	2.0	-0.5	٥.	1.0	0.	01.0 0.5	0.	0.	0.	0.	0.	12.235
CAT DISPLAYS	0.	0.	0.	0.	0.	0.	0.	0.	<b>c.</b>	2.0 -2.0 0.	1.0	0.	0.5	0.	0.	** . 233
DIRECT LIFT CONTROL	0.	0.	٥.	0.	0.	-0.5	0.	0.	0.	1.0 -1.0 0.	-0.5	1.0	1.0	0.	0.5	11.206
HCRLB BICIP ENGINE	0.	0.	0.	0.5	2.0	0.	0.	0.5	C.	00.5 0.5	0.	0.	0.	0.	0.	13.631
MICROWAVE LANDING SYSTEM	0.	0.	0.	0.5	0.	0.	1.0	0.	٥.	01.0 0.	0.	C.	1.0	<b>e.</b>	0.	9347
NAVSTAR/GPS	0.	0.	ο.	0.	0.	0.	٥.	C.	0.	1.0 -1.0 0.	0.	0.	1.0	0.	0.	8.256
BRST/TEAR RESIS FUEL TKS	0.	3.0	0.	0.	0.	-1.0	0.	0.	0.	01.0 0.	Q.	0.	0.	0.	0.	7.593
ICEPHOBIC SEC COATINGS	-0.5	0.	0.	-0.5	0.	-1.0	٥.	0.	0.	01.0 -0.5	5 O.	0.	3.0	0.	B.	5.096
ACTIVE LAMINAR FLOW CTL	2.0	0.	2.0	-2.0	0.	0.	٥.	2.0	0.	03.0 3.0	-3.0	٥.	0.	6.	1.0	3.841
INTEGRATED YAW DAMPER	0.	0.	s.	C.	٥.	0.	0.	0.	0.	1.0 -2.0 0.	0.	2.0	c.	0.	0.	2,539
MICRO HUD	0.	0.	٥.	-0.5	0.	-0.5	э.	0.	0.	2.0 -2.0 0.	-1.0	0.	2.0	0.	0.	1.535
AC ELECTRICAL SYSTEMS	0.	0.	0.	0.	0.	0.	0.	0.	0.	0. 0. 0.	0.	0.	0.	0.	0.	0.
ACT CTLS FOR RLX STOLTY	0.5	0.	0.5	1.0	0.	1.0	٥.	1.0	0.	03.0 1.1	-1.0	0.	0.	0.	0.	-0.030
DUCTED PROPULSORS	-0.5	0.	-1.0	-1.0	0.	-1.0	3.0	-1.0	3.0	01.0 -2.0	0.	9.	0.5	0.	2.0	-1,705
SGI LEVER THRUST/DRG CTL	0.	0.	0.	0.	0.	0.	э.	0.	9.	2.0 -1.0 0.	-1.0	0.	0.	٥.	0.	-2.671
VINGLETS	0.5	0.	0.	0.5	0.	-2.0	٥.	0.5	0.	01.0 0.5	i 0.	0.	0.	0.	0.	-2.384
FLUIDIC AUTO FLT CTL SYS	-0.5	0.	0.	0.	0.	-9.5	0.	-0.5	<b>0.</b>	01.0 -0.9	1.0	0.	0.	6.	0.	-12.200
LOW-BRAG SFC COATINGS	0.	0.	0.	-0.5	0.	-0.5	0.	0.5	0.	01.0 0.	٥.	0.	0.	0.	-0.5	-13.997
DIRECT SIDE FORCE CTL	0.	0.	0.	-0.5	٥.	-0.5	0.	0.	0.	0.5 -3.0 0.	-0.5	3.0	0.	٥.	0.	-14.098
ACT RIDE SMOOTHING	0.	٥.	0.	-0.5	0.	-0.5	Ο.	0.	0.	0.5 -3.0 0.	-0.5	3.0	0	9.	e.	-14.010
LASER GYROS	0.	٥.	0.	-0.5	0.	0.	0.	0.	0.	03.0 0.	1.0	0.	0.5	0.	C.,	-15.933
LOW-LEVEL PRESSURIZATION	0.	0.	0.	-9.5	0.	-0.5	0.	-0.5	2.0	01.0 -0.	5 -1.9	0.	0.	0.	0.	-17,141
HEADS-UP BISPLAY	0.	0.	0.	-1.0	0.	-1.0	0.	0.	0.	2.0 -3.0 0.	-1.0	0.	1.0	8.	0.	-22.525
LORAN C	c.	0.	0.	-0.5	0.	-0.5	٥.	0.	0.	01.0 0.	-0.5	0.	-0.5	٥.	0.	-23.7%
FIBER OPTICS (BATA TRNS)	0.	0.	٥.	0.	0.	0.	0.	0.	0.	03.0 0.	0.	0.	0.	0.	0.	-25.545
DREGA	0.	0.	0.	-0.5	0.	0.	٥.	0.	0.	01.0 0.	-1.0	0.	-0.5	0.	0.	-26.833
BOPPLER NAVIGATION	0.	٥.	0.	-2.0	0.	-0.5	٥.	0.	0.	1.0 -2.0 0.	-1.0	0.	0.	0.	0.	-36.517
ACT FLUTTER SUPPRESSION	0.	η.	0.	-0.5	0.	0.	0.	0.	0.	03.0 0.	-1.0	٥.	D.	٥.	0.	-30.353
ACT GUST ALLV	0.	9.	0.	-1.0	0.	0.	0.	0.	0.	03.0 0.	-1.0	0.	9.	0.	8.	-42.305
FLY-BY-WIRE	0.	0.	0.	-1.0	0.	-1.0	0.	0.	0.	03.0 0.	-1.0	0.	0.	0.	9.	-46.645
FLY-87-LIGHT	σ.	0.	٥.	- 1. 0	<b>0.</b>	-1.0	0.	0.	0.	03.0 0.	-1.0	0.	0.	0.	ø.	-45.645
INERTIAL NAVIGATION	0.	0.	٥.	-3.0	0.	-0.5	0.	0.	0.	1.0 -3.0 0.	-1.0	e	0.	0.	0.	-52,938

72

.

化合成化 医外间的 网络加加

Table 4.11. Overall Technology Ranking For Airplane B

AIRPLANE D

FIG OF 41.772 32.714 32.419 31.226 29.114 28- 952 25.531 23.597 20.554 20.527 18.646 57.942 48.097 34.18 23.786 55.85 38.340 25.323 20.843 18.350 16.13 2 1.0 2.0 ď . 4 å CATEGORY WEIGHTS 5.45 7.33 7.48 9.45 2.14 5.44 5.41 8.61 7.45 7.00 7.76 7.18 9.55 7.24 9.54 6.71 7.41 5 . 2 2 4 ď CEIL WINY SPED DOC SION WENT MOIS EFFC NOIS WAKL PACE AGE GLY GIT SFTT CHFT PEAF ď ð ď ď -0-5 ð ċ ð 0.S .. .. 2.0 5.0 . , 2 5.2 2.2 . 5 . å ď . å 0.5 2 2 ď 6 å ď . 6 ð -2.0 • à 9 9 ď ď -+ 0 5.5 -1.0 -1.0 2.2 5 2 ŝ • • 5 . . °. -0.5 -0.5 1.0 3.0 -1.0 1.0 -0.5 -1-0 -0-5 3.0 -2.0 2 2.0 -1.0 -0.5 . 0.5 0. . ð å ? . . . • . ć 3.0 • . • 0.1 2.2 3.0 -0.5 1.0 5.0 2 **0**.5 ••• 0.5 0.5 0.5 . 6 . 6 • 5 . 5 • 6 2.0 • • • ° • . • . . • . • • . • . . 2.0 3.0 3.0 -1.0 ? -1.0 5.5 0.5 ? • ŝ • . ð • **.** • 5 • • , , . • • ; • • • 5 • • 5 • • . • . \*\*\*TABULAR RESULTS FOR ALL TECHMOLOGIES FOLLOW: 0.5 o. s ... °. 2.5 : 0.5 **°** ••• 2.2 -1.0 -0.5 0.5 .. 3 • • • 5 6 5 ď ••• 1.0 0.5 ••• 0.0 ••• 0.5 0.5 • • . • • . . • . . . . 3.0 3.0 **.** . • **;** . 2.0 2.0 -0.5 2.0 ... 1.0 0.5 • 5 • • 6 • • • ď . • 5 • • INTEG AVIONICS AND DSPTS STAT CHG ROT COMB ENGINE ALCROWAVE LANDING SYSTEM LOW/MEDR SPEED AIRFOILS SEPARATE SFC TECHNOLOGY ICEPHOBIC SFC COATINGS FIBERGLASS COMPOSITES QUIET EFFICIENT PROPS MAT LAM FLOW AIRFOILS LEADING EDGE DEVICES STS STATUS DISPLATS LOAD LIPITING SEATS GRAPHITE COMPOSITES IMPROVED RESTRAINTS *<b>CIGITAL BATA LINKS* KEVLAR COMPOSITES BUCTED PROPULSORS FOULER FLAPS CRT DISPLAYS GATE ENGINE SPOILERS

73

Table 4.11. Overall Technology Ranking For Airplane B (concluded)

 $F_{e}^{i}$ 

-57.013	ġ	ò		5	-1.0	ċ	.0 -3.0	,	•	•		9	•	-3.0	0	•	•	INERTIAL MAVIGATION
-45.180		ċ	•	•	-3.0	1.0	3.0		0 0	-	•	6	•	-3.0	•	•	2.0	ACTIVE LAMINAR FLOW CTL
- 39. 602	•	ò	•	•	-1.0	•	.0 -2.0	,	•	•		Ģ	•	-2.0	•	•	0	DOPPLER MAYIGATION
-37.556	•	ď	•	•	-1.0		3.0	•	•	•	0	9	•	-0-2	•	•	<b>.</b>	ACT FLUTTER SUPPRESSION
-35.043	<b>.</b>	ď	•	1.0	-1.0	•	3.0		•	•	•	0		1.0	<b>.</b>	•	•	ACT GUST ALLEVIATION
-52.734		ð	ď	5	ď	å	13.0		•	•	6	•	•	-1-0	•	0	•	ドレマーロアージェルビ
-31.577	.0	6	-1-0		-1.0	6	11.0	,	•	•	<b>°</b>	•	•	-0-3	•	•	6	GREGA
\$84*42-	•	<b>°</b>		5	•	ò	3.0	е •	•	0	- -	0	•	-1.0	<b>.</b>	•	5	FLY-8Y-LIGNT
-29-632	<b>.</b>	6	-1.0		-0-5	6	11.0		•			ġ	•	-0-3	•	•	5	LORAN C
-24.396	0	6	1.0	•	-1.0	•	.0 -3.0		•	•	0		•	-1.0	•	6	5	HEADS-UP DISPLAT
-23,263	9	6	•	•	å	5	3.0		•	•	<b>.</b>	0	•	•	•	•	•	FISER OFTICS (DATA TRNS)
-17.652	<b>.</b>	5	•	•	-1.0	-0.5	11.0	o,	~ ~	<b>?</b>		0-	•	-0.3	•	•	•	LOW-LEVEL PRESSURIZATION
-15.130	 	ċ	•	•	ď	ò	0.5	,	•	•	5 0	0-	•	-0-3	•	•	5	LOW-BRAG SFC COATINGS
-15.489	ď	•	0.5	•	1.0	ò	13.9		•	G	<b>5</b>	ò	•		<b>.</b>	•	5	LASER GTROS
-11,048	ġ.	0	5	3.0	-0.5		3.0		•	•	ò	<b>°</b>	•	-0-3	•	•	5	DIRECT SIDE FORCE CTL
-3,733	•	ò	5	•	ď	0.5	0.10	•	••		0	~	•	0.5	<b>.</b>	•	0.5	¥146LE75
-3.312	å	ò	ò		-1.6		0-1-0-1		•	0	ö	•	•	•	•	•	5	SGL LEVER THRUSTIDRG CTL
-1.070	<b>°</b>	ò	•	ď	1°C	-0.5			~ ~	?	0	9	•	•	•	•	-0-5	FLUIDIC AUTO FLT CTL SYS
3.209	•	ð	2.0	•	-1.0	•	.0 -2-0		•	•	ė	ò		-0-3	•	•	•	41C40 MU8
3.491	•	<b>.</b>	•	3.0	-0.5	•	1.0 -2.0	-	•	•	<u>.</u>	6	•		•	•	6	ACT RIDE SMOOTNING
4.529	o,	ð		5	-1.0	1.0	13.0		•	-		2	•	1.0	0.5	•	0.5	ACT CILS FOR RLX STOLTY
5.952	°.	5		2.0		•	.0 -2.0		•	•	<b>ы</b>	9		6	9			INTEGRATED YAN DARPER
7.741	<b>.</b>	0	•	5	ö		1.0		•	0	• •	•	о.	5		•	0	INTEG LOW-COST NG LVLER
8.581	e,	•	•	5		•	0.10		•	°	0		•	6	•	3.0	5	BAST/TEAA RESIS FUEL TKS
8.761	<b>.</b>	ď	1.0	•	9	•	1.0 -1.0	-	•	•	ò	9	•	6	<b>.</b>	•	<b>.</b>	KAVSTAR/GPS
12.120	0.5	•	1.0	1.0	-0.5	9	1.0 -1.0		•	•		0	•	6	•	•	5	DIRECT LIFT CONTROL
15.207	<b>.</b>	•	2.0		ď		10.5		•			ò	0	5	9	6	•	IMPR STLL/SPNAERO TLRG
15.290	<b>.</b>	•	•	•	;	°	0.5		с ,	•		9		ò	•	3.0	5	ENERGY ABSONDING FLOOR
16.097	å	ð	3.0		ď			•					•	\$		•	5	
16.904	å		•	•	<b>.</b> . 2.9	5	•		•	•	<b>.</b>	ġ	•••	¢	¢			ACTIVE STALL PREVENTION

	Figures of Merit for						
Technology (9)	Airplane A 56/-53/85*	Airplane B 51/-57/58*					
Natural Laminar Flow Airfoils	58	25					
Spoilers	57	38					
Fowler Flaps	48	24					
Low/Medium Speed Airfoils	29	31					
Leading Edge Devices	24	29					
Improved Stall/Spin Aerodynamic Tailoring	24	15					
Active Laminar Flow Control	4	-46					
Winglets	- 3	- 4					
Low Drag Surface Coatings	-14	-15					

Table 4.12 Aerodynamics

\*Total number of technologies/lowest FM/highest FM

Ľ.

4.5.2.1 <u>Winglets</u>. The advantages of winglets are not yet clear. Although they promise to reduce induced drag while increasing wing bending moments when applied as a modification to existing aircraft, there is little data to suggest that they would offer improvements to aerodynamic efficiency that could not be gained through increased aspect ratio and wing twist. Difficulties involving flutter and large sideslip angles require further study. Reference 156 describes theoretical improvements in range of 6% to 8% over a similar wing without winglets. Although this article cites large stabilizing moments in yaw at sideslip angles of  $6^{\circ}$  to  $8^{\circ}$ , the addition of winglets also resulted in a weakly divergent dutch roll mode.

The baseline against which this technology was evaluated for both Airplane A and Airplane B was a high-aspect-ratio, twisted wing. Small benefits in cruise performance were offsat by weight penalties to compensate for substantial increases in root bending moments.

4.5.2.2 Spoilers. Spoilers alone offer only small advantages to general aviation aircraft by eliminating adverse yaw. However, they offer the promise of major improvements in performance by freeing the entire wing trailing edge for high lift devices (full span flaps). As mentioned previously in Section 4.3.2, this additional maximum lift capability can be used to either improve takeoff and landing performance or, at the other extreme, improve cruise performance by increasing wing loading (W/S). Since increased wing loading offers substantial benefits in fuel efficiency and ride qualities with only a small penalty (if any) in cruise ceiling, this technology is evaluated from the standpoint of increased W/S. It should be noted that since typical single engine aircraft operate at W/S  $\simeq$  730 to 1220 N/m<sup>2</sup> while commuters operate at W/S  $\approx$  2390 to 2870 N/m<sup>2</sup>, Airplane A will reflect larger impacts from this technology (W/S for heavy transports are typically on the order of 4790 to 6700  $N/m^2$ ).

Although cruise performance promises to reflect substantial gains, the incorporation of spoilers does have attendant risks.

Nonlinear control response, sluggishness, control reversal (due to flow reattachment), increase of authority with full span flaps deployed, and loss of control authority at negative angles of attack are but a few which are identified in the literature. Laundry, however, has expressed the view that the limited use of spoilers is attributable not so much to the lack of spoiler handbook data, but rather to the vast amount of aileron handbook data available (Ref. 121).

Still unavailable is a verified thin-airfoil method for predicting 2D characteristics of a wing-spoiler-slotted-flap configuration. However, Parkinson has reported on research using a numerical thick airfoil method utilizing a 2-source model to predict  $c_{l}$ and  $c_{l}$  vs.  $\alpha$  for an airfoil with a spoiler and slotted flap  $m_{o}^{0}$  (Ref. 159).

Furthermore, Reference 111 reports favorably on flight test results of spoiler characteristics on a modified Cessna 177 Cardinal (Redhawk program) and Reference 110 does likewise for spoilers on a modified PA34-200 Piper Seneca (ATLIT program). Reference 183 provides an overview of the state of the art of general aviation spoilers in 1974.

4.5.2.3 Fowler Plaps. Fowler flaps offer significant improvements in maximum lift coefficient ( $C_L$ ) when compared to the plain, max split, or single slotted flaps which are typical of the single engine general aviation (GA) fleet. However, their real benefits are more fully exploited when used as full span flaps in conjunction

with spoilers. Although this idea is not new (References 80, 122, 224), difficulties associated with integrating the spoilers have delayed the widespread use of this concept.

Paulson (Ref. 161) reports an increase in  $C_{L}$  of 96% when max Fowler flaps were fully deflected on a wind tunnel model wing. In a comparison between plain, slotted, and Fowler flaps, where the plain and slotted flaps had flap spans approximately 50% of the wing semi-span while the Fowler flaps had flap spans approximately 75% of the wing semi-span, Paulson (Ref. 162) reports  $\in C_{L}$  for max the Fowler flap approximately 48% greater than the plain or slotted flap. Under the conditions of this latter investigation, the Fowlerflapped configuration could either land at a speed 18% lower than its counterpart or reduce its wing area by 33%. Wentz (Ref. 224) reports a  $C_{L}$  = 3 for a one-quarter scale ATLIT wing equipped with max full span Fowler flaps.

For the evaluation at hand, Airplane A is presumed to have plain or single slotted half span flaps and Airplane B is presumed to be equipped with half span Fowler flaps. The anticipated effects of full span Fowler flaps is reflected in the evaluation. Hence, this technology is considered to be intergrated with spoilers. 4.5.2.4 Low Drag Surface Coatings. This technology is described in Reference 22, where an investigation was conducted by the Boeing Commerical Airplane Company of nine liquid coatings and 60 film/ adhesive systems. This investigation is part of the Energy Efficient Transport (EET) element of the Aircraft Energy Efficiency (ACEE)
program and was performed under contract to NASA-Langley Research Center. Three liquid coatings and four film/adhesives are currently undergoing further testing and evaluation.

Motivation for this research was supplied by the 1973 fuel crisis, and preliminary research by NASA of a T-33 wing reflected a 12% reduction in drag when skin joints, hinge lines, etc. were covered with a smooth thin film. Boeing's research indicates, however, that only a 1.6% reduction in drag coefficient ( $C_D$ ) can be expected due to the existance of unsealed gaps for high lift devices and control surfaces. Still, this 1.6% reduction translates to a fuel savings of 128,690 % (34,000 gals) per Boeing 727 per year.

Analysis of the data reveals that although almost all of the drag associated with roughness is eliminated from wing and tails, only 25% of the drag associated with gaps was eliminated. Hence, application of a low drag surface coating to a wing with protruding rivets makes little sense. From a mission profile consideration, Airplane A could benefit more from this technology than Airplane B because its mission is cruise dominated whereas Airplane B's is characterized by more time in climbs and approaches where dragdue-to-lift is usually more significant than profile drag.

However, even if one considers an advanced airplane characterized by flush rivets, bonded surfaces, or composite construction, the propulsion system will invariably include a propeller, and system efficiencies will dictate lower cruise speeds closer to

 $L/D_{max}$  than for a turbojet or turbofan counterpart. Realizable savings due to small reductions in profile drag are probably more than offset by penalties in weight, initial cost, and DOC for both Airplane A and B.

i i i Li i

> 4.5.2.5 <u>Active Laminar Flow Control</u>. This technology is receiving renewed interest in light of soaring fuel costs. As Shevell points out in Reference 191, nothing else has the potential of reducing drag to as great an extent. An example in his article, based on an aircraft operating at Reynolds numbers of 20 to 70 x  $10^6$ with 200 passengers and a range of 5500 nautical miles, illustrates his point. If this vehicle operated with 75% of its wing and tail surfaces effectively laminarized, drag would be reduced by 30% and L/D would increase by 30%. For a similar type of vehicle (N=.8, 200 passengers, JT9D engines), Maddalon points out that laminar flow control (LFC) energy requirements decrease as AR increases above 7 (Ref. 129). For this type of air carrier operation, DOC decreased despite a 17% penalty in maintenence costs and a 3% penalty in purchase price.

Although LFC offers a far greater potential for parasitic drag reduction than low drag surface coatings, it still promises significant penalties for Airplane A and Airplane B. As with all systems which reduce parasitic drag, the benefits realized by a propeller-powered airplane may be expected to be less than for a jet or fan powered vehicle. Also, fuel costs represent a significantly smaller percentage of DOC for Airplane A, and the

mission profile of Airplane B does not allow it to take full advantage of the system. The low altitude environment of Airplane B and the associated higher susceptibility to insects, dust, and other debris suggest higher maintenance costs for Airplane B than for a high-altitude cruise vehicle.

It should be reiterated at this point that one factor which this evaluation process significantly does not measure is risk. For the case at hand, an airplane optimized for LFC operation from the beginning of the design process will have markedly different characteristics than a non-LFC counterpart. Degradation of the LFC system promises significant (although yet unquantified) penalites in performance and operating costs.

4.5.2.6 Leading Edge Devices. These devices, which are found on almost every contemporary commercial jet liner, are only rarely found in GA aircraft. Fixed devices offer the same low performance advantages as retractable Krueger flaps or slats. However, cruise performance is degraded in exchange for the simplicity of a fixed system. Reference 111 presents some flight test results of a Krueger system installed on the Redhawk. Although they displayed excellent characteristics ( $\Delta C_{L_{max}}$  = .59 for the Krueger flaps with Fowler flaps deployed 40°), it was suggested that such a system was probably too heavy and too complex for a light GA single engine aircraft.

In exchange for the increase in  $C_{max}$ , W/S could be increased even further to improve cruise performance. However, these benefits would be offset by a probable degradation in ceiling.

An interesting application of a retractable leading edge device would be in their use on the outboard section of the aft wing of a canard configured vehicle. Such an application could offset the effects of increased upwash from the canard during low speed high angle-of-attack operations.

4.5.2.7 Advanced Low and Medium Speed Airfoils. These airfoils are characterized by turbulent flow conditions and were initially designed as low speed airfoils to provide low cruise drag, high climb lift-to-drag ratios, high maximum lift, and well behaved stall characteristics. Results of these efforts are documented by McGhee, Beasley and Whitcomb in Reference 139 and are shown in Figures 4.3 and 4.4, where the LS(1) series encompasses both the GA(W)-1 and GA(W)-2. As shown, these low speed airfoils have better  $c_{\ell}$ than more conventional NACA sections and work done to reduce pitching moments of a 17% thick airfoil as well as increase the lift-to-drag ratio of a 21% thick airfoil. These low speed airfoils all have a design  $c_{g}$  of 0.4. Two medium speed airfoils are described where  $c_{l} = .3$ ,  $R_{e} = 14 \times 10^{6}$ , and M = .72and .68. The design family of airfoils is shown in Figure 4.5. These airfoils, however, are characterized by higher pitching moments and higher drag than, for example, 6-series airfoils. Accordingly, an effort by Hicks and Schairer to increase the maximum lift coefficient of 632-215 airfcil is described in Reference 84. Figure 4.6 shows the modification and Figure 4.7 shows one of

several figures depicting typical results. This may be compared to Figure 4.8 from Reference 139 which shows low and medium speed airfoil data.



Fig. 4.3.  $c_{\ell}$  vs t/c of NASA Low-Speed and NACA Airfoils.



l

Fig. 4.4. Effect of t/c on  $c_{\ell}$  and  $\ell/d$  for NASA Low-Speed Airfoils. max

One difficulty which is posed for the thinner airfoils  $(t/c \approx .13)$ is that these airfoils do not lend themselves to easy configuration integration in GA aircraft. Thicker airfoils  $(t/c \approx .17)$  offer more wing volume for fuel and control linkages. Also, they suffer a smaller weight penalty in construction.



'Fig. 4.5. Design Family of NASA Low-Speed Airfoils.



Fig. 4.6. Modification to NACA 63<sub>2</sub>-215 Airfoil.



Fig. 4.7. Experimental Results of a Modification to an NACA  $63_2$ -215 Airfoil.

Difficulties with implementation of the LS(1) series into a configuration are suggested by the highly aft-loaded airfoil pressure distribution and suggest that a moderate amount of tailoring may be required at the wing-body junction to preclude premature flow separation of both the wing and fuselage in this region.

For the technology evaluation, it was assumed that neither Airplane A or B employed low or medium speed airfoils. Airplane B should suffer a smaller penalty in parasitic drag since its mission profile is more dominated by climbs and approaches where drag-dueto-lift is more dominant than in cruise.



Fig. 4.8. Experimental Results of Low and Medium Speed Airfoils.

4.5.2.8 Advanced Natural Laminar Flow Airfoils. Present efforts to develop a natural laminar airfoil are aimed at achieving a  $c_{l_{max}}$ at least as great as that associated with the GA(W)-2 while maintaining pitching moment coefficients lower than those of the same airfoil. Preliminary work suggests sectional reductions in drag by as much as 25% at design  $c_{g}$ . Improvements in performance are offset by a requirement to maintain a smooth surface, and relatively severe penalties in drag appear with the introduction of roughness. This suggests that flush riveting, bonding, or composite materials may be required to eliminate protruding rivet heads. Also, the user may expect to be confronted with a requirement to maintain a relatively insect-free surface.

This technology proved fairly difficult to evaluate. Airplane A could benefit rather significantly from this technology, but purchase cost penalties could outweigh benefits if a baseline of present manufacturing processes is upgraded to total flush riveting. For purposes of this evaluation, it will be assumed that this technology will be integrated with a composite wing since the latter offers reduced manufacturing costs. In the absence of an insect-free surface technology, Airplane B will encounter DOC expenses and reliability penalties associated with its more severe operating environment. 4.5.2.9 <u>Improved Stall/Spin Characteristics Through Aerodynamic Tailoring</u>. Solutions to stall/spin accidents range from better pilot stall recognition training to better designed vehicles. This evaluation addresses only the latter solution (in a rather restricted sense.

Anderson notes that solutions to the design problem appear to lie in (1) providing good handling qualities up to and beyond maximum lift, (2) making the aircraft spin resistant, and (3) providing stable static and dynamic stability characteristics with

good control about the pitch, roll, and yaw axes (Ref. 4). Three approaches which appear feasible are:

- Control stall progression on the wing to provide greater poststall roll damping and natural buffet stall warning.
- (2) Limit longitudinal control power to prevent complete wing stall.
- (3) Minimize adverse cross coupling by automatic (SAS) means.

Item 2 is addressed by attempts to reduce the relative horizontal tail power through (1) tailoring, (2) active controls (Ref. 29) or (3) employing a canard configuration (Ref. 185). Item 3 has already been successfully demonstrated by the military. Certainly, separate surface technology as discussed elsewhere in this paper should be investigated. Artificial stall warning devices appear to provide significant contributions, and Ellis notes that a large portion of stall, mush, and spin accidents occur in airplanes which are not equipped with these systems (Ref. 49). Stick shakers appear to be the most effective warning device. Aural warnings also provide a certain margin of safety, but visual-only angle-ofattack indicators appear ineffective.

Having discussed certain problems and solutions which appear in the literature, emphasis is now focused on methods of controlling stall progression through aerodynamic tailoring. Feistal, Anderson, and Kroeger have reported on promising results based on a leading edge discontinuity as illustrated in Figure 4.9 (Ref. 85). Figure 4.10 depicts an essentially flat-topped lift curve out to



Fig. 4.9. Leading Edge Modification for Stall/Spin Alleviation

an angle-of-attach of 32°. Significantly, rolling moment excursions as well as yawing moments were acceptable throughout the range of angle-of-attacks tested. Hicks and Henne have reported on a numerical optimization technique where a low speed wing (GA(W)-2 section) could theoretically be modified to move the initial stall to the wing root without affecting either span loading or induced drag characteristics (Ref. 83).

For this study, only the technologies representing efforts to control separation are evaluated. Both Airplane A and B achieve substantial benefits in safety at little cost.



Fig. 4.10. Flat-Topped Lift Curve from Leading Edge Modification for Stall/Spin Alleviation.

#### 4.5.3 Aircraft Systems

Only two systems as shown in Table 4.12 were evaluated. Surprisingly, Icephobic Surface Coatings did not fare as exceptionally well as one might expect. Here, minor penalties in weight (which affected ceiling and range to a lesser extent) and cost almost totally negated the +3 relative benefit awarded for safety. Airplane B appears to benefit substantially when pneumatic boots or hot air systems are replaced.

Technology (2)	Figures of Merit for	
	Airplane A 56/-53/85*	Airplane B 51/-57/58*
Icephobic Surface Coatings	5	33
AC Electrical Systems	0	17

### Table 4.13. Aircraft Systems

"Total number of technologies/lowest FM/highest FM

4.5.3.1 Icephobic Surface Coatings. Very little was found in the literature on icephobic surface coatings. Although Reference 18 mentions olefin plastics on helicopter blades, no mention is made of the success or failure of this research. Certainly, the rewards obtainable from a surface coating of this type demand further research. When one consideres that vertical and horizontal tails, wheel pants, struts, etc., could all be protected from ice accumulation in addition to the more conventional and limited GA appliccation to wing leading edges only, the potential benefits of such a system become staggering. It is suspected that in-house, unpublished research has already been performed by interested agencies. However, no information relating to efforts or results surfaced in the data gathering trips performed as part of this research.

For this technology evaluation, the baseline for Airplane A is not considered to be equipped with an anti-icing system, while the baseline for Airplane B is presumed to be equipped with a pneumatic system consisting of either boots or ducted hot air.

Airplane A is penalized for weight and cost but receives a higher benefit in safety. On the other hand, Airplane B receives an improvement in reliability, cost, and weight with only small improvements in safety.

4.5.3.2 <u>AC Electrical Systems</u>. Major research activity in shifting from DC to AC electrical systems was not detected either in the literature or during data gathering trips. However, at least one manufacturer cited a potential for large weight savings in their aircraft product line.

For the present, neither Aircraft A's nor Aircraft B's technology baselies is expected to benefit to any great extent. Airplane A is characterized by limited weight penalties associated with wiring, while Aircraft B is known to favor panel-mounted vs remotemounted avionics due to cost considerations (Ref. 167). Although newer commuters are usually equipped with autopilots and flight director systems, it is suspected that the majority of operational commuters are still hand flown on conventional round dial instruments with cross pointers. Hence, the potential weight savings are not as great as for a current business jet.

#### 4.5.4 Crashworthiness

Four technologies as shown in Table 4.14 were investigated. These technologies are considered to be extremely attractive. As explained earlier, their figures of merit are relatively low because they affect only a few categories and specifically do not influence DOC, Safety, or Reliability. Also, as shown in Table 4.3 and 4.4,

the category weighting for Crashworthiness was 6.816 (#11 of 17) for Airplane A and 7.333 (#9 of 17) for Airplane B.

Technology (4)	Figures of Marit for	
	Airplane A 56/-53/85*	Airplane B 51/-57/58*
Load Limiting Seats	17	19
Energy Absorbing Floor	14	15
Improved Restraints	14	15
Burst/Tear Resistant Tanks	8	9

Table 4.14. Crashworthiness

\* Total number of technologies/lowest FM/high-st FM

4.5.4.1 Load Limiting Seats. Efforts to improve occupant survivability in the event of a crash are becoming more refined with the correlations of test data obtained from the NASA Impact Dynamics Research Facility (IDRF) and the FAA Civil Aeromedical Institute (CAMI) with analytical data from a modified computer algorithm known as MSOMLA (for Modified Seat-Occupant Model for Light Aircraft). Fasanella and Alfaro-Bou note good agreement between CAMI sled data and full scale results from the IDRY. MSOMLA (which uses a spring-damper model as opposed to the finite element model in SOMLA) also showed good agreement (Ref. 53).

Having noted that there is much which can be done towards Improving survivability through improved seat design, three different types of improved seats as shown in Figures 4.11 (b), (c), and (d) are being investigated. The wire bending load limiter is shown in Figure 4.11 (a).



Fig. 4.11. Load Limiting Seat Concepts.

During stroking, the wire loop is translated along the wire under a constant force of 4.4 kN (1000 1bf). Tests as of April 1979 indicate that the ceiling mounted seat reduced hongitudunal pervic accelerations by 40% during a 15 m/s (50 fps) sled pulse and vertical accelerations by 50% during a 13 m/s (42 fps) sled pulse (Ref. 53). This seat, which has a 9.1 kg mass, suffers from added installation weight and a possible loss of stroking distance resulting from cabin deformation at impact. From an applications/integration viewpoint, it appears that the floor mounted seat (10 kg or 23 1bm) may offer more promise. However, preliminary test results indicated a requirement for further development work for the floor mounted and rocker action seats. Thomson and Goetz (Ref. 210) note in a similar report that the human tolerance of 25 g's can be met with a 30 cm (12 in) stroke dissipated over 0.1 sec. Such a system places an upper bound of 12.2 m/s (40 fps) on the seats.

Since the typical general aviation seat of 11 kg (25 lb) dissipates energy through seat deformation and leg buckling, load limiting seats offer great potential for improving vehicle crashworthiness and occupant survivability.

4.5.4.2 <u>Energy Absorbing Floor</u>. Several schemes which offer promise in application are described in Reference 210 and depicted in Figure 4.12. Assuming a 15 cm (6 in) available stroke (through floor deformation), Thomson and Goetz point to an upper limit of 8.2 m/s (27 fps) which can be dissipated by the floor structure through controlled collapse. Despite information obtained from

airframe manufacturers which indicates that this type of occupant protection is already being introduced in a few product lines, widespread incorporation is not yet noted. Tests and data correlation with analytical models indicate that three computer codes (KRASH, ACTION and DYCAST) demonstrate good quantitative results in predicting dynamic vehicle response to impact loadings (Ref. 79). The continued development of these tools promises to give further insight to floor and cabin deformation. Fasanella and Alfaro-Bou report, for example, that IDRF data showed a GA aircraft cabin during a 27 m/s (89 fps) impact deforming and become 21 cm (8.3 in) wider and 23 cm (9 in) lower. The apparent upheaval of the center floor section was actually the result of the cabin side walls forcing the floor down at the wall junction. Incorporation of the energy absorbing floor into future GA aircraft takes on added significance in view of such recent test data.



Fig. 4.12. Energy Absorbing Floor Concepts.

Both Airplane A and B will gain significantly in crashworthiness from this technology at an undetermined cost in weight. (Note that all of the five concepts depicted in Figure 4.12 incorporate an energy-dissipating foam.)

4.5.4.3 Improved Restraints. When an analysis of business aircraft accident investigations showed that the use of shoulder belts reflected a marked increase in survival, the researchers concluded that this use is considered to be the single most important means of improving occupant crash impact protection (Ref. 194). This report also noted that restraints attached to the seat provided little protection when the seat failed and separated from the floor. Further research by others indicates that a single shoulder strap allows the occupant to roll out of the restraint in some accidents. Hence, a restraining system consisting of two shoulder straps, two lap belts, and crotch strap, all connected at a single point centered on the lap belts appears to offer major improvements in crashworthiness. (Fig. 4.13(a) depicts such a system.) Such a system would incorporate locking inertia reels in the shoulder harness and thereby afford the passenger more freedom of movement during normal operations than a diagonal single shoulder strap would. Tests to date indicate that a tensile force of .54 kN (122 1b) may be encountered in the harness system during a 27 m/sec (89 fps) impact. Hence, metal-to-metal buckles are preferred to metal-to-fabric cam systems.

This technology promises significant benefits to both Airplane A and B. Anticipated penalties lie in increased costs.



Fig. 4.13. Five Point Improved Restraint vs Conventional Restraint

4.5.4.4 <u>Burst/Tear Resistant Fuel Tanks</u>. Efforts to develop a crash/tear resistant fuel tank were encouraged by the success of Army helicopter programs. However, the tanks developed under the Army programs are too heavy and expensive for general aviation aircraft (Bergey notes in Ref. 18 that a 130 liter helicopter tank may cost on the order of \$10,000). In an attempt to simplify the system, ballistic protection was eliminated as a criterion, and 1-, 2-, and 2 ply tanks with frangible fittings were manufactured and tested in a full size airframe which was accelerated using a

catapult. Reference 48 hotes that the lightest tank which performed satisfactorily with no leakage after a 29 m/s (96 fps) impact was one manufactured by Uniroyal Corporation from a single ply with a fabric weight of 43.23 gram/m<sup>2</sup>. None of the 2- and 3-ply tanks failed, but two other (lighter) 1-ply tanks and the original tank did. Figure 4.14 shows the installed tank, and Figures 4.15 and 4.16 show the crash site and test where the test vehicle impacted an earthen hill equipped with sunken steel tubes and rock piles. Figure 4.17 shows the external damage done to the wing where the single ply tank did not fail. (Note that a system to shear the fuel line from the tank was incorporated, and the frangible fittings did not leak.)



Fig. 4.14. Modified Fuel Tank Installed in Wing.

WHINAL PAGE I



Fig. 4.15. Airframe on Catapult with Earthen Hill, Steel Tubes, and Rockpiles Visible.

Since annual GA accidents over the ten year period between 1966 and 1976 numbered from 4,200 to 6,100 with post crash fires reported in approximately 350 of the accidents annually, the potential for improved safety for GA aircraft offered by these tanks is significant.

Weight and cost appear to pose disadvantages for both Airplans A and B. The 1-ply test tank weighed 9.4 kg (21 lb) including the fittings while the original tank weighed 4.4 kg (9.6 lb). In spite of a volume penalty of 5.3 liters (1.4 gals) per tank, production tanks may be expected to have a volume difference of less than 3.8 liters (1 gal) from conventional bladder tanks. These weight and volume penalites are expected to be greater when compared to a wet wing concept. However, a wet wing may be expected to suffer from post crash impact failures at least as often as conventional bladder tanks.



Fig. 4.16. Typical Impact of Airframe Following Catapult Acceleration.



Fig. 4.17. Wing Damage Where a 1-Ply Tank Did Not Leak.

## 4.5.5 Flight Control Systems

Fourteen technologies as shown in Table 4.15 were evaluated for Airplanes A and B. Here, the integrated, low-cost wing leveler appears attractive for Airplane A (see Table 4.10) but does not appear so for Airplane B. The major difference for the ratings lies in the fact that Airplane A is not equipped with an autopilot while B is. When a similar scenario is offered for Separate Surface Technology, Airplane A's benefits to Pilot Workload and Ride Quality offset the initial cost and DOC penalties. Airplane B,

ORIGINAL PAGE IS OF POOR QUALITY on the other hand, suffers no penalty in costs (autopilot replacement) and smaller benefits in Pilot Workload and Ride Quality.

٦,

	Figures of Merit for	
Technology (14)	Airplane A 56/-53/85*	Airplane B 51/-57/58*
Integrated Low Cost Wing Leveler	39	8
Separate Surface Technology	22	24
Active Stall Prevention	13	16
Direct Lift Control	11	12
Integrated Yaw Damper	3	6
Act Ctls for Relaxed Stability	-0	5
Single Lever Thrust/Drag Control	-3	-3
Fluidic Automatic Flight Controls	-12	-1
Active Ride Smoothing	-14	4
Direct Side Force Control	-14	-11
Active Flutter Suppression	-38	-38
Active Gust Alleviation	-42	-35
Fly-by Wire	-47	-33
Fly-by-Light	-47	-30

# Table 4.15. Flight Control Systems

\* Total number of technologies/lowest FM/highest FM

Active controls, fly-by-wire, and fly-by-light are extremely unattractive due to cost, where the flight environment considered for Airplane A and B (low cruise speed) does not take full advantage of the benefits available to aircraft with high cruise speeds.

4.5.5.1 Fly-Ry-Wire. This technology is considered to be fairly well developed but very costly. Papers presented at an AGARD Conference in 1974 (Ref. 25 and 57) noted then that the methods of designing a fly-by-wire (FBW) system were well understood but that safety considerations alone (without regard for certification requirements) cause the systems to be very complex and expensive. The economic justification for such a system is very much dependent on the aircraft mission. Hence it is difficult to visualize their incorporation into the GA fleet within the next ten years, especially when one considers the availability and cost of maintenance for such systems. Some points regarding such a system deserve mention, however. For example, the integration of FBW is considered mandatory if and when active flight controls prove economically justifiable. Such flight controls include those designed for relaxed longitudinal static stability, gust alleviation, and flutter suppression. All of these technologies promise weight savings and improved operational efficiencies which lower direct operating costs (DOC) for commercial air transport. An intriguing research program which is currently being pursued by the NASA/Langley Research Center together with Princeton University involves Langley's Digital Avionics Research system (DARE), and Princeton's Avionics

Research Aircraft (ARA). The ARA is a fully instrumented, fivedegree-of-freedom, FBW, low-wing, single engine GA aircraft. This vehicle, which has already been used to investigate flying qualities, human factors, and control, will be used with DARE to expand on the above investigations as well as those dealing with advanced digital control concepts (Ref. 45).

4.5.5.2 <u>Fly-By-Light</u>. This technology complements the capabilities and possible active controls applications of a FBW system because of the following characteristics attributable to fiber optics:

- (1) Quicker data transmission than possible with wires.
- (2) Large bandwidth capability offers the potential for replacing several wires with one fiber.
- (3) Fibers are non-inductive and non-conductive.
- (4) Provides better signal isolation than wires by decreasing "crosstalk."
- (5) Does not present either electrical or fire hazards.

However, fly-by-light (FBL) systems represent a somewhat significant cost penalty over FBW systems. As much, their integration into the general aviation fleet appears even more remote than FBW systems. At the same time, it should be noted that Bell Helicopter has logged several hundred flight test hours on their Model 206 equipped with a fiber optic yaw SCAS which includes an optical encoder, liber optic link, and an optical receiver and decoder. Bell also employed five complete fiber optic systems in the control of the sw shplate of an iron bird configuration

demonstrator. In this configuration, failure of three systems still allowed control authority over the main rotor. 4.5.5.3 <u>Active Gust Alleviation</u>. This technology could offer potentially large performance benefits resulting from reduced structural weight for those high speed aircraft which operate out of short fields on the order of 610 m (2000 ft). These aircraft will require relatively low wing loadings on the order of 1.9  $kN/m^2$ (40 lb/ft<sup>2</sup>), and structural weight penalties may be incurred when decreased field length capabilities are sought. Active gust alleviation holds promise for configurations subjected to gust load factors on the order of five or above. When integrated to reduce structural weight (and strength), however, the system becomes safety-of-flight critical and requires costly redundancies and certification testing. A unique passive system for single engine GA aircraft is described in Reference 173 where auxiliary aerodynamic surfaces were attached to a Cessna 172 to sense angle-of-attack and drive the flap system through a direct linkage. Although this system successfully attenuated gusts up to 3 m/s (10 fps) in the frequency range between that of the phugoid and short-period mode, it did so at the expense of reducing  $\rm C_m$  . Consequently, further studies where a linkage to an endow opentrols can be investigated appear warranted. An important point to be made here is that gust alleviation becomes attractive when significant structural weight savings can be realized. Since these savings usually represent the elimination of penalties incurred through strength requirements which are dictated by

adverse gust loads in cruise, any system or technology which increases C<sub>L</sub> of a wing (and thereby allows for higher W/S) will max probably be more cost effective for the low-speed GA fleet. Active gust alleviation is not expected to offer quantifiable benefits to either Airplane A or B due to the relatively low cruise speeds (low gust factors) of these vehicles.

4.5.5.4 <u>Active Ride Smoothing</u>. This technology is differentiated from that of active gust alleviation in the sense that this system does not provide for reduced structural weights. As such it is not safety-of-flight critical.

Primary benefits result from improved handling qualities and ride comfort. Reference 34 suggests that additional cost penalties for a commercial transport would be approximately 2% to 5% of total testing and certification expenses of a new vehicle with an integrated system. Retrofit of the system would incur higher cost penalties. Feasibility studies using a deHaviiland DH6 have been performed and indicate that total system weight should not exceed 2% of the aircraft's gross weight while total power requirements would not exceed 0.3% of total engine power. However, an effective ride smoothing system will require relatively large direct lift and side force surfaces located near the aircraft center-of-gravity (Ref. 34). No major reliability or maintenance problems are forseen.

For the present evaluation, both Aircraft A and B will incur cost penalties in exchange for improvements in ride and handling qualities.

4.5.5.5 <u>Active Flutter Control</u>. This technology's attractiveness is dependent largely on the amount of weight which could be saved at the expense of torsional stiffness. When short haul airplanes with their attendant low W/S and high AR are considered, advocates of active control point to wing structural weight savings on the order of 40% attributable directly to gust alleviation systems. This weight savings, however, may reduce torsional stiffness and flutter speed, and mandate active flutter control systems.

GA aircraft, on the other hand, already operate at low W/S and relatively low cruise speeds. As AR is increased, more attention will have to be focused on problems associated with aeroelastic phenomena such as flutter. Active flutter control is attractive when otherwise realizable structural weight savings result in an unacceptable degradation of torsional stiffness.

For the present evaluation, both Airplane A and B are presumed to possess sufficient torsional stiffness to negate a requirement for flutter suppression. As with active gust alleviation, this system is safety-of-flight critical since its anticipated use would be to reduce structural weights by reducing aerodynamic loads. It must be realized that the evaluation ratings will change considerably if an unanticipated increase in torsional stiffness is required as a result of increases in a spect ratio. 4.5.5.6 <u>Active Controls for Relaxed Inherent Stability</u>. This concept holds promise for reduced DOC through reduced fuel consumption. Like all active control systems, however, it poses

very high penalites in terms of front end (purchase) costs. Unlike gust alleviation and flutter suppression, where aerodynamic efficiencies result from decreased structural weights, this technology offers a direct payoff by allowing for reduced drag resulting from relaxed (natural) stability constraints. This is achieved by sizing the tail to meet control constraints instead of stability constraints and usually results in smaller tails with dec \_ased (parasitic) downloads when the main (forward) wing is moved forward. Results from studies of medium and heavy commercial transports were very promising. Reference 107 showed that in the case where takeoff gross weight was kept constant, either payload was increased by 15% or range was increased by 20%. Reference 144 noted that in the case where the mission (payload and range) was kept constant, relaxed static statility could yield a 10% reduction in gross weight and a 5% increase in cruise L/D. Reference 118 notes that a study involving the NASA Jetstar airplane, where active gust alleviation and relaxed static stability were investigated, showed tail surface areas reduced by 40% and fuel consumption reduced by 21%. This latter example should be interpreted with caution, however, since the effect of the gust alleviation system was to allow for a substantial decrease in wing sweep with an increase in AR form 5.3 to 9, which in itself offered a significant improvement in L/D.

4.5.5.7 <u>Integrated Yaw Damper</u>. Yaw dampers typically are employed on high altitude, high speed airplanes, although they also may be applied effectively for yaw damping at low speeds in general

aviation aircraft. Typically, they may be employed whenever a poorly damped dutch roll mode attributable to low values of the this type of system is difficult with present autopilots, however, because yaw damper functions are typically fed back to the GA pilot through the controls as annoying distractions, usually during the approach-to-landing phase. This may lead pilots to turn the system off at a time when they need it the most. Hence, a system integrated through separate surface technology where the pilot receives no feedback and system failures do not result in critical situations appears attractive. Yaw dampers will allow aircraft vertical control surfaces (rudders) to be optimally sized for particular flight conditions while retaining superior handling qualities over the aircraft's entire flight envelope. 4.5.5.8 Integrated Low Cost Wing Leveler. Bergey notes that an automatic and low cost wing leveler that does not depend on auxiliary power would be extremely valuable to the GA community (Ref. 18). While investigating inflight airframe failures for the period 1966-1975, Stapleford (Ref. 197) notes that a lack of spiral stability is probably a key factor in determining the frequency of loss of control. Two flight test programs (one by the FAA using a Beech Debonair A-33 and one by NASA using a Mooney M20) clearly demonstrated the serious problems encountered by a noninstrument-rated pilot who ventures into IFR conditions. Stapleford goes on to note that the absence of any airframe failures

by the Mooney M20 during the ten years investigated and encompassing over six million flight hours points clearly to the benefits of a wing leveler. (The M20 was equipped with a wing leveler as standard equipment.) The attractiveness of a wing leveler, then, is predicated on enhancing spiral stability and reducing the potential for airframe failures resulting from recovery procedures (or lack of them) from unusual attitudes. Under more favorable flight conditions, a wing leveler still offers significant benefits in reduced pilot workload.

At present, there is a low cost (≈ \$100) fluidic system available to the GA community. Also, a separate surface system retrofitted to a Cessna 172 has been satisfactorily demonstrated (Ref. 176). For the evaluation at hand, data on forecast autopilot use appears to indicate that half the aircraft in Airplane A's category will have autopilots, and almost all new commuters (Airplane B) will be so equipped. Hence, it will be arbitrarily assumed that Airplane A does not have a basic autopilot and that Airplane B does.

4.5.5.9 <u>Separate Surface Technology</u>. This technology has particular significance to the GA community due to the nature of controls (reversible) incorporated by most single engine and commuter airplanes. In such systems, autopilot functions are mechanized by tying a servomotor into the primary surface cable controls. Hence, all autopilot functions are fed back to the pilot and, with the system off, stability augmentation capabilities which

may be incorporated into the autopilot are lost. Separate surface technology, on the other hand, can provide continuous wing leveler and yaw damper functions with appropriate wash out circuits incorporated so as not to interfere with pilot control, and bence remain totally transparent to the pilot. An autopilot function could also be incorporated which, too, would be transparent. Such a system (SSSA) has been flight tested in a Cessna 172 in a wing leveler mode and a Beech M99 in a three axis autopilot attitude command mode with excellent results. Costs of such a system are expected to be similar to present autopilots, with improved safety. Hard-over failures result in the pilot flying the aircraft in an out-of-trim condition with no requirement to override an autopilot servomotor since one is not tied to the primary cable system. References 175, 176, 181, and 182 provide more details on SSSA, and Reference 20 notes that the development of samarian cobalt motors makes SSSA appear even more attractive. It should be noted that SSSA has recently been incorporated into the GA fleet in the form of a yaw damper on the Mitsubishi Diamond I. As with the wing leveler evaluation, Airplane A is considered to not have an autopilot while Airplane B does. 4.5.5.10 Direct Side Force Control. The primary application of this type of system would be to augment active ride smoothing systems. Lapins and Jacobson (Ref. 119) noted that side force controllers are more effective than rudders alone in alleviating the effects of turbulence through active ride smoothing systems.

This same requirement was noted by Conner and Thompson in Reference 85. When used for ride smoothing, such asystem will not be considered safety-of-flight critical. Still, heavy cost penalties are expected for both Airplane A and B, although Airplane B should benefit considerably through greater passenger satisfaction in what promises to be a more turbulent flight environment.

4.5.5.11 Direct Lift Control. Several applications of direct lift control (DLC) are envisioned, and all are mechanized through spoilers. In the more sophisticated systems, DLC will be required for ride smoothing in order to offset vertical loads. In some large commercial aircraft, DLC may be required for adequate flight path control as W/S is increased. When such a system was employed in the Redhawk (modified Cessna 172), more precise, easier, and apparently safer approaches resulted since flight path angles could be controlled without changing aircraft attitude. Likewise, landing ground roll can be reduced substantially in those cases where the approach path is constrained by obstacles. This would allow descents beyond the obstacle without a requirement to lower the aircraft nose and increase airspeed. Also, when the spoilers are maintained in a partially deployed configuration, a more conventional approach may be flown at higher airspeeds without incurring a landing performance penalty. This latter application promises improved safety since a greater airspeed stall margin can be maintained.

4.5.5.12 <u>Single Lever Thrust/Drag Control</u>. The use of spoilers to control flight path angle during approach to landing is known
to improve the landing performance of most pilots. When such a system is mechanized through the throttle, pilot workload is reduced both during the approach and during any required goarounds. Improvements in reduced pilot workload were noted even in the case where the throttle lever retains its individual function but has the DLC spoiler control mounted on it as a thumb wheel (as was done on the Redhawk).

It should be mentioned that single lever thrust control alone offers promising advantages in reduced workload. Teledyne Continental Motors has incorporated a governor into a single lever control which effectively combines the functions of throttle, mixture, and RPM. This system, although not yet in production, has undergone over three years of development and is presumed ready for production. In its present configuration, the throttle is used to set RPM, and the governor mechanization acts to control manifold pressure and fuel flow.

4.5.5.13 <u>Fluidic Automatic Flight Control Systems</u>. The fluidic three axis system designed for NASA by Honeywell and subsequently flight tested in an Aero Commander 680 FP showed excellent reliability and functioned very similarly to conventional autopilot systems (Ref. 222). Altitude hold, however, was degraded above 1,830 m and power recovery of the fluidic servo amplifiers was only 40%. Yet, systems such as these demonstrate a clear capability for further developemnt. The continuing decrease in the cost of microprocessor logic, however, has done much to offset the initial

advantages of lower cost and high reliability attributed to fluidic systems. On the other hand, developments in low-cost fluidic sensors such as the low speed airspeed indicator, the vortex rate sensor, and fluidic stall sensor, offer significant benefits at reduced cost. One of the most advantageous systems to result from efforts in fluidics was a wing leveler which incorporated a laminar flow proportional fluid amplifier with a very high signal-to-noise ratio that was developed by NASA Langley personnel.

4.5.5.14 <u>Active Stall Prevention</u>. This technology offers great potential for reducing stall/spin accidents. Work by Chevalier where a spoiler is added to the lower surface of the horizontal tail to prevent the attainment of the stall angle-of-attack (Ref. 29) appears encouraging. Since it appears that a significant number of stall-related accidents occur in airplanes which are not euqipped with artificial stall warning systems (Ref. 49), the need for an effective stall warning/prevention device appears well justified. Several warning concepts (stick shaker, audible horn, visual angle of attack indicator) are technologically mature. Of these, the stick shaker appears to be the most effective. The possibility of incorporating a stick pusher also appears attractive.

One detraction to active prevention systems lies in their accurate operation. Although the incorporation of active prevention in light GA single engine aircraft appears warranted, the question of degraded control authority below the stall angle-of-attack

detracts from the otherwise undisputed utility of such a system. Cost appears to be another possible detraction. In order to achieve widespread acceptance, such a system must either be mandated or offered at very low cost.

#### 3.5.6 Information Systems.

Eight technologies were evaluated under this grouping and are shown in Table 4.16. As illustrated, Digital Data Links, Integrated Avionics and Displays, and Systems Status Displays are all fairly attractive technologies. The Micro HUD, although presently reflecting no benefit to the user, deserves mention here. This system, discussed further in Section 4.5.6.5, promises to provide all HUD functions without either a CRT or the associated optical system. Presently, its main penalty lies in cost. Widespread use of such a system could drive costs down significantly and make this technology extremely attractive.

4.5.6.1 <u>Digital Data Links</u>. This concept entails the communcation of data (which is presently rendered verbally) between the airplane and various agencies in a digitized format much as present data links are used for encoding altimeters. Three types of systems are addressed in this section. The Digital Data Broadcast System (DDBS) is intended to provide RNAV systems with information required to satisfactorily navigate both preplanned direct routes and "whatever charted routes which are retained as an integral part of the ultimate area navigation environment" (Ref 97). Data would be broadcast in repeating data streams for

117

;

specified station or route coverage and would be accessed by tuning in the appropriate VORTAC frequency. It should be Loted that this system would marry RNAV to the existing VORTAC route structure instead of providing the more flexible option of inputting geographic coordinates. The present status of this system is unknown, and the last time it appeared in the literature used in the present research was in 1976 (Ref. 97).

Technology (9)	Figures of Merit for		
Technology (8)	Airplane A 56/-53/85*	Airplane B 51/-57/58*	
Digital Data Links	33	38	
Integrated Avionics and Displays	32	42	
Systems Status Displays	25	25	
CRT Displays	11	21	
Micro HUD	2	2	
Laser Gyros	-16	-14	
Heads Up Displays	-23	-24	
Fiber Optics (Data Trns)	-26	-23	

Table 4.16. Information Systems

\* Total number of technologies/lowest FM/highest FM

The second system to be discussed here is the Discrete Address Beacon System (DABS), which formed a significant aspect of the upgraded third generation ATC system (UG3RD) as discussed by the DOT Air Traffic Control Advisory Committee in December 1969. This system has continuously appeared in the literature and receives further attention in a report prepared as part of the 1979 Summer Faculty Fellowship Program in Engineering Systems Design (Ref. 148). As discussed in that reference, DABS will accommodate the following information between the aircraft and its ATC environment:

- (1) Clearances.
- (2) Runway surface winds to include wind shear and wake vortex information.
- (3) Weather information.
- (4) Minimum safe altitude warning.
- (5) Confirmation of assigned altitude.
- (6) Automated Terminal Information Service (ATIS).
- (7) Runway Visual Range (RVR).
- (8) Holding instructions.
- (9) Approach and departure clearance.
- (10) Conflict alert and resolution instructions.
- (11) Instructions as to proper heading, speed, altitude, and the time to execute the ATC instructions.

The third system to be discussed is the Automatic Communication and Reporting System (ACARS), which is used in conjunction with existing VHF radio equipment and allows for both voice as well as digitized information communication to enhance air-ground operational control communications. Although this type of system has been available for ten years, it has not been implemented as rapidly as hoped due primarily to its \$5,000 price tag (Ref. 148).

\*\*\*\* 1

It appears significant at this time to point out that studies of future ATC environments point to a congested environment as a very real possibility, and this suggests that the pilot could well become saturated with communications functions alone. At a NASA-sponsored Avionics and Controls Research and Technology Workship (Ref. 147), the team discussing "General Aviation and Short Haul" recommended that a principal focus for research is suggested to:

- Minimize or eliminate the requirement for communication (talk) to as great an extent as possible.
- (2) Allow IFR flights of the future to be performed as easily as VFR flights are today.

In view of this observation/recommendation, DABS appears to offer the most significant benefits of the three systems discussed. It should be noted that DABS is crucial to the Intermittent Positive Control (IPC) conception for Collision Aviodance Systems (CAS). A basic DABS unit may be expected to cost about the same amount as present transponders (\$750) or \$2000 with an IPC display (Ref. 97).

For the present evaluation, both airplanes are presumed to already be equipped with encoding altimeters but are penalized for the IPC display. Substantial benefits in improved pilot workload and safety are realized.

4.5.6.2 <u>Cathode Ray Tube (CRT) Displays</u>. CRT displays may be evaluated against mechanical, conventional instruments, or against other electronic displays. For purposes of this evaluation, they

will be evaluated against the former because other electronic systems are not envisioned to be competitive from a cost standpoint.

Still, some discussion appears warranted for various electronic display methods. The following data of 1975 vintage is condensed from Reference 196 and shown here as Table 4.17.

System	\$/Character	Volts/Panel
CRT	< 1	10,000-25,000
Plasma Panel	< 3	100
LED	< 4	
LC		< 6

Table 4.17. 1975 Characteristics of Some Electronic Displays

The display application also needs to be addressed and is summarized in Table 4.18 in terms of their 1979 status.

The versatile capabilities and low cost of the CRT tend to more than offset its disadvantage of relative size and high voltage, although the latter can pose a safety problem. A flat panel CRT, developed by Northrop for the Army and whose rights were subsequently sold to Texas Instruments, greatly reduces the problem of size but was found to be prohibitively expensive due to high manufacturing costs (Ref. 219). Present CRT's rely on a \$130 million annual market in computer terminals (which is doubling every three years) to offer very low costs. Indeed, one source (Tannas, Ref. 219), suggests OEM costs of \$9.00 for black and white units and \$36.00 for color in very large quantities. When present CRT capabilities are used as a baseline, the relative performance of other electronic displays in six major problem areas may be briefly tabulated as in Table 4.19.

TADIE 4.10. Electionic Display lectmology relapecti	Table 4	1.18.	Electronic	Display	Technol	ogy	Perspectiv
---	---------	-------	------------	---------	---------	-----	------------

Display Application	Available	In R & D
Discretes, Meters, & Legends	Electromechanical Galvanometers Incandescent Light Emitting Diode	Liquid Crystal
Alphanumeric •	Cathode Ray Tube Electromechanical Incandescent Light Emitting Diode Liquid Crystal Plasma	Chemoluminescent Electrochromic Electroluminescent Electrophoretic Ferroelectric
Vectorgraphic	Cathode Ray Tube Plasma Panel	Electroluminescent Light Emitting Diode Liquid Crystal
Video	Cathode Ray Tube	Electroluminiscent Ferroelectric Laser Liquid Crystal Plasma Panel

(data from Reference 219)

Armed with this background information, CRT's are evaluated now against their mechanical conventional counterparts. These systems offer reduced pilot workload in that scanning tasks can be

# Table 4.19. Six Major Display Problem Areas & Their Impact on New Technologies

.

(X designates problem with respect to current CRT)

	Light Emitting Diodes	Plasma Discharge	Flat CRT	Electro- luminescence	Liquid Crystal	Electro- phoretics	Electro- chromics
Luminous Efficiency	x	x					
Natrix Addressing					x	x	x
Duty Cycle		х			x	x	x
Uniformity/ Gray Scale	x	x				x	x
Full Color	x	х		x		x	X
Cost (includ- ing Electronics)	x	x	x	x	x	x	x

(data from Reference 219)

٩.

reduced from a 22 cm (9 in) radius about the artificial horizon to 6.4 cm (2.54 in). This, of course, implies an integrated avionics package capable of providing continuous systems monitoring and, inherently, warning-by-exception. Since the CRT's eliminate mechanical flags, etc. (as many as five or more in an integrated attitude director) reliability will be increased. (One avionics manufacturer indicated that possible sticking needles and flags on a production ADI were corrected through the incorporation of servomotors to drive the flags.) For those airplanes using remote mounted avionics, substantial weight savings can result by reducing the number of signal paths.

Both Airplane A and B should receive significant benefits in reduced pilot workload and safety. (Recall that a display device is essential to DABS/IPC implementation.) 4.5.6.3 <u>Heads Up Displays (HUD)</u>. HUD systems offer the potential for reducing pilot workload during approaches to landing under both VFR and IFR conditions. Currently, they display information such as course guidance, airspeed, angle-of-attack, altitude, etc., via a CRT through a lens system/combining screen which collimates the display on the aircraft windscreen. Thus the pilot may devote his attention to external cues from his landing environment and still receive flight data without having to redirect (and re-focus) his attention to within the cockpit. Associated with this capability, however, is the problem of cost. HUD systems are envisioned to cost on the order of \$25 to \$45

thousand for GA applications while they may not several \$100 thousand in military applications. Other data which tends to detract from HUD implementation are the observations that:

- At the NASA-sponsored General Aviation Avionics Workshop, no general agreement could be reached on the question of whether HUD studies for GA should be pursued (Ref. 196).
- (2) A USAF study concerning the use of HUD's as a primary instrument reference in the A-7D, F-15, and F-111D in 1976 resulted in mixed responses from 123 pilots on the question of using HUD's during approach to landing. Although a significant majority of A-7D and F-15 pilots said the HUD enhanced IFR operations, only a small percentage preferred its use during approaches. Most F-111D pilots preferred not to use the HUD for normal operations where weapons delivery was not involved (Ref. 15). The major complaint of the USAF pilots was that erroneous information could be displayed without warning.

For both Airplane A and B, improvements in pilot workload and safety were offset by penalties in maintenance cost, empty weight, reliability, and purchase price.

4.5.6.4. <u>Micro HUD</u>. A micro HUD, developed by Bell Helicopter Textron, offers HUD capability at an anticiapted 50% of the cost of regular HUD's. Although this cost penalty is still significant for GA, the system merits discussion due to its unique implementation. The system uses a micro-processor to drive a fiber optic

symbol generator which presently has the capability to generate a large range of dynamic symbols and numbers. (When the signal generator was viewed during July 1979, the microprocessor was not fully implemented and consequently only a limited set of symbols were generated. This generator is illustrated in Figure 4.18.) The most unique aspect of the system however, lies in the absence of the large and heavy optical system required of conventional HUD's. This system uses a pair of eyeglasses with a minute mirror centered in the eyepiece as shown in Figure 4.19. Signals are transmitted to the glasses via a fiber optic bundle and collimated to display images as shown in Figure 4.20. Two similar systems are currently being tested by the Army.

The benefits of this system over conventional HUD's lie in the elimination of (1) the high voltage and higher power requirements of a CRT, and (2) the elimination of costly and hea/y associated optical equipment.

Although the same benefits for the HUD were credited here, severe cost penalties were also awarded on the basis of an estimated \$20,000 price tag, which couldnot be supported by either Airplane A or B.

4.5.6.5. <u>Systems Status Displays</u>. Comprehensive engine health monitoring is significantly absent in present GA aircraft and, consequently, pilots are confronted with having to derive this information from existing displays (when sufficient raw data is available). The current DAAS system promises to rectify this situation by sensing:



and the second product of the second

\$

4

Fig. 4.18. Preliminary Fiber Optic Symbol Generator (Operational)

127



Fig. 4.19. Micro HUD with Fiber Optic Link and Mirror.

- (1) Manifold pressure
- (2) Engine RPM
- (3) Fuel Flow
- (4) Fuel Quantity
- (5) 0il Temperature
- (6) Oil Pressure
- (7) Cylinder head temperature for each cylinder
- (8) Exhaust gas temperature for each cylinder
- (9) Exhaust gas oxygen for each cylinder
- (10) Cowl flap position
- (11) Auxiliary fuel pumps

1

ORIGINAL PAGE IS OF POOR QUALIT



Fig. 4.20. Micro HUD Image Display.

plus numerous other aircraft system parameters. This system will alert the pilot to any critical out-of-tolerance condition and advise corrective action in some cases. It also presents commands to the pilot for setting the mixture lever (Ref. 39). Such enhanced capabilities do much to relieve the pilot from scanning and provides warning signals on an exception basis. The incorporation of an annunciator of some sort appears warranted for such a system, and it is noted that Princeton's Avionics Research Airplane (ARA) is so equipped. Reference 31 points to integrated multifunction displays as having the highest priority of ten equipment disciplines for NASA funding/study. This report (done in 1974) identifies a desired or expected purchase price of such displays at \$500-\$1500 for a single encine airplane and \$3000-\$7000 for a turboprop system.

When sufficiently integrated, system status displays promise more advanced warning of possible engine failures, more information to assist in selecting optimal power settings, and reduced pilot workload. Engine TBO's may have the opportunity for being improved through better engine management on the part of the pilot. These benefits appear to apply equally to loth Airplane A and B. 4.5.6.6 Integrated Avionics and Displays. This concept has received considerable interest and publicity over the past decade, due largely to the anticipated effects of increased air traffic congestion and the attendant increased pilot workload associated with a more restrictive flight environment, particularly under single-pilot IFR operations. NASA efforts to identify and develop an orderly investigation involving industry, educational institutions, and other agencies, are reflected in the Langleysponsored Avionics and Controls Research and Technology Workshop (Ref. 147) and the 1979 Summer Faculty Fellowship Program in Engineering Systems Design which was co-sponsored by ASEE (Ref. 148).

Ames efforts have been centered on the Advanced Avionics Systems studies which recently resulted in a Honeywell/King Radio contract in August 1978 to design and build a Demonstration Advanced Avionics System (DAAS) which will eventually be flight tested in a twin engine aircraft. Highlights of this program include several studies and a workshop as follows:

- (1) An ATC environment forecast (Ref. 97).
- (2) An electronics technology forecast.
- (3) A GA advanced avionics workshop (Ref. 196).
- (4) Preliminary Candidate Advanced Avionics Systems (PCAAS) study contracts to Southern Illinois University (Ref. 133) and Systems Technology Incorporated (Ref. 208).
- (5) In-house efforts at Ames investigated, among other things, different low-cost options to solve mavigation algorithms and to define a low-cost option to gyro sensors.

Interesting results to number five above involve the incorporation of a phase-locked-loop to enhance VOR scanning capabilities and the incorporation of magnetometers to solve the attitude sensing problem. Preliminary results comparing INS data with derived data using magnetometers is presented in Figure 4.21.

Under the present contract, the Honeywell/King DAAS effort will incorporate the following functions (Ref. 39):

- Automated guidance and navigation using VOR/DME navigational facilities.
- (2) Flight planning.



.

Fig. 4.21. Comparison of Pitch, Heading, and Roll Computations.

\$

- (3) Weight and balance and performance calculations.
- (4) Monitoring and warning.
- (5) Storage of normal and emergency checklists and operational limitations.

Pilot interface with this system is through a touch panel which in turn provides inputs to the Integrated Data Control Center (IDCC). The system is organized around an IEEE 488 bus and includes five microprocessors. Conventional pilot displays are incorporated with the exception of two CRT's. One serves as an IDCC interface with the pilot and the other serves as an Electronic Horizontal Situation Indicator (EHSI) with an electronic map capability. Unfortunately, no cost forecasts for this system were readily available. The STI study (which was dependent on the availibility of a \$6000 OMEGA system) penalized an unsophisticated avionics user \$4900 for a \$13,275 total avionics package and a sophisticated user \$13,300 for a \$62,645 system. The SIU study, on the other hand, forecast a cost of \$23,850 for a full IFR system in a Cessna 402 and \$9962 for a single engine new installation. (The STI PCAAS system used cockpit display formats similar to DAAS but with seven microprocessors, and the SIU system replaced all conventional instruments with plasma panel displays.)

It should be mentioned that the GA avionics community is not in total agreement with the single bus architecture, and at least two manufacturers which were visited expressed concern over possible single point failures which would render the entire system

either inoperative or unacceptably degraded. On the other hand, the concept of an integrated avionics system with or without the bus structure offers certain advantages which would otherwise be delayed. For example, the powerplant health monitoring capability of DAAS is seen as a major advance over present systems. Here, the advantage of DAAS is that it allows for more types of information to be displayed to the pilot through the IDCC, where present configurations do not allow for either the required panel display space or separate display costs to be efficiently absorbed.

A forseeable spin-off from a system which incorporates microprocessor logic systems lies in the further introduction of these systems into GA aircraft. Certainly, one of the most beneficial applications of such logic systems lies in the electronic control of propulsion fuel systems to allow for more efficient operation. For example, aivanced positive displacement engines will almost invariably require a high-pressure fuel injection system where the fuel is injected directly to each cylinder. Such a system will benefit substantially from an electronic control which senses induction air density and modulates as well as times the fuel flow. (This is to be contrasted to low-pressure continuous-flow systems in operation today.) Since the possibility of failure of such an electronic fuel injection system may be unacceptably high, the system could be mechanically driven and only modulated by an electronic logic circuit in order to achieve the highest

efficiencies. The fact that DAAS already incorporates a propulsion monitoring function (no control) could lead to a reduction of inflight engine failures by providing advance warning or indication of impending failure.

4.5.6.7 <u>Fiber Optics for Data Transmission</u>. Fiber optics have the capability to:

(1) Transmit data faster than by wires.

 $\lambda_{0}$ 

- (2) Replace several wires with one fiber since fibers have a larger bandwidth capability.
- (3) Function without the inductive or conductive properties of wires.
- (4) Retain better signal isolation between fibers than possible between wires.
- (5) Function without introducing electrical and/or fire hazards.(Ref. 158.)

However, no appreciable weight savings or benefits may be anticipated for retrofit systems, especially where data transmission rates are lower than one megabyte/sec.

Hence, fiber optic data transmission systems are not envisioned to offer advantages significant enough to justify their cost in the GA fleet.

4.5.6.8 Laser Gyros. These systems, developed by Honeywell, contain no moving parts and sense accelerations by measuring fringe shifts between two laser beams. Reliability is expected to be very high, but costs are expected to be equally high. Hence, their incorporation into the GA fleet is not envisioned at this time.

## 4.5.7 <u>Navigation</u> Concepts

Six technologies as shown in Table 4.20 were evaluated in this technology group. Microwave Landing Systems appear more attractive to Airplane B than A due to a larger number of IFR terminal operations. Of the three area navigation concepts investigated, none appeared to provide significant benefits. This was due largely to the fact that both airplanes were presumed to already have full IFR capability, and Airplane A was also equipped with an existing RNAV system. The two velocity information sources (Doppler and Inertial systems) were both extremely unattractive to Airplanes A and B.

Toobroloon (6)	Figures of Merit for			
rechnology (6)	Airplane A 56/-53/85*	Airplane B 51/-57/58*		
Microwave Landing Systems	9	21		
NAVSTAR/CPS	8	8		
Loran C	-24	-30		
Omega	-26	-32		
Doppler Navigation	-37	-40		
Inertial Navigation	-53	-57		

Table 4.20. Navigation Concepts

Total number of technologies/lowest FM/highest FM

4.5.7.1 <u>NAVSTAR/GPS</u>. This system is a global, satellite-based navigation system being pursued by the U.S. Air Force. Six of the originally-planned 24 satellites are already in orbit (Ref. 148), but Reference 8 indicates that the total number has been reduced to 18, with a planned operational date of 1987 to 1990. This system is extremely attractive to civil aviation because it provides a (forecast) reasonably priced navigation system capable of

(1) Extreme accuracies in x, y, z and t reference frames.

- (2) Area navigation.
- (3) Global coverage down to ground level.

(4) Collision avoidance.

When one considers that a general aviation GPS system is expected to possibly cost as little as \$2800 (Ref. 1 and 102) to \$3600 (Ref. 148) while an RNAV system comprised of 1 VOR, 1 DME and 1 RNAV computer will cost \$3700 (Ref. 148) while still being dependent on an ability to receive an acceptable VOR/DME signal, the attractiveness of GPS becomes clearly evident. It should be noted that the VOR/DME/RNAV system iescribed above still cannot provide altitude information, and velocity information is highly dependent upon accurate VOR/DME position fixes which must be differentiated.

Three levels of system accuracy are provided, where the first two are protected by the military. The third, envisioned to remain unprotected for civil aviation use, was recently found to be "too accurate" and consequently will probably be degraded

to offer 200 meter (660 ft) accuracies with a 50% confidence level (Ref. 8). In order to realize how accurate GPS is, one should note that tests performed at 6 km (20,000 ft) by a C-141 using either one or both of two satellites which were then in orbit reflected errors on the order of 4 m (13 ft) in x, y, and z (Ref. 1).

It is significant that a GPS-equipped aircraft will never be threatened by a signal-saturated environment because airborne equipment is totally passive, i.e. no signals are transmitted from the aircraft as is the case with DME. When a digital data link is added (Time-Division-Multiple-Access Link) together with a groundbased GPS transmitter, accuracies sufficient for terminal area guidance results, together with a collision avoidance capability.

In view of the above, benefits are expected for both Airplane A and B in terms of safety. Since the baselines for both airplanes include IFR avionics, only minor cost penalties are expected. It should be noted here that the forecast data on costs is considered fairly optimistic and did not appear to be supported by manufacturers. Hence, a penalty of -1 was assessed. If, on the other hand, costs do attain forecast levels, then a +1 might be warranted. In such a case, the figures of merit for both airplanes would be on the order of +24.

4.5.7.2 <u>Inertial Navigation Systems (INS)</u>. Inertial navigation provides the attractiveness of a self-contained navigational system. Also, it provides a capability of computing reasonably accurate velocities which normally could not be obtained by other means

without enhancement, e.g., ILS signals are too noisy to be differentiated with any confidence (Ref. 155). When INS is used with radio ranging and a Kalman filter, an Aided IN' results, which has applicability to short haul transports by providing 4-D or twosegment approach guidance (Ref. 193). However, INS purchase prices are on the order of \$100,000 in larger commercial aricraft and are not expected to fall below \$30,000 for the GA market (Ref. 97). The Aided INS system is an exception since it is predicated on low cost gyros. Annual maintenance costs are also exorbitant. In a study of relative avionics costs to the user, one investigator penalized INS \$9000/year while choosing not to assess such costs against other systems (Ref. 195).

Based on the above, both Airplane A and B will suffer severe penalties with only negligible gains in their operating environment. GPS, for example, can provide global coverage with velocity capabilities for a fraction of the cost. 4.5.7.3 <u>Doppler Navigation</u>. This navigation mode, like INS, is self contained and therefore not dependent on ground based systems. When used in conjunction with current navigation aids such as VOR/DME, it offers the potential for 4-D navigation in the terminal area. Like INS, however, it is an expensive alternative to an augmented GPS terminal environment. In the enroute phase of flight, the differentiation of DME signals when proceeding to/from a DME station or the integration of RNAV capabilities with DME and, say, three VOR stations should provide a satisfactory

velocity computation in either the airway or area navigation environment. Certainly, GPS should provide satisfactory velocities in its conventional (unaugmented) mode during enroute flight.

Doppler navigation systems are therefore expected to assess high cost penalites to users where the same capabilities are expected to be available at much lower costs.

4.5.7.4 Microwave Landing Systems (MLS). MLS offers an attractive alternative to conventional Instrument Landing Systems (ILS) due to an ability to handle several nircraft flying at different airspeeds in the approach phase. This capability offers promise even today to the GA aircraft operating in high-density, mixedtraffic, terminal environments. Figure 4.22 shows a conventional ILS profile and Figure 4.23 shows an MLS profile where the advantages of such a system are more obvious. If one visualizes a GA aircraft operating at 40% to 60% the approach speed of larger commercial jets and trying to execute an approach to the same runway as his jet counterparts are, the advantages of the MLS system becomes evident. Certainly, significant savings in fuel expended in the terminal area may be expected, particularly by commuter operations whose flight profiles dictate that larger amounts of their total flight time between refuelings is spent in the approach to landing phase.

Although Madden and Desai (Ref. 130) noted in 1973 that it was possible to track a curved approach path using MLS and DME to accuracies within the resolution of ATC radar, Hoffman and Hollister (Ref. 97) noted in 1976 that pilots expressed a reluctance to:



• • · · · ·

4

\*

Section on the destriction of section and section of the section of

Fig. 4.22. Typical ILS Implementation.

1.

1

141

•

•





- fly steep approaches which do not level off to the more conventional 3<sup>o</sup> prior to the final flare, and
- (2) fly curved approaches beyond the point where the 3<sup>o</sup> glide slope is intercepted.

A reasonable plan appears to be one where the curved and steeper flight paths are used to within 150 m (500 ft) AGL and a transition is made to a conventional glide slope.

Current ILS systems are characterized by reliability, safety, and efficient service. Probably, its only drawbacks are limited available channels (40, although only 20 are used currently) and performance degradation resulting from terrain irregularities and heavy snow. (Terrain effects stem from the fact that ILS glide slope signals require radiated energy to be reflected from the ground plane.) MLS incorporates a microprocessor as well as a digital data link and costs are forecast to be on the order of \$2000 for GA and \$34000 for the commercial jet carrier by the year 2000.

MLS on the other hand, offers five times the channel availability and is not affected by terrain features. Also, a spin-off from the versatility of the MLS is that noise abatement procedures can be more easily integrated than under present ILS systems.

Hence, MLS offers improvements in efficient terminal operations and possible noise benefits for some penalty in cost to both Airplane A and Airplane B.

4.5.7.5 LORAN C. LORAN C is a low-frequency, long-range, allweather radio navigation system with absolute accuracies on the order of 0.5 km (1600 ft). This accuracy may be improved by a factor of 2 or 3 by implementing more sophisticated user equipment. Use of LORAN is predicated on having velocity inforwation, and this implies either differentiating the LORAN C time difference measurements (TD) or implementing an INS. LORAN provides area navigation capabilities down to ground level, but at present does not have full coverage of the Continental United States (CONUS). GA user costs for a receiver (without INS) are expected to be as low as \$3000 (Ref. 195), while military systems may cost \$20,000 (Ref. 148). The receiver unit weighs approximately 11 kg.

LORAN C is being considered as a possible GA navigation aid primarily because it is being considered as a replacement for the present VOR/DME network which is characterized by extremely high operations and maintenance costs. The LORAN C ground-based system costs are expected to be an order of magnitude smaller than the present VOR/DME network.

As presently conceived, Airplane A and B will both suffer cost penalties if LORAN C is implemented. LORAN C does not provide precision terminal guidance, although ILS systems do. In evaluating this technology, the system is considered without an INS. Hence, the cost penalty is not as severe but is traded for deficiencies in reliability and safevy.

As this report was being finalized, new information regarding a joint test effort between DOT, NASA and the State of Vermont surfaced. The interested reader in referred to <u>Aviation Week and</u> Space Technology, March 24, 1980, pgs. 51-58.

4.5.7.6 <u>OMEGA</u>. Like LORAN C, OMEGA is a low-frequency, longrange, all weather navigation system. Unlike LORAN, however, it does have complete CONUS coverage. Its greatest deficiency is that of accuracy, and CEP's on the order of 4 km (13,000 ft) are reported in Reference 1. When an accurate INS is integrated with the system, errors on the order of 2 km (6500 ft) are quoted. Consequently, no terminal guidance is provided. Cost for an inexpensive OMEGA system is forecast to be on the order of \$30CO (Ref. 195) without the INS. However, it should be noted that 1977 receiver costs were on the order of \$20,000 to \$59,000 (Ref. 208).

Like LORAN C, this system's attractiveness lies in operational and maintenance costs of ground facilities which may be two orders of magnitude below that of the current VOR/DME network. This is better understood when one considers that only eight stations are used to provide complete global coverage.

For the present evaluation where neither Airplane A nor Airplane B is used over water, OMEGA (without INS) offers penalties in cost and safety.

#### 4.5.8 Noise

The three technologies investigated here are shown in Table 4.21. Quiet, Efficient Propeller technology appears fairly attractive to both Airplane A and B, due primarily to a promise for reduced interior noise. Ducted Propulsors appear attractive to Airplane B due to the maximum scores (+3) awarded for interior and exterior noise and this vehicle's unique flight profile.

Technology (2)	Figures of Merit for		
Technology (3)	Airplane A 56/-53/85*	Airplane B 51/-57/58*	
Quiet Efficient Propellers	24	29	
Ducted Propulsors	- 2	18	
Low Level Pressurization	-17	-18	

Table 4.21. Noise

Total number of technologies/lowest FM/highest FM

4.5.8.1 <u>Quiet Efficient Propeller Technology</u>. The fact that current propellers already operate at efficiencies on the order of 87% may lead one to doubt the existence or significant payoffs in this particular field of endeavor. A review of the literature, however, quickly identifies shortcomings: most of the propellers in use today are based on WW II technology, with appropriate refinements resulting from "cut and try" processes epplied to a basic design. More recently, however, the requirements for reduced fuel consumption and noise pollution have led to renewed interest in propeller technology.

۰.

NASA efforts to aid in the development of improved propellers are focused in the General Aviation Propeller (GAP) Technology program, with a goal toward reducing fuel use by 8% to 9%, lowering noise by 5 to 10 db, and improving safety. This ambitious program involves propeller and airframe manufacturers, consultants, and several universities.

Essentially, the major problem associated with propellers lies in the fact that they have been traditionally designed for performance, with little regard for noise. As one attempts to reduce noise, penalties in performance and/or weight usually materialize. However, since the propeller produces approximately 85% of powerplant noise, the attenuation of this noise will do much to improve community relations and ride comfort.

A major hurdle in reducing noise lies in its accurate prediction. Succi (Ref. 205) recently reported on the accurate prediction of the sound field using the Ffowcs-Williams Hawkins equation modified for computational simplicity. Having achieved this, he then reported on different means of reducing flyover noise: (1) Reducing propeller radius by 20% resulted in an 8 dbA reduction with a 4 1/2% loss in efficiency.

(2) Altering the radial load concentration from 80% to 60%by

- (a) re-twisting resulted in a 4.2 dbA reduction with a3.9% loss in efficiency.
- (b) changing planform resulted in a 4.8 dbA reduction and a 1% loss in efficiency.

Two additional methods are also discussed (increasing the number of blades and the blade sweep) but results were not quantified as above.

Other developments which promise to increase fuel efficiencies lie in airfoil development. The ARA-D, reported in Reference 38, appears particularly promising for turboprop application. Briefly, this particular airfoil retains its takeoff performance instead of showing the typical reductions in thrust which accompany reductions in activity factor. When one considers that higher cruise efficiencies are obtainable at lower activity factors (and what has traditionally implied lower takeoff performance), the ARA-D looks very attractive.

Increasing the number of blades has been recognized as an effective means of reducing noise. However, this presents a weight penalty which is compounded possibly by increased strength requirements resulting from increased vibration. In this light, the Kevlar composite propeller developed and certified by Hartzell is extremely attractive. This propeller has a 50% blade weight reduction with a 100% increase in strength. Unfortunately, its price was increased by a factor of 2.5. As composites are better understood and manufacturing processes refined, significant weight and price reductions a pear inevitable. In light of the above discussion, both Airplane A and B should receive significant benefits in noise and weight reductions as well as safety, with a modest penalty in cost. 4.5.8.2 Low Level Pressurization. This technology was envisioned as employing a low  $\Delta p$  on the order of 14 kN/m<sup>2</sup> (2 psi) for the sole purpose of attenuating noise. However, the addition of such a system will still require that a pressurization system be added at a distinct cost and weight penalty. Also, it was quickly noted that although this  $\Delta p$  could reduce an ambient altitude of 4.87 km (16,000 ft) to a cabin altitude of 3.05 km (10,000 ft), it would also exert an outward force of 9.61 KN on a 0.762 x 0.914 m door (2160 lbf on a 2.5 x 3 ft door). Hence, even "low level pressurization" will impose significant loads on the aircraft structure, implying further (structural) weight penalties in addition to those of the system.

٠

Both airplanes may be expected to be heavily penalized for a reduction in interior noise.

4.5.8.3 <u>Ducted Propulsors</u>. Ducted propulsors are investigated here as a possible means for reducing noise. Associated with much improved interior and exterior noise characteristics are improved takeoff and landing performance characteristics. However, cruise performance is degraded and significant weight penalties may be incurred. Hence, ducted propulsors offer Airplane B greater benefits intakeoff and landing performance due to its mission profile.

An interesting proposal toward attenuating noise is the Q-Fan as reported in Reference 230. However, this system is not considered compatible with the low speed characteristics of Airplane A and B.

## 4.5.9 Propulsion

Five propulsion systems and two related technologies were considered for Airplane A, while only two propulsion systems were examined for Airplane B. As shown in Table 4.22, the GATE Engine had the highest figure of merit for Airplane A, and the Stratified Charge Rotary Combustion Engine was second. As shown in Table 4.10, the third technology for Airplane A had a figure of merit of 58. The ratings for both engines considered for Airplane B are encouraging since the baseline for this vehicle already included a turboprop.

In view of the particularly high figures of merit attained by this technology group, it appears appropriate to reiterate the fact that the present evaluation technique provides a measure of potential benefit to the user as opposed to a benefit/risk evaluation.

4.5.9.1 <u>GATE Engine</u>. The General Aviation Turbine Engine studies which were begun in 1977 involved Detroit Diesel Allison, Garrett AiResearch, Teledyne CAE, and Williams Research, while management of the program was provided by the NASA Lewis Research Facility. This study was devoted to investigating opportunities for advanced
Technology (8)	Figures of Merit for	
Technology (8)	Airplane A 56/-53/85*	Airplane B 51/-57/85*
GATE Engine	85	32
Stratified Charge Rotary Combustion Engine	84	21
Advanced Diesel Engine	36	
Improved Turbocharging	31	
Liquid Cooling	13	
Stratified Charge Reciprocating Engine	12	
HCRLB Reciprocating Engine	11	

## Table 4.22. Propulsion

Total number of technologies/lowest FM/highest FM

technologies in small turbine engines below the 750 kw size class, and one key issue involved the question of how to make the turbine engine cost-competitive with reciprocating engines. The results of the study were extremely encouraging and have been reported in the literature (Ref. 12, 65, 123, 204). Briefly summarized, turbine engines in general traditionally suffer from penalties in purchase price and fuel consumption and consequently have not penetrated the cost-conscious GA market. Their dominance in larger sircraft can be attributed in large part to their attractive features which include lower weight, apparently better safety, improved ride comfort, and improved aircraft performance expressed in terms of higher ceilings and cruise speeds. In addition, the JP4 which they utilize has an approximate 10% advantage in energy content and a 10% price advantage when compared with avgas (Ref. 204). Interestingly, three of the four study participants forecast competitive purchase prices which were the result of lower component manufacturing costs and increased sales volume. The fourth participant elected to increase sophistication and efficiencies to the extent that purchase price was raised but operating costs were much lower.

Two major advantages of turbines which may not be readily apparent lie in its three-to-one weight advantage and potential for attaining greater cruising speeds. In a cruise dominated mission this results in a markedly smaller airplane for the same payload and range as a reciprocating-powered counterpart due to the cascading effect that lower engine weight and cleaner installations have on vehicle size. As an illustration, consider that the lighter engine allows for a smaller and lighter vehicle, which allows for less drag. This allows for a reduction in required fuel volume which results in an even smaller airplane, which requires even less fuel.

For the present evaluation, Airplane A was considered to be powered by a conventional reciprocating engine. A maximum purchase price penalty (-3) was assessed primarily because the market forecast used in the GATE studies appeared somewhat optimistic. Airplane B, on the other hand, was given a purchase

price advantage due to an anticipated significant reduction in manufacturing cost. Recall that this airplane's baseline was presumed to already be powered by a turboprop. Hence, performance improvements in speed and ceiling were not allowed although improvements in BSFC and purchase price were credited. 4.5.9.2 Stratified Charge Rotary Combustion Engine. Research in this particular field shows promising results. The rotary combustion engine, possesses several inherent features which make it extremely attractive for application as an aircraft powerplant. For example, it is simple, lightweight, and compact. Since power generation is not based on reciprocating pistons it is smoother. Liquid cooling allows for quieter operation, safer cabin heat and a probable reduction in cooling drag (since present air cooled installations are typically somewhat inefficient with unnecessary drag penalties). The absence of valves and cams also promises quieter operation and improved maintenance. Whereas homogeneous charge fuel efficiencies were less than competitive some years ago, (BSF  $\approx$  .328 kg/kW/hr or .54 lb/hp/hr), stratified charge BSFC's of .262 kg/kW/hr (.43 1b/hr/hr) have recently been achieved and compare very favorably with present reciprocating engines. Projections for the future are for BSFC's on the order of .231 kg/kW/hr (.38 1b/hp/hr).

A significant feature of this system lies in its multi-fuel capacity. One engine presently operational at Curtiss-Wright burns JP4, diesel fuel, alcohol, or avgas. As such, it is extremely attractive in that transitions from avgas to JP4 with

changes in market trends which may be forced through fuel shortages will have minimal effects on this powerplant. The key to this versatility lies in the air motion within the combustion chamber which apparently is conducive to stratified charge operation.

However, certain drawbacks to this system may be deduced from present and forecast levels of the state-of-the-art. Specifically, high pressure, timed fuel injection and timed ignition systems are expected to be required. These represent fairly high cost technologies. Also, although the induction manifold is smaller than for reciprocating counterparts (leading to lower turbocharging requirements), an improved turbocharging capability will probably be required. This alone can represent significant increases in cost, particularly if the production base is not supported by automotive engine requirements.

For this evaluation, Airplane A is presumed to be powered by a conventional reciprocating engine and Airplane B by a current technology turboprop.

4.5.9.3 <u>High Compression Ratio Lean Burn Engine (HCRLB)</u>. Increasing compression ratio and leaning the fuel mixture are the most effective ways of improving fuel economy. When this process is applied to improve the performance of a current technology, homogeneous charge reciprocating engine, the type of engine identified by the acronym HCRLB results. High compression ratio offers increases in thermal efficiency of both air standard cycles as well as fuel-air cycles. However, when applied to a homogeneous

charge engine, an upper limit is established by fuel octane number. (Present engines operate at CR's on the order of 8.5:1 and have BSFC's on the order of .255 to .268 kg/kW/hr (.42-.44 lb/hp/hr)). When improved fuel injection (timed, moderate pressure) and improved cylinder cooling techniques are applied, leaner fuel/air mixtures may be obtained. Reference 169 points to BSFC improvements on the order of 4% to 5% when leaner operation incorporating improved fuel injection and cooling methods are applied to a TCM IO-520 engine. However, costs to the user are expected to be p.ohibitively high as recertification costs are recouped. The reader is referred to References 169 and 170, both by Rezy, Stukes, Tucker, and Meyers, for excellent discussions and analytical results of very current concepts regarding the emissions and fuel consumption of reciprocating engines.

Only Airplane A was considered in the present evaluation, where B was considered to be powered by a turboprop.

4.5.9.4 <u>Stratified Charge Reciprocating Engine</u>. Much of the discussion material presented in this section resulted from information received from Teledyne Continental Motors and is unpublished. Their assistance is deeply appreciated and acknowledged. Interpretations of the general information acquired, on the other hand, are solely those of the present research team.

Reciprocating engine stratified charge systems are not new. Three concepts presently being pursued by industry are (1) the Honda Compound Vortex Controlled Combustion (CVCC) system, (2) the Texaco Controlled Combustion System (TCCS), and (3) The

Ford Programmed Combustion System (PROCO). These three systems operate under different combustion processes and are briefly described below.

Stratified Charge systems differ from homogeneous charge (current) systems in that the fuel/air mixture of the former is not homogeneous within the combustion chamber. Instead, a rich zone is maintained at the point of ignition, and the flame front then progresses into a lean region which would not normally be expected to sustain combustion alone. The overall effect is a leaner combustion process which yields lower emissions and better fuel efficiency. Two methods of charge stratification are currently visualized. In one case, stratification is achieved physically by injecting the rich mixture into a prechamber where it is ignited. The second method relies on obtaining an air flow pattern within the cylinder itself to maintain stratification and hence is called an open chamber system. The CVCC is based on a prechamber system while the PROCO and TCCS employ open chamber methods.

At present, prechamber technology appears to have progressed to the point where a GA powerplant could be put into production with a certain degree of confidence. Such a system, when compared to open chamber systems, appears to have less efficient output which would result in a heavier engine for a given power level. Also, the chamber itself will probably require additional cooling, and this may present a problem in an air-cooled configuration. On

the other hand, the lean fuel-air ratio can probably be extended, and  $NO_x$  will be greatly reduced while HC and CO may be expected to increase.

The PROCO and TCCS. both open chamber systems, differ primarily in fuel injection time. PROCO uses early injection near BDC, while TCCS uses late injection near TDC. PROCO, as a result, offers the possibility of better air utilization but has no multi-fuel capability. TCCS, while retaining a multi-fuel capability, can suffer from a smoke problem due to incomplete air utilization. Both systems operate at very high compression ratios with TCCS at 12:1 and PROCO at 11:1.

Of the above systems, TCCS appears to be more attractive for GA application due to its multi-fuel capability. Also, due to its late injection feature, turbocharging and compression ratios are limited by structural considerations rather than combustion considerations. When tested in a jeep, TCCS showed BSFC reductions on the order of 35% under part load operation. At the anticipated higher power loadings, differences between homogeneous and stratified charge operations indicate a 25% improvement with a turbocharged TCCS displaying BSFC's on the order of .25 kg/kW/hr (.41 lb/hp/hr) at BMEP's of 552 to 689 kN/m<sup>2</sup> (80 to 100 psi). Turbocharging the TCCS is expected to increase output without reducing fuel economy or multi-fuel capability. If air throttling is employed, its application will probably be used to improve emissions control at low power settings.

With respect to present GA engines, TCCS will require a high pressure, timed fuel injection system and a variable timing ignition system. Less effective air utilization points to heavier engines while complexity points to increased costs. Improvements to the reciprocating engine, then, tend to detract from its presently undisputed cost advantage, while its advantage in fuel efficiency appears to be eroding. On the other hand, one cannot ignore the fact that this powerplant has established itself in the GA market, and it is difficult to visualize other engine types replacing it in the next decade.

4.5.9.5 <u>Advanced Diesel Engine</u>. The diesel engine appears attractive for GA application due to its low-cost fuel (20% advantage) and its low BSFC's. Its fuel characteristics alone (where it is not octane/detonation limited) allow for higher compression ratios and higher efficiencies.

At present, Teledyne Continental Motors, General Products Division, is investigating advanced two-stroke diesel e.gine concepts which are being applied to a six cylinder, radial, 300 kW (400 hp) engine and a four cylinder, radial, 150 kW (200 hp) engine. The 300 kW engine is considered to involve higher risk than the 200 kW engine, and includes those associated with (1) ceramic components, (2) operation in an unfinned cylinder environment (no cooling air), (3) high speed turbo starter/alternator, and (4) catalytic combustors. Impressive performance characteristics of this engine include takeoff BSFC's on the order of .225 kg/kW/hr

(.37 lb/hp/hr) and cruise BSFC's of .195 kg/kW/hr (.32 lb/hp/hr). Since this engine already enjoys a fuel price advantage, fuel costs alone could experience a 40% reduction over present day GA propulsion systems. Other attributes of this system include what has occasionally been called "uncooled operation." However, although the cylinders are unfinned, the injector pumps and aftercooler require cooling, and it is evident that cooling air of some sort must be provided for. This requirement is nevertheless anticipated to allow for lower cooling drag losses than presently experienced by the GA fleet of air-cooled engines. Weight advantages of the system could not be ascertained. Although the dry engine weight of the 300 kW engine is listed as 207 kg (457 lb) vs 262 kg (578 lb) for a comparable GTSI0-520-H, the diesel engine is known to require oil cooling, and this suggests that installed weight advantages may fail to materialize if the oil system weight is high. The purchase price disadvantage has been estimated to be on the order of 20%.

۰.

٠.

Airplane A should receive advantages in DOC due to fuel efficiencies but is penalized for purchase price. No data for emissions characteristics was found, but the quieter operation attributed to the absence of valves was considered to be only a small advantage.

4.5.9.6 <u>Liquid Cooling</u>. This technology is considered mature but due to risks involved with single point failures such as those associated with leaks, thermostat failures, and pump failures, has disappeared from the GA reciprocating engine powered fleet. Still, the advantages attributed to its incorporation are significant enough that its possible re-implementation is investigated here. Only a few of the more salient characteristics associated with its advantages and disadvantages are listed here. Advantages:

- More uniform cylinder temperatures can be maintained, thereby relieving thermal stresses and improving TBO.
- (2) Engines can be manufactured with closer clumpances by reducing the effect of thermal stresses, particularly those associated with idle power rapid letdowns. This could aid in TBO improvement.
- (3) More freedom is gained in engine/airframe configuration integration.
- (4) Engines can be more compact due to the absence of fin spacing requirements.
- (5) Critical cooling requirements such as required for exhaust valves can be more easily met.
- (6) Fuel consumption during climbs should improve on the order of 10% due to the elimination of the requirement for cooling through rich operation.
- (7) Cooling drag associated with inefficient methods often encountered in air-cooled installations can be significantly reduced. (This drag can be significant as reported in References 36 and 142.)

- (8) Quieter operation may be expected due to the water jacket provided about the engine.
- (9) Cabin heat can be provided much more safely through a liquid heat exchanger than through present methods. The latter suffers from the distinct possibility of introducing exhaust fumes into the cockpit should the exhaust pipe heat exchanger fail.

#### Disadvantages:

٠.

1

- Single point failures resulting from a loss of coolant, thermostat failures in the closed position, or pump failures, are serious.
- (2) Warm up times will be extended. In cold weather, this could result in severe engine wear, with the distinct possibility of condensed products of combustion contributing to cylinder corrosion.
- (3) Deterioration of system efficiency through scaling suggests problems with system maintenance.
- (4) A vapor pocket in the coolant system can result in localized interruptions in cooling.
- (5) Engine maintenance costs may be expected to increase due to an increase in manhours required to remove or disassemble the engine. Air cooled systems offer the possibility of removing single cylinders from the engine.
- (6) Although dry weights may be expected to be on the order of air cooled systems, installed (wet) weights will be greater.

For the present evaluation, both reliability and safety were penalized by -1, thereby reflecting some optimism in the ability of technological improvements to offset Disadvantage No. 1. If this assumption should prove to be invalid, the severe penaltv which would result will make liquid cooling extremely unattractive. 4.5.9.7 <u>Improved Turbocharging</u>. Improved turbocharging promises to improve fuel efficiency and increase climb and cruise speed. by increasing the critical altitude and aircraft ceiling, significant improvements on the order of 10% to 15% can be expected in range.

However, procurement costs can be expected to be significantly higher if the improvements in turbocharging are dictated for the GA fleet alone. Today, for example, procurement costs are kept low due to a very high production base provided by the automotive and truck engine markets. A major manufacturer, for example, cites a production base on the order of one million units/year, with only 8,000 units being used by the GA fleet. The specific technology which promises to have the greatest impact on improved turbocharger performance appears to be that of air bearings.

It seems safe to predict that improved turbochargers will not find their way into the GA fleet until automotive requirements dictate their production. However, efforts toward improving fuel efficiencies (including the possibility of production of the PROCO or TCCS engines) may dictate their improvement. Accordingly, only a modest penalty in cost is assessed for this technology.

# 4.5.10 Structural Materials

Three technologies were considered and, as shown in Table 4.23, appear extremely attractive. Here, all three were evaluated on the basis of anticipated maturity in certification procedures and small purchase price penalties. Hence, in view of presently increasing manpower costs which would be significantly offset by composite manufacturing processes, the user was forecast to suffer no significant purchase price renalties. The advantages of composites which lie in cleaner aerodynamic shapes and weight reductions, allow for improvements in ceiling, cruise speed, fuel efficiency, and range.

	Figures of Merit for	
Technology 3	Airplane A 56/-53/85*	Airplane B 51/-57/58*
Fiberglass Composites	55	58
Kevlar Composites	51.	56
Graphite Composites	43	48

Table 4.23 Structural Materials

Total number of technologies/lowest FM/highest FM

4.5.10.1 <u>Fiberglass Composites</u>. These composites are attractive because they promise significant weight reductions at lower cost and risk than those associated with Kevlar or Graphite, although the latter two show greater tensile strength. Advantages noted also include the promise of reduced production cost, improved drag characteristics, low corrosion, and greater strength than aluminum. In one corrosion test, epoxy-fiberglass was noted to not corrode when in contact with anodized, primed, or painted aluminum surfaces. In fact, its inert and durable characteristics allow it to be used as one of several isolators for graphite, which reacts very strongly as a cathode to almost every metal. In one test, epoxy fiberglass was exposed to a 5% salt spray for 5000 hours and showed only superficial corrosion (Ref. 126). In another, a helicopter blade fabricated by Bell Helicopters was cracked and then run for an additional 900 hours without further failure (Ref. 120).

Its use in primary aircraft structures has seen only limited application in Rutan's canard configured vehicles, the Windecker Eagle, and the Bellanca Skyrocket, but this is seen to be largely the result of the over-designing required to certify a new production process, where weight advantages are quickly nullified. 4.5.10.2 <u>Kevlar Composites</u>. These aramid fibers have greater tensile strength than either fiberglass or graphite and are inherently inert. Current costs are on the order of \$33 per kg as a woven cloth vs graphite, which costs on the order of \$110 per kg in cloth form.

Hartzell's certified composite propeller is made from Kevlar. This blade is 50% lighter and twice as strong as its conventional counterpart but costs 2.5 times as much. As certification procedures are refined to allow the full potential of composites to be exploited, both cost and weight should be considerably reduced.

4.5.10.3 Graphite Composites. Most of the literature on composites published recently appears to deal with graphite. This composite has the strength and stiffness of steel, but only 60% of the density. Although it does not possess the tensile strength of Kevlar, its compressive strength is greater, and offers the potential for a 30% weight reduction in certain components. Although essentially inert when isolated, it acts as a strong cathode when in contact with aluminum and must be isolated through the use of other materials such as fiberglass. It is by far the most expensive of the three composites investigated, although costs have dropped from \$550-\$1200 per kilogram in 1968 to \$55-\$110 per kilogram currently. Here, the lower figure is for single fiber filaments while the upper figure represents the cost for its woven cloth form. As in many of the other technologies, its cost is expected to drop substantially if the auto industry forms a production base for its use.

As its characteristics become better understood, its intrusion into the GA market appears inevitable. At present, it is expected to be used extensively in the Lear Fan.

## 4.6 OTHER PROMISING TECHNOLOGIES

۰.

As one might suspect, certain promising technologies were omitted from the present evaluation due primarily to application considerations for airplane type as discussed previously in Section 4.2.3. In other words, a technology applicable to the jet fleet but not applicable to the single engine, light twin, or commuter

165

\$

airplane fleet could not be evaluated utilizing the present technique since the category weights were developed for specific mission profiles.

Two technologies receiving considerable attention in the current literature and which were not considered to be a part of the present research are the Prop Fan and the Quiet, Clean General Aviation Turbofan (QCGAT). Both of these technologies appear very promising for long-haul high speed cruise, and are discussed here.

4.6.1 Prop Fan

The now familiar 8-bladed propeller which has appeared in several periodicals offers extreme promise for the present turbofan-powered commercial fleet. It was not evaluated as part of Airplane A- or Airplane B-related technologies due to the fact that it is presently being developed for Mach numbers on the order of 0.8.

Presently, both an 8-bladed and a 10-bladed version have been investigated with extremely promising results. Fuel efficiency improvements on the order of 22% are forecast, with DOC improvements of 7% over comparable turbofans possible with improvements in design for maintainability (Ref. 46). The fuel figure is particularly impressive when one realizes that a 1% improvement provides an annual savings of 100 million gallons of aircraft fuel (Ref. 46) for the present civil fleet.

The major disadvantage of the prop fan is noise, and extensive studies are being performed to rectify this difficulty. When one

considers that military long endurance aircraft can realize 35% fuel savings and reduce gross weight by 25%, the additional weight penalty for noise treatment appears to be a small price. However, exterior noise still would not be resolved. (Note that noise cancelling techniques already employed through blade sweep have decreased sound pressure levels on the order of 6 db.) The interested reader is directed to References 46, 100, 101, and 141 for further information regarding prop fans.

#### 4.6.2 Quiet, Clean, General Aviation Turbofan (QCGAT)

AVCO Lycoming and Garrett AiResearch have both performed research in this area for NASA Lowis with promising results. As was the case for the prop fan, this technology was not evaluated as part of the present study due to the lack of applicability of turbofans to either Airplane A or B. In the research performed, Garrett chose to synthesize a stretched Lear 35 with a range of 3300 km (1780 nm) at an altitude of 12,200 m (40,000 ft) and M = 0.8 for its modified TPE 731-3. AVCO, on the other hand, had Beech synthesize an aircraft for its modified LTS 101 which resulted in a vehicle capable of 2780 km (1500 nm) at an altitude of 10,000 m (33,000 ft) and M = 0.6. It is emphasized that although the airplanes were synthesized, the modified engines were operational (although not flightworthy) and have been delivered to and tested by NASA-Lewis personnel. Results of the program ware extensively discussed at the General Aviation Propulsion Conference held at the Lewis Research Facility in Novembur 1979, and indicated

167

........

that major goals were met. Noticeably TSFC goals were not met under most conditions. However, all noise goals (which represented a 90% reduction in noise footprint) and most emissions goals were met.

Since a major thrust of the QCGAT program was to demonstrate large-engine technology transfers to small-engines, it appears that the program was quite successful. Certainly, the noise goals achieved are extremely impressive.

#### 4.7 SUMMARY OF THE TECHNOLOGY EVALUATION

ĥ

Several significant points made earlier in this chapter are repeated here for emphasis. Finally, a list of attractive and unattractive technologies will be disucssed.

# 4.7.1 Features of the Evaluation Technique

- (1) The rating yardstick used for Airplane A and Airplane B reflect a concensus of opinion both in the identification of relevant categories as well as in their quantification.
- (2) Responses to the three rounds of the survey to quantify the categories (Survey 2) ranged from 70% to 95%.
- (3) Only comparable technologies should be evaluated under the present technique. For example, high pressure, timed, fuel injection should not be compared with stratified charge reciprocating engines.
- (4) The relative figures of merit generated by different raters display significant stability in ranking different technologies despite differences in magnitude noted between raters.

- (5) Large changes in relative figures of merit between technologies are not observed when relative benefits are perturbed about an expected value. However, such changes may be expected if the state-of-the-art improves and results in an unforeseen application scenario.
- (6) The four technologies in the crashworthiness group appear somewhat underrated. This is attributable to the fact that improvements are restricted to only the crashworthiness category, and that this category had relatively low weights for both Airplane A and Airplane B.
- (7) Small differences in figures of merit should not be interpreted to rank one technology over another.
- (8) The ranking of technologies by figures of merit does not reflect risk in the present analysis.

# 4.7.2 Attractive Technologies

5

- (1) The GATE engine and the stratified charge rotary combustion engine dominate the rankings for Airplane A as shown in Table 4.9 and offer significant advantages to Airplane B as shown in Table 4.10.
- (2) All propulsion engines offer benefits to both airplanes.
- (3) Fiberglass, Kevlar, and graphite composites are all extremely attractive provided certification procedures are refined and e modest decrease in price is noted.

- (4) Aerodynamic concepts such as natural laminar flow airfoils, spoilers with full span flaps, and low/medium speed airfoils appear extremely attractive.
- (5) Integrated avionics systems which alleviate pilot workload such as DAAS and Digital Data Links are very attractive.

A very important consideration which deserves mention at this point is that many technologies which were eliminated from consideration due to the criteria of Section 4.2.3 are directly inferred from the list of attractive technologies just defined. For example, improved fuel injection, improved turbochargers, variable timed ignition systems, and ceramic technology are all implied for the improved positive displacement engines. Also, improvements in computational aerodynamics are required in order to permit three dimensional wing-body-tail analyses as well as wing-spoiler-flap analyses.

# 4.7.3 Unattractive Technologies

- Area navigation concepts such as LORAN C and OMEGA are fairly unattractive to the user although they offer significant savings in operations and maintenance expenses for the ground sites.
- (2) Active controls, Fly-by-Wire, and Fly-by-Light are extremely unattractive to both Airplane A and Airplane B, as are HUD's, Doppler, and Inertial Navigation.

#### CHAPTER 5

# DESIGN STUDIES

The design studies illustrate the direct and synergistic impacts of advanced technologies by comparing advanced and current technology airplanes that have been designed to the same specifications.

#### 5.1 DESIGN SPECIFICATIONS

The statement of work specified that conventional and canard configurations be developed for two types of airplanes incorporating appropriate advanced technologies. The preliminary specifications given are as follows:

(1) Design conventional and canard configurations for:

- (a) A small airplane with at least a 6-passenger cabin (including pilots), and
- (b) A large airplane with at least a 12-passenger cabin (excluding pilots).

(2) Performance objectives are:

- (a) Maximum L/D > 18
- (b) Cruise speed > 250 kt
- (c) Landing speed  $\leq$  60 kt

For design purposes, the two airplane types were more narrowly defined. The "small" airplane is defined as a 6-passenger (incluiing pilots) single engine business and/or personal airplane and

corresponds to Airplane A of Chapter 4. The "large" airplane is defined as a 19-passenger (excluding pilots) commuter airliner and corresponds to Airplane B of Chapter 4. The two types of airplanes are refered to as the "6-passenger" and "commuter" in this chapter.

2

#### 5.1.1 6-Passenger Design Specifications

Characteristics of existing 6-passenger airplanes were used as guidelines in completing the 6-passenger design specifications. As shown in Table 5.1, the resulting specifications were split into "required" and "desired" sets. This was done because  $L/D \ge 18$ strongly conflicts with  $V_{stall} \le 60$  kt. Furthermore, since it was felt that landing and takeoff distances are much more important than the speeds involved, the stall speed requirements were relaxed.

# 5.1.2 Commuter Design Specifications

The characteristics of existing commuter aircraft were used as guidelines in establishing the commuter design specifications. Again, the resulting specifications were divided into "required" and "desired" sets. The "desired" requirements for the commuter were relaxed extensively because of its operating envelope. Commuter aircraft operate at low altitudes, making a desired L/D of 18 extremely difficult to attain. Furthermore, a landing speed of less than 60 knots is somewhat unrealistic in the size of commuters typical of this study. Balanced field length, together with landing distance, is more important than landing speed. The resultion specifications follow in Table 4.5.2.

# Table 5.1. 6-Passenger Design Specifications

Required:	(1)	Seating for 6 persons (including 2 crew)
		with baggage.
	(2)	$v_{\text{cruise}} \geq 250$ kt at gross weight.
	(3)	Range $\geq$ 900 nm with 45 min. reserves
		carrying 6 passengers and baggage.
	(4)	Takeoff distance over a 10.7 m obstacle
		$\leq$ 610 m, sea level (ISA) and maximum
		gross weight.
	(5)	Landing distance over a 15.2 m obstacle
		$\leq$ 610 m, sea level (ISA) and maximum
		gross weight.
	(6)	Meet FAR 23 requirements except for
		V <sub>stall</sub> as noted in (8).
Desired:	(7)	Maximum $L/D \ge 18$ cruise.
	(8)	$v_{stall} \leq 60$ kt in the landing configura-
		tion at maximum gross weight and sea
		level (ISA).

# 5.2 TECHNOLOGY INTEGRATION

A major preliminary task performed during the course of the design study was the identification of those technologies to be incorporated into the designs. The technology evaluation of Chapter 4 was helpful in this respect and served as a guideline in selecting particular technologies.

#### Table 5.2. Commuter Design Specifications

(1) Seating for 21 persons (including 2 crew) Required: with baggage. (2)  $V_{cruise} \ge 250$  kt at maximum takeoff weight at 3050 m MSL (ISA). (3) 2000 nm ferry range where ferry range is determined for full fuel + 2 crew (only) at 250 kt with 45 min. reserves. (4) Segmented range of nine 87 nm legs without refueling with a 60% load factor (12 pax) and with 45 min. reserves. (5) Single leg range of at least 400 nm at 100% load factor with 45 min. reserves. (6) Balanced field length of 1370 m or less at sea level (ISA) and maximum takeoff weight. (7) Meet FAR part 25 requirements. (8) (a) Maximum  $L/D \ge 18$  at cruise. Desired: (b)  $V_{stall} \leq 60$  kt in the landing configuration at maximum gross weight

It should be noted that the impact of many of the evaluated technologies cannot be determined by design studies. For example, while the impact of composite materials may be evaluated, the effects of load limiting seats and CRT displays are much harder to quantify. In general, only technologies which directly influence the mission performance of an airplane are capable of being analyzed.

# 5.2.1 <u>6-Passenger Technology Integration</u>

As one would expect, the highest ranking technologies idantified in the analysis of Chapter 4 were in the areas of propulsion, structures, and aerodynamics. Specific technologies from toch of these areas were selected and are termed "Primary Technologies" in that they directly influence airplane performance. Other rechnologies which do not directly influence airplane performance were also incorporated and are termed "Secondary Technologies." Table 5.3 lists the Primary and Secondary Technologies which were incorporated into the advanced technology 6-passenger airplanes.

The reasons for selecting the technologies in Table 5.3 are as follows:

(1) Spoilers and Full Span Fowler Flaps

Trade studies (see Section 5.3.1) showed that high wing loadings (1915 to 2394 N/m<sup>2</sup>) markedly improve cruise efficiency but degrade takeoff and landing performance. The obvious solution here is to utilize spoilers for primary roll control and free the trailing edge of the wing for full span flaps. Fowler flaps are used because they provide high lift and low drag at small deflections (for takeoff) and higher lift and drag at large deflections (for landing).

# Table 5.3. 6-Passenger Advanced Technologies

ż

Primary Technologies	Secondary Technologies
Spoilers for roll control	Load limiting seats
Full span Fowler flaps	Energy absorbing floor
Advanced natural laminar flow airfoil	Improved restraints
Kevlar and/or graphite composite structure	Integrated avionics and displays
Stratified charge, highly turbocharged rotary combustion engine	Integrated low-cost fluidic wing leveler
Quiet, efficient propeller technology	

Primary Technologies are those which directly influence airplane performance while Secondary Technologies do not.

# (2) Advanced Natural Laminar Flow Airfoil

Preliminary data show that computer generated natural laminar flow (NLF) airfoils have low-drag characteristics approaching those of NACA 6-series airfoils and maximum lift coefficients resembling those of the turbulent-flow NASA LS-series airfoils, which are also computer generated. Furthermore, the advanced NLF airfoils are far less sensitive to roughness (in terms of maximum lift coefficient) than the NACA 6-series airfoils.

(2) Kevlar and/or Graphite Composite Structure Advanced composites can significantly reduce structural weight while providing a stiff and aerodynamically smooth

surface. Several studies indicate that major cost savings are possible due to a reduction in the number of parts and man-hours required for production.

(4) Stratified Charge, Highly Turbocharged Rotary Combustion (RC) Engine

This engine is characterized by a high power to weight ratio, good specific fuel consumption, and liquid cooling. The RC engine was selected instead of the GATE turboprop primarily because the former is undergoing tests while the latter is a paper study. Furthermore, the GATE engine is evolving into a higher horsepower category than required for the 6-passenger airplane.

(5) Quiet-Efficient Propeller Technology

This includes computer generated blade sections, optimized planforms, composite blades, etc. Cruise efficiency gains are relatively small (2-3%) but significant improvements in takeoff and climb efficiencies and noise characteristics are possible.

(6) Load Limiting Seats and Energy Absorbing Floor These technologies can reduce peak vertical g-loadings by 50% and therefore greatly reduce the risk of death or injury in the event of a crash. Little or no weight and cost penalties are expected.

(7) Improved Restraints

Improved restraints are a minor technology in terms of airplane weight and cost but greatly enhance survivability by

insuring that the occupant remains in his seat during a crash. If load limiting seats are used, effective restraints become mandatory. The only problems identified appear to be those assoicated with comfort and ease of use.

(8) Integrated Avionics and Displays

While not directly influencing performance, this technology can improve the utility and flexibility of an airplane and can be considered to increase safety since it has the potential to substantially reduce pilot workload. The degree of sophistication of the system as applied to the 6-passenger airplane has not been defined but the capabilities of the technology in general are illustrated by the NASA Ames PCAAS and DAAS studies.

(9) Integrated Low-Cost Fluidic Wing Leveler This wing leveler is visualized as a basic part of the airplane. Because most airplanes are not spirally stable, this technology can significantly improve safety, particularly in IFR conditions by reducing pilot workload. This system is commercially available to the homebuilt market for less than \$200.

# 5.2.2 Commuter Technology Integration

The discussions of advanced technologies integrated into the 6-passenger designs apply to the commuter designs. Identical technologies were integrated with two exceptions: (1) powerplant

and (2) airfoil. Table 5.4 presents the advanced technologies incorporated into the commuter design.

* Primary Technologies	Secondary Technologies *
Spoilers for roll control	Load limiting seats
Full span Fowler flaps	Energy absorbing floor
Advanced low-speed airfoil	Improved restraints
Kevlar and/or graphite composite structure	Integrated avionics and displays
GATE technology turboprop engine	
Efficient propeller technology	

Table 5.4. Commuter Advanced Technologies

Primary Technologies are those which directly influence aircraft while Secondary Technologies do not.

The technologies not previously discussed but applicable to the commuter are:

(1) GATE technology turboprop engine.

This engine was selected because of its high power-to-weight ratio, low purchase price, and good specific fuel consumption. Other advantages attributed to turboprops include high reliability and long engine lifetimes.

Ş

(2) Advanced low-speed airfoil.

The advanced NASA low-speed airfoils (LS-series) have characteristics similar to the original GA(W) series of

airfoils and exhibit good maximum lift characteristics together with improved pitching moment characteristics. Also, the thickness of the low-speed section assists in maintaining low wing weights while providing a high volume for fuel.

#### 5.3 PARAMETRIC TRADE STUDIES

Parametric trade studies were conducted to establish the impact of basic airplane parameters (wing loading, aspect ratio, power loading, etc.) on mission performance.

#### 5.3.1 6-Passenger Trade Studies

Three types of trade studies were conducted covering cruise, taekoff, and landing performance. Detailed derivations may be found in Appendix C.1.

The cruise trades were formulated to show the affective power loading ( $[n_p P/W]$ , hp/lb) required to cruise at 250 kt as a function of wing loading (W/S), effective aspect ratio (Ae), zero lift drag coefficient ( $C_{D_o}$ ), and altitude. It quickly became apparent that the high L/D requirement necessitated high cruising altitudes. Figures 5.1 and 5.2 present the cruise trades for altitudes of 6100 m and 9140 m, respectively.

The takeoff trades were expressed in terms of wing loading (W/S) and power loading (P/W) required for a 610 m takeoff distance over a 15.2 m obstacle. (This requirement was later changed to a 610 m takeoff distance over a 10.7 m obstacle.) The requirement proved to be a very strong function of the maximum

takeoff lift coefficient (C ) and a very weak function of  $\max_{\text{max}_{TO}}$  aspect ratio (A) and zero lift drag coefficient (C ). These characteristics led to a fairly simple representation of the takeoff problem as shown in Figure 5.3.

Velocity = 250 kt Altitude = 6100 m



Fig. 5.1. 6-Passenger Cruise Trade Studies (at 6100 m)



Fig. 5.2. 6-Passenger Cruise Trade Studies (at 9140 m)



Fig. 5.3. 6-Passenger Takeoff Trade Studies

The landing trades show wing loadings and maximum landing lift coefficients (C ) required for a 610 m landing max\_LDG distance over a 15.2 m obstacle and for a stall speed of 60 kt. Figure 5.4 gives the results of the landing trade study.

The trade studies of Figures 5.1-5.4 are summarized in Table 5.5.

As a result of the trade studies, it was decided that relaxing the  $V_{stall} \leq 60$  kt requirement was justified since it would otherwise arbitrarily restrict the application of advanced technologies. Airplane parameters and design goals for the advanced technology 6-Passenger airplanes were established as follows:

Parameters:

W/S = 2155 N/m<sup>2</sup> (45 1b/ft<sup>2</sup>) A = 11 h<sub>cruise</sub> = 9144 m (30,000 ft)

Goals:

$$C_{D_{O}} \leq .02$$
  
 $e \geq .75$   

$$C_{L_{max_{TO}}} \geq 2.5 \text{ and } C_{D_{O}} \leq .10$$
  

$$C_{L_{max_{LDG}}} \geq 3.0$$

5.3.2 Commuter Trade Studies

Two types of trade studies were conducted for the commuter aircraft and consisted of cruise and balanced field length performance. Detailed derivations are included in Appendix C.2.



. .

Fig. 5.4. 6-Passenger Landing Trade Studies

## Table 5.5. 6-Passenger Trade Study Summary

CRUISE TRADE STUDY

Current high performance single engine airplanes typically have  $C_D \approx .02$ , Ae  $\approx 5.5$ , and W/S  $\approx 1005$  N/m<sup>2</sup> which results in L/D =

11 at 9140 m.

L/D = 18 requires much higher W/S (2155 N/m<sup>2</sup>) and larger Ae (80) for  $C_D = .02$ , and  $C_D = .02$  is very difficult to achieve with  $W/S = 2155 N/m^2$ .

TAKEOFF TRADE STUDY

P/W required for constant W/S is a very strong function of с<sub>г тах</sub>то

Because cruise requirements demand high W/S, advanced flap technology (full span Fowler flaps) becomes a necessity if reasonable P/W is to be maintained.

LANDING TRADE STUDY

 $V_{stall} \leq 60$  kt is a very limiting requirement compared to the 610 m landing distance requirement.

If the V<sub>stall</sub> requirement is adopted, it becomes the critical factor in selecting W/S. If the landing distance requirement is used instead, the takeoff distance requirement becomes critical.

Similar to the 6-passenger studies, the commuter cruise studies show the effective power loading ( $\eta_{\rm p}~P/W)$  required to cruise at 250 kt and 3050 m as a function of wing loading (W/S), effective aspect ratio (Ae) and zero lift drag coefficient ( $C_{D_{o}}$ ). Figure 5.5 presents the cruise trade studies.


Fig. 5.5. Commuter Cruise Trade Studies

•

•

The balanced field length studies were developed in terms of  $C_{D_o}$ , effective aspect ratio Ae, maximum lift coefficient in taksoff configuration ( $C_{L_{max_{TO}}}$ ), and effective power loading ( $n_p$  P/W). The max<sub>TO</sub> studies show that balanced field length is dependent primarily on  $C_{L_{max_{TO}}}$  and ( $n_p$  P/W). Results of this study are presented as max<sub>TO</sub> Figures 5.6, 5.7, and 5.8, and were developed for values of  $C_{D_o} = .075$  and Ae = 10.

:



Fig. 5.6. Commuter Balanced Field Length Studies



•,

Fig. 5.7. Commuter Balanced Field Length Studies



Fig. 5.8. Commuter Balanced Field length Studies.

Observations resulting from an examination of these figures appear in Table 5.6.

Table 5.6. Commuter Trade Study Summary

CRUISE TRADE STUDIES Current commuters have values of  $C_{D_0} \approx .025$ , Ae  $\approx 6.0$ , and W/S  $\gtrsim 2155 \text{ N/m}^2$ , at a cruise altitude of 3050 m. L/D = 18 demands extremely low values of  $C_{D_0} (\approx .02)$ , coupled with high wing loadings ( $\approx 3352 \text{ N/m}^2$ ). As was the case with the 6-passenger design a  $C_{D_0} \approx .02$  appears very difficult to achieve under conditions of high wing loadings.

BALANCED FIELD LENGTH

Because of the high wing loadings desired in cruise flight, advanced flap technology is required (full span Fowler flaps). High C in takeoff allows lower power loadings for takeoff max and thereby maintains a more reasonable difference between cruise and takeoff power requirements.

These results dictated relaxation of two of the original specifications. First, the requirement for  $V_{stall} \leq 60$  kt is not practical for the commuter aircraft since these aircraft (in the 18-19 passenger range) typically are not constrained by this requirement. A survey for a commuter traffic model presented in Reference 44 showed that 93% of the airports in use had effective runway lengths of 1220 m or greater, where effective length is defined as the actual runway length corrected to sea level standard conditions. Hence, balanced field length as

opposed to stall speed is more critical in the determination of accessible airports. Second, the L/D requirement of 18 was relaxed to a value of 12 due to a desired cruise altitude of 3050 m.

This altitude was selected because available data indicated that the bulk of the commuter market could be accessed without having to cruise at higher altitudes. Figure 5.5 clearly indicates that  $L/D \approx 12$  is a more realistic goal. The design goals established for the commuter aircraft were:

#### Parameters:

 $W/S = 3112 N/m^2$ 

A = 12

Goals:

$$C_{D} \approx .020 - .025$$
  
e  $\approx .75$   
$$C_{L} = 2.5$$
  
max<sub>TO</sub>

## 5.4 CONCEPTUAL DESIGNS

This section describes the conceptual design of conventional and canard configurations for the 6-passenger and commuter airplanes which incorporate the advanced technologies discussed in Section 5.2. Current technology baseline  $\varepsilon$ irplanes were also established to provide a basis for comparison of the advanced technology designs.

The project staff was aided in this effort by the senior design class of the Aerospace Engineering Department at the University of Kansas. The additional manpower proved valuable to the design effort and assisted in identifying unforeseen difficulties associated with technology integration and configuration analysis.

#### 5.4.1 6-Passenger Designs

A current technology baseline, an advanced technology conventionally configured vehicle, and a canard configuration were synthesized.

All three designs utilize the same basic cabin, which seats six passengers (all facing forward), and has a baggage area behind the rear seats. The cabin, which is illustrated in Figure 5.9, is somewhat idealized since it was common to all three designs. A more involved design effort would probably result in a more rounded cross-section with contours more closely matched to the configuration in question.

Design of the 6-Passenger airplanes followed the methods outlined below. Detailed procedures may be found in Appendix C.

(1) Preliminary Sizing

An initial gross weight was estimated using the Breguet range equation and the wing was sized accordingly. Tail areas were sized with tail volume coefficients obtained from similar types of existing airplanes.

(2) Weight and Balance

Component weights were calculated giving an improved weight estimate and providing center of gravity information.

(3) Horizontal Tail Sizing

The horizontal tail area required was determined from static margin and takeoff rotation requirements.



Fig. 5.9. 6-Passenger Cabin Layout

(4) Aerodynamics

Airplane wetted area and zero lift drag coefficient were calculated.

(5) Final Sizing

GASP (General Aviation Synthesis Program) was used to "fly" the airplane. Engine size and gross weight were adjusted to meet the design specifications. Prior to the actual sizing, GASP weight and drag routines were calibrated against the manual calculations.

The three 6-Passenger designs are discussed separately in the following sections.

5.4.1.1 <u>Current Technology Baseline</u>. The 6-Passenger current technology baseline (6PAXBL) provides a basis of comparison for the advanced technology airplane.

In designing 6PAXBL, characteristics of current high performance, single engine, general aviation airplanes were retained as much as possible. The major exception was that a 7620 m cruise altitude was assumed so that the power required at cruise would not result in an exceptionally large engine. Specifications for this airplane are those shown in Table 5.1 with the exception of the L/D > 18 goal. 6PAXBL has the following characteristics:

Geared, fuel injected, turbocharged, air-cooled, 6-cylinder
 (opposed) reciprocating engine.

(2) Conventional aluminum structure.

- (3) Partial span, single-slotted flaps.
- (4) Conventional control surfaces.
- (5) Cruise altitude of 7620 m; pressurized to maintain a 2440 m cabin.
- (6) NACA 64<sub>2</sub>-A215 airfoil.
- (7) W/S = 1053.4 N/m<sup>2</sup> (22 1b/ft<sup>2</sup>).
- (8) A = 7.5.

The design procedures discussed in Section 5.4.1 resulted in the 6PAXBL design shown in Figure 5.10 and Table 5.7 summarizes the characteristics and performance of this airplane.

Referring to Table 5.7 one notes that 6PAXBL is a high performance airplane which has speed, payload, and range characteristics more similar to those of the Bellanca Skyrocket than to those of more conventional general aviation airplanes. A comparison of 6PAXBL to the advanced technology designs and to existing airplanes is discussed in Section 5.5.

5.4.1.2 <u>Advanced Technology Conventional Configuration</u>. The 6-Passenger conventional configuration, advanced technology airplane (6PAXAD) incorporates the technologies listed in Table 5.3. The trade studies of Section 5.3.1 were used to define the basic airplane parameters.

No real problems were encountered in the conceptual design of 6PAXAD but detailed design of the flap and spoiler systems would be complicated by the small size of the wing. The general characteristics of this airplane are as follows:



•



Table 5.7. 6-Passenger Current Technology Baseline Characteristics

\*

```
WEIGHTS
Gross = 1859.8 \text{ kg} (4100 \text{ 1b})
                                       Payload = 544.3 kb (1200 lb)
Empty = 1051.4 kg (2318 lb) Fuel (max payload) = 264 kg (582 lb)
    ENGINE
Geared, fuel injected, turbocharged, air-cooled 6-cylinder
(opposed) reciprocating
   Max Power = 317 \text{ kW} (425 \text{ hp})
   Critical Altitude = 7620 \text{ m} (25,000 \text{ ft})
   Cruise BSFC = .274 \frac{\text{kg}}{\text{kW-hr}} (.450 \frac{1\text{b}}{\text{hp-hr}})
   Dry Specific Weight = .894 kg/kW (1.47 lb/hp)
   TBO = 1800 hr
   PERFORMANCE
V cruise = 250 kt (75% power, gross weight)<sup>1</sup>
Range = 900 nm + 45 min. reserves (250 kt, 544.3 kg (1200 lb)
   payload)<sup>2</sup>
L/D = 11.7 (250 \text{ kt, gross weight})^{1}
V = 58.3 kt (flaps 40°)
Takeoff dist. over 10.7 m (35 ft) obstacle = 540 m (1771 ft)
    (flaps 10^\circ)<sup>2</sup>
Landing dist. over 15.2 m (50 ft) obstacle = 351 \text{ m} (1152 ft)
    (flaps 40^{\circ})^2
```

<sup>7</sup> 7620 m (25,000 ft) altitude

<sup>2</sup> Sea level, gross weight

- Geared, stratified charge, highly turbocharged, liquid cooled, 2-rotor, rotary combustion engine.
- (2) Composite (Kevlar and/or graphite) structure.
- (3) Full span single-slotted Flower flaps.
- (4) Spoilers for roll control.
- (5) Cruise altitude of 9140 m; pressurized to maintaina 2440 m cabin.
- (6) NASA advanced natural laminar flow airfoil.
- (7) W/S = 2154.6 N/m<sup>2</sup> (45  $1b/ft^2$ )
- (8) A = 11

Here, the design procedures previously discussed in Section 5.4.1 resulted in the 6PAXAD design shown in Figure 5.11 and Table 5.8 summarizes its characteristics and performance.

Table 5.8 shows that the 6PAXAD design meets the required specifications of Table 5.1 but does not have the desired  $L/D \ge 18$  or  $V_{stall} \le 60$  kt. A detailed discussion of 6PAXAD together with a comparison with 6PAXBL is presented in Section 5.5. 5.4.1.3 <u>Advanced Technology Canard Configuration</u>. The 6-Passenger canard configuration, advanced tehenology airplane (6PAXC) incorporates the technologies listed in Table 5.3 and the trade studies of Section 5.3.1 were again used to define the basic airplane parameters.

Several problems were encountered in the design of 6PAXC. First, methods for sizing the canard and for calculating induced drag were lacking. Second, GASP will not handle canards. For



Fig. 5.11. Advanced Technology Conventional Configuration 6-Passenger Design

# Table 5.8. 6-Passenger Advanced Technology Conventional Configuration Characteristics

```
WEIGHTS
Gross = 1258.7 kg (2775 1b)
                                      Payload = 544.3 kg (1200 lb)
Empty = 582 kg (1283 1b) Fuel (max payload) = 132.5 kg (292 1b)
   ENGINE
Geared, stratified charge, highly turbocharged, liquid-cooled
2-rotor, rotary combustion
   Max Power = 205 \text{ kW} (275 \text{ hp})
   Critical Altitude = 9144 \text{ m} (30,000 \text{ ft})
   Cruise BSFC = .231 \frac{kg}{kW-hr} (.38 \frac{1b}{hp-hr})
   Dry Specific Weight = .45 kg/kW (.74 lb/hp)
   TBO = 3000 hr
   PERFORMANCE
V<sub>cruise</sub> = 250 kt (75% power, gross weight)<sup>1</sup>
Range = 900 nm + 45 min. reserves (250 kt, 544.3 kg (1200 1b)
   payload)<sup>2</sup>
L/D = 14.2 (250 \text{ kt, gross weight})^1
V_{stall} = 64.5 \text{ kt} (flaps 40^\circ)^2
Takeoff dist. over 10.7 m (35 ft) obstacle = 610 m (1999 ft)
   (flaps 20°)<sup>2</sup>
Landing dist. over 15.2 m (50 ft) obstacle = 406 m (1332 ft)
    (flaps 40^\circ)^2
```

<sup>1</sup> 9144 m (30,000 ft) altitude <sup>2</sup> Sea level, gross weight these reasons, 6PAXC cannot legitimately be compared to 6PAXBL or 6PAXAD, and as presented here, must be considered only as a possible configuration and not as a refined design. These problems are discussed in greater detail in Section 5.4.3.

General characteristics of the canard design are the same as those of 6PAXAD as follows:

- Geared, stratified charge, highly turbocharged, liquid cooled, 2-rotor, rotary combustion engine.
- (2) Composite (Kevlar and/or graphite) structure.
- (3) Full span single-slotted Fowler flaps.
- (4) Spoilers for roll control.
- (5) Cruise altitude of 9140 m; pressurized to maintaina 2440 m cabin.
- (6) NASA advanced natural laminar flow airfoil.

(7) 
$$W/S = 2154.6 \text{ N/m}^2$$
 (45 1b/ft<sup>2</sup>)

(8) A = 11

I

The full span flaps called for may create a trim problem but the extent of the problem, if it exists, was not investigated due to the previoually mentioned lack of reliable analysis techniques.

Figure 5.12 presents a 3-view of the advanced technology canard configuration.

## 5.4.2 Commuter Designs

An advanced technology conventional configuration and an advanced technology canard configuration were synthesized. A third existing aircraft was selected as a baseline for comparison.



. .



The existing aircraft selected has a 19 passenger capability and was selected because it is one of very few aircraft specifically designed as a commuter.

Both advanced technology conceptual designs utilize the same fuselage interior. Accommodations are included for 19 passengers (sitting three abreast) and two pilots. Additional space is included for cabin furnishings and a flight attendant. A baggage/cargo area is included behind the passenger area, with access through an upward opening cargo door. The passenger area of the aircraft is entered through a smaller door at the front of the cabin. Figure 5.13 presents this interior arrangement.

Design of the advanced technology aircraft followed the procedures outlined below. Detailed data and calculations may be found in Appendix C.2.

(1) Preliminary Sizing

The gross weight of each of the advanced designs was matched to the baseline aircraft. Tail areas were determined initially by minimum static stability requirements.

(2) Weight and Balance

A detailed weight and balance calculation was performed for each of the designs because interior furnishings and accommodations for these aircraft constitute a large portion of the aircraft weight. ?

(3) Horizontal Tail Sizing

The horizontal tail was sized for static margin and rotation requirements.



.

. .

Fig. 5.13. Commuter Cabin Layout

-

#### (4) Aerodynamics

Airplane wetted area and zero lift drag coefficient were calculated.

(5) Final Sizing

GASP was used to perform the mission analysis of the aircraft. Here, gross weight was maintained while engine size was adjusted to meet design specifications. Two specific missions were analyzed. The first involved a single leg mission at gross weight, while the second was characterized by nine 87 nm legs at a 60% load factor. Both missions were flown at 3048 m. Prior to mission analysis, GASP was calibrated to obtain the drag and weight characteristics which were predicted through manual calculations.

The two advanced designs and the baseline aircraft are discussed separately in the following sections.

5.4.2.1 <u>Current Technology Baseline</u>. The aircraft utilized for the current technology baseline (COMBL) appears in Figure 5.14.

The characteristics of COMBL are:

- (1) Twin turboprop engines.
- (2) Conventional aluminum structure.
- (3) Partial span, double-slotted flaps.
- (4) Conventional control surfaces.
- (5) Cruise altitude = 3048 m.
- (6) NASA 65<sub>2</sub>A215 at the root, NACA 64<sub>2</sub>A413 at the tip.
- (7) W/S = 2157 N/m2
- (8) A = 7.71



Sauffilisis.

- vat enformation for all react

۰ ۳



and the second s

The performance characteristics of COMBL are summarized in Table 5.9.

# Table 5.9. Commuter Current Technology Baseline Characteristics

-			
	WEIGHTS		
	Gross = 5670 kg (12500 lb)		
	Empty = 3379 kg (7450 lb)		
	Max Fuel weight = 1969 kg (4342 1b)		
	ENGINES		
	Turboprop with emergency methanol injection		
	Max Power 708.4 kW (950 hp)		
	Cruise BSFC = .335 $\frac{\text{kg}}{\text{kW-hr}}$ (.55 $\frac{1\text{b}}{\text{hp-hr}}$ )		
	Dry Specific Weight = 4.36 kW/kg (2.65 hp/1b)		
	TBO = 3000 hr		
	PERFORMANCE		
	V <sub>cruise</sub> = 250 kt @ 60% power, 3048 m (10,000 ft),5670 kg (12,500 1b)		
	L/D = 10.44 [ 3048 m (10,000 ft), 5670 kg (12,500 lb) ]		
	V <sub>stall</sub> = 87 kt (flaps 36°, sea level, 5670 kg (12,500 lb))		
	Takeoff Distance over 10.7 m (35 ft) obstacle = 914 m (3000 ft) (flaps 0°, sea level, 5670 kg (12,500 lb))		
	Landing Distance over 15.2 m (50 ft) obstacle = 897 m (2944 ft) (flaps 36°, sea level, 5670 kg (12,500 lb))		
	<pre>Balanced Field Length = 1149 m (3771 ft)(takeoff configuration, 5670 kg (12,500 lb))</pre>		
	Single Engine Service Ceiling = 3962 m (13,000 ft) (5670 kg (12,500 lb))		

The performance characteristics of COMBL are summarized in Table 5.9.

#### Table 5.9. Commuter Current Technology Baseline Characteristics

WEIGHTS Gross = 5670 kg (12500 lb)Empty = 3379 kg (7450 lb)Max Fuel weight = 1969 kg (4342 lb)ENGINES Turboprop with emergency methanol injection Max Power 708.4 kW (950 hp) Cruise BSFC = .335  $\frac{\text{kg}}{\text{kW-hr}}$  (.55  $\frac{1\text{b}}{\text{hp-hr}}$ ) Dry Specific Weight = 4.36 kW/kg (2.65 hp/1b) TBO = 3000 hrPERFORMANCE V cruise = 250 kt @ 60% power, 3048 m (10,000 ft),5670 kg (12,500 1b) L/D = 10.44 [ 3048 m (10,000 ft), 5670 kg (12,500 lb) ] V stall = 87 kt (flaps 36°, sea level, 5670 kg (12,500 lb)) Takeoff Distance over 10.7 m (35 ft) obstacle = 914 m (3000 ft) (flaps 0°, sea level, 5670 kg (12,500 lb)) Landing Distance over 15.2 m (50 ft) obstacle = 897 m (2944 ft) (flaps 36°, sea level, 5670 kg (12,500 lb)) Balanced Field Length = 1149 m (3771 ft)(takeoff configuration, 5670 kg (12,500 lb)) Single Engine Service Ceiling = 3962 m (13,000 ft) (5670 kg (12,500 lb))

5.4.2.2 <u>Conventional Configuration Advanced Technology Commuter</u>. The conventional configuration, advanced technology commuter (ADCOM) incorporates the technologies listed in Table 5.4. The trade studies of Section 5.3.2 were used as guidelines for the definition of the basic airplane parameters.

2

The conceptual design of ADCOM proceeded rather smoothly, and only one major difficulty was encountered. Because of the small wing size, the wing volume available for fuel became critical. However, a manual calculation using conservative approximations revealed that available wing volume could accommodate the mission fuel. Figure 5.15 presents a 3-view of ADCOM.

Also, the characteristics of ADCOM are presented in Table 5.10.

5.4.2.3 <u>Canard Configured Advanced Technology Commuter</u>. The problems encountered with the 6-Passenger canard design also hampered the commuter canard design. Initial sizing and wing/ canard location were determined utilizing conventional configuration analysis methods. For this reason, the advanced technology canard commuter (ADCOMCN) should closely approximate the results which could be expected from more sophisticated methods. However, lacking verification, this canard should not be compared with the advanced technology conventional configuration (ADCOM) or the baseline aircraft (COMBL).

Figure 5.16 presents a 3-view of ADCOMCN, which is characterized by the following:



The second second second



:

# Table 5.10. Advanced Technology Conventional Configuration

```
Commuter Characteristics
   WEIGHTS
Gross = 3706 kg (12,580 1b)
Empty = 3058 kg (6,742 lb)
Max Payload = 1724 kg (3800 1b)
   ENGINES
GATE Turboprop
   Kax Power = 457 kW (613 hp)
   Cruise BSFC = .274 \frac{\text{kg}}{\text{kW-hr}} (.45 \frac{1\text{b}}{\text{hp-hr}})
   Dry Specific Weight = 3.29 kW/kg (2.0 hp/1b)
   TBO = 3000 hr
   PERFORMANCE
V<sub>cruise</sub> = 250 kt @ 70% power, 3048 m (10,000 ft), 5706 kg (12,580 lb)
L/D = 12.5 @ 5706 kg (12,580 lb), 3048 m (10,000 ft)
V<sub>stall</sub> = 84 kt (flaps 40°, sea level, 5706 kg (12,580 lb))
Takeoff Distance over 10.7 (35 ft) obstacle = 1047 m (3600 ft)
   (flaps 20°, sea level, 5706 kg (12,580 lb))
Landing Distance over 15.2 m (50 ft) obstacle = 588 m (1930 ft)
   (flaps 40°, sea level, 5606 kg (12,580 lb))
Balanced Field Length = 1097 m (3600 ft) (flaps 20°, sea level,
   5706 kg (12,580 1b))
Single Engine Service Ceiling = 2475 m (8120 ft) 5706 kg
   (12,580 1b)
```



-

144

Fig. 5.16. Advanced Technology Commuter Canard Configuration

(1) Twin turboprop engines.

- (2) Composite (Kevlar and/or graphite) structure.
- (3) Full span, single-slotted Fowler flaps.
- (4) Spoilers for roll control.
- (5) Cruise altitude = 3048 m.
- (6) LS(M) 0417 low-speed airfoil.
- (7)  $W/S = 3259 N/m^2$ .
- (8) A = 12

ġ

Both the conventional and canard advanced technology configurations utilize the same fuselage, with the same geometrical parameters for the wing, with the exception of sweep.

The canard configuration does offer some interesting side effects in configuration analysis. Conventional methods of analysis indicated that wing sweep was required to properly balance the aircraft while maintaining an aerodynamic center location that was acceptable from a static longitudinal stability standpoint.

Although wing weight generally increases with the addition of sweep, increasing the wing root chord such that the wing trailing edge sweep may be reduced to zero (as when a "Yehudi" is incorporated) will allow the relocation of the main landing gear from within the torque box to a position aft of the rear spar. The resulting torque box weight savings may be expected to offset the weight penalty incurred through wing swcep.

Additional characteristics of the canard are examined in Section 5.4.3.

## 5.4.3 <u>Canard Analysis Problems</u>

Development of the 6-passenger and commuter canard configurations was halted because of an absence of reliable canard analysis methods. The difficulty here lies in the fact that present methods do not account for the effects of wake deformation from the canard on the main (aft) wing. Although this situation does not pose severe limitations at low angles-of-attack or when the canard is not highly loaded, it was suspected that non-shortcoupled canards under either of these conditions would display characteristics which would be significantly different than those predicted by present methods.

:

Initial attempts to rectify the situation involved the use of the Quasi Vortex Lattice Method (QVLM) of Lan (Ref. 116). Here, the configuration of Figure 5.17 (Ref. 68) was used as a validation case. This particular configuration was wind tunnel tested at the NASA Langley Research Center under conditions similar to the flight regime of the advanced technology canard configurations. Although QVLM does not model a deformed wake, this effect was expected to be minimal at low angles-of-attack due to the short-coupled nature of the test configuration. Leading edge separation was assumed at the higher angles of attack, and the strakes were not included in the QVLM-tested configuration.

Figure 5.18 illustrates the results of the computer analysis where fairly good agreement with the tunnel data is noted. Here, the observed differences are believed to result from augmented vortex lift generated on the wing by canard vortices. However,



**4.** #

Fig. 5.17. QVLM Validation Configuration



Fig. 5.18. Comparison of Experimental and OVLM Results for the Validation Configuration

further attempts to test QVLM on non-short-coupled configurations were hampered by an absence of published wind tunnel results for these configurations. Furthermore, an investigation of conventional vortex lattice methods (VLM) indicated that all of those codes which were examined employed undeformed, planar wakes. These observations, together with the following three points, led

eventually to a decision to halt further canard analysis.

- GASP results for the conventional configurations reflected angles-of-attack on the order of 8° to 10° under certain low speed flight conditions.
- (2) Preliminary canard configurations appeared to favor nonshort-coupled canards.
- (3) Current successful high aspect ratio canard configurations are characterized by very heavily loaded canards.

#### 5.5 DESIGN STUDY SUMMARY

This section summarizes the results of the design study. Advanced and current technology airplanes are compared using GASP and a parametric analysis. As previously mentioned, the canard configurations were not fully developed and are therefore omitted in these comparisons.

## 5.5.1 GASP Comparisons

The impact of advanced technologies on the airplanes which were analyzed proved quite dramatic. GASP provided some very interesting and useful information, especially since it accounts for synergistic effects.

5.5.1.1 <u>6-Passenger Designs</u>. Data for the current and advanced technology designs were presented in Tables 5.7 and 5.8 respectively. Table 5.11 summarizes the characteristics of the two airplanes, and emphasizes those quantities which illustrate the impact of advanced technologies.

	Current Technology	Advanced Technology
Wing Loading	$1053.6 \text{ N/m}^2 (22 \text{ 1b/ft}^2)$	2154.6 N/m <sup>2</sup> (45 1b/ft <sup>2</sup> )
Power Loading	57.56 N/kW (9.65 1b/hp)	59.65 N/kW (10,0 1b/hp)
Aspect Ratio	7.5	11
Wetted Area	67.71 m <sup>2</sup> (723.8 ft <sup>2</sup> )	40.37 m <sup>2</sup> (434.5 ft <sup>2</sup> )
Gross Weight	1859.8 kg (4100 1b)	1258.7 kg (2775 1b)
Empty Weight	1051.4 kg (2318 lb)	582 kg (1283 1b)
Empty/Gross	.565	.462
V 1 stall	108 km/hr (58.3 kt)	110.5 km/lir (64.5 kt)
Takeoff Dist. <sup>2</sup>	540 m (1771 ft)	609 m (1999 ft)
Landing Dist. <sup>1</sup>	351 m (1152 ft)	406 m (1332 ft)
Lift/Drag <sup>3</sup>	11.7	14.2
Fuel Required <sup>4</sup>	264 kg (582 lb)	132.5 kg (292 1b)
Avg. Cruise Fuel Eff.5	.84 km/kg (12.1 $\frac{nm}{gal}$ )	1.80 km/kg (25.9 nm gal)

Table 5.11. 6-Passenger Comparison by GASP

<sup>1</sup> Sea level, gross weight, landing flaps.

<sup>2</sup> Sea level, gross weight, takeoff flaps.

<sup>3</sup> Cruise altitude, 250 kt, gross weight.

<sup>4</sup> 900 nm + 45 min. reserves, 250 kt, 544.3 kg (1200 1b) payload

<sup>5</sup> Cruise altitude, 250 kt, mid-weight.

Refering to Table 5.11, the integration of advanced technologies (see Table 5.3) had the following results:

(1) Wetted area is reduced by 40%.

¢,

(2) Maximum gross weight is reduced by 327.

- (3) Empty to gross weight ratio is lowered by 18%.
- (4) Lift to drag ratio is increased by 21%.
- (5) Fuel required is reduced by 50%.
- (6) Average cruise fuel efficiency is increased by 114%.
- (7) Takeoff distance is increased by 132".
- (8) Landing distance is increased by 16%.
- (9) Stall speed is increased by 11%.

Requirements were still met or exceeded.

In the current analysis, it is not possible to single out exactly how much a given technology benefits the airplane; the effects are highly synergistic. For example increasing wing loading (with help of full span flaps) decreases wetted area and wing weight which allows a smaller engine which decreases weight and fuel consumption, which again allows a smaller wing, etc.

Ideally, GASP could be used to investigate the impacts of individual technologies and various combinations of technologies but this proved to be beyond the scope of the current study. The problem was overcome by using a parametric method of comparison as discussed in Section 5.5.2.

5.5.1.2 <u>Commuter Designs</u>. Table 5.12 summarizes the configuration characteristics of the two designs which were analyzed and displays those quantities most affected by advanced technology integration.

Design Gross Weight				
	Current Technology	Advanced Technology		
Wing Loading	2155 N/m <sup>2</sup> (45 1b/ft <sup>2</sup> )	3279 N/m <sup>2</sup> (68 1b/ft <sup>2</sup> )		
Power Loading	4.07 kg/kW (6.69 1b/hp)	6.63 kg/kW (10.9 1b/hp)		
Aspect Racio	7.71	12		
Lift/Drag	10.44	12.55		
Wetted Area	165.2 m <sup>2</sup> (1778.7 ft <sup>2</sup> )	136.5 m <sup>2</sup> (1469.2 ft <sup>2</sup> )		
Max. Gross Weight	5670 kg (12,500 1b)	5706 kg (12,580 1b)		
Empty Weight	3379 kg (7,450 1b)	3058 kg (6,742 1b)		
Empty/Gross	. 596	.536		

#### Table 5.12. Commuter Comparison by GASP

32 42 42 4 4 5 5 6 1 1/ · · ·

in the first of the first of the second state of the second state of the second state of the second state of the

all is a line of the second second

The results of technology integration into the commuter aircraft are:

- (1) Wetted area is reduced by 17%.
- (2) Empty to gross weight ratio lowered by 10%.
- (3) Lift to drag ratio increased 20%.

Performance comparisons of the aircraft were obtained from the GASP analysis of the multi-leg mission, where a 60% load factor was assumed for each aircraft and the remaining weight was allowed for fuel. Recent FAA regulations (Ref. 215) allow COMBL a zero-fuel weight of 5675 kg, or a gross weight of 6356 kg. A comparison of ADCOM with respect to COMBL (with additional fuel) allows a realistic evaluation of advanced technology incorporation

into a commuter operation. Table 5.13 summarizes the performance characteristics of the two aircraft.

	COMBL	ADCOM
Max. Gross Weight	6350 kg (14,000 lb)	5289 kg (11,660 1b)
Payload	1089 kg (2,400 lb)	1089 kg (2,400 1b)
Takeoff Distance	1113 m (3,650 ft)	1091 m (3,580 ft)
Landing Distance	967 m (3,173 ft)	579 m (1,900 ft)
Balanced Field Length	1345 m (4,412 ft)	1006 m (3,300 ft)
Single Engine Service Ceiling	3094 m (10,150 ft)	3094 m (10,150 ft)
Fuel Required 45 min. reserve	1311 kg (2,890 lb) 216 kg res (477 lb res)	782 kg (1,723 lb) 127 kg res (279 lb res)
Time Required	4.84 hrs	5.0 hrs

Table 5.13. Comparison of Baseline and Advanced Technology Commuters for the Multi Mission\*

Nine 87 nm legs, including taxi, takeoff, climb, and cruise.

The effect of technology integration on the performance characteristics of the commuter aircraft are:

- (1) Maximum gross weight is reduced by 17%.
- (2) Takeoff distance is virtually matched.
- (3) Landing distance is reduced 40%.

- (4) Balanced field length is reduced 25%.
- (5) Single-engine service ceiling is matched.
- (6) Fuel required is reduced 40%.
- (7) Mission time is increased 3%.

## 5.5.2 Parametric Comparisons

A measure of airplane cruise efficiency termed "Range Factor" (kF) was derived using the Breguet range equation with an expression for lift to drag ratio. RF is defined as pound (payload) miles per gallon of fuel and is a function of range (R), altitude (h), velocity (V), empty to gross weight ratio ( $W_{\rm g}/W_{\rm G}$ ), effective aspect ratio (Ae), zero lift drag coefficient ( $C_{\rm D}$ ), effective specific fuel consumption (sfc/n<sub>p</sub>), and wing loading (W/S).

Since the independent variables are functions of technology the RF will illustrate the impacts of advanced technologies. Derivation of RF may be found in Appendix C.

5.5.2.1 <u>6-Passenger</u> <u>Designs</u>. The RF comparisons were done for four different technology baselines defined as follows:

- (1) Current technology non-turbocharged baseline
  - non-turbocharged reciprocating engine.
  - conventional aluminum structure •
  - conventional controls and control surfaces.
- (2) Current technology turbocharged baseline
  - turbocharged reciprocating engine.
  - conventional aluminum structure.
  - conventional controls and control surfaces.
- (3) Low risk advanced technology baseline
  - Advanced turbocharged reciprocating engine; lower sfc and higher critical altitude.

- conventional aluminum structure.
- spoilers for roll control; full span single-slotted Fowler flaps.
- moderately increased wing loading and aspect ratio.
- (4) High risk advanced technology baseline
  - advanced, stratified charge, highly turbocharged rotary combustion engine.
  - composite structure.

ŝ

- spoilers for roll control; full span single-slotted Fowler flaps.
- high wing loading and aspect ratio.

The RF variables for baselines (1) and (2) are determined by averaging data for various existing 6-hassenger single engine airplanes. No designs were established for baseline (3) so the RF variables were estimated using available data. The RF variables for baseline (4) are obtained from the GASP designed advanced technology airplane (6PAXAD). Table 5.14 presents the data used in the RF comparisons.

Range factor is plotted as a function of speed for each of the technology baselines in Figure 5.19. Range has been normalized to 900 nm + 45 min. reserves to make the comparisons consistent.

Variables	Technology Baselines				
	(1)	(2)	(3)	(4)	
h (m)	2,133.6	6,096	7,620	9,144	
(ft)	7,000	20,000	25,000	30,000	
₩ <sub>E</sub> /₩ <sub>G</sub>	. 590	. 595	.565	. 462	
A	6.57	7.08	10	11	
e	.7	.7	.7	.7	
с <sub>р</sub> о	.0226	.0226	.0250	.0286	
SFC ( <u>kg</u> kW-hr)	.260	.271	.231	.231	
$(\frac{1b}{hp-hr})$	.428	.445	. 380	.380	
η P	.85	.85	.85	. 85	
W/S (N/m <sup>2</sup> )	969.56	1035.63	1635.63	2154.57	
(1b/ft <sup>2</sup> )	20.25	21.63	35	45	

## Table 5.14. Data Used for 6-Passenger Range Factor Comparisons

The technology impacts illustrated in Figure 5.19 are very interesting and are summarized as follows:

- Turbocharging allows baseline (2) airplanes to cruise at higher altitudes than baseline (1) airplanes and thus increases RF at typical cruise speeds. Maximum RF occurs at a higher speed but is slightly lower.
- (2) The low risk technologies, baseline (3), increase maximum RF very significantly and cause  $RF_{max}$  to occur at a higher speed.



Fig. 5.19. 6-Passenger Range Factor Comparisons

Land .

- (3) The high risk technologies, baseline (4), further increase  $RF_{max}$  and shift  $RF_{max}$  to a higher speed.
- (4) The GASP designed current technology baseline (6PAXBL) is much more efficient at 250 kt than baseline (2) due to lower  $C_{D_{e}}$ , lower  $W_{E}/W_{G}$ , and higher cruise altitude.
- (5) GASP comparisons between the advanced technology (6PAXAD) and current technology (6PAXBL) designs as discussed in Section 5.5.1.1 are validated.

5.5.2.2 <u>Commuter Designs</u>. RF comparisons were made for four technology baselines:

- (1) Current technology baseline
  - turboprop engines.
  - conventional aluminum structure.
  - conventional control surfaces.
- (2) Low risk advanced technology baseline
  - turboprop engines.
  - conventional aluminum structure.
  - spoilers for roll control, full span single-slotted Fowler flaps.
  - High wing loading and high aspect ratio.
- (3) Medium risk advanced technology baseline
  - turboprop engine.

- composite structure.
- spoilers for roll control; full span single-slotted Fowler flaps.
- high wing loading and high aspect ratio.

- (4) High risk advanced technology baseline
  - GATE technology engines.
  - composite structures.

h

 spoilers for roll control; full span single-slotted Fowler flaps.

• high wing loading and high aspect ratio.

The variables for the RF baseline, (1), were taken from COMBL. The high risk baseline, (4), was developed using the characteristics of ADCOM. The low risk technology baseline, (2), was developed using variables common to the other two baselines by integrating the drag characteristics and high wing loadings of ADCOM with the propulsion and structural characteristics of COMBL. The medium risk technology baseline, (3), was developed by integrating composite structures into baseline (2).

The variables used in the RF analyses follow in Table 5.15.

The results of the comparison appear in Figure 5.20, where range factor RF is shown as a function of speed for a single leg range of 800 nm.

The technology impacts are:

- The current technology baseline state-of-the-art is fairly high. Incorporation of low risk technologies will not substantially benefit the performance of the aircraft.
- (2) High risk technologies offer substantial performance improvements for this size of commuter aircraft. Fuel efficiency is more than doubled, while low risk technology offers only a 30%

improvement over the current state-of-the-art. The GATE engine provides a greater improvement in efficiency than the integration of composites.

.

S.

	Technology Baselines				
Variadies	(1)	(2)	(3)	(4)	
h (m)	3048	3048	3048	3048	
(ft)	10,000	10,000	10,000	10,000	
₩ <sub>E</sub> /₩ <sub>G</sub>	.596	. 596	.536	. 536	
A	7.71	12	12	12	
e	.81	.71	.71	.71	
с <sub>р</sub>	.0253	.0268	.0268	.0268	
sfc ( <u>kg</u> )	.335	.335	.335	.274	
$(\frac{1b}{hp/hr})$	.551	.551	.551	-450	
η p	.88	.88	.87	.87	
W/S (N/m <sup>2</sup> )	2154	3256	3256	3256	
(1b/ft <sup>2</sup> )	45	68	68	68	

Table 5.15. Data Used for Commuter Range Factor Comparisons



Fig. 5.20. Commuter Range Factor Comparisons

# CHAPTER 6

### GENERAL AVIATION SYNTHESIS PROGRAM

This program, developed by the NASA-Ames Research Center under the supervision of Thomas L. Galloway, has been discussed in the literature as early as 1973 (Ref. 64), been utilized in several vehicle trade studies of 1975 vintage (e.g. Ref. 80 and 153), and more recently been used by some of the manufacturers who participated in the General Aviation Turbine Engine (GATE) studies. As of this writing, it is still receiving significant attention, and ongoing improvements to the code appear to signal even greater versatility and applicability to the particular synthesis problems faced by general aviation.

# 6.1 PROGRAM FAMILIARITY

Gaining familiarity with the program's capabilities was given a high priority during the early phases of the research effort. Here, the continuous and extensive support rendered by Galloway proved invaluable. Through his assistance, extensive familiarity with the program logic was gained in a relatively short period of time.

The program is particularly versatile and the user is offered a wide range of input variables with which to define significant vehicle, propulsion system, and mission parameters. Vehicle sizing as well as propulsion system sizing are available

231

PRECEDING PAGE BLANK NOT FILMED

as standard options, and user-specified inputs allow for various capabilities of the program to be exercised. For example, during the actual employment of GASP, sizing the vehicle and propulsion system vs mission parameters without actually exercising the mission trajectory options proved to be particularly cost effective.

Three uniquely different types of program executions will be referred to throughout this chapter. Accordingly they will be defined here for clarification.

- Short Run execute only the vehicle weight, drag, and propulsion system sizing options without trajectory definition. This is a standard program option.
- (2) Standard Run proceed with the short run and continue with trajectory definition. This is also a standard program option.
- (3) Commuter Run execute successive legs where gross takeoff weight and fuel remaining at the beginning of each leg are derived from the preceeding leg. This is a non-standard option and represents a modification by the present research team as discussed in Section 6.2.3.

# 6.1.1 Canard Configurations

GASP does not possess the capability to handle canards. By this, it is meant that GASP will not allow for the investigation of possible reductions in induced drag due to the upload on the forward tail nor will it make any allowance for structural weights which may differ due to different span loadings and/or fuselage bonding moments. The underlying factor for this shortcoming is simply a lack of canard data which would allow for the development and incorporation of empirical weight and/or aerodynamic equations.

# 6.1.2 Propulsion System Sizing

GASP synthesizes vehicles with either a "rubber" engine or a fixed propulsion system based on user-specified inputs. The very attractive "rubber" engine feature of this program allowed for rotary combustion, reciprocating, and turbine powered configurations to be matched with a propeller and sized for different vehicles.

# 6.1.3 Drag and Weight Calculations

The program will compute both a drag polar and a component weight breakdown. It also provides the user with sufficient input capability to generate specific drag and weight characteristics. For the evaluation at hand, aerodynamic and weight characteristics were manually calculated and then compared against results generated by GASP. Although aerodynamic coefficients showed reasonably good agreement, structural component weights for the current technology 6-passenger airplane did not. Since consistent results were required for the technology evaluation comparisons, it was decided to incorporate manually calculated drag and weight characteristics for all computer runs. This required that several short runs be made to size the

233

CONTRACTOR OF A CONTRACTOR OF A

propulsion system as well as obtain desired weight and drag characteristics, where the latter characteristics were obtained through inputs already provided for in the program.

## 6.2 PROGRAM MODIFICATIONS

Three modifications were made to GASP in order to allow for its maximum utilization.

### 6.2.1 Twin Engine Centerline Thrust

The first involved the addition of an input variable which would allow for a twin engine vehicle to be configured with centerline thrust in a tractor-pusher arrangement. No aerodynamic changes were accounted for. However, this modification allowed for the retention of preliminary tail sizing and longitudinal balancing options which already exist in GASP. Structural weights and the drag of the wing and fuselage could be expected to change when engines are removed from the wing and added to the fuselage.

## 6.2.2 Brake Specific Fuel Consumption (BSFC)

GASP computes a scaled BSFC for a given propulsion system type and applies this figure to compute range and/or fuel used during various mission segments. Although one manufacturer provided a subroutine which reflected the SFC's of its GATE proposal, a decision was made to modify the program and allow a user-specified input to correct the computed BSFC to a desired level. Hence, after the propulsion system was sized over several short runs, this input was applied to adjust the BSFC to different levels based on technology forecasts for different engines.

#### 6.2.3 Commuter Mission Profile

The two most attractive features of GASP lie in its propulsion system sizing algorithm and its fairly sophisticated trajuctory analysis for takeoff, climb, and acceleration. This latter feature is lost, however, when a commuter is analyzed because only one leg can be flown at a time. Although the program does possess the versatility to examine missions at other than takeoff gross weights with full fuel, subsequent commuter legs must be individually loaded and executed after previous leg weights have been determined.

To overcome this difficulty and still utilize the full potential of GASP, an extensive modification was made to the program. Three major goals of this modification were to (1) insure that every GASP computational methodology be retained, (2) insure that all GASP capabilities be retained, and (3) keep the modification totally transparent to the user.

At present, all three goals appear to have been achieved. One variable are been added to the original namelist called \$INGASP and acts as a switch. Test runs of the program with this switch on or off allows all of those program capabilities actually used in the present research to be accessed and utilized. With the switch on, an additional namelist containing only seven variables (three are 10-element arrays) is read. Here, up to ten legs at different altitudes, Mach numbers, and ranges may be specified together with any range-correcting options. Once GASP enters the mission profile analysis phase, each specified

leg is analyzed utilizing only GASP methods. Three fuel weights for each leg are computed as in the original program, and the remaining design fuel quantity together with the landing weight are transferred to the next leg as initial values. Failure to meet all specified legs results in an error message identifying the accumulated successful leg completions and the point of fuel exhaustion or, if elected, vehicle resizing together with a further analysis of the mission trajectory. Although simple in concept, the reader who is familiar with the lateral and highly interrelated structure of GASP will recognize that this modification ultimately controls the majority of the 72 subroutines in the program.

Numerous commuter runs were made to test the validity of this modification. The most successful of these involved a nineleg specification of 161 km (87 nm) each for the Swearingen Metro. Each subsequent leg reflected anticipated reductions in time-to-climb and range-in climb, and the final leg resulted in a 5.6 km (3 nm) difference from an expected range which had been predicted in a scenario generated for a commuter study performed in Reference 215.

#### 6.3 PROGRAM BENCHMARKS

In order to validate the use of GASP, the program was benchmarked against two vehicles whose characteristics were manually calculated.

In the case of the 6-passenger airplane, a new vehicle representing current reciprocating engine technology with a conventional aluminum structure was designed and analyzed. Several short runs were made to verify weight, drag, fuel capacity, and propulsion characteristics. The final standard run over a desired 1667 km (900 nm) range resulted in a 9 km (5 nm) deficiency.

When a similar analysis was performed for the commuter, the resulting vehicle had almost identical characteristics with those shown for Example A in Volume I of the published GASP documentation. This latter airplane had a gross weight of 5675 kg (12,500 lb) and a range of 1117 km (603 nm) at 3048 m (10,000 ft) with a 45 minute reserve. The example data were subsequently modified by calculating available fuel volume and then limiting gross weight to approximately 6356 kg (14,000 lb) based on compromises between available fuel volume and a zero fuel weight of 5675 kg (12,500 lb) with a payload of 19 passengers. Commuter runs were then made with excellent agreement between results and manual calculations.

In all cases, significant differences between expected and GASP-generated structural weights were noted. Some concern was expressed by the present research team over the unusually high aspect ratios (on the order of 12) which were being examined. Manual computations, however, reflected good agreement between the methods of Nicolai (Ref. 154), Torenbeek (Ref. 214), and those of one airframe manufacturer. Consequently, inputs to GASP were adjusted over several short runs until desired component weights resulted. These correction factors were then considered constant for subsequent investigations of advanced technology

vehicles. It must be noted that this decision to use manually generated data does not reflect an opinion that the GASP methodology is incorrect. Rather, since comparisons between current and advanced technologies were being conducted, it was felt that the al. procedure would yield more consistent results.

Differences in drag characteristics between expected and GASP-generated values were small. However, drag computations were also adjusted for the baselin: vehicles over several short runs until desired values were obtained. These correction factors were then also considered constant for the investigation of advanced technology vehicles.

It is significant that these correction factors are already provided for in the GASP input list. Hence, the adaptability of GASP is not degraded due to input limitations.

#### 6.4 USE OF GASP

The results obtained from the use of GASP are reflected in the performance and trade studies discussed previously in Chapter 5. Additional data are presented in Appendix C.

#### 6.5 RESULTS OF GASP IMPLEMENTATION

The use of GASP contributed markedly to the productivity of this research effort and allowed for the analysis of several airplane configurations in a rapid and orderly manner. The wide range of inputs already provided in the basic program presents the user with attractive options for vehicle definition, and the

propulsion sizing and trajectory definition capabilities offer significant contributions to General Aviation aircraft synthesis capabilities.

----

1 8

٩.

### CHAPTER 7

#### RESULTS

# 7.1 OVERVIEW OF TASKS PERFORMED

This research, which spanned 12 months, was pursued in an attempt to identify those advanced technologies which, when integrated into general aviation airplanes, would offer significant improvements in safety, performance, efficiency, and utility. In order to achieve this primary goal, several diverse and otherwise independent tasks were performed. These may be broadly grouped as follows:

- (1) A thorough and in-depth data base was established.
  - (a) 31 manufacturers were visited.

£

- (b) 3 NASA research centers were visited.
- (c) 6 professional meetings were attended.
- (d) A detailed literature search was performed.
- (2) A technology evaluation technique was developed, tested, and evaluated. This technique needed to reflect the consensus of the general aviation community, be broad enough to consider all aspects of a technology's impact, be relatively sensitive to differences in a technology's potential benefit, and display a degree of stability.
  - (a) Survey 1 was a Delphi survey used to identify the measurements of a technology's impact (category identification). 17 categories were identified after four rounds

241

PRECEDING PAGE BLANK NOT FILMED

of the survey were completed.

- (b) Survey 2 was a Delphi survey used to quantify the measurements (category rating). Three rounds were used to establish the ratings (category weights).
- (c) Pessimistic, Likely, Optimistic, and Expected relative benefits were determined and examined to establish the stability of the evaluation technique (PLOE studies).
- (d) A factor analysis was performed for the relative benefit matrix to determine the suitability of the selected categories.
- (e) The effect of different raters was examined.
- (f) An analysis of variations in the Empty Weight category rating was performed because it appeared that the definition of this category and/or its impact on payload was not adequately defined to Survey 2 participants.
- (3) The effects of integrated technologies on two classes of airplanes were examined. Both airplanes were specified to cruise at 250 knots. Airplane A was synthesized as a 6passenger, high performance, personal/business airplane. Airplane B was synthesized as a 19-passenger commuter.
- (4) The General Aviation Synthesis Program (GASP) was evaluated, modified, and used to investigate the effects of technology integration on Airplanes A and B. It is emphasized here that the modifications were made only to adapt the program to the study at hand and do not reflect deficiencies in #

the program.

(5) A modified version of the Quasi Vortex Lattice Method (QVLM) was used to investigate canard configurations. Here, the modification allowed the investigation of the effects of a forward tail on an aft main wing. Fairly good agreement was noted for a short-coupled canard, but non-short-coupled configurations were not evaluated due to an absence of wind tunnel data.

#### 7.2 RESULTS OF TASKS PERFORMED

The results of the separate tasks identified in Section 7.1 are summarized below according to major areas of emphasis.

## 7.2.1 Manufacturer and Research Center Visits

All visits contributed immeasureably to the data base required for the technology evaluation. Manufacturers contributed significantly by sharing their views of both advanced technology development as well as of NASA research in general. Comments regarding technologies requiring further study and recommendations for further NASA research may be found in Appendix A. However, the following opinions and perceptions of the manufacturers are emphasized here.

- NASA research should be confined to basic research, whereas product development should be left to manufacturers.
- (2) High risk technologies such as those encountered in propulsion require NASA funding if they are to materialize as hardware.

(3) Certification and product liability costs are often far greater than those costs incurred in the actual development of a technology.

#### 7.2.2 Technology Evaluation

An evaluation technique based on a linear compensatory model was successfully developed, tested, and implemented. This model reflects the opinions of the general aviation community both in the identification of categories to be used as measurement devices as well as in the determination of the relative importance of these categories.

Two sets of category weights were developed as part of this research. The first was for a 6-passenger, high performance, personal/business airplane, and the second was for a 19-passenger commuter.

7.2.2.1 <u>Attractive Technologies</u>. The following technologies appear to be attractive to general aviation.

- All propulsion technologies. The GATE turboprop and Stratified Charge Rotary Combustion Engine were distinctly ahead of all other technologies considered for Airplane A.
- (2) Fiberglass, Kevlar, and graphite composite materials. Certification procedures must reflect the latest composite test data, however, to preclude overly restrictive requirements from nullifying otherwise significant weight reductions. Designs will also need to incorporate proven methods for lightning protection. Only areas most vulnerable to lightning

strikes need protection, such as tips of wings and tail surfaces or any relatively sharp pointed projection. Promising methods for protection include flame sprayed aluminum coating, external conductors or diverters, and embedded aluminum or copper wire mesh.

- (3) Natural laminar flow airfoils, spoilers with full span Fowler flaps, and low/medium speed airfoils. Development of these technologies requires improvements in computational aerodynamics.
- (4) Advanced integrated avionics systems which reduce pilot workload.

7.2.2.2 <u>Unattractive Technologies</u>. The following technologies appear unattractive.

(1) Area navigation concepts such as Loran C and Omega.

(2) Active controls, fly-by-wire, and fly-by-light.

(3) HUD's, Doppler navigation, and inertial navigation.

7.2.2.3 <u>Technologies Not Examined</u>. Technologies with significant potential but not evaluated due to their inapplicability to Airplane A and/or B include the following:

(1) Propfan

(2) Quiet, Clean General Aviation Turbine engine (QCGAT).

#### 7.2.3 Design Studies

Design trade studies for both airplanes reflect significant performance gains resulting from the integration of new technologies. Here, the synergistic effects of composites, aerodynamics, and propulsion technologies were examined. The resulting airplanes displayed warked improvements in performance, operating cost, and empty weight and served to verify the predictions evolving from the Evaluation Technique. Lighter aircraft with higher wing loadings and improved propulsion systems showed 40% to 50% reductions in fuel used.

# 7.2.4 GASP

The use of GASP aided immensely in the evaluation of the impact of advanced technologies. Here, manual calculations for weight and drag were input, and the program was used to size the vehicle and propulsion system, and to evaluate mission performance. Its versatility promises significant benefits for general aviation synthesis.

#### 7.2.5 QVLM

Although this analysis tool reflected fairly good agreement for the one configuration evaluated, the inability of this method as well as conventional VLM methods to model a deformed wake was perceived to be a shortcoming to the investigation of non-shortcoupled canard configurations.

## CHAPTER 8

## CONCLUSIONS AND RECOMMENDATIONS

#### 8.1 CONCLUSIONS

The identification and integration of certain advanced technologies can have significant and profound impact on future general aviation ai planes. The results of the present study suggest that it may be possible to integrate certain high-risk technologies into two low-speed general aviation airplanes and realize substantial fuel savings while improving safety, comfort, and performance. These fuel savings alone represent significant reductions in the cost of owning and operating an airplane.

The initial cost of the airplanes, however, is expected to be high. Developments which would substantially lower purchase price include:

- A NASA commitment to develop advanced propulsion system technologies specifically for general aviation.
- (2) A requirement by the automotive and trucking industry for more sophisticated and efficient turbochargers.
- (3) A similar requirement by the automotive industry for large supplies of Kevlar and graphite composites.
- (4) The development of a substantial data base for composites which would lead to better-defined certification requirements.

The key to the improved performance and efficiency of the 6-passenger light single-engine airplane investigated lies in increased wing loading. Higher aspect ratios, a NASA natural laminar flow airfoil, lighter weight through the use of composites, and improved propulsion systems also contributed significantly. Since wing loading is constrained by stall speed requirements of 61 kt for this class of airplanes, a re-examination of this requirement appears warranted.

Improvements in commuter efficiencies, while not as great as experienced by the 6-passenger airplane, were still dramatic. Improved propulsion systems, higher aspect ratios and wing loadings, lighter structural weights through the use of composites, and the incorporation of a NASA low-speed airfoil, all contributed significantly to the improvements enjoyed by this vehicle.

#### 8.2 RECOMMENDATIONS

Those technologies identified in Chapter 4 and repeated in Section 7.2.2.1 should be pursued. Additionally, those technologies which improve the crashworthiness of an airplane (scats, restraints, floor, fuel tanks) should receive continued attention by both NASA and the FAA.

Computational methods which allow for the more accurate aerodynamic analysis of wing-body-tail configurations (both fore and aft tail) should be developed. Finally, verified prediction techniques for wing-spoiler-flap configurations should also be developed.

Integrated avionics systems and related technologies designed to reduce pilot workload must be pursued. This implies that a vigorous human factors research program directed towards an analysis of the pilot-airplane-flight environment interface should also be conducted.

While the Ames DAAS concept and the Princeton Advanced Research Airplane appear well suited to investigate the effects of advanced avionics, advanced aerodynamic technologies must be actively and carefully investigated and flight tested in order to establish a verified data base for the integration of these technologies. Furthermore, although advanced propulsion systems play a key role in the present research, significant benefits to general aviation appear possible through the incorporation of only aerodynamic technologies.

The research of advanced technology general aviation airplanes should be continued. While the separate technologies all require development, an integrated test vehicle should be pursued in order to realistically investigate the actual benefits realizeable through the synergistic effects of the different technologies. In view of the fairly long lead times which are characteristic of the aviation industry, research should be directed as soon as possible to a high-aerodynamic-technology vehicle. Such a vehicle would be characterized by composite structures (for both weight and surface finish), advanced airfoils, higher wing loadings, and higher aspect ratios.

#### BIBLIOGRAPHY AND REFERENCE LIST

- Alberts, R.D. and W.H. Ruedger. <u>Continued Study of NAVSTAR/</u> <u>GPS for General Aviation</u>. (Final Report). NAS CR-159145. Prepared by Research Triangle Institute, Research Triangle Park, North Carolina, for National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. December 1979.
- Alberts, R.D. and W.H. Ruedger. <u>Preliminary Study of</u> <u>NAVSTAR/GPS</u> for <u>General</u> <u>Aviation</u>. (Final Report). NASA CR-145050. Prepared by Research Triangle Institute, Research Triangle Park, North Carolina, for National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. November 1976.
- 3. Alfaro-Bou, Emilio and Victor L. Vaughan, Jr. Light Airplane <u>Crash Tests at Impact Velocities of 13 and 27 m/sec</u>. NASA Technical Paper 1042. National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. November 1976.
- 4. Anderson, S.B. <u>An Historical Overview of Stall/Spin Character-istics of General Aviation Aircraft</u>. Paper no. 78-1551. Presented at the AIAA Conference on Air Transportation: Technical Perspectives and Forecasts, Los Angeles, California. August 21-24, 1978. American Institute of Aeronautics and Astronautics, New York, N.Y.
- 5. Anon. "Business, Utility Aircraft Shipments, December, 1979." Aviation Week and Space Technology, 112:103 (February 11, 1980).
- 6. Anon. Jane's All the World's Aircraft, 1978-1979. Edited by John W.R. Taylor. New York: Franklin Watts, Inc., October 1978.
- 7. Anon. "NASA Tests Crash Effects on Airframes." Aviation Week and Space Technology, 111:88-89 (October 8, 1979).
- 8. Anon. "NAVSTAR delay, Satellite Cut Planned." Aviation Week and Space Technology, 112:16 (March 10, 1980).
- 9. Anon. "The Sperry SPZ-600 Autopilot." Business and Cormercial Aviation, pgs 68-72 (December 1977).
- 10. Anon. "A Worldwide Rotary Update." <u>Automotive Engineering</u>, 86:31-42 (February 1978).

251

PRECEDING PAGE BLANK NOT FILMED

- 11. Baerst, C. F. and D. G. Furst. <u>General Aviation Propulsion</u> for the <u>Ninetics</u>. Paper no. 79-1158. Paper presented at the AIAA/SAE/ASME 15th Joint Propulsion Conference, Las Vegas, Nevada, June 18-20, 1979. American Institute of Aeronautics and Astronautics, New York, N. Y.
- 12. Baerst, C. F. and D. G. Furst. <u>General Aviation Turbine Engine</u> (GATE) <u>Study Final Report</u>. NASA CR-159482. Prepared by AiResearch Manufacturing Company of Arizona, A Division of the Garrett Corporation, Phoenix, Arizona for National Aeronautics and Space Administration, Lewis Research Center, Cleveland, Ohio. Febraury 5, 1979.
- Barber, M. R. and Jack Fischel. "General Aviation -- The Seventies and Beyond." <u>Vehicle Technology for Civil Avia-</u> tion -- The Seventies and Beyond. NASA SP-292. Proceedings of a conference held at Langley Research Center, Hampton, Virginia. November 2-4, 1971.
- Barber, Marvin R., et al. An Evaluation of the Handling Qualities of Seven General-Aviation Aircraft. NASA TN D-3726. National Aeronautics and Space Administration, Flight Research Center, Edwards, California. November 1966.
- Barnette, James F. <u>Role of Head-Up Display in Instrument</u> <u>Flight</u>. Report No. IFC-LR-76-2. USAF Instrument Flight Center (USAFIFC), Randolph AFB, Texas. August 18, 1976.
- Bateman, L. F. "An Evolutionary Approach to the Design of Flight Decks for Future Civil Transport Aircraft." <u>Aircraft</u> Engineering, 50:4-10 (July 1978).
- Baum, J. A., et al. <u>Prop-Fan Data Support Study, Technical</u> <u>Report.</u> NASA CR 152141. Prepared by Hamilton Standard, Division of United Technologies Corporation, Windsor Locks, Connecticut, for National Aeronautics and Space Administration, Ames Research Center, Moffett Field, California. February 28, 1978.
- Bergey, Karl H. Assessment of New Technologies for General Aviation Aircraft. Report No. FAA-RD-78-132 (Final Report). Prepared for U. S. Department of Transportation, Federal Aviation Administration, Systems Research and bevelopment Service, Washington, D. C. September 1978.
- Berkowitz, M., et al. <u>Performance, Emissions, and Physical</u> <u>Charcteristics of a Rotating Combustion Aircraft Engine.</u> NASA CR-135119. Prepared by Curtiss-Wright Corporation, Woodridge, New Jersey for National Aeronautics and Space Administration, Lewis Research Center, Cleveland, Obio. December 1976.

- Bird, Daniel K. <u>Electromechanical Actuation for Business</u> <u>Aircraft.</u> Paper No. 790622. Presented at the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas. April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 21. Black, D. M., R. W. Menthe, and H. S. Wainauski. <u>Aerodynamic Design and Performance Testing of an Advanced 30° Swept Eight Bladed Propeller at Mach Numbers from 0.2 to 0.85</u>. NASA CR 3047. Prepared by United Technologies Corporation, Windsor Locks, Connecticut, for National Aeronautics and Space Administration, Lewis Research Center, Cleveland, Ohio. 1978.
- 22. Boeing Commercial Airplane Company. <u>Aircraft Surface Coatings</u> <u>Study</u>. NASA Contractor Report 158954 (released for early domestic dissemination). Energy Efficient Transport Program. Contract NAS1-14742, Task 4.1.3. National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. January 1979.
- Bolton, W. R. <u>Status Report</u>: <u>Separate Surface Stability</u> <u>Augmentation Simulation and Flight Test</u>. Flight Research Laboratory, CRINC, University of Kansas, Lawrence, Kansas. November 1973.
- 24. Borst, Henry V. <u>The Design and Selection of Optimum Propellers</u> for <u>General Aviation Aircraft</u>. Paper no. 790575, Paper preented at the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas. April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 25. Broihanne, G., R. Deque, and M. Bossard. "Design of an Entirely Electrical Flying Control System." <u>AGARD Conference Proceedings No. 157 on Impact of Active Control Technology on Airplane</u> <u>Design.</u> Paper presented at a Joint Symposium of the Flight Mechanics Panel and Cuidance and Control Panel of the Advisory Group for Aerospace Research and Development (AGARD) held in Paris, France. October 14-17, 1974.
- 26. Castle, Claude B. and Emilio Alfaro-Bou. Light Airplane Crash Tests at Three Flight-Path Angles. NASA Technical Paper 1210. National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. June 1978.
- 27. Cetron, Marvin J. and Thomas I. Monahan. "An Evaluation and Appraisal of Various Approaches to Technological Forecasting" in <u>Technological Forecasting for Industry and Government --</u> <u>Methods and Applications</u>, edited by James R. Bright. Englewood Cliffs, New Jersey: Prentice-Hall, Inc., 1968.

- 28. Chambers, Joseph R. "NASA's Stall/Spin Program for Light General Aviation Aircraft." <u>Proceedings of the First FAA General</u> <u>Aviation Research and Development Conference</u>. Paper presented at a conference conducted by the Department of Transportation, Federal Aviation Administration, National Aviation Facilities Experimental Center (NAFEC), Atlantic City, New Jersey. August 17-18, 1977.
- Chevalier, H. L. Some Theoretical Considerations of a Stall <u>Proof Airplane</u>. Paper no. 790604. Presented at the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas. April 3-6, '979. Society of Automotive Engineers, Inc., Warrendale, nnsylvania.
- 30. Church, George W. <u>Advances in Avionics</u>. Paper no. 79-0562. Presented at the AIAA 15th Annual Meeting and Technical Display, Washington, D. C., February 6-8, 1979. American Institute of Aeronautics and Astronautics, New York, N. Y.
- 31. Cohn, David M., et al. <u>Executive Summary Analysis of Technology</u> <u>Requirements and Potential Demand for General Aviation Avionics</u> <u>Systems for Operation in the 1980's</u>. Prepared by Decision Sciences Corporation, Jenkintown, Pennsylvania, for National Aeronautics and Space Administration, Ames Research Center, Moffett Field, California. June 1974.
- Collins, D. J. <u>Status Report: Separate Surface Stability</u> <u>Augmentation Design and Development</u>. Flight Research Laboratory, CRINC, University of Kansas, Lawrence, Kansas. November 1973.
- Comptroller General of the United States. <u>Navigation Plan-</u> ning -- <u>Need for a New Direction</u>. Report to the Congress of the United States. LCD-77-109. PB 278 627. March 21, 1978.
- 34. Conner, D. W. and G. C. Thompson. "Potential Benefits to Short-Haul Transports Through Use of Active Controls." <u>AGARD</u> <u>Conference Proceedings No. 157 on Impact of Active Control</u> <u>Technology on Airplane Design</u>. Paper presented at a Joint Symposium of the Flight Mechanics Panel and Guidance and Control Panel of the Advisory Group for Aerospace Research and Development (AGARD) held in Paris, France. October 14-17, 1974.
- 35. Cooley, William W. and Faul R. Lohnes. <u>Multivariate Data</u> <u>Analysis</u>. New York: John Wiley and Sons, Inc., 1971.
- 36. Corsiglia, V. C., J. Katz, and R. A. Kroeger. <u>Full-Scale Wind</u> <u>Tunne! Study of Nacelle Shape on Cooling Drag</u>. Paper no. 79-18:0. Paper presented at the AIAA Aircraft Systews and Technology Meeting, New York, N. Y. August 20-22, 1979. American Institute of Aeronautics and Astronautics, New York, N. Y.

- Covault, Craig. "NASA Defining Fuel-Efficient Program." <u>Aviation Week and Space Technology</u>, 111:74-75 (October 8, 1979).
- 38. Davis, D. G. M. <u>The Impact of Noise Regulations on Propeller</u> <u>Design</u>. Paper no. 790593. Paper presented at the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas. April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 39. Denery, D. G., et al. <u>A Demonstration Advanced Avionics</u> <u>System for General Aviation</u>. Paper no. 790569. Paper presented at the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas. April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 40. Denery, D. G., et al. <u>Integrated Avionics for Future General</u> <u>Aviation Aircraft</u>. Paper no. 78-1482. Paper presented at the AIAA Aircraft Systems and Technology Conference, Los Angeles, California. August 21-23, 1978. American Institute of Aeronautics and Astronautics, New York, N. Y.
- 41. de Silva, B. M. E. and R. T. Medan. "Camard-Wing Vortex Effects in Subsonic Flow." Journal of Aircraft, 17:5-6 (January 1980).
- 42. de Sury, G. <u>Head-Up Piloting Aids</u>. NASA Technical Translation. NASA TT F 17,389. National Aeronautics and Space Administration, Washington, D. C. January 1977.
- Dodge, Steven Malcolm. <u>A Comparative Analysis of trea Naviga-</u> tion System in General Aviation. NASA CR 132504. Prepared by Massachusetts Institute of Technology, Flight Transportation Laboratory, Cambridge, Massachusetts. M.S. Thesis, June 1973.
- Bouglas Aircraft Company. <u>Analysis of Operational Requirements</u> for <u>Medium Density Air Transportation</u>. <u>Summary, Volume I</u>. NASA CR-137603. Prepared for National Aeronautics and Space Administration, Ames Research Center, Moffett Field, California. March 1975.
- 45. Downing, D. R., W. H. Brvant, and R. F. Stengel. <u>NASA/Prince-ton Digital Avionics Flight Test Facility</u>. Paper presented at the 3rd Digital Avionics Systems Conference, Fort Worth, Texas. November 6-8, 1979.
- 46. Dugan, James F. Jr., Bernard S. Gatzen, and William M. Adamson. <u>Prop-Fan Propulsion -- Its Status and Potential</u>. Paper no. 780995. Presented at the SAE Aerospace Meeting, San Diego, California. November 27-30, 1978. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.

- 47. Econ, Inc. An Assessment of the Benefits of the Use of NASA <u>Developed Fuel Conservative Technology in the U. S. Commercial</u> <u>Aircraft Fleet</u>. NASA CR 148148. Prepared by Econ, Inc., Princeton, New Jersey.
- 48. Edwards, W. Thomas and William M. Perella, Jr. <u>Crash-Resistant Fuel Systems for General Aviation Aircraft</u>. Paper no. 790592. Paper presented at the SAE Business Aircraft Maeting and Exposition, Wichita, Kansas. April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- Ellis, David R. <u>A Study of Lightplane Stall Avoidance and</u> <u>Suppression</u>. Report no. FAA-RD-77-25. Prepared by Flight Research Laboratory, Aerospace and Mechanical Sciences Department, Princeton University, for U. S. Department of Transportation, Federal Aviation Administration, Systems Research and Development Service. February 1977.
- 50. Elson, Benjamin M. "Inexpensive Advanced Avionics Concepts Being Sought." <u>Aviation Week and Space Technology</u>. (Reprint from the August 1, 1977 issue).
- Embraer. <u>Bandeirante Commuter Airliners</u>. Sales brochure published by Embraer - International Sales Division - P. O. Box 343, Sao Jose dos Campos, CEP 12200, SP Brazil. February 1979.
- 52. Erdos, Paul L. <u>Professional Mail Surveys</u>. New York: McGraw-Hill Book Co., 1970.
- 53. Fasanella, Edwin L. and Emilio Alfaro-Bou. NASA General Aviation Crashworthiness Seat Development. Paper no. 790591. Paper presented at the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas. April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 54. Fehrle, A. C. and R. L. McDougal. <u>Bonding and Durability</u>. Paper no. 709561. Paper presented at the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas. April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 55. Feistel, T. W., S. B. Anderson, and R. A. Kroeger. <u>A Method</u> for Localizing <u>Wing Flow Separation at Stall to Alleviate Spin</u> <u>Entry Tendencies</u>. Paper no. 78-1476. Paper presented at the AIAA Aircraft Systems and Technology Conference, Los Angeles, California, August 21-23, 1978. American Institute of Aeronautics and Astronautics, New York, N. Y.
- Fink, Martin R. and Robert H. Schlinker. "Airframe Noise Component Interaction Studies." Journal of Aircraft, 17:99-105 (February 1980).

- 57. Flannigan, J. and J. Emfinger. "Production Design Requirements for Fly-By-Wire Systems." <u>AGARD Conference Proceedings No. 157</u> <u>on Impact of Active Control Technology on Airplane Design.</u> Paper presented at a Joint Symposium of the Flight Mechanics Panel and Guidance and Control Panel of the Advisory Group for Aerospace Research and Development (AGARD) held in Paris, France. October 14-17, 1974.
- 58. Forsyth, Donna L. and John D. Shaughnessy. <u>Single Pilot IFR</u> <u>Operating Problems Determined from Accident Data Analysis</u>. NASA TM 78773. National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. September 1978.
- 59. Foster, Charles R. "Emission Standards and General Aviation." <u>Proceedings of the First FAA General Aviation Research and</u> <u>Development Conference</u>. Paper presented at a conference conducted by the Department of Transportation, Federal Aviation Administration, National Aviation Facilities Experimental Center (NAFEC), Atlantic City, New Jersey. August 17-18, 1977.
- Callington, Roger W. <u>Pneumatic Angle of Attack Sensor</u>. Paper presented at the 1978 Air Data Systems Conference, USAF Academy, Colorado. May 2-5, 1978. Department of Aeronautics, USAF Academy, Colorado.
- 61. Gallington, Roger W. and James W. Christian. <u>Pressure Driven</u> <u>Angle of Attack Indicating System</u>. (Final Report: March 1972-March 1975). AFATL-TR-76-10. Prepared by Department of Aeronautics, USAF Academy, Colorado for Air Force Armament Laboratory, Armament Development and Test Center, Elgin Air Force Base, Florida. January 1976.
- 62. Galloway, Thomas L. "Small Transport Aircraft Technology." Astronautics and Aeronautics, 18:26-35 (February 1980).
- Galloway, Thomas L. A Study of Commuter Aircraft Design. Paper no. 77-CT-36. Presented at the Gas Turbine Conference and Product Show, Philadelphia, Pennsylvania. March 27-31, 1977. Society of Mechanical Engineers, New York, N. Y.
- 64. Galloway, Thomas L. and Mark H. Waters. <u>Computer Aided</u> <u>Parametric Analysis for General Aviation Aircraft</u>. Paper no. 730332 presented at the SAE Business Aircraft Meeting in Wichita, Kansas, April 3-6, 1973. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 65. Gill, J. C., et al. <u>Study of an Advanced General Aviation</u> <u>Turbine Engine (GATE)</u>. NASA CR-159558. Prepared by Detriot Diesel Allison, Division of General Motors Corporation, Indianapolis, Indiana, for National Aeronautics and Space Administration, Lewis Research Center, Cleveland, Ohio, April 10, 1979.

- 66. Gilson, Richard D. and Robert W. Fenton. <u>Development of Stall</u> <u>Deterrent Device for Small Airplanes</u>. FAA-RD-75-53. Prepared by the Ohio State Research Foundation, Columbus, Ohio, for the Department of Transportation, Federal Aviation Administration, Washington, D.C. June 1975.
- 67. Giulianetti, Demo J. and Louis J. Williams. "Toward New Small Transports for Commuter Airlines." <u>Astronautics and</u> <u>Aeronautics</u>, 18:16-25 (February 1980).

ł

- 68. Gloss, Blair B. Effect of Wing Planform and Canard Location and Geometry on the Longitudinal Aerodynamic Characteristics of a Close-Coupled Canard Wing Model at Subsonic Speeds. NASA TN D-7910. National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. June 1975
- 69. Goodman, James R. <u>Optical Control Technology</u>. Paper presented at the Helicopter Flight Controls Specialist Meeting of the American Helicopter Society. Bell Helicopter Textron, Fort Worth, Texas. October 1978.
- 70. Graves, G. Barry, Jr. "Advanced Avionic Systems." <u>Vehicle</u> <u>Technology for Civil Aviation -- The Seventies and Beyond.</u> NASA SP-292. Proceedings of a conference held at Langley Research Center, Hampton, Virginia. November 2-4, 1971.
- Green, Paul E. and Donald S. Tull. <u>Research for Marketing</u> <u>Decisions.</u> (3rd Edition). Englewood Cliffs: Prentice-Hall, Inc., 1975.
- 72. Greene, R.A. The Effects of Low-Level Wind Shear on the Approach and Go-Around Performance of a Landing Jet Aircraft. Paper no. 790568. Paper presented at the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas. April 2-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 73. Grosveld, F.M.W.A. The Application of the Prop-Fan Concept in Preliminary Design of a Very Advanced Technology Light Twin (VATLIT '85). Paper no 79-1343. Paper presented at the AIAA/SAE/ASME 15th Joint Propulsion Conference, Las Vegas, Nevada. June 18-20, 1979. American Institute of Aeronautics and Astronautics, New York, N.Y.
- 74. Hammond, Alexander D. "High-Lift Aerodynamics." <u>Vehicle</u> <u>Technology for Civil Aviation -- The Seventies and Beyond.</u> NASA SP-292. Proceedings of a conference held at Langley Research Center, Hampton, Virginia. November 2-4, 1971.

- 75. Haney, H. P., R. R. Johnson, and R. M. Hicks. <u>Computational</u> <u>Optimization and Wind Tunnel Test of Transonic Wing Designs</u>. Paper no. 79-0080. Presented at the AIAA 17th Aerospace Sciences Meeting, New Orleans, Louisiana. January 15-17, 1979. American Institute of Aeronautics and Astronautics, New York, N. Y.
- 76. Harder, R. D., et al. <u>A-7 Airborne Light Optical Fiber</u> <u>Technology (A'OFT) Demonstration Project. NELC-TR-2024.</u> Performed by Naval Electronics Laboratory Center, San Diego, California, for Naval Air Systems Command, Washington, D. C.
- 77. Hare, E. W. "COMED -- The Cockpit Display of the Future." Aircraft Engineering, 50:17-19 (July 1978).
- 78. Harned, Malcolm S. "General Aviation Aircraft A Forecast of 1990 Designs and Capabilities." <u>ICAO Bulletin</u>, pgs. 14-20 (August 1979).
- 79. Hayduk, R. J., et al. <u>Nonlinear Structural Crash Dynamics</u> <u>Analyses</u>. Paper no. 790588. Paper presented at the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas. April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 80. Heldenbrand, R. W., G. L. Merrill, and G. A. Burnett. <u>Study</u> of <u>Small Civil Turbofan Engines Applicable to Military Trainer</u> <u>Airplanes -- Final Report</u>. NASA CR 137575. Prepared by AiResearch Manufacturing Company of Arizona, Phoenix, Arizona, for National Aeronautics and Space Administration, Ames Research Center, Moffett Field, California. April 1975.
- 81. Henderson, William P. The Effect of Canard and Vertical Tails on the Aerodynamic Characteristics of a Model with a 59° Sweptback Wing at a Mach Number of .30. NASA TMX-3088. National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. September 1974.
- Hicks, Raymond M., et al. <u>An Assessment of Airfoil Design</u> by <u>Numerical Optimization</u>. NASA TMX-3092, National Aeronautics and Space Administration, Ames Research Center, Moffett Field, California, 1974.
- Hicks, Raymond M. and Preston A. Henne. "Wing Design by Numerical Optimization." Journal of Aircraft, 15:407-412 (July 1978).
- 84. Hicks, Raymond M. and Edward T. Schairer. <u>Effects of Upper Sur-face Modification on the Aerodynamic Characteristis of the NACA 632-215 Airfoil Section</u>. NASA TM-78503, National Aeronautics and Space Administration, Scientific and Technical Information Office.

- 85. Hicks, Raymond M. and Garrett N. Vanderplaats. <u>Application of Numerical Optimization to the Design of Supercritical Airfoils Without "Drag-Creep</u>". Paper no. 77044. Presented at an SAF meeting, Feb 28-March 4, 1977. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 86. Hill, Gary C. and Jeffrey V. Bowles. <u>Study of a Very Low Cost</u> <u>Air Combat Maneuvering Trainer Aircraft</u>. NASA TMX-73,162. National Aeronautics and Space Administration, Ames Research Center, Moffett Field, California. August 1975.
- 87. Hill, Gary C. and Michael Harper. Design, Cost, and Advanced <u>Technology Applications for a Military Trainer Aircraft</u>. NASA TMX-62,469. National Aeronautics and Space Administration, Ames Research Center, Moffett Field, California. August 1975.
- 88. Hinton, David A. and John D. Shaughnessy. Adaptation of Time Line Analysis Program to Single Pilot Instrument Flight Research. NASA TM 78748. National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. August 1978.
- 89. Hoadley, Arthur W. Conversion of Wing Surface Pressures into Normalized Lift Coefficient. Paper no. 790567. Paper presented to the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas, April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 90. Hoerner, Sighard F. Fluid Dynamic Drag. Bricktown, New Jersey: Hoerner Fluid Dynamics, 1965.
- 91. Hoffman, Daniel J. <u>Environmental Exposure Effects on Composite Materials for Commercial Aircraft</u>. First Quarterly Progress Report. D6-44815. Prepared by the Boeing Commercial Airplane Company, Seattle, Washington for National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. February 1978.
- 92. Hoffman, Daniel J. <u>Environmental Exposure Effects on Composite</u> <u>Materials for Commercial Aircraft</u>. Second Quarterly Progress Report. D6-44815-2. Prepared by the Boeing Commercial Airplane Company, Seattle, Washington for National Aeronautics and Space Administration, Langley Research Center, Hampton Virginia. May 1978.
- 93. Hoffman, Daniel J. Environmental Exposure Effects on Composite <u>Materials for Commercial Aircraft</u>. Third Quarterly Progress Report. D6-44815-3. Prepared by the Boeing Commercial Airplane Company, Seattle, Washington, for National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. August 1978.
- 94. Hoffman, Daniel J. Environmental Exposure Effects on Composite <u>Materials for Commercial Aircraft</u>. Fourth Quarterly Progress Report. D6-44815-4. Prepared by the Boeing Commercial Airplane Company, Seattle, Washington for National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. December 1978.
- 95. Hoffman, Daniel J. <u>Environmental Exposure Effects on Composite</u> <u>Materials for Commercial Aircraft</u>. Fifth Quarterly Progress Report. D6-44815-5. Prepared by the Boeing Commercial Airplane Company, Seattle, Washington for National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. February 1979.
- 96. Hoffman, Daniel J. <u>Environmental Exposure Effects on Composite</u> <u>Materials for Commercial Aircraft</u>. Sixth Quarterly Progress Report. D6-44815-6. Prepared by the Boeing Commercial Airplane Company, Seattle, Washington for National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. May 1979.
- 97. Hoffman, William C. and Walter M. Hollister. Forecast of the General Aviation Air Traffic Control Environment for the 1980's. NASA CR-137909. Prepared by Aerospace Systems, Inc., Burlington, Massachusetts for National Aeronautics and Space Administration, Ames Research Center, Moffett Field, California. June 1976.
- 98. Hurkamp, C. H., et al. <u>Technology Assessment of Advanced General Aviation Aircrait</u>. NASA CR-114338. Prepared by the Advanced Concepts Department, Lockheed-Georgia Company, for National Aeronautics and Space Administration, Ames Research Center, Moffett Field, California. June 1971.
- 99. Hurkamp, C. H., W. M. Johnston, and J. H. Wilson. <u>Technology</u> <u>Assessment of Advanced General Aviation Aircraft</u>. (Final Report). NASA CR-114339. Prepared by Lockheed-Georgia Company, Marietta, Georgia for National Aeronautics and Space Administration, Ames Research Center, Moffett Field, California. May 1971.
- 100. Jackson, A. H. Jr., and B. S. Gatzen. <u>Multi-Mission Uses for</u> Prop-Fan Propulsion. Paper presented at the AGARD Conference on Variable Geometry and Multicycle Engines, Paris, France. September, 1976. Hamilton Standard, Division of United Technologies Corporation, Windsor Locks, Connecticut.
- 101. Jeracki, Robert J., Daniel C. Mikkelson, and Bernar. J. Blaha. <u>Wind Tunnel Performance of Four Energy Efficient Propellers</u> <u>Designed for Mach 0.8 Cruise</u>. Paper no. 790573. Presented at the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas, April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.

- 102. Joglekar, Amil N. and Karl Seiler, III. <u>Economic Analysis of</u> <u>Future Civil Air Navigation Systems.</u> (Final Report for CY-1977). Report no. FAA-EM-78-6. Propared by MITRE Corporation, METREK Division for U. S. Department of Transportation, Federal Aviation Administration, Office of Systems Engineering Management, Washington, D. C. December 1977.
- 103. Jones, Charles. <u>Advanced Rotary Engine Studies</u>. Paper presented at the General Aviation Propulsion Conference, National Aeronautics and Space Administration, Lewis Research Center, Cleveland, Ohio. November 28-29, 1979. Curtiss-Wright Corporation, Wood-Ridge, New Jersey. November 1979.
- 104. Jones, Charles. <u>A Review of Curtiss-Wright Rotary Engine</u> <u>Developments with Respect to General Aviation Potential</u>. Paper no. 790621. Presented at the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas. April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 105. Jones, Charles, et al. An Update of the Direct Injected Stratified Charge Rotary Combustion Engine Developments at Curtiss-Wright. Paper no. 770044. Paper presented at the SAE International Automotive Engineering Congress and Exposition, Detroit, Michigan. February 28-March 4, 1977. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 106. Kirsch, Richard. "Improved Survivability for General Aviation Aircraft." <u>Proceedings of the First FAA General Aviation</u> <u>Research and Development Conference</u>. Paper presented at a conference conducted by the Department of Transportation, Federal Aviation Administration, National Aviation Facilities Experimental Center (NAFEC), Atlantic City, New Jersey. August 17-18, 1977.
- 107. Klug, H. G. "Transport Aircraft with Relaxed/Negative Longitudinal Stability -- Results of a Design Study." <u>AGARD Conference Proceedings No. 157 on Impact of Active Control</u> <u>Technology on Airplane Design</u>. Paper presented at a Joint Symposium of the Flight Mechanics Panel and Guidance and Control Panel of the Advisory Group for Aerospace Research and Development (AGARD) held in Paris, France. October 14-17, 1974.
- 108. Knox, Charles E. and Desmond Hartnell. <u>A Comparison of Two</u> <u>Commercial and the Terminal Configured Vehicle Area Navigation</u> <u>Systems</u>. NASA TMX-72683. National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. June 17, 1976.

- 109. Koenig, R. W. and G. K. Sievers. <u>Preliminary QCGAT Program</u> <u>Test Results</u>. Paper no. 790596. Paper presented at the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas. April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Penncylvania.
- 110. Kohlman, David L. <u>Flight Evaluation of a Spoiler Roll Control</u> <u>System on a Light Twin-Engine Airplane</u>. NASA CR 2935. Prepared by University of Kansas Center for Research, Inc., Flight Research Laboratory, Lawrence, Kansas for National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. January 1978.
- 111. Kohlman, David L. Flight Test Results for an Advanced Technology Light Airplane. Paper no. 77-1217. Presented at the AIAA Aircraft Systems and Technology Meeting, Seattle, Washington. August 22-24, 1977. American Institute of Aeronautics and Astronautics. New York, N. Y.
- 112. Kowalski, S. H. <u>Avionics Cost Development for Civil Appli-</u> <u>cation of Global Positioning System</u>. (Interim Report). Publication of Transportation, Federal Aviation Administration, Office of Systems Engineering Management, Washington, D. C. July 1978.
- 113. Kroeger, R. A. and T. W. Feistel. <u>Reduction of Stall-Spin</u> <u>Entry Tendencies through Wing Aerodynamic Design</u>. Paper no. 760481. Presented at the SAE Business Aircraft Meeting, Wichita, Kansas, April 6-9, 1976. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 114. Laitone, E. V. "Positive Tail Loads for Minimum Induced Drag of Subsonic Aircraft." Journal of Aircraft, 15:836-843 (December 1978).
- 115. Lamar, J. E. <u>A Vortex Method for the Mean Camber Shapes of</u> <u>Trimmed Non-Coplanar Planforms with Minimum Vortex Drag.</u> NASA TN D-8090. National Aercnautics and Space Administration, Langley Research Center, Hampton, Virginia. June 1976.
- 116. Lan, C. E. "A Quasi-Vortex-Lattice Method in Thin Wing Theory" Journal of Aircraft, 11:518-527 (September 1974).
- 117. Lang, James D. and Michael S. Francis. Dynamic Loading on an <u>Airfoil due to a Growing Separated Region</u>. Paper no. 27. Presented at the AGARD Conference Proceedings No. 204 on Prediction of Aerodynamic Loading held at National Aeronautics and Space Administration, Ames Research Center, Moffett Field, California. September 1976.

- 118. Lange, R. H., at al. <u>Applications of Active Controls</u> <u>Technology to the NASA Jetstar Airplane</u>. NASA CR-2561. Prepared by Lockheed Georgia Company for National Aeronautics and Space Administration, Flight Research Center, Edwards, California. June 1975.
- 119. Lapins, Maris and Ira D. Jacobson. <u>Application of Active</u> <u>Controls Technology to Aircraft Ride Smoothing Systems</u>. NASA CR-145980. Prepared by University of Virginia, Charlottesville, Virginia. May 1975.
- 120. Larson, George C. "Composite Materials and General Aviation." Business and Commercial Aviation, pgs. 86-98 (September 1979).
- 121. Laundry, William E. <u>A Review of Spoiler-Type Ailerons for</u> <u>General Aviation Aircraft</u>. LTR-LA-199. Paper presented at the 1976 National Business Aircraft Meeting and Engineering Display, Wichita, Kansas. April 6-8, 1976. National Research Council Canada, National Aeronautical Establishment, Ottawa, Canada. April 1976.
- 122. Lays, E. J. <u>Turbine Engines in Light Aircraft</u>. Paper presented at the National Conference on Energy Conservation in General Aviation, Kalamazoo, Michigan, October 10-11, 1977.
- 123. Lays, E. J. and D. L. Murray. <u>General Aviation Turbine</u> <u>Engine (GATE) Study -- Final Report</u>. NASA CR-159603. Prepared by Williams Research Corporation, Walled Lake, Michigan, for National Aeronautics and Space Administration, Lewis Research Center, Cleveland, Ohio. June 26, 1979.
- 124. Lew, James N. "Planning a General-Aviation Product." <u>Astro-</u> nautics and Aeronautics, 15:76-82 (January 1977).
- 125. Linstone, Harold A. and Murray Turoff. <u>The Delphi Method</u> --<u>Techniques and Applications</u>. Reading, <u>Massachusetts</u>: <u>Addison-</u> <u>Wesley Publishing Company</u>, 1975.
- 126. Lovell, D. T. and R. V. Carter. <u>New Materials for Future</u> <u>Commercial Aircraft</u>. Paper no. 79-1804. Paper presented at the AIAA Aircraft Systems and Technology Meeting, New York, N. Y. August 20-22, 1979. American Institute of Aeronautics and Astronautics, New York, N. Y.
- 127. Ludwig, Walter, Herman Erbacher, and Joseph Visconti, "Design, Fabrication and Test of the B-1 Composite Horizontal Stabilizer." <u>Third Conference on Fibrous Composites in Flight Vechile</u> <u>Design - Part I.</u> NASA TMX-3377, pgs. 51-70. Conference cosponsored by the National Aeronautics and Space Adminstration and the U. S. Air Force, Williamsburg, Virginia, November 4-6, 1975.

- 128. Mack, M. D., et al. <u>Aerodynamics of Spoiler Control Devices</u>. Paper no. 70-1873. Presented at the AIAA Aircraft Systems and Technology Meeting, New York, N. Y. August 20-22, 1979. American Institute of Aeronautics and Astronatuics, New York, N. Y.
- 129. Maddalon, Dal V., et al. <u>Energy and Economic Trade Offs for</u> <u>Advanced Technology Subsonic Aircraft</u>. NASA TMX-72833. National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. April 1976.
- 130. Madden, Paul and Mukund Desai. Nonlinear Trajectory Following in the Terminal Area: Guidance, Control and Flight Mechanics Concepts using the Microwave Landing System. Paper no. 73-903. Presented at the AIAA Guidance and Control Conference, Key Biscayne, Florida. August 20-22, 1973. American Institute of Aeronautics and Astronautics, New York, N. Y.
- 131. Magnavox Government and Industrial Electronics Company. <u>Design Study of a Low Cost Civil Aviation GPS Receiver System.</u> NASA CR-159176. Prepared by Magnavox Government and Industrial Electronics Company, Advanced Product Division, Torrance, California, for National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. December 1979.
- 132. Margason, R. J. Vortex-Lactice Fortran Program for Estimating Subsonic Aerodynamic Characteristics of Complex Planforms. NASA TN D-6142. National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. February 1971.
- 133. McCalla, T. M., et al. <u>Preliminary Candidate Advanced</u> <u>Avionics System for General Aviation</u>. NASA-CR-15205. Prepared by Southern Illinois University, Carbondale, Illinois, for National Aeronautics and Space Administration, Ames Research Center, Moffett Field, California. July 1977.
- 134. McCormick, Barnes W. <u>Aerodynamics</u>, <u>Aeronautics</u>, and <u>Flight</u> Mechanics. New York: John Wiley and Sons, 1979.
- 1.5. McCormick, B. W., et al. <u>The Analysis of Propellers Including</u> <u>Interaction Effects</u>. Paper no. 790576. Paper presented at the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas. April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 136. McCullough, Bruce and Lloyd W. Bingham, Jr. <u>A New Light Twin</u> <u>Using Bonded Metal Construction</u>. Paper no. 790603. Presented at the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas. April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.

- 137. McDonnell Douglas Corporation. USAF Stability and Control. DATCOM. Prepared under contract AF 33(616)-6460, October 1969, and revised under contract F33615-74-C-3021, January 1975. Flight Controls Division, Air Force Flight Dynamics Laboratory, Wright-Patterson Air Force Base, Ohio.
- 138. McGhee, Robert J. and William D. Beasley. Low-Speed Aerodynamic Characteristics of a 17-Percent-Thick Airfoil Section Designed for General Aviation Applications. NASA TN D-7428. National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. December 1973.
- 139. McGhee, Robert J., William D. Beasley and Richard T. Whitcomb. <u>NASA Low- and Medium-Speed Airfoil Development</u>. NASA TM 78709. (Paper released for early domestic dissemination.) National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. Marcy 1979.
- 140. Mertens, Randy. "General Aviation Avionics: 1997." <u>Aircraft</u> Owners Bulletin, 2:8-11 (June 15, 1979).
- 141. Metzger, F. G. and C. Rohrbach. <u>Aeroacoustic Design of the Prop-Fan</u>. Paper no. 79-0610. Presented at the AIAA 5th Aeroacoustic Conference, Seattle, Washington, March 12-14, 1979. American Institute of Aeronautics and Astronautics, New York, N. Y.
- 142. Miley, S. J., et al. <u>Determination of Cooling Air Mass Flow</u> for a <u>Horizontally</u> - <u>Opposed Aircraft Engine Installation</u>. Paper no. 790609. Presented at the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas. April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 143. Miller, Robert W. "How to Plan and Control with PERT." Harvard Business Review, 40:93-104 (March 1962).
- 144. Mooij, H. A. "Handling Quality Criteria Development for Transport Aircraft with Fly-by-Wire Primary Flight Control Systems." <u>AGARD Conference Proceedings No. 157 on Impact of</u> <u>Active Control Technology on Airplane Design. Paper pre-</u> sented at a Joint Symposium of the Flight Mechanics Panel and Guidance and Control Panel of the Advisory Group for Aerospace Research and Development (AGARD) beld in Paris, France. October 14-17, 1974.
- 145. Morgan, Homer G. "Trends in Aircraft Noise Alleviation." <u>Vehicle Technology for Civil Aviation -- The Seventies and</u> <u>Beyond.</u> NASA SP-292. Proceedings of a conference held at Langley Research Center, Hampton, Virginia. November 2-4, 1971.

266

- - L

- 146. Nagao, Akitoshi, et al. <u>Adhesive Bonded Structure of New</u> <u>Pressurized Piston Twin Aircraft</u>. Paper no. 790563. Paper presented at the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas. April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 147. National Aeronautics and Space Administration. Avionics and Controls Research and Technology. Edited by Herman A. Rediess, NASA Headquarters, and Duncan E. McIver, NASA-Langley Research Center. NASA Conference Publication 2061. Proceedings of a workshop sponsored by the NASA Office of Aeronautics and Space Technology and hosted by the Langley Research Center, Hampton, Virginia. June 27-29, 1978.
- 148. National Aeronautics and Space Administration. Avionics: <u>Projections for Civil Aviation 1995</u> - 2000. Edited by Patricia Carlson and John Krobock. NASA CR-159035. Report compiled and written by 23 authors who were participants in the 1979 NASA-ASEE Summer Faculty Fellowship Program in Engineering Systems Design. National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia.
- 149. National Aeronautics and Space Administration. <u>General</u> <u>Aviation Propulsion Conference</u>. Figure preprints from a conference held at the Lewis Research Center, Cleveland, Ohio. November 28-29, 1979.
- 150. National Transportation Safety Board. Special Study --Single-Engine, Fixed-Wing General Aviation Accidents, 1972-1976. Report no. NTSB-AAS-79-1. Bureau of Technology, National Transportation Safety Board, Washington, D. C. May 31, 1979.
- 151. Neal. G. L. <u>Electronic System Safety Testing Reality</u>. Paper no 790570. Paper presented at the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas. April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 152. Nelson, Richard W. "Cabin Safety by Crash Survival." Proceedings of the First FAA General Aviation Research and Development Conference. Paper presented at a conference conducted by the Department of Transportation, Federal Aviation Administration, National Aviation Facilities Experimental Center (NAFEC), Atlantic City, New Jersey. August 17-18, 1977.

- 153. Newman, Michael and George Huggins. <u>Conceptual Design of</u> <u>Single Turbofan Engine Powered Light Aircraft</u>. NASA CR-151973. Prepared by Cesana Aircraft Company, Pawnee Division, Wichita, Kansas, for National Aeronautica and Space Administration, Ames Research Center, Moffett Field, California. March 31, 1977.
- 154. Nicolai, Leland M. <u>Fundamentals of Aircraft Design</u>. Fairborn, Ohio: E. P. Domicone Printing Services, 1975. Distributed by the School of Engineering, University of Dayton, Ohio 45469.
- 155. Niessen, Frank R. <u>A Low-Cost Inertial Smoothing System for Landing Approach Guidance</u>. NASA TN D-7271. National Aeronautics and Space Administration, Washington, D. C. June 1973.
- 156. North, David M. "Winglets Boost Performance of Learjet." <u>Aviation Week and Space Technology</u>, 111:39-47 (August 20, 1979).
- 157. Nunnally, Jim. <u>Psychometric Theory</u>. New York: McGraw-Hill Book, Co., 1967.
- 158. Parker, D. E., W. L. O'Brien, and W. R. Johnston. "The Realization of Cost and Weight Savings by the Application of Advanced Composites to the B-1 Vertical Stabilizer." <u>Third Conference</u> on Fibrous Composites in Flight Vehicle Design - Part I. NASA TMX-3377, pgs. 33-50. Conference co-sponsored by the National Aeronautics and Space Administration and the U. S. Air Force, Williamsburg, Virginia, November 4-6, 1975.
- 159. Parkinson, G. V., et al. <u>A Prediction Method for Spoiler</u> <u>Performance</u>. Paper no. 770459. Presented at the Business Aircraft Meeting, Wichita, Kansas. March 29-April 1, 1977. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 160. Patton, James M. Jr., et al. Spin Flight Research Summary. Paper no. 790565. Paper presented at the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas. April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 161. Paulson, John W. Jr. <u>Wind-Tunnel Investigation of a Fowler</u> <u>Flap and Spoiler for an Advanced General Aviation Wing</u>. NASA TN D-8236. National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. June 1976.

- 162. Paulson, John W. Jr. <u>Wind-Tunnel Test of a Conventional Flap</u> and <u>Aileron and a Fowler Flap and Slut-Lip Aileron for an</u> <u>Advinced General Aviation Wing</u>. Paper 750501. Presented at the 1975 National Business Aircraft Meeting, Wichita, Kansas. April 8-11, 1975. National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia.
- 163. Pasmany, L. <u>Future Trends in General Aviation</u>. Paper no. 70-1220. Presented at the AIAA 7th Annual Meeting and Technical Display, Houston, Texas. Octobe: 19-22, 1970. American Institute of Aeronautics and Astronautics. New York, N. Y.
- 164. Pinsker, W. J. G. "Active Control as an Integral Tool in Advanced Aircraft Design." <u>AGARD Conference Proceedings No.</u> <u>157 on Impact of Active Control Technology on Airplane Design.</u> Paper presented at a Joint Symposium of the Flight Mechanics Panel and Guidance and Control Panel of the Advisory Group for Aerospace Research and Development (AGARD) held in Paris, France. October 14-17, 1974.
- 165. Pitman, W. Andrew. "Cost Reduction with Composites in B-l Slats." <u>Third Conference on Fibrous Composites in Flight</u> <u>Vehicle Design - Part I.</u> NASA TMX-3377, pgs. 243-258. Conference co-sponsored by the National Aeronautics and Space Administration and the U. S. Air Force, Williamsburg, Virginia. November 4-6, 1975.
- 166. Polhamus, Edward C. "Subsonic and Transonic Aerodynamic Research." <u>Vehicle Technology for Civil Aviation -- The</u> <u>Seventies and Beyond</u>. NASA SP-292. Proceedings of a conference held at Langley Research Center, Hampton, Virginia. November 2-4, 1971.
- 167. Pratt and Whitney Aircraft of Canada, Ltd. <u>Aviation Fuels for</u> <u>the Future</u>. Unpublished Pamphlet. <u>Fratt</u> and Whitney Aircraft of Canada, Ltd., Longueuil, Quebec. June 13, 1979.
- 168. Rainey, A. Gerald. "Advanced Active Controls Technology." <u>Vehicle Technology for Civil Aviation -- The Seventies and</u> <u>Beyond.</u> NASA SP-292. Proceedings of a conference held at Langley Research Center, Hamptom, Virginia. November 2-4, 1971.
- 168. Rannenberg, G. C. <u>Chilled Recirculation ECS for Aircraft</u>. Paper no. 70-ENAs-5. Paper presented at hte 9th Intersociety Conference on Environmental Systems, San Francisco, California. July 16-19, 1979. The American Society of Mechanical Engineers, New York, N. Y.
- 169. Rezy, Bernard J., et al. Concepts for Reducing Exhaust Emissions and Fuel Consumption of the Aircraft Piston Engine. Paper no. 790605. Presented at the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas. April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.

- 170. Rezy, Bernard J., et al. <u>Screening Analysis and Selection of</u> <u>Emission Reduction Concepts for Intermittent Combustion Air-</u> <u>craft Engines</u>. NASA CR-135074. Prepared by Teledyne Continental Motors, Aircraft Products Division, Mobile, Alabama for National Aeronautics and Space Administration, Lewis Research Center, Cleveland, Ohio. November 1976.
- 171. Rich, Melvin J. and Raymond L. Foye. "Low Cost Composite Airframe Structures." <u>Third Conference on Fibrous Composties in</u> <u>Flight Vehicle Design - Part I.</u> NASA TMX-3377, pgs. 225-242. Conference co-sponsored by the National Aeronautics and Space Administration and the U. S. Air Force, Williamsburg, Virginia. November 4-6, 1975.
- 172. Rodgers, D. L. <u>Development and Flight Testing of a Fluidic</u> <u>Flight Control System</u>. NASA CR-913. Prepared by Honeywell, Inc., Minneapol's, Minnesota for National Aeronautics and Space Administration, Flight Research Center, Edwards AFB, California. October 1967.
- 173. Roesch, Phillippe and Raymond B. Harlan. <u>A Passive Gust</u> <u>Alleviation System for a Light Aircraft</u>. NASA CR 2605. Prepared by MIT Measurement Systems Laboratory, Cambridge, <u>Massachusetts for the National Aeronautics and Space Administration</u>, Washington, D. C. October 1975.
- 174. Roskam, Jan. <u>Airplane Flight Dynamics and Automatic Flight</u> <u>Controls.</u> Part I and II. Route 4, Ottawa, Kansas: Roskam Aviation and Engineering Corporation, 1979.
- 175. Roskam, Jan. <u>Design Philosophy and Hardware Implementation</u> of <u>Separate Surface Automatic Flight Control Systems</u>. Paper presented at the Technological University of Delft, The Netherlands, May 21, 1974. Department of Aerospace Engineering, University of Kansas, Lawrence, Kansas.
- 176. Roskam, Jan. Flight Test Results of a Separate Surface Wing-Leveling System. Paper no. 74-0369. Presented at the SAE Business Aircraft Meeting, Wichita, Kansas. April 2-5, 1974. Society of Automotive Engineers, Warrendale, Pennsylvania.
- 177. Roskam, Jan. Forward Swept Wings and Business Airplanes. Paper presented at the SAE Turbine Powered Executive Aircraft Meeting, Scottsdale, Arizona. April 9-11, 1980. Society of Automotive Engineers, Warrendale, Pennsylvania.
- 178. Roskam, Jan. <u>Handbook of Aircraft Design Data</u>. Part I, II, and III. Lawrence, Kansas: Roskam Aviation and Engineering Corporation. August 1975.
- 179. Roskam, Jan. <u>Methods for Estimating Drag Polars of Subsonic</u> <u>Airplanes</u>. (Second Printing). Lawrence, Kansas: Roskam Aviation and Engineering Corporation, 1971.

180. Roskam, Jan. <u>Methods for Estimating Stability and Control</u> <u>Derivatives of Conventional Subsonic Airplanes</u>. (3rd printing). Published by the author, Lawrence, Kansas: 1971.

4

- 181. Roskam, Jan. <u>Simulation and Simulator Development of a</u> <u>Separate Surface Attitude Command Control System for Light</u> <u>Aircraft.</u> Paper presented at the AGARD Flight Mechanics Panel and Guidance and Control Panel Joint Symposium, "Flight Simulation/Guidance Systems Simulation," The Hague, The Netherlands, October 20-24, 1975.
- 182. Roskam, Jan. <u>Utilization of Separate Surface Control Systems</u> on <u>General Aviation Aircraft</u>. Paper no. 770471. Presented at the SAE Business Aircraft Meeting, Wichita, Kansas. March 29-April 1, 1977. Society of Automotive Engineers, Warrendale, Pennsylvania.
- 183. Roskam, Jan, Kohlman, D. L. and W. H. Wentz. <u>Spoilers for</u> <u>Roll Control of Light Airplanes</u>. Paper no. 74-861. Presented at the AIAA Mechanics and Control of Flight Conference, Anaheim, California. August 5-9, 1974. American Institute of Aeronautics and Astronautics, New York, N. Y.
- 184. Royston, W. R. "Military Aircraft Flight Component Development." <u>Third Conference on Fibrous Composites in Flight</u> <u>Vehicles - Part I.</u> NASA TMX-3377, pgs. 191-224. Conference co-sponsored by the National Aeronautics and Space Administration and the U. S. Air Force, Williamsburg, Virginia. November 4-6, 1975.
- 185. Rutan, Burt. <u>Defiant Introductory Bulletin</u>. Rutan Aircraft Factory, Mojave, Calfornia. September 1, 1978.
- 186. Sakata, I. Frank and Robert B. Ostrom. <u>Study on Utilization</u> of <u>Advanced Composties in Commercial Aircraft Wing Structures</u>. <u>Volume 1: Executive Summary</u> (Final Report). NASA CR 154381-1. Prepared by Lockheed-California Company for National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. August 1978.
- 187. Sanders, Karl L. <u>High-Lift Devices, A Weight and Performance Methodology</u>. Technical Paper no. 761. Presented at the SAWE 28th Annual Conference, San Francisco, California. May 1969. The Society of Aeronautical Weight Engineers, Inc., Los Angeles, California.
- 188. Schaibly, J. H., et al. <u>Simulated and Projected Performance of the NAVSTAR GPS Control Segment</u>. Paper presented at the AIAA Guidance and Control Conference, San Diego, California. August 16-18, 1976. American Institute of Aeronautics and Astronautics, New York, N. Y.

189. Scott, Jerome E. and Peter Wright. "Modeling an Organizational Buyer's Product Evaluation Strategy: Validity and Procedural Considerations." <u>Journal of Marketing Research</u>, 13:211-224 (August 1976).

1

- 190. Shannon, R. W. <u>Lessons Learned from PABST</u>. Paper no. 790560. Paper presented at the SAE Busness Aircraft Meeting and Exposition, Wichita, Kansas. April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 191. Shevell, R. S. "Technological Developments of Transport Aircraft -- Past and Future." Journal of Aircraft, 17:67-80 (February 1980).
- 192. Shrager, J. J. <u>Head-Up Displays: A Literature Review and</u> <u>Analysis with an Annotated Bibliography</u>. FAA-RD-78-31. Performed by Federal Aviation Administration, National Aviation Facilities Experimental Center, Atlantic City, New Jersey. April 1978.
- 193. Smith, Gerald L. <u>Ames-Aided Inertial Navigation Work -- The</u> <u>First Two Years of Progress</u>. NASA TMX-62199. National Aeronautics and Space Administration, Ames Research Center, Moffett Field, California. April 1973.
- 194. Snyder, Richard G. and Thomas J. Armstrong. <u>Crashworthiness</u> <u>Analysis of Field Investigation of Business Aircraft Accidents.</u> Paper no. 790587. Paper presented at the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas. April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 195. Solomon, H. L. <u>Economic Requirements Analysis of Civil Air</u> <u>Navigation Alternatives</u>. <u>Volume I</u>. Report No. FAA-ASP-78-3. <u>Prepared by Systems Control, Inc. (Vt.), Palo Alto, California</u> for U. S. Department of Transportation, Federal Aviation Administration, Office of Aviation System Plans, Washington, D. C. April 1978.
- 196. Stanford University. <u>Transcription of the Workshop on General</u> <u>Aviation Advanced Avionics Systems</u>. NASA CR 137861. Edited by Michael Tashker, Stanford Research Institute. Prepared by Stanford University, Stanford, California for National Aeronautics and Space Administration, Ames Research Center, Moffett Field, California. May 1976.
- 197. Stapleford, Robert L., et al. <u>A Study of the Effects of Aircraft Dynamic Characteristics on Structural Loads Criteria.</u> Report no. FAA-RD-78-155. Prepared by Systems Technology, Inc., Hawthorne, California for U. S. Department of Transportation, Federal Aviation Administration, Systems Research and Development Service, Washington, D. C. November 1978.

- 198. Stein, Kenneth J. "Advanced Systems Aid Profile Descents." <u>Aviation Work and Space Technology</u>, 111:57-62 (August 20, 1979).
- 199. Stein, Kanneth J. "New Procedures Key to Fuel Savings." <u>Aviation Week and Space Technology</u>, 111:105-111 (August 27, 1979).
- 200. Stewart, Warner L., et al. <u>NASA Research on General Aviation</u> <u>Power Plants</u>. Paper no. 79-0561 presented at the AIAA 15th Annual Meeting and Technical Display, Washington, D. C. February 6-8, 1979. American Institute of Aeronautics and Astronautics, New York, N. Y.
- 201. Stoecklin, Robert L. <u>737</u> Graphite Composite Flight Spoiler Flight Service Evaluation. NASA CR 145207. Prepared by Boeing Commercial Airplane Company, Seattle, Washington, for National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. August 1977.
- 202. Stone, R. H. Flight Service Evaluation of Kevlar-49/Epoxy Composite Panels in Wide-Bodied Commercial Transport Aircraft. Third Annual Flight Service Evaluation Report. NASA CR-145141. Prepared by Lockheed-California Company, Burbank, California for National Aeronautics and Space Administration, Langley Research Center, Hampton, Virgina. March 1977.
- 203. Stough, H. P. III and J. M. Patton, Jr. <u>The Effects of Con-figuration Chauges on Spin and Recovery Characteristics of a Low-Wing General Aviation Research Airplane</u>. Paper no. 79-1786. Presented at the AIAA Aircraft Systems and Technology Meeting, New York, N. Y. August 20-22, 1979. American Institute of Aeronautice and Astronautics, New York, N. Y.
- 204. Strack, William C. <u>New Opportunities for Future Small Civil</u> <u>Turbine Engines -- Overviewing the GATE Studies</u>. Paper no. 790619. Presented at the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas. April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 205. Succi, George P. Design of Quiet Efficient Propellers. Paper no. 790584. Paper presented at the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas. April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 206. Tauschek, Max J. "Tomorrow's Spark-Ignition Engine." in <u>Powerplants for Industrial and Commercial Vehicles - A Look</u> <u>at Tomorrow</u>. SAE 270. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania. April 1965.

- 207. Teledyne Continental Motors. <u>Lightweight Diesel Aircraft</u> <u>Engines for General Aviation</u>. Unpublished pamphlet. Teledyne Continental Motors, General Products Division, Musekgon, Michigan.
- 208. Teper, Gary L, Rober H. Hoh, and Richard K. Smyth. <u>Prelim-inary Candidate Advanced Avionics System (PCAAS) Final Report.</u> NASA CR-152026. Prepared by Systems Technology, Inc., Haw-thorne, California for National Aeronautics and Space Admin-istration, Ames Research Center, Moffett Field, California. September 1977.

.

- 209. Thomas, James L. and David S. Miller. <u>Numerical Comparisons</u> of <u>Panel Methods at Subsonic and Supersonic Speeds</u>. Paper no. 79-0404. Presented at the AIAA 17th Aerospace Sciences Meeting, New Orleans, Louisiana. January 1979. American Institute of Aeronautics and Astronautics, New York, N. Y.
- 210. Thomson, Robert G. and Robert C. Goetz. <u>NASA/FAA General</u> <u>Aviation Crash Dynamics Program -- A Status Report</u>. <u>AIAA Paper no. 79-0780</u>. Paper presented at the AIAA/ASME/ <u>ASCE/AHS 20th Structures, Structural Dynamics, and Materials</u> <u>Conference, St. Louis, Missouri. April 4-6, 1979</u>. <u>American</u> <u>Institute of Aeronautics and Astronautics, New York, N. Y.</u>
- 211. Timmons, Lawrence M. Improving Business Jet Performance: <u>The Mark Five Sabreliner</u>. Paper no 790582. Paper presented at the SAE Business Aircraft Meeting and Exposition, Wichita, Kansas. April 3-6, 1979. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.
- 212. Tischler, Mark B., et al. <u>Flight Research Group Report on the Applications of Spoiler Systems to Twin Engine Aircraft.</u> Unpublished report prepared by the University of Maryland Aerospace Engineering Department's Flight Research Group. August 7, 1978.
- 213. Tomazic, William. "Alternate General-Aircraft Engines." <u>Aircraft</u> Piston Engine Exhaust Emissions Symposium. NASA CP-2005, pgs. 315-328. Conference co-sponsored by the Environmental Protection Agency, Federal Aviation Administration, and National Aeronautics and Space Administration, and held at the Lewis Research Center, Cleveland, Ohio. September 14-15, 1976.
- 214. Torenbeek, E. <u>Synthesis of Subsonic Airplane Design</u>. Delft, Netherlands: Delft University Press, 1976.
- 215. Trammell, Archie, Robert Parrish, and Arnold Lewis. "The Commuter Airlines -- The Growth Potential, Regulatory Outlook, and Equipment Needs." <u>Business and Commercial Aviation</u>, pgs. 73-96 (February 1978).

- 216. Taeka, J. and G. C. Letton, Jr. <u>Variable Geometry Air Cycle</u> <u>Machine</u>. Paper no. 77-EMAs-7. The American Society of Mechanical Engineers, New York, N. Y.
- 217. Tulinius, Jan R. and Richard J. Margason. <u>Aircraft Aerodynamic Design and Evaluation Methods</u>. Paper no. 76-15. Presented at the AIAA 14th Aerospace Sciences Meeting and 12th Annual Meeting: Aerospace '76, Washington, D. C. January 26-30, 1976, American Institute of Aeronautics and Astronautics, New York, N. Y.
- 218. University of Celifornia, Los Angeles. <u>Biomedical Computer</u> <u>Programs</u>. Edited by W. J. Dixon. Los Angeles: University of California Press, 1976.
- 219. University of California, Los Angeles. <u>Display Systems</u> <u>Engineering</u>. Unpublished notes collected and bound for a five day short course coordinated by Dr. Harold R. Luxenberg. University of California, Los Angeles - University Extension, Continuing Education in Engineering and Mathematics, Los Angeles, California. June 18-22, 1979.
- 220. Vaicaitis, R. "Noise Transmission into a Light Aircraft." Journal of Aircraft, 17:81-86 (February 1980).
- 221. Vaughan, Victor L. Jr. and Emilio Alfaro-Bou. Impact Dymanics Research Facility for Full-Scale Aircraft Crash Testing. NASA TN D-8179. National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. April 1976.
- 222. Weathers, Terry M. <u>NASA</u> <u>Contributions to Fluidic Systems</u> --<u>A Survey</u>. NASA SP-5112. Prepared by Computer Sciences Corporation, Falls Church, Virginia, for National Aeronautics and Space Administration, Technology Utilization Office, Washington, D. C. 1972.
- 223. Wentz, W. H. Jr. Effectiveness of Spoilers on the GA(W)-1 <u>Airfoil with a High Performance Fowler Flap</u>. NASA CR-2538. Prepared by Wichita State University, Wichita, Kansas for National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. May 1975.

i

224. Wentz., W. H. Jr. <u>Reflection-Plane Tests of Spoilers on an</u> <u>Advanced Technology Wing</u>. Paper no. 760482. Presented at the <u>Business Aircraft Meeting</u>, Wichita, Kansas. April 6-9, 1976. Society of Automotive Engineers, Warrendale, Pennsylvanis.

- 225. Wents, W. H., Jr. and H. C. Sestheren. <u>Development of a</u> <u>Powler Flap System for a High Performance General Aviation</u> <u>Airfoil</u>. NASA CR-2443. Prépared by Wichita State University for the Center for Research, Inc., University of Kansas, for National Aeronautica and Space Administration, Langley Research Center, Hampton, Virginiz. December 1974.
- 226. Werner, J. and A. R. Krenkel. <u>Slender Body Theory Programmed for Bodies with Arbitrary Cross Section</u>. NASA CR-145383. Prepared by Polytechnic Institute of New York, Farmingdale, N. Y. for National Aeronautics and Space Administration, Langley Research Center, Hampton, Virginia. February 1978.
- 227. Westfield, William. "Near-Term Engine Emission Reduction Investigation." <u>Proceedings of the First FAA General</u> <u>Aviation Research and Development Conference</u>. Paper presented at a conference conducted by the Department of Transportation, Federal Aviation Administration, National Aviation Facilities Experimental Center (NAFEC), Atlantic City, New Jersey. August 17-18, 1977.
- 228. Williams, P. R. G. and B. S. Campion. "Impact of Active Control Technology on Aircraft Design." <u>AGARD Conference</u> <u>Proceedings No. 157 on Impact of Active Control Technology</u> <u>on Airplane Design.</u> Paper presented at a Joint Symposium of the Flight Mechanics Panel and Guidance and Control Panel of the Advisory Group for Aerospace Research and Development (AGARD) held in Paris, France. October 14-17, 1974.
- 229. Willis, Edward A. <u>General Aviation Internal-Combustion</u> <u>Engine Research Programs at NASA-Lewis Research Center</u>. NASA TM-78891. National Aeronautics and Space Administration, Washington, D. C. July 1978.
- 230. Wrobel, Rose and Millard G. Mayo. <u>Q-Fans for General Avia-tion Aircraft</u>. NASA CR 114665. Prepared by Hamilton Standard Division of United Aircraft Corporation, Windsor Locks, Connecticut, for National Aeronautics and Space Administration, Ames Research Center, Moffett Field, California. December 1973.
- 231. Yamamoto, Kenichi and Takumi Muroki. <u>Development on Exhaust</u> <u>Emissions and Fuel Economy of the Rotary Engine at Toyo Kogyo</u>. Paper no. 780147. Paper presented at the SAE Congress and Exposition, Detroit, Michigan. February 27-March 3, 1978. Society of Automotive Engineers, Inc., Warrendale, Pennsylvania.

- 232. Yulo, Carlo. An Overview of Enroute Radio Nevigation Services for Civil Aviation. Report no. FAA-RD-76-187. Final Report. Prepared for U. S. Department of Transportation, Federal Aviation Administration, Systems Research and Development Service, Washington, D. C. August 1976.
- 233. Zelon, C. C., J. E. Cassidy and R. G. Shipley. <u>Fiber Optics</u> <u>Cost Analysis Program (FOCAP)</u>. AFAL-TR-77-190. Prepared by Rockwell International, Los Angeles Division, for the Air Force Avionics Laboratory, Air Force Systems Command, Wright Patterson AFB, Ohio. September 1975.

5 1)

ì

.\*

# APPENDIX A NOTES FROM VISITS TO MANUFACTURERS AND RESEARCH CENTERS

This appendix contains notes generated by the research team during the course of visits to 31 manufacturers and three research centers. The notes have been edited to a certain degree and the numes of manufacturers have been deleted. They are included as a part of this report because comments, ideas, and opinions gained from the visits were of extreme value to the present research effort. Also, it is hoped that certain views expressed from time to time may now be made a matter of public record in a form more accessible to the general aviation community than previously possible.

It must be emphasized that this appendix contains the views of those persons visited and do not necessarily reflect the opinions of the authors of this report. In those cases where technologies which were evaluated as part of this research are discussed, the reader is advised to refer to Chapter 4 for further information. Also, when a statement attributing some characteristic of "Company B" appears under a discussion for "Company A", the statement reflects Company A's view of Company B's position, involvement, or action.

HINCEBING PAGE PLANK MED PLANT

## A.1 AIRFRAME MANUFACTURERS

## A.1.1 Company A

## I. Powerplants and Propulsion

The real need here is powerplants up to and including the 746 kw (1000 HP) range.

Liquid Cooling

The major advantage is as a configuration tool. The engine and radiator can be placed independently of each other in the aircraft. Might possibly reduce cooling drag.

More efficient; cooling liquid can be used to heat the cabin.

On a direct weight and cost comparison with air-cooled engines, liquid cooling will weigh more and cost more. Excess cost can be returned in performance gains and configuration allowances.

Increases TBO of the engine by cooling the cylinders more evenly and thereby allowing closer tolerances between components.

## **Diesel Engines**

High fuel tolerance. Will burn practically anything.

Excellent fuel efficiency.

Could suffer a weight problem.

#### Stratified Charge Engines

Higher fuel efficiency.

Low emissions.

#### Rotary Engines

Offers advantages in size and smoothness of operation.

۰.

\$

Coupled with a stratified charge mixture, fuel consumption can be lowered to conventional aircraft engine levels.

#### Turbocharging

Current aircraft turbochargers are not matched to engines. They are simply diesel truck turbochargers which are "tacked on" to airplane engines.

If turbochargers were optimized for aircraft, 3047 m (10,000 ft) could be added to current ceilings.

This improvement is one of the most realistic and achievable goals for the present.

Increased TBO's

Turbine engines have the potential for 5000 to 7000 hour TBO's with careful monitoring. Without it, 3500 hour TBO's are still possible. However, a strict maintenance schedule may be too expensive and time consuming.

Current reciprocating engines have TBO's of only 2000 hours. The high-powered, 6 cylinder injected engine is a real maintenance problem.

Liquid cooling is essential to increasing TBO's. It allows closer tolerances. If not liquid cooled, TBO's of current reciprocating engines could still be increased by the installation and incorporation of larger bearings and by adding 50 lbs of structural "beef-up" to the engine.

Improved Ignition and Carburetion

Current aircraft reciprocating engines need spark advance incorporated into them. Presently, they have none. At least mechanical advances should be used.

Electronic control with a manual backup for throttle and mixture would increase range by 20%. (Note that best power and economy occur a: two different operating points.)

A combination of these would greatly improve engine life and fuel economy.

Propellers

Propellers can be improved by utilizing thinner airfoils tailored to the span of the blade.

Composite construction of blades would increase the damage tolerance of the propeller.

(Company A)

Neise reduction is required. One way of accomplishing this is to sweep the blade tips.

Emphasis was placed on economics, schedules, and judgement vs analysis during the design phase. Cost is the most important underlying factor.

The skill of the labor force is an important consideration. "Planes must be built by folks, not craftsmen."

## II. Structures

ì

Metal-to-metal bonding

This is a current technology for general aviation, both on secondary and primary structures.

This company currently bonds 4000 to 6000 subassemblies per month.

Composites

Composites are used extensively, but in secondary structure only.

Use of composites in primary structures is constrained by the FAA. The FAA requires two times the normal factor of safety of 1.5. This added strength requirement eliminates any weight savings from composites.

Aluminum is still used because of its low cost.

Prediction Methods

A flight test flutter prediction method is needed. Must allow real time flutter analysis.

A good noise prediction method is needed, both for internal and external noise.

A high frequency loads analysis method is needed for the 70 to 100 Hz range.

## III. Aerodynamics

**Required Prediction Methods** 

Tail effectiveness as a function of tail and wing location, both longitudinally and vertically.

#### (Company A)

A method for examining the fuselage-nacelle channel on twins is required. Providing low-speed inboard lift is a key requirement. Must eliminate tail buffet. A good potential flow model would probably solve the problem.

An accurate method for predicting hinge moments for ailerons, flaps, and spoilers is needed.

## Tailored Airfoil Sections

Sections should be generated for the specific task instead of being picked out of a handbook. Tailoring the section can reduce drag by as much as 35 counts.

GA(W) wing (airfoil) is difficult to manufacture.

OSU airfoil service is not used because of an inhouse capability for generating airfoil sections.

IV. Systems

Suggested Methods for Anti-Icing

Microwave, but interferrence with MLS may be possible.

Sonic and pulsating methods.

Avionics

Primary emphasis in this area should be placed on reducing pilot workload. This requires simplified avionics.

Engine Condition Monitors

Some are currently in operation.

Suggested engine particles be monitored, and that microphones be used for detecting unusual sounds.

V. <u>Canards</u>

۰,

1

Not a good airplane and doesn't compete with this manufacturer's product line. Present canards are "neat" airplanes, but shouldn't be considered a utility airplane.

Locating the landing gear and fuel will be difficult.

Trim difficulties expected when flaps are incorporated on the aft wing.

## A.1.2 Company B

## I. Propulsion

ì

# NASA-Lewis Research Programs

Company B would like to see the four major programs for General Aviation continued: (1) GATE, (2) Stratified Charge Reciprocating Engine, (3) Stratified Charge Rotary Combustion Engine, (4) Advanced Diesel.

Also want to see continued propeller studies, with emphasis on weight, composites, noise, and efficiency.

Currently is involved with the GAP program: (1) Agricultural aircraft, (2) Turboprops, (3) Light single engine aircraft.

Mentioned the Ames cooling drag project and expressed interest in it.

Noise

Keenly interested in exterior and interior noise research. Supports the noise reduction tests at KU and LaRC.

Suggested possibility of electro-static cancellation of interior noise.

## Turbocharging

Current turbochargers need to be optimized.

Fuels and alternate power sources

More research needed on alternate fuels for General Aviation.

Suggested NASA investigate alternate power sources.

Lockheed is doing work on battery development: Lithium Hydroxide, Hydrogen Peroxide. Might have significant advantages for aviation applications, especially if electric propulsion becomes a possibility.

Liquid Cooling

Sees major advantages here.

50% increase in TBO's is required in order to make this attractive.

(Company B)

Improved Ignition and Carburetion

Sees a definite need here, but feels that very basic research is required to fully assess benefits of advancing timing as well as better mixture and throttle control.

This company was involved in a timing change study.

## Turboprops and Turbofans

Visualizes the 149 kw - 373 kw (200 -500 HP) range becoming a reality for turbines.

Turbofans cannot match the efficiency of turboprops below a 400 knot flight regime.

## Rotary Combustion Engine

Not much interest here.

## II. Crashworthiness and Safety

Crashworthiness

Further work required here, particularly with respect to composites. Composites splinter and release energy all at once as opposed to aluminum which (in crush and dissipate loads.

Safety

An investigation of the stall/spin problem is needed now. Since the simplest way to prevent a spin is to prevent the stall, incorporate active stall prevention. Don't give the pilot the ability to stall.

#### III. Aerodynamics

h,

GA(W) sections

Feels as though this section was oversold. Wing-body interference problems outweigh benefits. Aft loaded sections cause problems with wing-body interference/ separation as well as with trim drag due to pitching moment characteristics. Control surfaces tend to float.

OSU Computer Service

Company B uses the service.

This company would like to see results of programs like ATLIT and Redhawk disseminated more quickly.

(Company B)

## IV. Structures

#### Composites

FAA certification is a problem, but this is to be expected because there are severe technical problems with the incorporation of composites.

A General Aviation composites study is required (comprehensive) over a 15 year period. All aspects, such as manufacturing, aging, environmental effects, should be addressed. Moisture absorption is a problem.

NASA should support composite tests for General Aviation much like the programs for commercial transports.

Composites are utilized in secondary structure only. Lack of sufficient data precludes use in primary structure.

Stiffness characteristics as opposed to strength should be verified.

Investigate possible implementation in engines. Good potential for weight savings there.

Has great concern for crashworthiness aspects.

## V. Systems

Information -----

Aircraft systems, if improved, offer potential for substantial gains. Tradeoffs in dollars, weight, and side effects are significant.

Electrical

Areas such as generators, alternators, and starters need more attention. Antennas can be greatly improved.

Fly-by-Wire and Fly-by-Light

Despite anticipated certification difficulties, outlook is optimistic for fly-by-wire.

Fly-by-light has little application.

SSSA and Active controls

Gust alleviation systems offer substantial benefits. Ride Quality is second to gust alleviation. Active flutter suppression might be attractive.

(Companies B & C)

SSSA would be good as a wing leveler, yaw damper, or total autopilot control. Certification problems might be encountered.

## Avionics

I

Digital data links between the aircraft and ground are required for reducing cockpit noise, voice communication, and pilot workload.

A real need exists for advanced integrated displays in the cockpit.

Digital data links with the ground could also provide weather information and collision avoidance.

CRT displays could make IFR flight as easy as VFR.

Key to advanced avionics is effective integration of advanced microprocessors. Required if advanced systems like engine controls are to be incorporated.

More efficient heating and cooling for the cabin is desirable. Better air cycle systems should be developed with better efficiencies. The same is needed for electric, hydraulic, and pneumatic systems.

## VI. Canards

This company is neutral toward canard configurations. Some points expressed include:

Might be an attractive solution to stall/spin problems.

Might impair visibility from the cockpit.

Notes: (1) Be energy conscious.

(2) Research institutions should solve the technical problems. Industry will solve the production and certification problems if the technology has enough merit.

#### A.1.3 Company C

I. Cost

New Technologies

New technologies are attractive only if low cost.

## (Company C)

An increase in purchase price is allowable only when a decrease in life cycle costs are realized. Price is not too important in the high performance category of airplanes.

## II. Flight Demonstrator

Reduced Noise

Use mufflers.

Explore Boeing method for acoustic paneling in the inlets of turbine engines.

Drag Reduction

Increase aspect ratio. Negligible benefit expected for small aircraft due to increase in weight.

Perform a general clean-up of the exterior surfaces.

Configuration

Explore canards and tandem wings and determine if the benefits claimed really exist.

Winglets

Utilize for lateral-directional control.

Explore the effects of variable geometry winglets.

High Lift Devices

Explore different high lift devices and determine what is most effective for reducing approach and landing speeds.

## III. Structures

Composites

No composites are used by this company except for fiberglass radomes and fairings.

Graphite corrodes metal rivets.

Crashworthiness aspects need to be explored.

Major problem is a lack of data on fatigue, manufacturing methods, inspectability, and certification requirements.

(Company C)

Composites are a difficult endeavor for manufacturers to pursue due to costs and certification difficulties. Here, a manufacturer must gain experience with secondary structures before pursuing more ambitious projects.

#### Bonding

The military has had problems with field servicing and maintenance of bonded structures.

No bonding is used by this company.

## IV. Systems

Fly-by-wire and Fiber Optics

Fiber optics offer potential weight savings.

Interference problems hamper electronic systems. EMI shielding doubles conventional wire weight. Fiber optics, however, are free of radio interference.

## Active Controls

Hampered by expensive redundancies.

May pay off in ride control for aircraft with low wing loadings. Load control can be effective in reducing weight of the vertical fin and high aspect ratio wings.

This company is uncertain about the application of active controls to general aviation aircraft.

#### Batteries

New lead-acid batteries are better in cost and weight than NICADS.

## V. Computational Methods and Research

Airfoils

Work with NASA has been helpful in comparing codes.

The OSU facility has not been used.

#### NASTRAN

Used with good correlation noted.

## (Companies C & D)

Flutter

Free play and friction in flight controls and their effect on flutter need to be studied. What are allowable tolerances here?

Aileron "buzz" needs to be studied.

VI. Propulsion

TBO

One problem known to exist with a particular turbofan is a degradation in maintenance where typical values experienced reflect

- 300 hours between inspection and overhaul of the hot section.
- 900 hours between inspection and overhaul of the gearbox.

By comparison, a turbojet presently in use has a 4000 hour TBO.

<u>Note</u>: Engineering support costs and certification costs amount to 10 to 100 times the cost of the technology itself.

## A.1.4 Company D

## I. Propulsion

Liquid Cooling

No major advantages. While it would improve TBO, only a 1% improvement in cooling drag reduction could be attained.

Rotary Combustion Engine

Development of this engine should be pursued.

At present, the engine suffers from high SFC, high weight, and is not as smooth as expected. The following potentials could be realized if the rotary were pursued:

Weight savings. Fuel consumption comparable to current recips. Good turbocharging candidate.

(Company D)

**Diesel** 

Excellent fuel efficiency.

A 2-cycle supercharged diesel is known to be receiving attention.

GATZ Turbine Engine

Definitely worth pursuing.

The 373 kw (500 hp) size should be emphasized. A flat rating would make it compatible with present high performance single engine applications.

GATE resulted in little new technology identification.

Do not pursue very small turbines on the order of 75 kw (100 hp).

Cost is expected to be high, on the order of 5:1 vs reciprocating engines.

## II. Configurations

Configuration research is a waste of time. It is mature as a result of 75 years of evolution.

## IJI. Airfoils

GA (W)

Mixed views were expressed for this class of airfoils. The section is sensitive to contours, leakage, and interference, and is heavily aft-loaded.

OSU Airfoil Service

Never really supported the service.

Airfoils developed by Hicks at NASA Ames are good.

## IV. Structures

Composites

Very little composites are used by this company. Further research appears dictated before this technology gains acceptance in general aviation.

Companies D & E)

Bonding

Bonding is used on stringers, doors, cowls, and other secondary structures. It is not used on primary structures.

Advantages: smooth contours, better appearance, stiffer, and saves on labor.

Disadvantages: Costly to tool up for, and has field repair problems.

Few problems with delamination have been experienced.

## V. Controls

L,

l

Fly-by-wire and Fiber Optics

Not seen as a possibility in general aviation in the foreseeable future.

Weight savings would not offset the cost penalty for general aviation.

Not enthusiastic towards active control systems, SSSA, etc., at all.

However, a gust alleviation system to improve ride quality might be attractive to general aviation.

## VI. Avionics and Electronics

Integrated Avionics and Microprocessors

Expect cost disadvantages due to redundancy requirements.

Advanced avionics might be attractive to the market. On the other hand, today's avionics may already be too advanced.

# A.1.5 Company E

I. Structures

Metal Bonding

Prime fastening method to replace riveting.

Major advantages: reduction in man-hours and reduces surface roughness.

(Company E)

Major problems: lack of understanding of life cycle. Implementation by a manufacturer requires a large capital outlay. Environmental effects, especially humidity, are of major concern

Inspectability, repairability, and maintainability are also questionable.

No bonding is used by this company.

Better adhesives which are less pressure sensitive are needed, with attendant research of properties.

Bonding distributes loads more effectively and has better fatigue properties.

### Composites

ŀ

. .

Ŋ,

Use will possibly replace metal bonding, but is expected to occur later.

General aviation will see primary structures composed of composite materials in about 20 years.

Concern was expressed about carbon in a crash environment.

This company had a fiberglass door on one of their aircraft but had to replace it with a metal one because the original was not stiff enough.

Company B utilized a honeycomb structure in the belly of one of their aircraft for crashworthiness energy absorption.

## II. Propulsion

#### GATE

Company F is following the developments of this program.

Appears that fuel problems facing the country will have major impact on this program.

Expects to see smaller turbines in general aviation by the late 1980's.

Diesel Engines

Have much promise because of excellent specific fuel consumption characteristics.

(Company E)

Diesels might be more attractive and beneficial to general aviation than turbine engines.

Liquid Cooling

.)

١

Absolute necessity for higher power-density engines. 400 hp is the limit for air cooled engines. Uneven cooling is a serious problem with air cooled systems during rapid descents.

Should significantly increase TBO by reducing cyclic thermal loads and allowing the manufacture of a "tighter" engine.

Should significantly reduce cooling drag and simplify engine/airframe integration.

Development of this technology will probably require government funding.

Fuel Controls

Considered to be a very near term objective.

Examined an auto-mixture control for use but found the controls unreliable and inconsistent as shown below.



Rotary Combustion Engine

Unsure of real benefits in this area.

Stratified charge, however, has real potential.

i

Propellers

Has experienced problems with vibration and blade flutter on some propellers which required a change.

Turbocharging

Can anticipate installation problems on smaller engines. Not much of a selection available. They are not optimized. Current systems are structurally sound, and have displayed excellent reliability.

Turboprops

Does not see demand in general aviation airplanes under 6 place, pressurized, single engine versions.

Studies of weight savings of turbeprops is misleading because of the requirement for additional equipment such as radios, pressurization, etc. for the aircraft to operate in the ideal and efficient environment for the turboprop.

Engines currently used by this company are 4 and 6 cylinder models, with TBO's ranging up to 2000 hours. Turbocharging reduces TBO down to the 1000 to 1200 hour range.

## III. Aerodynamics

Wing Loadings

Expects higher W/S to appear in general aviation. A major problem with high W/S is the high power required to cope with the associated high  $C_t$ 's.

Airfoils

Aft loaded airfoils cause interference problems.

Likes the OSU center and is maintaining contact.

Not highly interested in airfoil refinements.

Computational Aerodynamics

Methods should be used to design aft fuselage sections and examine cooling drag.

## Stall/Spin

If aircraft are made stall-proof, they will not spin. Question of how to stall-proof an airplane is significant. Incorporating a stick pusher, or limiting elevator power are effective but limit part of the performance envelope.

## Canards

Believes the following advantages exist: reduced wetted area, possible gust alleviation, lower fuselage weight.

Company I has realized gains from reduction of cooling drag and relating engines to 75% power.

## IV. Flight Controls and Avionics

Very interested in microprocessor integration into aircraft systems.

Flight Controls

Impressed with the SSSA approach.

Gust alleviation is a good candidate for advanced active control.

Fly-by-wire and light offer substantial weight savings. Multiplexing with fiber optics would reduce wire bundle weight significantly.

Active controls will be needed in the future.

Avionics and Displays

In 1971-72, this company performed a study of a HUD with a side stick force controller. The study was constrained by cost. NASA could further investigate these areas.

The Omega navigation system is good but does not provide terminal guidance. It needs government support for a fully integrated system to be developed.

Single-pilot IFR workload is much too high. What is needed is a visual, automated communication mode rather than present verbal methods. This would improve the workload and accuracy of information transfer.

ŝ.

V. Crashworthiness

Definitely needs more work. Seats are not designed for energy absorption.

Ag planes are good crashworthiness examples. Roll cage, good seat belts, and shoulder harnesses.

More realistic design specifications are needed.

VI. Noise

More attention should be and will be paid to noise in the future design of aircraft. Low level pressurization offers some potential for interior noise reduction.

## A.1.6 Company F

General philosophies for building aircraft:

"Keep it simple. Eliminate systems. Make it easy to fly."

I. Structures

Composites should be pursued.

II. Configuration and Aerodynamics

Canard

Allows lower gross weight and less wetted area. The wetted area of a canard configuration can be 2/3 that of a comparable conventional aircraft.

For a push-pull canard configuration, the fuselage has two inflection points in bending moment. This allows for  $\epsilon$  much lighter fuselage.

Wing Sweep with Winglets

Allows winglets to be used for directional stability.

Get  $C_{l_{\beta}}$  without dihedral.

Sells airplanes (looks good).

Wing Strakes

Provides volume for fuel and interior cabin space for elbows.
Aerodynamics of Canard Configurations

The wing tip is in upwash and results in localized low to negative induced drag. With the tips loaded up, the winglets work well.

Care must be exercised in selecting the proper airfoil section for a highly loaded canard. W/S of the canard can be twice that of the main wing, and the canard and wing can carry a 50/50 load distribution.



α

When the trim requirements for a canard with the lift characteristics of the previous figure are examined, it can be seen that the main wing cannot be stalled.



(Companies F & G)

Efficiency Factor

Effective e > 1 when total area is counted in  $A = b^2/S$ .

 $b = b_w$  and  $S = S_{canard} + S_{wing}$ 

III. Engine Cooling

Updraft Cooling

Produces much more uniform cooling of the cylinders, which could increase TBO significantly. Possible problem with oil spray from the engine degrading cabin forward view.

The cooling air is only slightly affected when passing over the exhaust manifold first.



### IV. Avionics, Systems, and Displays

Very interested in HUD's, improved displays, warning annunciators. Much improvement is needed here.

Would like to see an angle-of-attack indicator.

Avionics need improvement in price and reliability.

Fuel indicators are terrible. Should be non-linear and very accurate for the last 30 to 60 minutes of available fuel.

- A.1.7 Company G
- I. Propulsion

South and the second second

GATE Studies

Particularly interested in a 280 kw (275 hp) unit and another on the order of 746 kw (1000 hp).

Endorses GATE with reservations. Recommends that the range of sizes of GATE be well defined.

Aircraft Engines

Believes that the low bypass turbofan still offers more efficient thrust at high operating altitudes.

A variable bypass engine is needed with a capability for bypass ratios of .2 to .3 at 41,000 ft.

Noise is a big problem with low bypass engines but variable bypass would help.

If the GATE studies are accurate, a successful single engine turboprop will materialize.

#### II. Aerodynamics and Stability and Control

Wing Development

A design by this company incorporated a supercritical high aspect ratio wing (8). The new wing design proved to be too expensive for production so an existing wing was modified. It was not clear if production of an entirely new wing was cost prohibitive, or whether production problems associated with the supercritical wing were prohibitive.

The wing modification for the design consists of the following: the aft 60% of the existing wing was retained. The front 40% was replaced with a computer generated section. Each wing tip was extended 3 feet, and winglets were added. These modifications resulted in a 17% increase in L/D at cruise.

When used, winglets accounted for a 3% to 4% reduction in drag, and modifications of an existing wing resulted in a 230 lb weight penalty.

Light Aircraft

A modified 64-A215 had leading edge stall problems. Hence, a highly cambered (drooped) leading edge was incorporated which corrected the problem by maintaining flow attachment at the leading edge.

The Whitcomb airfoils could have problems with spins because of high leading edge suction. The high suction on the drooped airfoil on one plane aggravates spin characteristics.

?

5

Expressed interest in an airfoil section developed at the University of Illinois. It is reported to produce leading edge stall without the normally associated effects, and could have possibilities with spin-proofing or control during post-stall maneuvers.

More work needs to be done with airfoil design for general aviation. Forward camber is more attractive than aft camber.

T-tails are good for resolving spin problems.

High Speed Aircraft Control

Spoilers are used for primary lateral control.

Although direct lift control is not used, it has excellent technical merit.

#### III. Structures

Bonding

No bonding is used at highly loaded joints.

Feels that the ideal extent of bonding should be about 1/2 of what is currently in use on a particular light airplane.

Delamination at the trailing edges of wings, etc., and its causes (e.g. moisture) must be examined.

Since large portions of an aircraft are bonded at once, a parts shortage can halt the bonding process.

Multiple stage bonding can weaken old joints.

Quality control is a major problem, and strict environmental conditions must be maintained at all times.

Bonding has allowed the construction of a very simple airframe which also costs less.

Composites

٢.

Certification is a major problem associated with composites. NASA needs to work on certification criteria that are suited to composites, particularly where primary structures are involved.

(Companies G & H)

s

#### Composites have primary advantages of

- (1) Weight reduction
- (2) Cost reduction
- (3) Ability to accommodate complex shapes

Composites in propellers will yield better uniformity.

#### Honeycomb

Used extensively in the cabin area of the fuselage on one light aircraft.

Only used in floor and radome of a particular high speed aircraft.

#### IV. Systems

Flight Controls

Digital flight controls look good but are not being pursued at the moment.

Avionics

Panel mounted systems are more attractive than remote ones in terms of lighter weight, less complexity and lower cost.

#### V. Research By NASA

Recommends methods be developed for determining the effects of humidity on flight test results. No methods currently exist, nor is it accounted for in certification.

### A.1.8 Company H

#### I. Aerodynamics and Configuration

**T-Tails** 

T tails were designed for market appeal, and any performance gained is an "extra".

T tails can provide for less trim change while improving power-on stability.

GA(W)-1

Designs can suffer from wing-body interference. However, they produce high  $C_{I}$ 's and gentle stall characteristics.

5

Canards

Not convinced about the benefits of a canard.

Cooling Drag

Twin engine aircraft with a third engine mounted in the nose of the aircraft can be (have been) used for cooling drag determination.

II. Bonding

Some bonding is utilized in cowls. Further bonding is being incorporated slowly.

III. NASA Involvement With General Aviation

NASA should be very selective in their entrance into general aviation and should confine their work to basic research as opposed to providing industrial guidance.

- A.1.9 Company I
- I. Propulsion

1

Air Cooled Engines

There are no near term alternatives to air-cooled engines.

Variable timing and electronic ignition will be utilized when required and when their flexibility and reliability are proven.

The automotive industry is not the leader of technology, but the leader of marketing.

An automatic system is needed to properly maintain correct mixture settings for best efficiency in cruise.

Operating an engine lean of peak results in cooler operation. (see sketch on next page)

Cooling design methods exist, but cooling requirements for general aviation engines are not specified in enough detail. Major problems with cooling drag lie in (1) inlets which are too large, (2) exhaust vents which are not well analyzed, and (3) lack of an efficient diffuser (plenum).



Engine Operating Characteristics

Definition of cooling drag: the momentum loss of air required to cool the engine.

Liquid Cooling

Could be done without a weight penalty over air cooled systems.

It is more efficient from a cooling drag standpoint.

The high reliability required results in increased cost.

Turbines

Use only for justified purposes.

The turbocharged reciprocating engine offers some advantages over the turboprop. As an example, see the figure at the top of the next page.

A larger turboprop must be used to retain hot day performance and to offset a relatively high lapse rate.



#### Altitude

Reciprocating Engine vs Turboprop Performance.

### II. Aerodynamics

High Wing Loadings

Face the following problems

- Public reluctance to accept the handling qualities associated with higher cruise speeds.
- (2) Landing and takeoff distances are increased.
- (3) High wing loading generally requires high power loading.

High W/S requires effective flap systems, and possibly even full span flaps with spoiler roll control.

The general aviation data base for full span Fowlers with spoiler roll control is small.

Some effective wing area is regained for takeoff with moderately deflected Fowlers, but is accompanied by a (small) drag increase.

Natural Laminar Flow

Studies need to be done on effects of contour shape tolerances and surface finishing.

Eppler's work on airfoils is encouraged and supported.

### III. Safety and Crashworthiness

Deicing and Anti-icing

More work should be done in the search for an icephobic surface coating.

Weather

Studies of severe weather structural response and damage limiting methods should be pursued.

More research regarding lightning strikes is needed.

Crashworthiness

Definitely needs more work. NASA should continue its present research in this area.

Specific areas which merit study: fuel containment, engine compartment fire containment, effect of composites, heat tolerant metals.

IV. Noise

Possible methods for reduction include

- (1) Vibration damping.
- (2) Quiet, yet power-effective mufflers.
- (3) Quieter propellers.
- (4) Sound damping in the cabin.
- (5) Variable transmission and/or gearing systems for the engine/propeller interface. Could be attractive for certain applications by holding propeller rpm constant while allowing engine rpm to vary.

### V. Operations

ATC

Would like to see NASA involved more in the conceptual stages of ATC planning.

New basic concepts which allow for the optimized use of inertial nav and RNAV are needed. These would help fuel efficiency quite a bit.

Flight Manuals

More standardization of flight manuals for aircraft is needed. GAMA is working on this.

٠

## VI. Flight Test

### Flight Test Pilots

The technical capabilities of test pilots needs to be improved through better training.

Management needs to be more aware of the functions, qualifications, expertise, and capabilities of test pilots.

### Instrumentation

Good thrust and torque meters are needed to determine performance accurately.

More development is needed on the vortex generating airspeed sensor. Airspeed measurement without disturbing the flow field requires development.

A humidity detection device is needed to improve the accuracy of engine operating parameters.

#### VII. General Comments

FAA regulatory procedures are overly restrictive. Also, the regions are autonomous, and certification requirements based on the interpretation of a regulation by one region may differ substantially from those of another region.

NASA may want to hold a seminar to educate management on the availability of high technologies.

NASA may also want to educate the younger engineering generation on older (pre-1954) documents.

Management and marketing decisions can conflict with technology and result in bad airplanes.

Most general aviation airplanes aren't designed. Rather, they are just built.

Product liability is the worst problem confronting the industry.

Regarding airfoils, NASA should publish data for realistic flight conditions in addition to those for "smooth airfoils", etc.

Rewrite the Pratt and Whitney reciprocating engine manuals (circa 1940-1945) for application to general aviation. They are good.

(Companies I, J, & K)

Anything that makes aero design more of a science and less of an art will be an important area of research.

### A. 1. 10 Company J

### I. Operational Environment

The critical area for advancement is not the vahicle, but the operational environment. i.e., the pilot's interface with the systems must be improved.

Pilots need clearances, weather, and traffic advisories, which are currently obtained verbally. These should ideally be communicated visually.

Solutions

Develop data links from the ground to the aircraft.

If these problems were solved using the data link and CRT display, it would eliminate unnecessary voice communications, radar sets, and possible collision avoidance equipment. The system would reduce work load and be more efficient.

### II. Composites

Cost and certification are the most serious problems.

#### III. Propulsion

Available turbochargers are not optimized for aircraft use. General aviation is constrained to off-the-shelf equipment.

IV. Liability

Plagued by nuisance suits. Even if the company is innocent, it still costs money. This is a big factor for a small company.

10% of the vehicle cost is liability insurance.

# A.1.11 Company K

Company K developed a single-engine, very high performance aircraft employing a composite structure.

#### I. Structures

Composite construction of the aircraft is not advanced technology. The material used (fiberglass) has been available since the 1950's.

(Companies K & L)

Composite structure allows a significant reduction in parts count. This should cut costs.

The fuselage is constructed in halves. The left half is outfitted with all bulkheads, engine mount, and internal hardware. After this is done, the right half is bonded to it. This method should reduce labor costs significantly since construction time is reduced. It allows the worker to outfit the aircraft easily without having to crawl through the fuselage shell as is required with conventional aircraft manufacturing processes.

The aircraft is not particularly lighter than existing aircraft.

The major construction material is a fiberglass honeycomb sandwich.

II. Aerodynamics

No advanced technology. All airfoil sections are NACA sections.

Wing loading is relatively low and on the order of 20 to 30 psf.

This airplane is aerodynamically clean. Lack of joint lines and protrusions, together with a "slick" surface finish which the composite construction allows, creates a very low-drag airframe.

With 298 (400 hp) and 6 passengers, the aircraft is capable of 261 kt cruise at 6096 m (20,000 ft).

#### A.2 PROPULSION MANUFACTURERS

A.2.1 Company L

#### I. Advanced Airfoil Sections

Currently working on an airfoil section with Ohio State to develop a thinner section that produces greater maximum lift to reduce weight.

Want to maintain current cruise efficiency  $(n_p = .9)$  while improving low-speed performance. The goal is an efficiency of .7 to .75 in climb.

Doesn't believe that cruise efficiency can be improved. Thus they are concentrating on the low-speed regime.

5

# II. Blade Planform/Configuration

Blade number is increasing in twins. Single angine aircraft will remain with 2 or 3 bladed propellers because of weight and engine characteristics.

The Q-tip reduces noise while generally maintaining rpm. Performance is not improved.

Proplets are not winglets. Rather, they are Q-tips bent in the opposite direction.

# III. <u>Materials</u>

Composite blades are made with Kevlar.

The composite blade was designed to twice the strength of a comparable aluminum blade. This conservatism in strength was a safeguard for certification. After certification, plans for reducing strength and weight to more realistic values may be undertaken.

The composite blade costs 2 1/2 times as much as an aluminum blade but weighs half as much.

Costs of the composite blade appear more favorable each day. The cost of Kevlar is declining and the price of aluminum is increasing. In addition, when strength is reduced to realistic levels, the removal of Kevlar will reduce costs even more.

For the composite propeller configuration, the propeller assembly weight is divided equally between hub and blade weights.

## IV. GAP Program

Five different blade designs are undergoing testing for acoustics. These propellers are all 1/2 scale, and are currently being wind tunnel tested. A full scale test is expected to follow.

# V. Operations

For reciprocating engine propellers, one must forego cruise airfoils to allow for Vibration. Example: Use Clark Y instead of a 16 series section.

For turbine aircraft, vibration is not as great a problem, and the most efficient airfoil can be used.

(Companies L & M)

Blade erosion has proven to be no more of a problem for the composite blades than for aluminum blades.

Five bladed propellers are performing well. When more than three blades are used, however, vibration considerations impact more heavily on the design.

#### VI. NASA Research

Need studies on propeller characteristics when used for reverse thrust.

### A.2.2 Company M

I. GAP Studies

Purpose

Identify technologies that will reduce fuel costs. Identify technologies that improve noise characteristics.

Technology Elements include weight, noise, cost, life, airframe integration, and emphasize "clean sheet" airplane design.

#### II. Technologies Applicable to Propellers

Advanced Airfoils

Looking at ARAD characteristics over their entire performance range.

Utilizing computer-aided design of airfoils.

Comparing L/D,  $C_L$ , etc. of Clark Y, GA(W), current in-house designs, and ARAD.

Current cruise efficiency stands at  $\eta \approx 85\%$  to 87%. Advanced airfoils offer 1% to 2% efficiency increase, and also help relieve compressibility losses.

Improved Propeller-Nacelle Integration

Looking primarily at a performance improvement.

Engine shape is a significant parameter.

Using potential flow analysis to determine what the nacelle effects are on the propeller flow field.

Possibly realize a 3% to 4% increase in efficiency.

### Design Optimization

Blade sweep. Also, reduce t/c through the use of composites.

Composite Materials

Screening materials. Looking for optimized materials and the manufacturing techniques required in a production environment.

Composites offer the greatest value in fatigue life and low weight but have large penalties in cost, both in acquisition and in processing.

Effects of tip speed: Diameter and rpm tradeoffs.

Effect of number of blades

Effect of blade loading on noise

Current blade loadings are designed for performance and are not optimized for noise.

III. <u>Areas Where Greatest Gains Can Be Made in Propeller</u> Technology

Advanced Airfoils

Propeller/Nacelle Integration

Compressibility Studies

Current losses are on the order of 1% to 5%.

Sweep of  $45^{\circ}$  at the tips can gain half of the losses back.

A combination of tip sweep and advanced airfoils should gain most of the losses back.

Believes 90% installed efficiency can be realized compared to the current levels of 85% to 87%.

Analysis of propellers done with a 3-D strip computer code, based on a 2-D method by North Carolina State and Lockheed.

(Companies M & N)

Ś

#### IV. NASA Research and Funding

Composites

Investigate materials and processes.

Establish time criteria for fatigue testing, etc.

Develop and substantiate data.

### A.2.3 Company N

I. Propulsion

Q-Fan

Work was halted because of lack of industry interest.

Prop-fan

Designed for higher Mach application: .55 and above.

Features

Area-ruled spinner. Swept blade shape. Integrated nacelle shape. High power loading.

**Objectives** 

5% to 10% savings in DOC. 20% reduction in fuel consumption at M = .8.

Sweeping blades appropriately can cancel noise at the source. Works well for the prop-fan, but the effects on conventional propellers is unknown.

The prop-fan doesn't really represent difficult structural problems. Metal spar, foam leading edge and trailing edge, composite shell.

It has better performance on takeoff than a turbofan.

The integral spar with shell is designed primarily for safety. Damage to a blade will not result in failure of the complete propeller.

#### II. Environmental Control Systems

#### Goals

Reduce power requirements. Reduce Life Cycle Costs (LCC) for entire aircraft system.

A variable air-cycle system was produced for a fighter aircraft, with a broad range of applicability in mind.

The system saved 817 kg on a 28150 kg aircraft, and also saved .37 cubic meter of internal space, or 40% of the environmental control system compartment.

Reliability of the variable cycle system is just about as good as a conventional system even though it has more moving parts.

Recirculation can save 30% of a 373 kw power requirement.

Air cycle is lighter than vapor cycle, but requires more power.

### III. Air Bearings

Offer great promise for high speed rotary machinery.

Have more load capability at high speeds, but tend to fail at low speeds (just the opposite of ball bearings).

Have the potential for greater reliability. No maintenance.

Systems are heavy and cost more than ball bearings.

#### IV. Micro-Electronics

Engine Control Systems

Total electronic controls for engines utilizing no mechanical parts or backup systems for operation.

Engine control systems will have to control more functions and more accurately.

For small engines, cost is the important factor. This company produces the engine controls for a current small turbofan at approximately \$3 to \$4 thousand per unit.

Controls synchronize, synchrophase, and control fuel flow of the engines. Results in a 7% to 15% improvement in sfc. Fly-by-wire and Fiber Optic Systems

For flight controls and engine controls.

Distributed microprocessors seem more likely than a single computer.

When developing fly-by-light and digital controls, the following points must first be considered.

- How much redundancy is required for the mission, and what methods for redundancy will work best for the situation.
- (2) Policing the system is difficult, because failure determination requires much complexity.

Advanced Data Diagnostic Systems

Airborne integrated data system. Monitors health of aircraft systems and engines.

Will also be applied to flight profile monitoring.

General aviation could use at least an engine monitoring system.

At first, the system will probably be ground based. As more electronic systems (microprocessors) appear on aircraft, onboard monitoring systems will emerge.

#### A.2.4 Company O

I. General Aviation Engine Market

The total piston engine market in general aviation is \$250 to \$300 million per year.

\$25 to \$30 million would be required to start from scratch with a new engine and take it to production.

Technology must be cost effective for the manufacturer.

Liability

Is a problem within the industry. A company may pay over \$3 million per year for liability insurance.

General aviation engines today are very sophisticated and not World War II products.

### II. New Technologies

Improved Fuel Injection

The fuel injection system currently used is simple, low cost, and reliable, but does not compensate for atmospheric density changes.

Fuel injection could be improved to provide for donsity compensation so that fuel consumption could be reduced 3% to 4% during takeoffs, landings, and climbs.

Variable Timing

Offers the potential of a  $3 \frac{1}{2}$  increase in fuel economy.

Metallurgy

Methods need to be developed for lower cost production of titanium. It is an abundant element.

If engines were designed with titanium, a 30% weight reduction over current engines could be realized.

If the goal of 1.6 kw/kg (1 hp/1b) is to be realized, titanium must be utilized.

III. Ideas On Advanced General Aviation Engines

Diesel Engine: No such thing as an "uncooled engine".

Rotary Combustion Engine: Will never be feasible because of the following reasons:

(1) low volume sales.

(2) complex tool-up and machining.

**GATE Engines:** 

The following is a comparison between a reciprocating ngine (turbocharged) and a GATE turboprop.

(Companies O & P)

Engine	Power, kw (HP)	Altitude
Company 0	242-231 (325-310)	Sea Level
Company 0	186 (250)	7620 m (25,000 ft)
GATE Turboprop	269 (360)	7620 m (25,000 ft)
GATE Turboprop	500 (670)	Sea level

Turbocharged Reciprocating Engine vs GATE Turboprop

By this comparison, GATE doesn't look that good.

Derating (flat rating) will already affect the turboprop. Utilizing it at an off design point degrades the fuel specifics.

The hot day performance of a turboprop is not as good as a turbocharged reciprocating engine.

The GATE engine costs are unrealistic. The projected market is totally unrealistic. For example, a currently produced comparable turboprop engine costs roughly \$50,000, compared to an existing reciprocating engine which is \$25,000. Both have similar sea level horsepower. The costs of the turboprop have been amortized for years, and it still costs a lot. How will a small turbine, utilizing advanced technology, cost anywhere near a current reciprocating engine?

The "real world" factors of marketing and aesthetics need to be accounted for in design.

A.2.5 Company P

I. Marketing

Technology of no use unless

It satisfies a need.

Developer can afford it.

It's profitable to the user.

It's introduced at the right time.

### Turboprop Mature Market Priorities

Price - Very sensitive at equal hp.

HP - Wants more horsepower at same \$/hp.

Reliability and product support.

SFC (but not so important, since other factors contribute to DOC.)

TBO and other maintenance costs.

Operating characteristics.

Weight.

Turboprop New Market Priorities

SFC - Emotional as well as economical issue.

Price.

Reliability and product support.

TBO and other maintenance costs.

Operating characteristics.

Weight.

Engine Targets : 1490 kw (2000 hp) -- .262 kg/kw-hr sfc 522 kw (700 hp) -- .27 kg/kw-hr sfc 224 kw (300 hp) -- .32 kg/kw-hr sfc

Turboshaft Priorities

SFC @ 60% power (installed), @ low altitude.

Weight (installed).

Price.

Reliability and product support.

TBO and other maintenance costs.

Operating characteristics.

Brand name.

Price importance is inversely proportional to gross weight.

Turbine Engine Market

Customers	<u>1978</u>	<u>1985</u>
Corporate	58 <b>X</b>	65%
Utility, commuter,		
& paramilitary	16%	27%
Military	16%	47
Civil Helicopters	10%	47

Sales: Forecasting a 5.8% growth rate for turbine aircraft over the next ten years.

**Turbofan Priorities** 

Thrust: Speed improvement.

SFC: Range improvement.

Price.

II. Advanced Design

**GATE Studies** 

Goal was to use high technology but obtain lower costs. This goal is contradictory.

Marketing Considerations

Not favorable for small turboprop -- high volume but low dollar sales.

Market penetration of a 224 kw (300 hp) engine at 5000 units/yr for \$20 thousand = \$100 million/yr. This is small for Company P considering an anticipated \$50 million development cost. For \$100 million, can develop a 1491 kw (2000 hp) turboprop, and make more on less volume of sales. The development costs are not much different for  $\omega$  small or a large engine.

The only way a small turboprop will be developed is through NASA sponsorship of research and development.

### III. Aerodynamic Research

Concerned with compressors, combustors, and turbines.

#### Compressors

Centrifugal Compressors

Pressure ratios of 7 to 11.

Current engine with centrifugal compressor gained four points in  $\eta$  and 7% in sfc.

Always looking for better shapes.

Does extensive computational aerodynamics work in the design and analysis of compressors.

Axial Compressors

Maximum pressure ratio for 2 stages is approximately 2.

Looking at a rotating hub with stators cantilevered from outer ring. This reduces the stall margin.

#### IV. Mechanical Research

Looking at: Structures. Cooled turbine. Fan blade foreign object damage. Reduction gearbox. Rotor fragment containment. High Dn (diameter x speed) roller bearing Technology. Journal bearing technology. Supercritical rotor dynamics technology.

Structures Technology

In-house finite element programs supposedly better than NASTRAN. Finite element programs used for:

Transient dynamic analysis. Contact analysis. Mode shapes.

Also, experimentally determining mode shapes via a holographic analysis is attractive. (Company P) Finite element program for turbine blades has 12,000 degrees-of-freedom.

Also can analyze effects of flange leakage and foreign object ingestion.

Cooling Technology

Used 12,000 degree-of-freedom model for one blade.

TIT 1170°C (2140° F) but blade temperature is 977° C (1790° F).



Cool turbine blade with integral air channels as shown above.

Gearboxes

Engine rpm's as high as 30,000.

Vibration is a big problem in gearboxes.

Matched gearing and indexing can reduce vibration by 50%.

Increasing the thickness of the shaft usually helps reduce overall flexibility.

IV. Prop-Fans and Variable Pitch Fans

trop-fan offers efficiencies of .8 at Mach = .8.

Also attractive for short to medium range aircraft.

Directly challenges turbofan.

Variable pitch fan (VPF) challenges medium-speed turboprops.

Recommendations



Range

Prop-Fan Savings vs Turbofan.

Investigate variable pitch fan application to commuter aircraft.





### VI. High Efficiency Propellers

#### Propeller Research

Blade design, afterbody effects, airfoils, nacelles, and inlet effects, control, hub design, materials, all require study.

Need propeller developments to match engine developments.

Tradeoffs

As velocity increases, propulsive efficiencies become more significant.

With conventional airfoils, a large diameter is required for high efficiency. For noise reduction, need to reduce diameter and/or rpm.

Commuters need high thrust/power for takeoff, possibly attained by compromising cruise performance.

Areas under review

More efficient airfoils.

Noise (FAR 36 near and far field).

Optimization.

Integration of spinner and nacelle.

Increasing number of blades

#### Advantages

#### Disadvantages

reduce diameter complex hub shorter undercarriage higher cost reduce engine-out problems higher weight (thrust-line closer to c.g.) increase high speed efficiency looks better lowers noise engine speed goes up but torque goes down

Anticipated work needed for

High speeds - increase propfan efficiencies. Low speeds - increase turboprop efficiencies.

Noise

```
Free-turbing propeller has cruise noise advantages (variable propeller speeds).
```

Research

Direct towards maximum fuel economy.

Apply (integrate) new technologies in areas of:

Airfoil shapes, Materials and construction, and Control concepts.

### VII. Fuels

A main thrust is that engines be able to use a wide range of fuels. Specifically applicable for turbines and diesels.

Tar sands may be important as a fuel source.

### VIII. Materials

Priorities

Safety, inspection, and simplicity.

Low first cost.

Weight.

Effects on fuel economy.

Durability/cost tradeoffs in defining desired TBO's.

#### Influencing Factors

Increased energy cost affects materials cost.

Increased fuel cost requires increased efficiency.

Supply of critical materials such as chromium, cobalt, and tautalum could be a problem.

Chromium - 95% comes from Rhodesia and South Africa. Used for corrosion resistance and strength.

Cobalt - replaceable at a cost.

Tantalum - used for oxidation resistance in turbine blades.

# New and/or Alternate Materials

Ceramics - Reliability.

Substitutes - Nickel for cobalt. Aluminum for chromium or Molybdenum for chromium. Columbium for tantalum. No substitute for titanium in impellers.

Reduce Input Materials for Lower Costs.

Powdered metal is expensive, and costs twice that of bar stock.

Castings - need better non-destructive testing even when using hot isostatic pressing.

Mixed processes - diffusion bonding or friction welding of castings and forgings.

Note: Forgings have a high rejection rate: Keep 15% and diseard 85%.

Improved Corrosion Resistance Methods

Rapid solidification rate (RSR) - single crystals.

Better coatings.

Areas for Further Research

Improve non-destructive testing.

Improve contidence in

Powdered metal technology.

Casting methods.

Extension of engine life.

### A.2.6 Company Q

l

### I. Significant points made during preliminary briefing

Reciprocating engine performance and operations are often adversely affected by enroute as well as destination weather.

Turbine engines are lighter, more reliable, faster, and less susceptible to enroute weather than their reciprocating counterparts. Also, they are not subject to cooling drag.

The disadvantages of a turbine include cost, and higher specific fuel consumption. The latter is offset somewhat by the elimination of cooling drag.

If turbines are to get any cheaper, broad usage must be found.

Turbines are still not the answer to all segments of general aviation: best over 150-186 kw (200-250 hp).

General aviation accident rates are much higher for reciprocating powered aircraft than for turbines.

### II. Comments on General Aviation

Businessmen like quiet working atmosphere in aircraft (more productive use of time).

Corporate passengers like long range, comfortable, over-the-weather flying capability.

General aviation's major drawback is cost. New technology must be incorporated without large price increases.

#### III. GATE Project

Design Philosophy

Traditional - reduce parts count, and make the remaining parts more efficient.

Cheaper production, commonality.

Cost Factors - high expense

Rotation speed - affects life, shaft dynamics and bearing suspensions

High tip speeds create high stresses in compressor and turbine blades.

Precision: surface finish is critical.

Low volume production, tooling costs, and required research and development (R & D).

Cost Solution

The solution is lower rotational speed.

Multi-stage axial compressor (5-6 stages).

Again, the low speed of the turbine reduces stresses (50% in the turbine blades).

Shaft fuel injection, with centrifugal fuel nozzels (low-pressure system).

The net result is a much lighter weight turboprop, compared to a reciprocating engine (a 224 kw reciprocating engine weights 272 kg). Smoother, quieter, no cooling drag, smaller nacelle, jet fuel, competitive BSFC, and longer TBO's, are all advantages over the reciprocating engine. Costs \$19,000 on a large-scale production basis.

Turbofan Concept

Turbofan utilizes a common hot core with the turboprop.

Add 3-stage low-pressure compressor and fan, and a 3-stage low-pressure turbine.

Bypass ratio = 5 to 5.5:1; thrust = 4.319 KN; weight = 84 kg; sfc = 40.5 kg/KN-hr at SL, 69.44 kg/KN-hr at 9140 m with 1.113 KN thrust.

Forecasting a TBO of 10,000 hours (aircraft life).

Priced at \$23,000 with several thousand/year production run.

GATE Realization

Minimum time required to get the GATE engine to a demonstrator level is 3-4 years (optimistic), 5-6 years for certification.

There are no plans for turboprop hardware at Company Q currently. The cost of development to demonstration would probably be under \$10 million.

However, if NASA doesn't fund additional GATE work, industry probably won't pursue it.

Small turbine development led to a low cost turbojet. The unit cost \$2000 for a 1000 unit production run in 1976. It was designed for 30 minutes to 1 hour of life at 35,000 rpm. The engine has been tested up to 18 hours, and has a TSPC of 132.6 compared to 117.3 kg/KN-hr of another similar engine. Life of the engine is limited by the use of grease-packed bearings.

60% of the cost of this turbojet was in the turbine.

Simple compressor concept is proven by this turbojet with only minimal compromises in aerodynamics. TSFC goes up, but not unreasonably. The reduction in performance results from lower pressure ratios.

Turbine design is flexible:

build to high stress - long life. build to low stress - lower initial cost.

GATE turboprop description

6-stage axial compressor, 1-stage centrifugal compressor, 4-stage turbine.

low gearing

1.02 m (40 in) long, weighs 72.6 kg (160 lb) without starter.

12.5 pressure ratio,  $1010^{\circ}$  C TIT, 280 kw (376 hp) and 0.356 KN (80 lb) thrust.

IV. Current Work

Military application

A broad production base for a current small turbofan might be realized from a likely contract.

Another small turbofan used in a different program is derated 10% to 15%.

Utilizes oil lubricated bearings and could benefit general aviation through spinoffs.

#### A.2.7 Company R

Developing rotary combustion engines for aircraft and other vehicles.

I. Rotary Configuration

Rotor system is simple and easily adaptable to different sizes.

Size (displacement) can be changed by

- (1) increasing width of rotor.
- (2) adding rotors.
- (3) increasing diameter of rotors.

Item 3 is the least attractive, because apex speed is a factor. Items 1 and 2 are easy.

4 rotors is an attractive maximum. Allows the engine case to be built in two parts, each with two rotors. In this manner, the crankshaft does not have to be split for the timing mechanism. This keeps cost down.

Wide rotors are not very desirable (current model is 3 inches wide) because of crankshaft bending under loads.

II. Seals

Unique apex and side seals are incorporated which resolved previous seal problems.

The seals are sintered ferrous metal, which is very compatible with the trochoidal surface coating. Compatibility is the key. Trochoidal coating is high cost and applied with a detonation gun.

Other less costly methods exist, including plasma spray and chrome.

In a test of seals, this company found

- (1) their trochoidal surface showed imperceptible wear.
- (2) their apex seal wear after 2000 hours was very low and forecast to last 5000 hours.

This test was, however, performed at a lower BMEP than expected for aircraft.

3

### III. General Comments on Rotary Combustion

(1) lower weight.

H

1

k

ł

1

- (2) lower cost.
- (3) multi-fuel capability not restricted to AVGAS.

3000 hour TBO for the aircraft rotary combustion engine is a viable goal.

Lower cost due to simpler system and easy adaptation to different sizes. Reciprocating engine manufacturing is cost-constrained (or cost-fixed) by lot sizes of production runs.

The rotary engine OEM cost will be better than for a similar technology level reciprocating engine. At the worst, it will be the same.

Now running a military contract for development of a rotary engine. In addition, they have contracted an airframe manufacturer to do sizing studies utilizing their aircraft rotary engine.

Engines are liquid cooled although air cooling has been demonstrated. The liquid cooled engines are lighter than equivalent reciprocating engines, even with coolant and radiators. Therefore, they offer advantages both in cooling drag and weight.

### IV. Charge Stratification

Main research effort.

Uses an unthrottled, direct fuel injection approach with much success.

Fuel injection is followed by secondary air ingestion.

Uses one injector for the pilot flame ( $\simeq$  10% of fuel) and a second injector for 90% of the fuel which controls power output.

Already demonstrated excellent emissions and fuel economy in a surface vehicle.

### V. Aircraft Engine

Was testing a 246 kw (330 hp), 127 kg (280 lb) engine, but not at the moment. It is expected to be run again.

#### (Companies R & S)

An earlier NASA report documents studies of an aircraft rotary engine which was

- (1) a gasoline engine.
- (2) not stratified-charged.
- (3) normally aspirated.

Present preliminary analysis indicates that the above engine, when modified to stratified charge capability, will meet emissions, have wide-cut fuel capability, and have a BSFC  $\simeq 0.24 \text{ kg/kw-hr}$  (.4 lb/hp/hr). (NASA goal is 0.23.)

VI. Suggestions for NASA (areas NASA could help in)

- (1) More mutual effort (or an atmosphere of) between NASA and the manufacturers.
- (2) Stratified charge combustion modeling.
- (3) Basic research on materials higher speeds and loads will require better wear rates.
- (4) Materials research of high temperature cast aluminum specific data on low cycle thermal fatigue.
- (5) Improved turbocharging.

### A.2.8 Company S

Background: Developing an advanced technology turboshaft engine for military helicopter applications.

One of two companies developing this engine.

I. Objectives

SFC = .55, at 358 kw (480 ESHP) (60%).

Relatively easy to change turbine and apply to a turbcprop vehicle for 75% power at cruise altitude.

Requirements - the engine component design should not contain any design features (inherent limitations) that would preclude the engine from obtaining an FAA certification.

Front drive characteristic specifically designed so that it could be adapted to turboprop application.

Present contract funds research through 500 hours of testing, but not through certification. Would like to see NASA fund the certification phase.

.

If certification is attempted, the engine could possibly be certified by 1984.

This is a high technology engine and, even if mass produced, could never get below \$56 to \$60/kw (\$75 to \$80/1b).

### II. Highlights

- Designed for SFC of .33 kg/kw-hr (.55 lb/hp/hr) power at sea level.
- (2) This 597 kw (800 ESHP) engine is to have a dry weight of 90.8 kg (200 lb). Includes a FOD device at inlet, which would not be required for civil use.
- (3) Must have provisions for customer bleed air (pressurization, etc.)
- (4) Must have 7.5 kw continuous drive pad for accessory pack.
- (5) Operation for 5 mins without oil at 75% power.
- (6) Modular assembly/disassembly.
- III. Noise

Comment: anything on a flight schedule (commuter) must meet already mandated emissions standards and about-to-bemandated noise standards.

### A.2.9 Company T

- I. Advanced Materials and Processes
  - Directionally solidified turbine blade. Exothermically cooled. Produces a 2.42% sfc improvement.
  - (2) Abradable seals. produces a 10:1 wear ratio with turbines, compressors, or any component that requires close tolerances.
  - (3) Single crystal turbine blades. Eliminates grain boundary strengthening alloys. Provides stronger blades. Allows higher blade and TIT temperatures (1092° C [2000° F] for blades), and no surface coatings or blade cooling is required.

### II. Turbine Engines

Compared to large turbine engines, small turbines have less efficient components. Manufacturers cannot afford the

complexity of weals, clearances, and components required to get the same efficiencies, temperatures, and pressure ratios as are found in larger turbine engines.

Turboprop business aircraft requirements:

- (1) increased altitude and horsepower.
- (2) increased component efficiencies and operating temperatures.
- (3) propeller noise reduction.
- (4) high horsepower propellers.
- (5) STAT studies (see Appendix A.5.1)

### III GATE Program

As engine horsepower is decreased, the turbine engine is less competitive with respect to the reciprocating engine. One company representative, however, believes that the light, fixed wing aircraft market is an ultimate customer for the turbine.

Application of the small turbine is best suited to the lightto-medium weight category of pressurized, twin engine aircraft.

Features:

;

- (1) laser hardened gears.
- (2) single stage compressor.
- (3) free turbine (see figure below). A single shaft is 2% to 3% less expensive than a twin shaft, free turbine configuration, but the twin shaft has a larger market with helicopter applications.



Twin Shaft, Free Turbine Configuration.
#### (Company T)

- (4) a turbine inlet temperature (TIT) of 1203° C.
- (5) an aircooled laminated centrifugal turbine. This technology allows the incorporation of very complex cooling air paths within the turbine, unattainable by other fabrication techniques.

Results of GATE study were:

- (1) overall efficiency  $(\Delta \eta)$  increased 9.8%.
- (2) SFC was reduced 7.4%.
- (3) unit cost was reduced 217. In a production run of 7,000 to 10,000 units per year, the projected cost of each engine was \$15,000. 40% of the total reduction is attributable to a 40% reduction of turbine cost, and 24% of the tota? reduction to the forecast production rate.
- (4) cost was predicted through GASP by the following proportionality:

 $cost \alpha (hp)$ .68

Four engines were examined in the study: a turbojet, turboshaft, turboprop, and turbofan. The turboprop is more applicable than the fan to twin engine aircraft in the 242-298 kw (325-400 hp) range.

Cruise and takeoff criteria favor the turboprop over the jets. SFC is worse than for reciprocating engines, but the difference is returned in weight and size reduction. Seatmiles/gallon increases substantially.

# IV. Recommended Areas of Emphasis for Research

- (1) Energy efficient engines with improved SFC.
- (2) Alternate fuels research.
- (3) Establish durability evaluation methods (Example: methods for predicting engine life and TBO).
- (4) Bird strike design methodology. A method is required for designing turbofans of all sizes to survive bird strikes and still operate.
- (5) Pursue GATE.

Propulsion groups should examine high performance, low cost turboprops, advanced engine cycles, and component improvements in efficiency and manufacturing techniques.

# A.3 AVIONICS AND AUTOPILOT MANUFACTURERS

## A.3.1 Company U

ł

I. Integrated Avionics

### Multiplexed systems

The military is evaluating the "1553" multiplexing bus.

Multiplexing should not be incorporated into general aviation avionics systems.



Multiplexed System (military)

(numbers indicate individual avionics units. i.e. navigation radio, communication radio, Automatic Direction Finder, etc.)



Broadcast System (ARINC)

Multiplexing has some serious disadvantages:

- (1) expensive.
- (2) if the controller fails, the entire system fails.
- (3) requires triplex lines and receivers for redundancy.

Multiple connection between data sources and sinks is not necessary as shown in the broadcast system. If one unit fails, the entire system does not fail.

## System costs

The primary costs in avionics are displays and sensors.

ŧ

Low volume production increases unit costs.

Cost reduction  $\exists a$  avionics demands that low cost sensors be developed.

New sensing methods should be explored: optical sensors and low cost solid state sensors.

The automotive industry has not been very helpful in these areas since they do not appear to be pursuing advanced technology.

#### II. Electronic Instrumentation

Displays

Color displays can be as economically produced as monochromatic displays.

Multifunction displays offer great promise.

Mean time between failure (MTBF) of the units can be doubled if operating temperatures are decreased 11° C (15° F).

Currently developing high resolution color CRT displays for commercial airlines.

Built-in-test

Has some advantages if its design is not overly complex.

Results obtainable depend largely on the innovation of the designer.

### III. Advanced Technologies

Digital technology will filter down to general aviation with low cost computers. As electronics cost decrease, capabilities increase.

Airframers still decide on which major systems to incorporate or pursue.

Electronic fuel controllers show promise if they can lower life cycle costs.

# IV. Navigation Systems

Omega - VLF

Not accurate enough for domestic service.

(Companies U & V)

Loran C

Incomplete airspace coverage.

In addition, 100 to 200 kHz power line transmission control signals interfere with its performance.

#### Global Positioning System

GPS has the ability to fulfill the roles of VOR, DME, RNAV, and possibly category I approaches.

GPS can be made attractive to a very large group of customers which includes general aviation. It is very cost effective.

It offers up to 10 meter accuracy with the most sophisticated equipment, but only 200 meters for the less sophisticated equipment.

The key to GPS is integrating it into the current navigation system.

In the 80's, a GPS system could possibly cost less than VOR-DME for equal performance.

Antennas are not a restrictive problem. A 15 cm diameter, 2.5 cm thick antenna should suffice for general aviation operations.

# V. Suggestions for NASA

NASA should sponsor research, but avoid in-house development. Industry (avionics) will develop technologies and concepts.

### A.3.2 Company V

#### I. Current Avionics Trends

Industry is leaning towards installation of complete systems produced by one manufacturer. Today's systems cannot be easily interfaced.

OEM's will set requirements for systems integration by the "common harness" concept.

Integrated systems (or units) are evolving and already appearing on the market. Example unit: VOR receiver, DME, RNAV (4 waypoint), ILS, Glideslope.

(Companies V & W)

#### II. Comments from the Company

Does not want standardization of avionics in general aviation. This tends to destroy the competitiveness of the avionics market. If all the units look and operate similarly, freedom of design and competitive motivation could be lost.

Avionics products are governed by the inertia of the pilot market. Pilots resist new changes to the current products, and will not buy units with radical differences from the norm because of their lack of understanding of the products. Because of this, avionics products that require operations training are not favored by the manufacturers.

## III. Research for NASA

Human factors research is needed on the cockpit environment.

Research is needed on "display by exception" because of the limited panel space in general aviation aircraft.

#### A.3.3 Company W

I. Integration

Integration of systems is inevitable, and is already occuring.

Integration of avionics has some serious drawbacks. For example, power supplies must be redundant.

# II. Cost Drivers in Avionics

Actual electronic component cost is decreasing.

The level of sophistication in usage requirements (50 to 25 kHz spacing, better filters) has increased. This increases costs.

Requirements for increased reliability of units, certification, and labor wages increase costs.

#### III. New Technologies

þ.

Health monitoring of the engine is not difficult, but low cost sensors must be developed to make the idea feasible and practical.

Monitoring by exception would greatly reduce pilot workload.

Digital operation is eminent and is currently being incorporated into avionics. However, Company U autopilots are analog systems.

# IV. Suggestions for NASA Research

- (1) Human factors and pilot workload in the cockpit.
- (2) Pilot displays.
- (3) Low cost sensors
- (4) Standardization of cockpit layout and controls.
- (5) However, since aviation is not a crucial national problem, NASA might better concern itself with energy research.

#### A.3.4 Company X

I. Displays

1

A market exists for better displays even at a higher price.

Visualizes CRT's replacing mechanical HSI's.

CRT's will become a part of this company's avionics products because more functions can be integrated into a compact display space. The cost will be high, however.

NASA should research methods for replacing expensive mechanical instrumentation.

Significant reductions in price would be achieved if the market volume was doubled or tripled.

10% of sales is devoted to research and development.

The DAAS program is good only for demonstration. Regardless of DAAS, integrated avionics technology will develop within the industry.

### II. Fiber Optics

Fiber optics show great promise by replacing heavy wire bundles and eliminating interference. However, fiber optics require a standard bus for operation.

Improvement of mechanical connectors for fiber optics will promote the integration of the technology.

(Companies X & Y)

### III. Electronic Flight Controls

Absolute dependence on electronics is becoming economically feasible. Quadruple redundancy can achieve extreme reliability.

Fly-by-wire technology can be expected in this market, but only slowly.

Separate Surface Stability Augmentation (SSSA) shows a lot of promise.

IV. Product Liability

This is a problem for this company (like any other company), but they have yet to pay any product liability claims.

#### V. FAA Certification

Has good experience with their FAA regional office. However, sometimes the FAA does not know how to approach new technology. They attempt to test new technologies with old methods, sometimes nullifying any advantages.

Lightning is not a problem for this company's products.

A.3.5 Company Y

Avionics is a \$435 million/year business. Company Y controls 45% of the market. Most customers are corporations.

Caters to the market and produces airline-quality products for general aviation users.

Specialities: attitude and heading displays and sensors air data computers, displays, and sensors automatic flight control.

I. Projections for Technology. The following are likely:

More integration of functional features with regard to information acquisition and displays.

More digital computation and interface.

Strapdown AHRS (Attitude Heading Reference System)

Flexible vs free floating gyros (for cost savings)

CRT displays will replace conventional ADI, HSI, and eventually the entire panel.

(Company Y)

#### Management systems

Long Range NAV

Energy management including autothrottle, fuel controls, and flight control systems.

GPS

Too expensive, though highly accurate.

Controlled by the military. Not available for 7-10 years.

Antennas are unwieldy.

Collision Avoidance System. Interesting because of computer application.

# II. Future in Flight Controls

Looks 5 years ahead. 10 years is too far downstream.

One of the most difficult tasks is to service obsolete systems. New systems appear continuously, and old systems are expensive to maintain.

In general aviation today, the airframer or the user integrates the system. In the future, only the airframe manufacturer will integrate systems due to complexity and lack of standardization.

Because analog system capabilities are almost saturated, digital technology is attractive because of its increased functional capability, reliability, and accuracy. Company Y does not produce digital systems currently, but is exploring the technology.

The emphasis on size and weight is a unique problem to general aviation. The bus interface will allow lower weight.

III. Navigation and Energy Management

Autoland

ţ,

Too expensive for general aviation. Category II approaches are impractical for general aviation pilots because of FAA maintenance and training requirements.

Stability augmentation/load alleviation requires faster actuators.

(Companies Y & Z)

.

Greater automation: digital systems appear promising, but requires more power.

Fly-by-wire and fiber optics are not being pursued. They are not cost effective. Even boosted systems require a cable backup.

Navigation management capabilities and results (result of onboard computers)

Integration of navigation sensors (INS, Omega, VOR/DME).

Full ATC data base, including charts and plates.

Flight planning.

Reduced workload.

Energy management capabilities (result of onboard computers)

Improved fuel and engine efficiency.

Flight manual storage.

Automatic engine control.

IV Air Data Systems: (see fellowing figures)

A.3.6 Company Z

I. Current Products and Technology

Manufactures flight control systems for general aviation, primarily for those aircraft weighing 5765 kg and under

Utilizes a single rate gyro, which is powered pneumatically and electrically for redundancy.

Invented the tilted gyro turn coordinator.

Double power system on gyro is more stable. Each power supply modulates the other when a power surge is experienced.

Uses 400 Hz A.C. electric drive and pneumatics. The gyros regulate valves which control pneumatic servos for the autopilot.

If the electrical system fails, the wing leveler function still operates because the entire system from gyro to servo is totally pneumatic.



.

. .

Navigation/Energy Management System



-

and a state of the second s

•

car deter an group complex genters



344



.

۰.

Digital Air Data Computer Basic Block Diagram

.

345

₹

.

•

#### II. New Technologies

Will not incorporate digital technology into their analog products. The cost of recertification is too prohibitive.

Digital systems do not offer the redundancy of pneumatic systems, where the former is only electrical and the latter is pneumatic-electric.

Cost also prohibits switching from pneumatic to electric actuators.

Pneumatic system has advantages: altitude compensation, no electric current drain, and smoother operation.

Familiar with the fluidic rate sensor research at Langley, but does not perceive future applications of the technology.

Advanced displays

HUD's, CRT's, are excellent for reducing pilot workload.

Transition from IFR to VFR flight is dangerous. CRT displays could alleviate this problem. A HUD allows for a safer transition.

Fiber optics. No foreseeable applications at this company.

### III. Certification

A serious problem.

Certification procedures make minor product improvement changes unattractive to this company due to excessive (and therefore expensive) delays.

Weeks of preparation by the company followed by delays and then a trivial flight test damage the image of the FAA and the certification process.

Has over 260 STC's. Any major change on the autopilots would require a new STC for each aircraft: phenomenal cost for a small company.

#### A.3.7 Company AA

#### I. Current Products

Produce flight directors, HSI's, autopilots, turn and bank indicators, and yaw dampers.

(Company AA)

Present products are analog-digital hybrids. Utilizes many integrated circuits.

Uses a strain-gage, piezo-resistive pressure transducer for altitude hold input.

Autopilot actuators are DC servos with infinite resolution and a slipclutch.

# II. Advanced Technologies and Problems

CRT's. Waiting to examine the success of this technology in the commercial aircraft market.

SSSA. Not pursuing this technology. Conservatism and product liability are the major drawbacks. (Conservatism on the part of the airframe manufacturers.)

Problems with new technologies:

Ŧ

Field service. Retards the incorporation of microprocessors into their products. The field maintenance skill level is not high enough to support this technology.

User misuse and lack of understanding. Pilots do not read the users manual and they do not know how to use some of the features of the autopilots.

Product liability. Liability cases are so far-reaching today that small companies must exercise caution when introducing new products. Product liability severely inhibits new technology, innovation, and advanced products. Recently hired a full-time accident investigator.

Flight Directors. One representative of this company was not convinced of the benefits of flight directors. Pilots are attracted to the HSI commands and do not maintain outside surveillance.

3% of sales is returned for internal research and development.

#### A.4 SYSTEMS MANUFACTURERS

#### A.4.1 Company BB

# I. Controls

Digital Flight Controls: Looking at a distributed microprocessor.

Fly-by-wire: Already developed a fly-by-wire system. Utilizing a failsafe 3 motor common control drive.

Fiber optics: A fiber optic control system has been developed and flown.

# II. Composite Structures

Control Hardware

Developed filament-wound graphite push-pull rods.

Developed injection molded composite bell crank. Cost analysis:

> Composite bellcrank -- \$11.15 Equivalent magnesium bellcrank -- \$75.00 Illustrates a savings of 85% due to composite material.

### Airframe

0

Currently under contract to NASA-Langley Research Center to evaluate composite airframe components in the field.

Evaluating graphite and Kevlar. Graphite is utilized in controls and surfaces. Kevlar application is primarily in doors and fairings. Stiffness is not a restrictive problem of these composites, but their non-isentropic properties must be recognized.

Graphite hazard: filaments of graphite are released in a crash environment and infiltrate electronics, causing short circuits.

#### III. Displays

Heads Up Displays (HUD) and Head Mounted Display (HMD)

HUD's are extremely useful in an IFR environment.

(Company BB)

HMD's are more versatile than HUD's. Allow unrestricted movement and are simpler and lighter.

Current CRT head mounted displays are heavy, cumbersome, require a helmet mounting, and are not commercially viable. They also block peripheral vision and have a monocular presentation.

Developed the micro-HUD. A small reflecting chip is mounted onto glasses, directly in front of the eye.



for photograph.

Method has broad applications. Has potential to cut cost over current HUD's by a factor of 4. Dynamic display capability obtained through a vibrating fiber optic display. (see figure on next page.) Mechanically vibrated optics, with timed light impulses, creates dynamic images, similar to CRT's.

# IV. Fiber Optics

Lightning strike benefits

Aircraft that have composite structures and utilize fly-by-wire are vulnerable to lightning strikes. Current will destroy the system.

Fiber optics are non-conducting and would not be as vulnerable to a current surge from lightning.

Structural monitoring

Fiber optics can be imbedded within the composite material matrix for structural monitoring.



Vibrating Fiber Optic System With Microprocessor Symbol Generator.

Scanning the fibers with a microprocessor, a signal delay or lack of signal from a fiber would indicate a possible failure developing within the material it was imbedded in.

Signal carrying

Fiber optics have a wider bandwidth, allowing more signals or data to be passed along them as compared to wires.

# A.4.2 Company CC

Deals with control of vehicles, from aircraft to mopeds.

I. Avionics

PCAAS

Projected technologies were studied, but current technology was utilized for the demonstration program to minimize costs.

**GPS** system

A GPS antenna may be too large for incorporation into a general aviation airframe.

### II. Control Systems

Microprocessor systems

Electronics for advanced control systems are available at low cost. The problem of cost for controls stem from the sensors, the displays, and the actuators.

NASA should investigate low cost sensors and actuators.

Fluidics

Are impaired by temperature variation because of thermistor control.

Because of sensitivity to temperature, fluidics cannot meet the TSO temperature requirements.

III. Safety

Spiral

A wing leveler should be provided for general aviation aircraft.

Some aircraft have apparent spiral modes with different time constants than the actual spiral mode. These result from lateral out-of-trim conditions.

An old NASF study suggests utilizing a centering spring to relieve the problem.

Very speed sensitive. Also, aircraft may diverge in one direction and not the other. Characteristics also differ from plane to plane.

FAA crash studies

Compiled complete crash statistics over a 10 year period for single engine, multi-engine, and retractable gear categories.

The single engine retractable gear aircraft was involved in the largest percentage of crash incidents. One aircraft, however, was equipped with a wing leveler and remained accident free.

(Companies DD & EE)

## A.4.3 Company DD

and the second second

# I. Turbocharging and Engine Control

The match point for aircraft engines is matching the turbocharger to the aircraft flight envelope.



The truck diesel engine turbocharger provides a production base of over 1 million units per year.

Microprocessors are applicable to aircraft engines only if total control is exercised: simultaneous control of mixture, manifold pressure, and rpm.

Turbochargers are generally overdesigned and will (must) match engine TBO.

The most significant improvement forecast for turbochargers is the incorporation of air bearings. However, most of the turbocharger advanced technologies (air bearings, ceramic wheels, sheet metal housings) are too costly.

Current aircraft turbochargers have been labeled "just a truck turbocharger". This turbocharger, however, covers a wide power range and allows for low cost because of the diesel engine production base.

# A.4.4 Company EE

Produces actuators for commercial aircraft and large business jets at the rate of 2 to 3 units per month for each type.

### I. Hydraulic Flight Control Actuators

A' tuators are "tailored" to each application. They are designed for a specific aircraft type. General "off the shelf" hardware is not available.



Turbocharger Design Logic

3

Ì

Actuators are very sophisticated because of redundancy requirements.

Actuators must be located at the surface for production of maximum power.

"Custom" actuators of these types and sophistication are too expensive for general aviation application.

#### A.5 NASA RESEARCH CENTERS

### A.5.1 Ames Research Center

I. Avionics

Improved guidance and navigation based on VOR/DME. Emphasis here is on improved systems which use VOR/DME.

System is based on 2 VOR/DME systems, where the receivers scan for close proximity stations. One difficulty lies in the fact that it takes 5 seconds to lock and interpret a VOR signal. Hence, the time to search for acceptal signals can be prohibitive. An alternative would be the incorporation of a computer which stores the last station and signal to speed up the scanning process.

A further difficulty lies in storage requirements. Inputting station longitudes and latitudes is very time consuming. VOR stations which broadcast their position (longitude and latitude) in addition to radial information eliminate the requirement to load this data to memory, and would greatly enhance the attractiveness of the scanning mode.

Scanning results could be incorporated with air data to greatly enhance navigation guidance.

Pilot Interface with Avionics

Each capability added to avionics today will affect pilot workload by a factor of 4. Hence, emphasis is being directed toward simplifying the pilot workload involved with the use of avionics.

For example, improved navigation capabilities (scanning) requires loading station locations. Development of an "electronic map" allows the pilot to input station locations by either a light pen, a matrixed optical system surrounding the map display, or by a touch-



sensitive screen. The map display system described here is currently flying in 2 Cessna 402 at Ames.

(Ref. 50)

This photo is of the light pen. Also demonstrated was a map overlay which is inserted over a touch-sensitive very thin plastic grid of wires. This grid sheet is laid over the CRT with the map over it. The sheet is manufactured by a company that makes touch-sensitive keyboards for microwave ovens. The large number of

ORIGINAL PAGE IS OF POOR QUALITY

grids per sheet and small number manufactured resulted in a cost of \$1200 for three grid sheets. Costs would be much lower for a production run.

This touch sensitive map has been installed in a Cessne 402.

To locate points on the map, the pilot touches the map location. The plotter arm moves to the input point and marks the map. Then a query is returned: OK? The pilot responds by touching "YES" or "NO" on the screen. If required, the pilot is allowed to improve his map point by responding to "LEFT/RIGHT" queries and "UP/ DOWN" queries. As such, he can quickly, accurately, and reliably load what amounts to an RNAV flight plan without having to load radial/DME waypoints.

The matrixed optical system, where light beams form a grid in front of the CRT screen, is shown on the following page. Placing a finger on the CRT display interrupts a light beam and results in an input to the computer. The weight and balance system is shown where (1) the particular airplane loading capabilities for the particular 402 exists in the software package (fuel/baggage mix in tip tanks and nacelles, and passenger seating configuration), (2) the white dot in the envelope as well as the airplane representation changes to show what (e.g.) 50 lbs in the left nacelle will do to the envelope as well as to the available volume in the nacelle. The system is virtually error proof, simple, and fast.

Replacement of gyros with other sensors

Replacement is accomplished with a 3 axis magnetometer which senses accelerations and integrates outputs about 3 axes to derive pitch, roll, and yaw.

Pressure transducers are also receiving attention

Silicon chip pressure transducers using piezcelectric c.ystals in membranes cost around \$12.

A shed vortex airspeed measurement device based on the principle of airspeed being proportional to the frequency of shed vortices is installed and working on the Cessna 402.

#### Common bus

Reduces weight, simplifies interface of components, and allows multiplexing of signals. Implies standard bus (note: general aviation is not required to meet ARINC standards).



(Ref. 50)

# II. Stall/Spin

SIL

The key to spin avoidance lies in the prevention of spin entry and not in spin recovery.

Research is based on leading edge gloves and gaps.

See sketch on next page.

# ORIGINAL PAGE IS OF POOR QUALITY



Vortices generated at higher angles of attack affect separation characteristics.

## Thus far:

R/C model with the modified wing will not spin.

Grumman American Yankee has vastly improved spin characteristics in flight test configuration.

Needs:

Theoretical basis needs to be developed.

Twin engine configurations need to be examined (for channel flow study).

III. Cooling Drag

Assess level of cooling drag

Cooling drag is defined as any drag resulting from pumping air through a nacelle (or cowl).

Utilizing a Seneca wing and a Cessna 402 nacelle for the test configuration. The baseline was established by (1) modulying the nacelle to eliminate trailing edge separation, (2) streamlining the front of the nacelle for low drag, (3) sealing the air inlets and the exhaust openings, and (4) removing the propeller. This configuration established the low drag baseline.

Compared to the original configuration, a 13% drag reduction was realized.

(Ames Research Center) Reducing the air mass flow rate required for cooling does not reduce drag much.

Optimize and Investigate different nacelle shapes

Eliminate cowl flap. Replace with louvered opening.

- louvered opening T

Evaluated three inlet sizes.

for the second sec



Single, low inlet produced the best results. better at high angle of attack.

allows optimized nacelle shape.



Better nacelle shape at front.



# IV. STAT

STAT is the Small Transport Aircraft Technology program which is aimed at investigating aircraft for the following range of specifications:

15 - 80 passengers

50 - 100 nm legs

Includes design studies of airframe, engine, and propeller, with specific studies of

high/low speed compatibility

good ride quality

low noise

identification of high cost features.

Overview of STAT

Will emphasize aerodynamic analysis and include:

Analytical methods for improved performance predictions for multi-element high lift devices.

Wind tunnel programs to examine turboprop slip stream effects on wing configuration.

Fropulsion system research to define improvements in turboprop and turboshaft engines.

Structural research to identify cost reductions in manufacturing through the use of advanced aluminum alloys.

Avionics research to yield improved IFR capabilities.

Aircraft flight control systems:

fly-by-wire
fly-by-light
integrate1 electronics
digital controls
fail safe and fail passive controls (already incorporated in V/STOL and helicopter aircraft)

Improved icing protection.

Ways to reduce commuter prices to \$55,000/seat.

May also examine bonding/composites because of the costs associated with fasteners (rive:s).

Typical specifications that STAT would cover: turboprop, 50 passengers, 250 kts.

#### V. Aerodynamics

General comments

Drag reduction at high speed involves delaying M<sub>div</sub>.

Drag reduction at low speed essentially lies in improving C<sub>L</sub>

Computer codes:

FLO 22 - transonic wing on wall

FLO 28 - transonic wing on body

Full potential flow wing-body with attached flow and no small disturbances.

Natural Laminar Flow Control (NLFC)

6-series airfoil sections are examples of NLFC.

NLFC is achieved by reducing adverse pressure gradients over the airfoil.

Surface coatings are implied.

Very Mach dependent.

6-series has good cruise with wide drag bucket, but adverse stall characteristics.

. . . . . . . . .

# (Ames & Langley Research Centers)

NLFC may not be practical due to bugs, insects, etc.

NLFC is difficult to evaluate in the wind tunnel due to tunnel effects.

Comments on GA(W) airfoil

Use of 2D data without regard for the wing-body junction may lead to flow separation there.

Best use of a GA(W) wing might be on a high wing configuration, where the wing upper surface doesn't intersect the fuselage.

# A.5.2 Langley Research Center

I. Crash Dynamics

Crashworthiness Design

Seats

Energy absorbing seats can reduce peak vertical "g" loadings by 50%.

Wire bending (translating a loop along wire) is a very efficient way to absorb energy.

One of the most promising crashworthiness features together with improved restraints.

Do not locate seats over wing spars.

Seat pan is very important. Current pans collapse. Must be designed with regard to the rest of the seat for good crashworthiness.

Industry builds seat for comfort (thick cushions compress too much and cause belts to loosen).

### Restraints

Must be anchored to seat with the seat firmly anchored to the floor.

Inertia reels are sometimes unreliable. Incorporated largely for ease of entry/exit. Ideally, a restraint holds the occupant firmly in his seat, and the seat and structure absorb energy.

Lateral movement of occupants is best controlled by a double harness (seat mounted).

(Langley Research Center)

Structure

Floor/subfloor should be made crushable to absorb energy. Aside from vertical stroking of the seat, this is the only other area where stroking distance exists to absorb energy.

Fire prevention

Military work on fuel containment is quite successful and technology transfer is needed.

Fuel treatment to reduce vaporization has some potential.

Foam in the fueltanks costs 1% to 2% of tank volume but can contain fuel and prevent spraying.

II. Compusites

Composite types

Advanced composites include graphite, Kevlar, and boron. Glass is not considered an advanced composite. Graphite and boron are probably too expensive for general aviation.

Testing

Flight Service: Aircraft with test samples include:

L1011, Boeing 737 - Kevlar. CH-54 - graphite stifferers. Bell 206 - Kevlar doors, fairing. C-130 wing box - boron/epoxy. DC-10 rudder, Boeing 737 spoiler skins - graphite/epoxy.

Worldwide ground based exposure.

Problems with composites

Lightning strike

Graphite suffers only from localized damage.

Kevlar and glass are more of a problem.

Composites do not attract lightning any more than metal structures.

(Langley Research Center)

Metal honeycomb with composite skins could compromise safety since a clear path for the strike to dissipate may not be available.

Solutions are possible by using any external conducting path such as wire screens.

Moisture absorption/high temperature

Matrix absorbs about 1% moisture by weight.

Paint absorbs about 3% moisture by weight.

High temperature and/or moisture weakens matrix.

Miscellaneous

3

ì

Generally impervious to fuels.

Composite manufacturers: Dupont, Hercules, 3M, Union Carbide.

III. 3D Aerodynamics

See next page for 3D aerodynamics progress.

Currently working on wing/canard design and wing/fuselage interaction.

IV. 2D Aerodynamics

Turbulent flow shapes

Tend to be aft loaded.

Control surface floating is not a problem.

Industry is wary of aft-loaded airfoils.

Natural Laminar Flow

Goal: Attainment of C<sub>L</sub> equivalent to that of the max turbulent flow airfoils while maintaining the low drag of the NACA 6-series sections.

Thickness and moment constraints also exist.

Bellanca Skyrocket airfoil: t/c = 15%,  $C_{D_o} = .006$ ,

c<sub>l</sub> = 1.8. max

(-5

			1	•		
1 K. 44 1 MET 191	na serena en antina.	a de <b>houghl</b> ooker die ei	••••	16 day 16 4 5 5 50	an 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1. 1.	
		۰.				

	Flow Type	Analysis					Design			
Shape		Steady			Unsteady			Steady		
		Sub	Trans	Super	Sub	Trans	Super	Sub	Trans	Super
Airfoils	Inviscid	+	+	÷	+	+	+	+	+	+
	Viscous	+	+	-	[+]	-	-	+	+	-
Wings	Inviscid	+	+	+	+	[+]	+	+	-	+
	Viscous	[+]	[+]	-	-	_	-	-	-	-
	LE Vortex	[+]	-	[+]	[+]	-	-	[+]	-	-
Complete Aircraft	Inviscid	+	[+]	÷	[+]	-	[+]	-	-	[+]
	Viscous	-	-	-	-	-	-	-	-	-
	LE Vortex	[+]	-	[+]	[+]	-	-	-	-	-

# Advances in 3-D Aerodynamics

.

+ Available

[+] Not mature yet
- Not available

. ...

.

-

.

∢

365

a) - e e

1

Nati 1 1 A A

• •

```
c, is not sensitive to roughness.
```

Drag is moderately to significantly sensitive to roughness.

Maintenance of airfoil contours (ripple-free) is more important than bugs, insects, dirt, etc. Flat spots are bad.

Low speed airfoils ( $M \leq .4$ )

Improve symmetrical airfoils with control surfaces.

Attempting to reduce pitching moment. One example (LS(M)-0417), when compared to the GA(W)-1, shows:

```
Δc<sub>l</sub> = +.22
max
c<sub>d</sub> is reduced
c<sub>m</sub> shows a 30% reduction
slightly worse stall
c<sub>d</sub> lower in climb.
```

Medium speed airfoils (M < .7)

Examing two airfoils: 13% and 17% t/c

Retains excellent low speed characteristics

Improved  $c_m$ , same  $c_l$ , and less  $c_d$  than old NACA max sections.

V. Avionics and Controls

Current programs

T-30 (Sabre business jet) will be equipped to study pilot-ATC interface. Digital and analog systems will be included.

The heart of the system is a digital Bendix 910.

A conventional analog autopilot is included.

Fully programmable color CRT display.

4

#### (Langley Research Center)

MLS, curved approaches, will require more than conventional flight director.

Fluidics

Low cost, highly reliable, rate sensors for autopilots.

The wing leveler is inexpensive, simple, and performs well. An \$85 model for homebuilts is available commercially.

Examining a 2-axis wing leveler and heading hold autopilot

Working with Cessna and Piper on a fluidic autopilot.

Looking at an all-fluidic airplane for demonstration.

Pilot workload - single pilot IFR

A simulator is an important tool for human factors research.

Unconventional controllers show promise. Example: side stick controller.

Integrate displays and instrumentation.

DABS - Discrete Address Beacon System - automatic uplink and downlink data flow.

Exploring the speech synthesis problem.

Langley has a general aviation simulator programmed to study these problems and possible solutions to them.

VI. Stall/Spin Research

Tools

.

. ۳۳. - Spin tunnel (Re  $\approx$  80,000 to 90,000) - Only steady-state spin conditions can be examined. Spin entry cannot. Have encountered problems with small scale effects of the test models.

R/C models, 1/5 scale.

Wind tunnel models.

Full scale wind tunnel.

Flight test.

(Langley Research Center) Simulation - A data base is required to develop this.

Airplanes

.

1 15

- #

Yankee Sundowner Ceasna 172 Piper T-tail prototype

Configuration Effects

Drooped leading edge

Full span application of this modification aggravated the spinning characteristics of the test aircraft. Spins became unrecoverable. .

¥

Can improve spinning characteristics if (1) the modification covers 50% - 60% of the outboard span and (2) ends in a sharp break near the mid-span of the wing.

The modification appears to require a chord extension.



More testing must be performed to establish the benefits of this modification to different aircraft types.

# A.5.3 Lewis Research Center

# I. GATE

21

4

Purposa

Study application of advanced technology to small turbines to determine utility of small turbines in general aviation.

1

Overall goal is 20% increase in performance and a 50% reduction in cost (to \$50/hp).

Company participants (4)

Detroit Diesel Allison

Garrett AiResearch

Teledyne Continental

Williams Research

Garrett, Williams, and Teledyne approached the problem with the low cost idea, and reduced the cost of turbine engines by 50% at forecast production levels.

Allison's front end costs were higher but DOC was greatly improved through sophistication.

Some viewpoints of GATE

Must be advocated to NASA headquarters.

Began two years ago. Now investigating a hardware program, costing \$60 million with two contractors producing test engines.

Low cost potential exists, although at high risk for the manufacturer. Hardware production will not occur soon, if at all, without NASA funding.

The demonstration engine would not run until 5 to 6 years after funds are allocated.

Main drawback is high risk, and that risk will not be acceptable to industry without NASA funding.

. . . . . . .
- •

### II. Positive Displacement Engines

General aviation positive displacement engine research

### Program goals

Improved fuel economy, weight and drag reduction.

Major Thrusts

Lean operation of conventional engines

Improved fuel injection

Improved cooling

Advanced combustion chamber and modeling

Definition of alternative engine systems (advanced spark ignition, light weight diesel, stratified charge rotary combustion)

Advanced combustion techniques. e.g. adiabatic

High speed/high pressure fuel injection

Continued upgrading of facilities, instrumentation, and analytical/diagnostic capabilities

Aircraft Fuel Injection

Sec. 1

Objective: Establish requirements of an improved fuel injection system for leaner operation.

Approach: Determine effects of spray quality on engine performance and emissions.

Droplet size, distribution, and velocity

Spray pattern

Injection timing and duration

Nozzle position.

How: Manifold flow visualization tests, injector characterization tests, engine combustion tests (using commercial and test engines.)

370

### (Lewis Research Center)

i

2

Advanced spark ignition aircraft piston engine design study.

Technology analysis and design candidate evaluation.

Define expected technology base for an advanced spark ignition aircraft piston engine for the late 1980 time period.

Evaluate candidate design items against criteria to choose most likely advanced technology for engine design.

Recommend new or augmented technology programs.

Fuels for advanced spark ignition aircraft piston engine.

Near term: 100 LL AVGAS or wide-cut version (homogeneous charge combustion)

Far term: kerosene base commercial jet fuel (Jet A). (stratified charge combustion).

Homogeneous charge.

ultre a de sta - -

a a substance a sub-

- 2 19.4 34 1 1

1<sup>2</sup> \*\*\* 5 \*

Conventional:

Compression ratio (thermal efficiency) is detonation limited to approximately 8.5-9.0 to 1.

Fuel economy at higher power is materials (temperature) limited.

High Compression Ratio Lean Burn (HCRLB)

Increased compression ratio (increased thermal efficiency) is possible with 100 octane fuel to approximately 12 to 1.

Lower exhaust gas temperatures improve valve and valve guide durability.

Minimal hardware changes to conventional engine produce significant results.

Expect greater than 10% increase in fuel economy.



~

Advanced Technology Base Items Hierarchical Structure.

ł,

T . Asta

(Teledyne Continental Motors)



الاسوم الي الألة

(Teledyne Continental Motors)

and the second second

۰.

- a - au

(Lewis Research Center)

Engine operational systems.

Fuel delivery systems.

Electronic fuel injection.

Electronic fuel control (mechanical fuel system, tweaked electrical system)

Electronic air control (controls engine performance by control of air flow).

Direct cylinder injection.

Ignition systems.

Y

Breakerless, continuously variable timing.

Breakerless, stepped timing.

Engine Auxiliary systems.

Single-lever control.

Servo-mechanical.

Servo-electrical/electronic.

Electric power generation.

High speed, brushless alternator.

Single-unit starter motor/generator.

Engine driven air conditioning

Rovac, air-to-air heat exchanger (rotary compressor, saves weight by 50%)

Design and technology features.

High Compression Ratio Lean Burn engine (HCRLB)

Six-cylinder, horizontally opposed, air/oil cooled.

12:1 compression ratio, lean burn, 100 octane avgas.

Turbocharged for critical altitude of 10,670 m.

Naturally aspirated version also made available (without turbocharger only).

# ADVANCED SPARK-IGNITION AIRCRAFT PISTON ENGINE ADVANCED TECHNOLOGY ITEM EFFORT VS. BENEFIT

;

1

	Low	Moderate	High
Fuel	Wide-cut AVG/	S Effort	
Combustion System	Homogeneous	Charge Lean Burn	Stratified Charge
Supercharging/ Turbocompounding/ Bottoming Cycle		Improved Supercharg	jing/Turbocompounding
Engine Operational Systems			Electronic Controls
Configuration		V	
Cooling			
Materials			
Manufacturing			
Engine Auxiliary Systems			
Lubricants	Muitiviscosit	y Grade Oils	

(Teledyne Continental Motors)

×.

### (Levis Research Center)

Electronic microcomputer single-lever control system.

Target weight of 373 kw version is 163 kg.

Gear-driven propshaft - 0.75:1 (partly for noise).

Minimum 2000 hour TBO.

Compatible with common-cylinder family concept (down to 4 cylinders).

### Stratified Charge

-

Six-cylinder horizontally opposed, air/oil cooled.

12:1 compression ratio, stratified charge, kerojet fuel.

Turbocharged for critical altitude of 10,670 m.

Naturally aspirated version also made available.

Electronic microcomputer single-lever control system.

Target weight of 373 kw version is 136 kg (highly turbocharged).

Gear-driven propshaft - 0.7:1

Minimum 2500 hour TBO.

Compatible with common-cylinder family concept.

Rotary engine program.

Candidate for future general aviation use.

Assess current state of the art.

-

Look into advanced rotary engine designs.

Program status:

RC 2-75 Curtiss Wright engine tests - completed. Modified RC 2-75 test contract - in progress. NASA in-house rotary engine test - just starting.

· · · · · · · · · · · · · · ·

# ADVANCED SPARK - IGNITION AIRCRAFT PISTON ENGINE TECHNOLOGY BASE CHRONOLOGY (OPTIMISTIC)

	80	81	82	83	84	85	86	67	88	89	90
Precise Engine Definition	7										
Time Available for Development of Advanced Technology Items											
Development of Engine Manufacturing Techniques											
Parts and Materials Procurement for Prototype											
Prototype Engine Build-up and Testing						E					
Engine and Engine/Air Frame Certification Testing	Te Le	echn evel I	ology Froz	y en \				e			
Marketing and Customer Acceptance Testing					Y						
Product Enters Market										¥	7

(Teledyne Continental Motors)

377

(Lewis Research Center) "clean sheet" rotary engine design - just starting Curtiss Wright 1119 kw stratified charge engine tests - in progress. (land vehicle) Stratified charge engine tests - future (aircraft) In-house rotary engine. 1978 Mazda, 2 rotors, 75 kw at 7000 rpm. Test Program: Baseline - carbureted. Tear down and install combustion instrumentation. Repeat baseline. Autotronic fuel control and multispark ignition. Leaning study. Ignition effects on combustion. Thermal barrier coatings. Turbocharging. Stratified charge - simplified system. "Clean sheet" rotary design st\_dy. Contractor: Curtiss-Wright Corporation Date: June 11, 1979 - June 11, 1980 Goals: 1 1b/hp at takeoff power. BSFC < .23 kg/kw/hr at 75% cruise. Operate efficiently on 110/130 octane plus one or more alternative fuels. Meet 1979 EPA piston aircraft emission standards.

¢

į





	EMISS	10NS-% OF	STD	WEIGHT**
	HC	CO	NOX	KG/KW
EXISTING ENGINE	139*	89	57	•77
ENGINE RANGE	60-240	180-360	10-50	.85 - 1.09

\* 85% from taxi & idle \*\* includes cooling system

Rotary Engine Characterization



(Lewis Research Center) Manufacturing costs comparable or less than current aircraft engines.

Maintenance and overall life cycle costs lower than current reciprocating engines.

Altitude capability equal to current engines.

Use SCRC 4-350 as baseline engine.

Conceptually design engine.

Engine/airframe integration.

Identify new technology items which offer a significant payoff.

Diesel engine program

<u>`</u>

×

Objective: Develop in-house research and technical base to assess potential alternative engine candidates for future general aviation use.

Potential benefits.

50% reduction in life cycle cost.

25%-40% reduction in fuel consumption.

107-30% reduction in specific weight.

50% reduced specific size.

Improved reliability through fewer moving parts.

Broad range of usable fuels.

Engine description.

2 cycle, radial diesel (6 cylinder rated at 298 kw (flat)).

3500 rpm geared to 2400 rpm for the propelier.

Uncooled, adiabatic engine. High technology insulated cylinders.

Minimum cooling configuration greatly reduces cooling drag. Cooling is required for the turbocharger and oil.

(Lewis Research Center)

2 cycle operation results in 93° C lower EGT.

Evaluating a 149 kw and a 298 kw aluminum finned, cooled engine. 149 kw version is 4 cylinder, with a BSFC of .213 to .219 kg/kw/hr.

For the 298 kw, uncooled version, cruise BSFC is .195 kg/kw/hr. At 6100 m and full power, BSFC is expected to be on the order of .213 to .219 kg/kw/hr.

The diesel operates best at full power but obtains its best BSFC at 75% to 80% power.

III. Propeller Technology

Candidate Propulsion systems (4)

Turboprop Propfan Variable pitch fan Turbofan

10

Parametric comparison of a small aircraft to a large aircraft:

	small	large
CL	1.7	3.0
W <sub>max</sub> (kg)	5675	363200
S (m <sup>2</sup> )	29.97	511
W/S (N/m <sup>2</sup> )	1867	6990
C <sub>L</sub> (min drag)	.7	.7
q <sub>stall</sub> (N/m <sup>2</sup> )	1101	2327
$q_{L/D_{max}}(N/m^2)$	2681	10055
L/D max	13	19

Relative to a large aircraft, general aviation aircraft require propulsion systems having:

high cruise thrust at altitude high engine thrust/weight ratio low specific fuel consumption

# INSTALLED PROPULSIVE EFFICIENCY AT CRUISE

۰,





TURBOFAN -HIGHEST THRUST/WEIGHT

.

384

- \* - \* \* \* \*



PROP FAN FUEL SAVING TRENDS RELATIVE TO TURBOFANS

٠.

÷.



# MAX. L/D SPEED VS. ALTITUDE FOR "SMALL" AND "LARGE" AIRCRAFT

1.

· •

5

45

. \*

### (Lewis Research Center)

Propulsion system comparisons

Propfans: High speed - challenges the turbofan.

Variable pitch fan: Intermediate speed - challenges the turboprop.

Recommend two feasibility studies

- Propfan application to small business aircraft (Mach = .7 to .8).
- (2) Variable pitch fan application to commuter aircraft.

General Aviation Propeller (GAP) program

Background

NACA propeller research ended in the 1950's.

Advanced turboprop of 1975 at M = .8.

GAP - technology for lower speed general aviation.

Goals

` + ` ......

Reduce fuel used by 8% to 9%.

Lower noise by 5 to 10 dB.

Improved safety through use of composites for strength.

Current design practice

40 years of experience.

Aluminum construction.

Few blade designs: cut down or extend tip. add blades.

Aircraft integration: "cut and try" process.

. .

Design Trends:



Program participants

Cost/benefit study - McCauley

Improved performance concepts - Purdue

Aero-acoustic methodologies - OSU

Advanced propeller balance - contract

McCauley GAP cost/benefit study

Task I: Identify advanced technologies and cost/benefit.

Task II: Optimum configurations and mission analysis.

Task III: Technology assessment and program plan.

Aircraft Categories:

(1) Reciprocating engine, 261 kw, 250 kts.

(2) Turboprop, 746 kw.

(3) Aerial application, 186-746 kw.

McCauley Study Team:

à

McCauley - Performance and coordination. Cessna Pawnee - Aircraft and mission analysis. Ohio State - Noise and performance. Materials Science Corporation - Composites.

(Lewis Research Center)

Improved Performance Concepts

Swept lifting line analysis.

Improved testing techniques - laser Doppler velocimeter.

New concepts - proplets, swept blades.

Propeller aero-acoustic methodologies

Phase I: Predict and test 1.52 m diameter models

Airfoil technology - Clark Y, NACA 16, GAW-2, ARA-D.

Phase 2: Flight test comparison.

Phase 3: Improved methodologies and verification.

Team: Ohio State Rockwell International Hartzell Hank Borst Dowty

Small Transport Aircraft Technology (STAT)

Lewis Research Center Capabilities

Propulsion system performance

Noise

Icing Studies

Propeller aeroelastics.

FY 1979 Plans

-#\_\_\_~

Engine and propeller studies.



n M 1 U U

390

÷,



HIGH SPEED PROPELLER PERFORMANCE SUMMARY

391

.



`-

# NOISE COMPARISON OF PROPELLERS

Mo<sup>\*</sup> 0.8 Design Conditions

### APPENDIX B

### TECHNOLOGY EVALUATION

Five significant aspects of the technology evaluation are discussed in this appendix. Although all are briefly mentioned in Chapter 4, they are amplified here to further define the basis for the results of the technology evaluation.

Section B.1 discusses the Category Selection Survey (Survey 1) and describes how the 17 evaluation categories were selected.

Section B.2 describes the Category Rating Survey (Survey 2) and describes how the relative importance of the 17 evaluation categories were established. Histograms of round 3 and mean category rating trends are included.

Section B.3 describes the Candidate Technology Identification process. 81 deleted technologies (137 Preliminary Candidate Technologies - 56 Final Candidate Technologies) are briefly discussed.

Section B.4 addresses the PLOE studies which were performed as part of the analysis of the selected Evaluation Technique. Complete tables for the pessimistic, likely, and optimistic scenarios are presented for both airplanes. Also, the results of this study are graphically presented for both airplanes.

Finally, Section B.5 examines the effect of Empty Weight category ratings of 0 and 9.5 vs the computed ratings of 4.340 and 5.657 (for Airplanes A and B respectively) on the technology rankings. 393

### **B.1 CATEGORY SELECTION SURVEY**

The category selection survey was based on the Delphi method which involves sequential rounds of written communication in which the group response from one round is fed back to participants for consideration in the next round. The participants were the project staff and members of the University of Kansas Aerospace Engineering Department Faculty. Four survey rounds were required to obtain satisfactory convergence and round-to-round feedback consisted of participant comments and participant voting (in the form of an "average" list) from the previous round.

The first round established a large category list with definitions for each category. This list and definitions were refined in subsequent rounds to arrive at the final category list of 17 categories. Participants rated their own confidence in their responses with a confidence level, CL, ( $0 \le CL \le 10$ ) which was used to weight the voting.

An average list was established after each round by compiling a list of the "N" most-voted categories where N was the average number of categories in the participant lists. When two categories in the average list appeared to directly overlap, the category with the smaller number of votes was deleted and the next most voted category was added.

Variations in the average list were small after the first round and convergence was assumed after the fourth round. Tables B.l through B.4 present the average lists of rounds 1 through 4 respectively.

394

	وأحصبها كالمترافق والمراجع والمراجع والمراجع والمترافع والمتعاد والمتحد والمتحد والمتحد والمتحد والمحادي والمراجع	فبسياق والا
Category	Votes	
Safety	5	
Fuel Efficiency	4	
Purchase Price	4	
Reliability	4	
Survivability	4	
Empty Weight	3	
Exterior Noise	3	
Handling Qualities	3	
Interior Noise	3	
Maintenance Cost .	3	
Pilot Workload	3	
Range	3	
Stall Speed	3	
Maximum number of votes = 8	(no confidence levels)	

Table B.1. Round 1 Average Category List

-

The final list of categories was obtained by checking the fourth round average list against the three selection criteria as discussed in Section 4.1.2. This resulted in a final category list which differs from the fourth round average list in three respects.

 The categories of "Payload" and "Empty Weight" overlap.
Hence "Payload", having fewer votes, was deleted from the list. Table B.2. Round 2 Average Category List

--

a. - : \*

Ŧ.

÷.

.)

ŗ.

ì

ł

.

7

Category			Vot	<b>es</b>	x
Purchase Price .	• • •	• • •	54	.5	
Exterior Noise .	• • •	• • •	53	.5	
Fuel Efficiency.	• • •	• • •	49	.0	
Emissions	• • •	• • •	46	.5	
Reliability	• • •	• • •	46	.5	
Direct Operating	Cost.	• • •	44	.5	
Payload	• • •	• • •	44	.5	
Overall Pilot Wor	kload	• • •	39	.5	
Safety	• • •	• • •	39	.5	
Survivability	• • •	• • •	39	.5	
Takeoff/Landing H	erform	ance.	39	.5	
Interior Noise .	• • •	• • •	39	.0	
Range	• • •		37	.5	
Cruise Speed	• • •	• • •	30	.0	
Ride Qualities .	•••	•••	30	.0	
Static Comfort .	•••	• • •	30	.0	
Empty Weight	• • •	•••	24	.0	
Lift/Drag Ratio.	• • •	•••	22	.0	
Maximum number of votes =	63.5 (	includ	es con	fidence	levels)

(2) "Overall Pilot Workload" was changed to "Pilot Workload" for clarity, but the former's definition was retained.

Category		Votes
Direct Operating Cost	• •	52.5
Emissions	• •	52.5
Exterior Noise	• •	52.5
Fuel Efficiency	• •	52.5
Overall Pilot Workload	••	52.5
Purchase Price	• •	52.5
Reliability	• •	52.5
Safety	• •	52.5
Takeoff/Landing Performance		52.5
Empty Weight	• •	43.5
Interior Noise	• •	43.5
Ride Qualities	• •	43.5
Static Comfort	• •	43.5
Survivability	••	42.5
Cruise Speed	••	35.5
Range	• •	33.5
Payload	• •	25.5
Maximum number of votes = 52.5(includ	ies c	onfidence levels)

Table B.3. Round 3 Average Category List

6.04

ł

1

5

and the second sec

ł

н<u>с</u>

(3) The definitions of all of the categories were reviewed and refined if necessary.

The final category list (17 categories) used in the evaluation technique is presented in Table B.5.

# Table B.4. Round 4 Average Category List

Ť**\** 

1

•

······································		
Category		Votes
Cruise Speed	• • • • •	53.5
Direct Operating Cost	• • • • •	53.5
Emissions	• • • • •	53.5
Exterior Noise		53.5
Fuel Efficiency		53.5
Interior Noise	• • • • •	53.5
Overall Pilot Workloa	d	53.5
Purchase Price	• • • • •	53.5
Reliability	• • • • •	53.5
Ride Qualities	• • • • •	53.5
Safety	• • • • •	53.5
Static Comfort		53.5
Takeoff/Landing Perfo	rmance	53.5
Empty Weight		44.5
Range	• • • • •	44.5
Ceiling	• • • • •	28.0
Crashworthiness	• • • • •	28.0
Payload		24.5
Maximum number of votes = 53.5	(includes	confidence levels)

398

Category	Definition
Ceiling	Altitude at which the maximum rate of climb is .51 m/s (service ceiling).
Crashworthiness	The characteristics of an air- plane which determine the level of occupant protection in the event of a crash.
Cruise Speed	Maximum continuous cruise speed.
Direct Operating Cost	All costs directly attributable to flying and keeping an air- plane operational. All schedul- ed and unscheduled maintenance costs and fuel costs are included.
Emissions	Pollutants produced during the operation of an airplane. Does not include noise.
Empty Weight	Airplane weight without fuel, crew, and payload.
Exterior Noise	Noise perceived at ground level due to the operation of an airplane.
Fuel Efficiency	Airplane cruise efficiency measured in air-miles per pound-fuel.
Interior Noise	Noise perceived by the occu- pants of an airplane.
Pilot Workload	The amount of time, concentra- tion, and effort a pilot must devote to the safe operation of an airplane. This includes the effects of airplane handling qualities.

# Table B.5. Final Category List Used In The Bvaluation Technique

· . .-

÷٦

Ī

\*

٠

399

. ...

... . . . . . . . . .

*,* •

Ì١.

F . . .

1997 - Angler a Bar

27

×

Category	Definition
Purchase Price	The price paid for a new air- plane by the user, including avionics and equipment costs.
Range	The distance an airplane can fly without refueling while allowing for appropriate fuel reserves.
Reliability	A measure of the probability of failure of an airplane component or system.
Ride Qualities	A measure of the effects of aircraft motion on the smooth- ness and comfort of the ride experienced by the occupants.
Safety	A measure of an airplane's inherent characteristics which reduce the probability of an accident.
Static Comfort	A measure of an airplane's inherent comfort. This includes roominess, seat comfort, venti- lation, decor, ease of entry, etc.
Takeoff/Landing Performance	This parameter includes takeoff and landing speeds, field length requirements, and rates- of-climb and descent.

## Table B.5. Final Category List Used In The Evaluation Technique (Concluded) and the set of the Evaluation Technique

• 1 • 1

# B.2 CATEGORY RATING SURVEY

Ĩ,

16 1<sup>1</sup> 1

**.** 

The category rating survey was based on the Delphi Method which involves sequential rounds of written communication in which the group response from one round is fed back to participants for consideration in the next round. The 42 participants involved in this survey represent General Aviation manufacturers, user groups, university faculty, NASA centers, and project staff. A complete listing of the participants is given in Chapter 4.

1.900

Three survey rounds were required and the round-to-round feedback consisted of participant voting and comments. Participant voting was represented by "mean" and "corrected mean" category ratings and voting distributions in the form of histograms.

The mean category ratings for each airplane are simply weighted averages computed using Equation B.1.

$$\overline{R}_{j} = \frac{\frac{\sum_{i=1}^{n} (R_{ij})(CL_{i})}{\sum_{i=1}^{n} (CL_{i})}$$
(B.1)

where  $\overline{R}_j$  = the mean rating  $(0 \le \overline{R}_j \le 10)$  of category j.

n = the number of participants.

 $R_{ij}$  = the rating (0  $\leq R_{ij} \leq 10$ ) given category j by participant i.

 $CL_i$  = the confidence level (0  $\leq CL_i \leq 1$ ) of participant i.

The corrected mean category ratings for both airplanes were obtained by multiplying the mean rating by the ratio of the maximum possible rating (10) to the maximum mean rating as shown by Equation B.2.

$$\overline{R}_{cj} = (-\frac{10}{\overline{R}}) \overline{R}_{j}$$

١

where  $\overline{R}_{cj}$  = the corrected mean rating ( $0 \le \overline{R}_{cj} \le 10$ ) of category j.

 $\overline{R}_{max}$  = the maximum mean category rating ( $0 \le \overline{R}_{max} \le 10$ )  $\overline{R}_{j}$  = the mean rating ( $0 \le \overline{R}_{j} \le 10$ ) of category j.

The histograms show the number of votes each rating received in a given category as shown by Equation B.3.

$$V_{jk} = \sum_{i=1}^{n} (T_{ijk}) (CL_i)$$
(B.3)

where  $V_{ik}$  = the number of votes in category j for rating k.

n = the number of participants.

T<sub>ijk</sub> = the voting indicator of participant i in category

j for rating k (T<sub>ijk</sub> = 1 for yes, 0 for no).

CL<sub>i</sub> = the confidence level of participant i.

Note: V<sub>ij</sub> is rounded to the nearest 0.5 for histogram plotting purposes.

Figure B.1 presents the mean and corrected mean category ratings and the histograms from round 3. The round-to-round trends of the mean ratings are illustrated in Figure B.2. **3** 6

ş

۰.







Fig. B.1. Results of Round 3 of the Category Rating Survey



Fig. B.1. Results of Round 3 of the Category Rating Survey (continued)

URIGINAL PAGE IS OF POOR QUALITY

The second second state of the second s

R

•

Airplane A

. .

Airplane B

. . .

المراجع المراجع المراجع المراجع المراجع

404



4

1.15.5.

ment danst

Fig. B.1. Results of Round 3 of the Category Rating Survey (continued)

405

- P. .

4

2v 11




· · · ·

A REAL MADE AND A

- 74

<u>Airplane</u> <u>B</u>

• •

٠.,

1.1



relief of the surveyor the second sec

#### Airplane A

։ ֆել էլնապես ⊭⊭։



RATING

· ·

્યેટ્ટર્ચ્યા આવ્યો સંસર્ધાર્થ્

1



anar s e S

Fig. B.1. Results of Round 3 of the Category Rating Survey (continued)

فالله عياست مطاميعها والأطباح ويدر تعديديها والاي والمستحدة بالمنصور والمستحدة والم

RATING

407



. . . .

سلى ي

• ",

Fig. B.1. Results of Round 3 of the Category Rating Survey (continued)

هيه الم

i kana ana

ை ், தோளை

.

. •

-----

OF POOR QUALITY



.

Airplane B

Fig. B.1. Results of Round 3 of the Category Rating Survey (continued)

409

1

in a fill of an a state

Airplane A

Airplane A

E MATRIA S. M. 1996

Airplane B

Ċ,





Fig. B.1. Results of Round 3 of the Category Rating Survey (concluded)



Fig. B.2. Mean Category Rating Trends.



Fig. B.2. Mean Category Rating Trends (continued)



Fig. B.2. Mean Category Rating Trends (continued)



Fig. B.2. Mean Category Rating Trends (continued)



Fig. B.2. Mean Category Rating Trends (continued)

×.



Ī

Ĩ

Fig. B.2. Mean Category Rating Trends (continued)



Fig. B.2. Mean Category Rating Trends (continued)

1.11

ł

۰.



Fig. B.2. Mean Category Rating Trends (continued)

;



Fig. B.2. Mean Category Rating Trends (concluded)

As mentioned previously, three rounds were required to obtain satisfactory convergence. The convergence criterion used is that suggested in Reference 125 and is based on the stability of participant voting. The procedure is as follows:

- Calculate the total number of rating changes (TNC) in a category between two successive rounds. TNC<sub>1</sub> is the <u>number</u> of <u>participants</u> who changed their rating of category i between two rounds.
- (2) Calculate the total percent change (TPC)

$$TPC_{i} = \left(\frac{TNC_{i}}{n}\right) 100 \qquad (B.4)$$

where  $TPC_{i}$  = the total percent change in category i.

TNC<sub>i</sub> = the total number of rating changes in category i. n = the number of participants.

(3) If  $\text{TPC}_{i} \leq 15\%$ , then category i can be considered stable. Note: Only those participants who responded in two successive rounds can be included in this analysis.

Some difficulty was encountered in applying the convergence criterion. First, some of the participants did not respond in every round. Second, the convergence criterion applies to individual categories rather than the survey as a whole. These problems were overcome by computing the average category stability and by taking the magnitude of the changes in the mean ratings into consideration. Table B.6 presents the round-to-round category stability and the computed average stability. As shown in this table, an average stability of 18% was achieved between rounds 2 and 3. Although larger than the desired 15%, the stability was considered satisfactory due to the fact that the mean category ratings were not changing significantly.

The data reduction, histogram plotting, and stability analyses were done with an HP-9825 minicomputer. The program used computes both weighted (confidence levels included) and unweighted (confidence levels excluded) mean ratings and histograms, but only the weighted results were used in the course of the survey.

	Total Percent C	hange
	Rounds 1 to 2	Rounds 2 to 3
Category	Airplane	Airplane
		A D
Ceiling	44.44 38.46	38.46 19.23
Crashworthiness	33.33 50.00	30.77 38.46
Cruise Speed	40.74 53.85	15.38 7.69
Direct Operating Cost	44.44 42.31	11.54 11.54
Emissions	40.74 42.31	26.92 38.46
Empty Weight	40.74 50.00	11.54 15.38
Exterior Noise	44.44 34.62	19.23 11.54
Fuel Efficiency	48.15 53.85	11.54 15.38
Interior Noise	48.15 42.31	19.23 7.69
Pilot Workload	51.85 38.46	23.08 19.23
Purchase Price	51.85 53.85	11.54 23.08
Range	55.56 42.31	11.54 23.08
Reliability	40.74 50.00	19.23 11.54
Ride Qualities	37.04 42.31	15.38 26.92
Safety	25.93 30.77	7.69 7.69
Static Comfort	29.63 42.31	19.23 19.23
Takeoff/Landing Performance	33.33 42.31	15.38 7.69
Averages	42.97%	17.99%

## Table B.6. Category Rating Survey Stability

- ۴

2

ŝ

:

٩

١

- . .

The category weightings used in the evaluation technique are the weighted mean category ratings from Round 3 and are given in Chapter 4, Table 4.3.

•

,

.

rs ar", ...

۰. .

#### **B.3** CANDIDATE TECHNOLOGY IDENTIFICATION

This appendix discusses the literature search, the preliminary candidate technology selection, the final candidate technology selection, and the deleted candidate technologies.

- (1) Literature Search The literature search was conducted using the Lockheed DIALOG system. The on-line search resulted in 1655 abstracts from which 107 articles were ordered from the National Technical Information Service (NTIS). In conducting the search, nine search topics were used and are as follows:
  - (a) Aircraft Design
  - (b) Aircraft Propulsion
  - (c) Aircraft Structures
  - (d) Flight Controls
  - (e) Navigational Aids
  - (f) Avionics
  - (g) Canard Configurations
  - (h) NAVSTAR/GPS

(i) GASP (General Aviation Synthesis Program)
The actual transcripts of the search are given in Tables B.7
and B.8 for topics a, b, c, and d, e, f, g, h, respectively.
Topic (i) yielded no new articles.

(2) Preliminary Candidate Technologies - The preliminary candidate technologies (137) were selected according to the criteria given in Chapter 4 and are presented in Table 4.6.

Table B.7. Search of A craft Design, Aircraft Propulsion, and Aircraft Structures.

.n 2

. \*

Set	Items	Descruption	40		TURNOUSONS	•		
1	52062	DESIGN?	48	101	TURBUPRUP?	Set	Items	Description
2	8333	SYNTHEST?	49	396	RECIP?	1	1885	AIRFRAME?
3	153	STZING	50	535		2	49696	MATERIAL?
<u>م</u>	9910	CONFIGURATO	51	926	TURBOFAN?	3	2588	BONDING
5	67520	1-4 (DR	52 1	249	PRUPELLER?	4	15251	COMPOSIT?
, , , , , , , , , , , , , , , , , , ,	26161	AIDCOAST	53	135	PROPULSIVE	5	6048 <b>6</b>	2-4/OR
	20101		54	1	PROPELLOR?	6	23605	FLIGHT
	2133		55 3	302	40-54/0R	7	26161	AIRCRAFT
	22005		56 26	161	AIRCRAFT	8	43598	6+7
	23605	6-0 (OD	57 2	193	AIRPLANE?	9	45145	STRUCTUR?
	44234	GAUDIO	59	33	AEROPLANE?	10	1051	FRANE?
1 1 1	11155		59 23	605	FLIGHT	11	45877	9+10
	51135		60 44	23-i	56-59/DR	12	697	11+8+5
13	2331	18405 /	61135	978	SYSTEM?	13	61124	MODEL?
	10375		62 97	C09	ENGINE?	14	24628	SIMULAT?
15	449		63 2	765	POWER(W) PLANT	15	0	SYNTRII?
10	1455	11111111111111111111111111111111111111	64 8	363	P"WED(W)PLANTS	16	8381	SYNTHES?
10	3097	TANKEDO	65210	253	61-64/OR	17	83794	13-16/OR
	558	IANKER /	66	955	5540060, nD65	18	625	12-17
19	12598	SPACECRAFT	67 13	425	SOLAR?	19	5331	TRADE?
20	5984	SUPERSUNIC	68 51	428	NUCLEAR?	20	13134	OPTIM?
21	3192	HTPERSUNIC	69	593	RAMUET	21	19	18+(19+20)
44	Ŭ		70 12	593	SPACECRAFT	22	0	LARGE(W)MILITARY
23	5		71 9	575	SPACE(W)VEHICLE	23	5984	SUPERSONIC
24	11	MILIIARY (W) IRANSPORT	72	538	SPACE (W) VEHICLES	24	3192	HYPERSONIC
25	2	MILLITARY (W) TRANSPORTS	73 10	097	TUREQUET?	25	17	21-(23+21)
20	312		74	61	ION(W)ENGINE	26	582	18-(23+24)
27	65	HPY /	75 3	399	ION(W)ENGINES	27	9	LARGE (W) TRANSPORT
28	281	REMUTELY(W)PILUTED	76	367	SCRAW?	28	3097	HEAVY
29	0	HIWAL	77	0	SCRAM(W)JET	29	579	26-(27+29)
30	11639	MISSIL	78 77	841	67-77/OR	30	18680	AIRCRAFT/TI, DE
31	429	VERTICAL(W)IAKE(W)UFF	79	788	66-78	31	16172	FLIGHT/TI, DE
32	628	VERTICAL(W)TAKEUFF	80 61	124 /	ODEL?	32	389	29*(30+31)
33	37642	16-32/08	81 246	528 9	SIMULAT?			
34	303	15-33	82 6	527	79-180+81)	Prir	nt 32/9	5/1-275
35	3640	HELICUPIER?	83 5	506	82-33			
30	40	MUVERCRAFT	84	11	UEL(W)PROPERTIES	Sear	ch Tir	ne: 0.378 Prints: 275
37	5/5	SPACE (W) VEHICLE	85	10	44+83			
38	538	SPACE IN / VLHIULES	86	84	(48+50) • 83			
39	4		87 2	200	(48+50+52)+83			
40	3	GROUDD(W)EFFECF(W)VLHICLES	88 1	22	87-86			
41	16	SUPPACE (W) EPPECF (W) VEHICLE						
42	58 6007	SURFACE (W)EFFECI (W)VEHICLES	Print 4	14/5/	/1-283			
43	4527	30-42/UR 24-42	Print 8	37/5/	(1-206			
44	283							
45	10/10	AIRUKATI/DE	Search	Time	: 0.140 Prints: 489			
40	145	40744						
4/	138	44-43						

424

I WHETHER PERSON AND A 19 YO REPORTED IN THE PROPERTY OF THE PERSON AND A 19 YO REPORTED AND A 19 YO REPORTED A

# Table B.8. Search of Flight Controls, Navigational Aids, Avionics, Canard Configurations, and NAVSTAR/GPS.

Set	Items	Description	
1	26161	AIRCRAFT	57 20012 FACTLITS
2	23605	FLIGHT	50 360 (140))(CANEE),(CANEE)
3	68	STABILITY(W)AUGMENTATION	50 305 (1+2)*(54+55)*(56+57) 60105224 34.00.60
4	14	ACTIVE(W)CONTROLS	591V9234 24+30+52 60 967 68-50
5	23	ACTIVE(W)CONTROL	0V 207 00-59
6	16640	STABILITY	CI 9 33760 CI 1508 ANIONIOC
7	2020	CONTROL (W) SYSTEM	62 TOLD INCOMPLETATION
8	14746	CONTROL (W) SYSTEMS	64 ETAD DISOLAND
9	1	AUTOMATIC(W)PILOT	65 0701 (1)01.(CO.CO.CA)
10	519	AUTOMATIC(W)PILOTS	65 1497 65-55
11	7	AUTO(W)PILOT	67 7007 WEATUED
12	0	AUTO(W)PILOTS	68 40603 MATERIALD
13	100	AUTOPILOT?	69 1330 66-(67.60)
14	31673	6-13/DR	70   0.01900 + (07400)
15	6019	(1+2)+14	71 14500 ELICUT (DE
16	6052	15+3+4+5	72 16710 ALPODACT (DE
17	65	RPV?	73 1070 (71,72).co
18	53	TFR	74 76700 15510
19	54	TERRAIN(W)FOLLOWING	75 44603 11545002
20	76	DATA(W)LINK	76 669 73-(7.1+75)
21	12607	SFACECRAFT?	77 2 76+60+50
22	575	SPACE(W)VEHICLE	78 43 76+53
23	538	SPACE(W)VEHICLES	79 554 76 - (38 + 30)
24	13492	17-23/OR	80 1267 79+60+53
25	5651	16-24	81 15513 SATELLIT2
26	45145	STRUCTUR?	82 165 CANADD
27	871	FLUTTER?	83 7 CANAROS
28	13	LOAD(W)ALLEVIATION	84 80 (82+83)+(1+2)
29	5212	25~(26+27+28)	85 22 42+84
30	29	V/STOL	
31	312	VIOL	Print 80/5/1-780
32	3640	HELICUPIER?	Print 85/5/1-22
33	7500	MISSILL	
34	3199	PTPERSUNIC?	Search Time: 0.967 Prints: 802
35	2330	20-25-00	
30	2060	30-337 UK	
30	5000	25-00	
30	52002	SVNIHESID	
40	5221	TRADE2	
41	13134	OPT IM?	
42	75053	38-11/0R	
43	040	42.37	
44	61124	MODEL?	Set Items Description
45	24628	SIMULATO	0 GLOBAL(W)POSITIONIG(W)SYSTEM
46	1455	FIGHTER?	2 34 GPS
47	2	MILITARY(W)TRANSPORTS	3 BI GLOBAL (W) POSITIONING (W) SYSTEM
48	3097	HEAVY	4 29 NAVSTAR
49	40	HOVERCRAFT?	5 9 2+3+4
50	173	SURFACE(W)EFFLCT	6 24 2•3
51	11	MILITARY (N) TPANSPORT	r 44 4+6
52	79717	44-51/DR	o 89 3+4
53	500	43-52	Print R/E/1-00
54	4746	NAVIGATION	Frint 8/5/1-89
55	4203	POSITION?	Seanch Tumot O 110 D
56	7806	AID?	Search time: 0,112 Prints: 89

?

- (3) Final Candidate Technologies The final candidate technologies (56) were selected according to the criteria given in Chapter 4 and are presented in Table 4.7.
- (4) Deleted Technology Discussion The reasons for deleting technologies from the preliminary candidate technology list in generating the final candidate technology list are now discussed for each deleted technology by area.

#### I. AERODYNAMICS

- Variable Geometry Winglets
   Deleted because it appears to be more of an idea than a technology. Data are lacking.
- Supercritical airfoils
   Deleted because it is not applicable to the 6-passenger or
   commuter airplanes. It does appear very promising for high
   subsonic cruise applications and is currently used on a few
   airplanes.

#### II. AIRCRAFT SYSTEMS

白素

- Microwave Anti-Icing Deleted because it appears to be an idea rather than a technology. Data are lacking.
- Lithium Hydroxide/Hydrogen Peroxide Batteries
   Research on advanced batteries is on-going, but data are
   somewhat limited. If the potential of these batteries is
   realized, they will certainly benefit general aviation
   airplanes. Unless an electric-powered airplane becomes
   feasible, impacts may be expected to be minimal.
- Air-Cycle Environmental Systems Deleted due to a lack of data. Also, improvements in most environmental systems can be expected to gain acceptance.
- Improved Lead-Acid Batteries Lead-acid batteries are an established technology. Improvements in this area can be expected to gain acceptance of their own accord.
- Variable Cycle Environmental Systems These systems do not appear to be well suited to general aviation airplanes.

\*Accurate Fuel Monitoring and Management This is a result of the application of various technologies. Improvements in this area do not appear to be cost-effective for most general aviation applications.

\*High-Speed Brushless Alternator This was combined with AC electrical systems due to technology level considerations.

Single Unit Starter/Generator Deleted due to technology level considerations. Also, viable developments in this area are expected to gain acceptance of their own accord. Although used on some turbine engines, gearing for reciprocating engines is expected to be heavy.

•Air Bearings

Deleted due to technology level considerations. However, this technology is considered to be the most important of several in order to develop improved turbochargers.

#### **III. COMPUTATIONAL METHODS**

This entire group was deleted for two reasons. First, all are second order technologies. Second, with the exception of CADCAM (which appears a long way off for general aviation), general aviation is increasingly using computational methods at present. Specific developments which must be realized are those codes with an ability to handle (1) canards where deformed wakes may be accounted for, (2) 3D analytical methods for advanced low-speed, medium-speed, and natural laminar flow wings, and (3) wing-spoiler-flap configurations.

#### **IV. CRASHWORTHINESS**

'Foam-Filled Fuel Tanks Deleted primarily due to a lack of data.

•Anti-Misting Fuel Treatment. Deleted primarily due to a lack of data.

•Frangible Fuel Fittings Current systems are quite expensive. If the costs could be reduced, widespread use could be expected. Note that these fittings are used in the Burst/Tear Resistant Fuel Tanks which are discussed in Section 3.5.4.4.

V. FLIGHT CONTROL SYSTEMS

Winglets for Lateral-Directional Control This is an unlikely application of winglets except for canard configurations. Deleted primarily because winglets are included in the aerodynamics technology area.  Force-Stick Controllers
 Deleted due to technology level considerations, and is actually implied in the technologies of fly-by-wire and fly-by-light.

•Digital Automatic Flight Controls Deleted because these technologies are already filtering into general sviation.

•Stick Shaker/Pusher and Stabilizer/Elevator Spoilers for Stall Prevention These two technologies were combined into one and are evaluated (Active Stall Prevention).

#### IV. INFORMATION SYSTEMS

•Flush Antennas This was deleted due to technology level considerations.

•Single-Function and Time Shared CRT Displays These were combined into "CRT Displays" and are evaluated.

•Warning Annunciators and Airplane Health/Diagnostic Systems These were combined into "Systems Status Displays" and are evaluated.

•Total Panel-Mounted Avionics Deleted because the trend is already towards panel-mounted avionics.

•Active Outside Imaging While this is possible, it will not be feasible for general aviation airplanes in the foreseeable future.

•Fluidic Shed Vortex Airspeed Sensors Deleted due to technology level considerations.

'Fluidic Rate Sensors Deleted due to technology level considerations.

•Multiplexing and ARINC-Type Broadcast Hierarchy Both were deleted because they represent design philosophies more than technologies. Either is applicable to integrated avionics packages. 1

ł

ţ

•Piezo-Resistive Pressure Transducers Deleted due to technology level considerations. • Liquid Crystal Displays

Deleted due to technology level considerations and the fact that non-mechanical displays in the near future will probably be CRT's in most applications. See the discussion of displays in Section 4.5.6.2.

#### • Flat Panel CRT Displays

Deleted because flat CRT's appear unlikely in general aviation applications in the foreseeable future. See the discussion of displays in Section 4.5.6.2.

- Touch Sensitive CRT Deleted due to technology level considerations. This could be a component of a number of advanced avionics systems, and was an integral part of PCAAS.
- Weather Radar and Radar Altimeter Deleted because they are already gaining acceptance in general aviation.
- Alternate Weather Detection Only one alternate weather detection device (aside from radar) was found and it is currently available (Ryan Stormscope).
- On-Board Computing Capability This was deleted because it is an integral part of many other advanced technologies such as integrated avionics and displays, systems management displays, etc.
- 3-Axis Magnetometer Acceleration Sensor Deleted due to technology level considerations. See related work discussed in Section 4.5.6.6 and Figure 4.21.
- Improved Stall Warning This is more of an idea than a specific technology. Active stall prevention is discussed in Section 4.5.5.14.

#### VII. MATERIALS/PROCESSES

- All Processes These were deleted because they are second order technologies.
- Metal/Metal Bonding This technology is already finding substantial acceptance.
- Corrosion Resistant Coatings Deleted due to technology level considerations.
- Honeycomb Core/Composite Skin Panels Considered as a possible use of all composite materials.

#### VIII. NAVIGATION CON . JPTS

.....

·VOR/DME RNAV This technology is already in use.

·Scanning VOR/DME RNAV

This extension of VOR/DME RNAV does not really change the fundamental technology of VOR navigation although it does represent an improvement in more efficient use and automation. Work done at Ames appears in Section 4.5.6.6.

•VLF NAVCOM Deleted because it is currently available.

•Differential OMEGA Delet 1 due to technology considerations. OMEGA is evaluated in Section 4.5.7.6.

• Inertial Smoothing This concept is not really applicable to general aviation airplanes although inertial navigation is evaluated. Ames has done work in this area for VTOL aircraft.

IX. NOISE

Noise Absorbing Materials
 No breakthroughs were found in this area.

• Improved Mufflers This is a concept rather than a technology. No specific examples were found.

•Variable Engine/Propeller Gearing This is a concept rather than a technology. No specific examples were found.

### X. PROPULSION

•Advanced Reciprocating Engine This was divided into the two technologies described as an Advanced Stratified Charge Reciprocating Engine and an Advanced High Compression Ratio Lean Burn Engine. Both are assumed to have variable ignition timing and fuel injection.

•Auto Engine Conversions This was deleted because it has limited application to general aviation although it is being examined for agricultural airplanes.

- Stratified Charge, Variable Timing, Electronic Ignition, Automatic Mixture Control, Lean Burn Combustion Chamber, Density Compensating Fuel Injection, and Total Microprocessor Engine Control.
   These technologies were either deleted due to technology level considerations or were considered to be included in one or more of the advanced technology engines.
- •Cooled Turbine Blades This technology is already being pursued.
- Ceramic Turbines
   Deleted due to technology level considerations. Ceramic components are considered a part of certain advanc^d engines.

Composite Propellers
This is included in quiet, efficient propeller (cohnology.

•Torsionally (aeroelastic) Tailored Propeller Blades This is more a concept than a technology.

•Single Lever Throttle/Mixture Control This has already been developed for general aviation.

•QCGAT Engine and Propfans Both of these technologies were deleted because they are not applicable to either the 6-passenger or commuter airplanes. However, both are very promising and are discussed in Section 4.6.

'Variable Bypass Turbofan and Variable Pitch Fan (turbofan) Neither one is applicable to the airplanes in question.

•Efficient Propeller Technology This was combined with Quiet Propeller Technology.

# B.4 <u>PESSIMISTIC</u>, <u>LIFELY</u>, <u>OPTIMISTIC</u>, <u>AND</u> <u>EXPECTED</u> (<u>PLOE</u>) <u>FIGURE</u> <u>OF MERIT STUDIES</u>

The rationale and methodology of the PLOE studies are presented in Chapter 4. The purpose of this appendix is to supply basic documentation for the analysis.

Tables B.9 through B.11 present the actual technology evaluations for pessimistic, likely, and optimistic relative benefits respectively. Figures B.3 and B.4 illustrate the PLOE figures of merit for Airplanes A and B respectively.

When reviewing the evaluations presented here, the reader is reminded that they are used only to test the evaluation technique and do not (in general) correspond to the final technology evaluation as far as the overall ordering of the technologies is concerned. Also, Efficient Propeller Technology and Quiet Propeller Technology are treated separately in the PLOE studies but were combined into a single technology in the final technology evaluation. Table B.9. Pessimistic Technology Evaluation

+++TABULAR RESULTS FOR ALL TECHNOLOGIES FOLLOW:

- **-**

CRSH CRUS EMIS EMIY EXTR FUEL INTR PILT PRCH RELI RIDE STAT TOLD FIG OF CEIL WINY SPED DOC SION WGNT NOIS EFFC NOIS WRK'. PRCE RGE BLTY WLTY SFTY CRFT PERF MERIT CATEGORY WEIGHTS 5.43 A.82 7.92 7.91 1.61 4.34 4.29 8.16 7.04 7.35 8.51 7.29 8.85 6.11 9.42 6.24 6.87

AIRPLANE A

. i ni

~

0.5 102.944 GATE ENGINE 0. ٥. 1.0 1.5 1.0 3.0 ٥. 2.0 0. 1.0 -1.0 1.0 2.0 1.0 2.9 0. 74.300 ο. 1.0 1.0 2.0 2.0 1.0 1.0 1.0 ο. 0. 1.0 0.5 0. STRT CHG ROT COMB ENGINE 0. 1.0 ٥. 1.0 KEVLAR COMPOSITES 0. 1.0 1.5 0. 2.0 ο. 2.0 ٥. 0. -1.0 2.0 0. 50,840 ٥. 0. 0. ۵. D. 47.408 ADVANCED DIESEL ENGINE 0. 0. ٥. 1.0 3.0 0. 1.0 1.0 1.0 0. -1.0 2.0 0.5 ٥. 0.5 0. 0. QUIET PROP TECHNOLOGY 0. 0. 0. 0. 0. 0.5 3.0 0. 3.0 0. 0. 0. 0.5 0. 0.5 0. 0. 45.280 GRAPHITE COMPOSITES 0. 1.0 1.5 0. 3.0 ٥. 2.0 0. 0. -3.0 2.0 ρ. ٥. 0. 40,845 0.5 ٥. ٥. STRAT CHG RECIP ENGINE 0. Ο. 1.0 3.0 0.5 0. 1.0 0. 0. ٥. 1.0 0.5 0. 0. 39.479 C. 0.5 ٥. HCRLB RECIP ENGINE 0. 1.0 3.0 0.5 0. 1.0 0. 0. 0. 1.0 0.5 0.5 39.479 0. 0. ٥. 0. ٥. EFFICIENT PROP TCHNLGY 0. 0. 0.5 0.5 0. 0.5 0. 0.5 0. 0. ٥. 0.5 1.0 0. 0.5 0. 1.0 38.247 33.749 FIBERGLASS COMPOSITES Ο. Ο. 1.0 0.5 0. 0.5 0. 1.0 0. 0. 0.5 1.0 0. ٥. ٥. 0. 0. SPOILERS 0.5 0. 1.0 0.5 0. ٥. ٥. 0.5 0. ٥. -0.5 0.5 ٥. 1.0 ٥. 0. 0. 24.165 IMPROVED TURBOCHARGING 20.406 2.0 0. 0. 0. 0. α. ٥. 1.0 0. 0. -1.5 1.0 0. 0. 0. 0. 1.0 SEPARATE SFC TECHNOLOGY 16.771 0. 0. ٥. 0. 0. 0. Ο. 0. 0. 1.0 0. 0. ٥. ٥. 1.0 0. 0. 16,171 LIQUID COOLING 0. Ο. 0.5 0. 2.0 -1.5 1.0 1.0 1.0 ٥. -0.5 e. -0.5 0. 0.5 ٥. 0. LOAD LIMITING SEATS 0. n. 0. 0. 0. 14,210 σ. 3.0 ٥. 0. 0. 0. 0. ٥. ٥. 0. -1-0 0. NAT LAM FLOW AIRFOILS 0. 0.5 ο. 13,916 0.5 0.5 0. 0. 1.0 0. 0. -1.0 0.5 ٥. ٥. 0. 0. 0. 0. ٥. 0. 0. SYS STATUS DSP 0. 0. 0. -0.5 0. 2.0 -1.0 0. ٥. 0. 1.0 0. 0. 11.652 ACTIVE LAMINAR FLOW CTL -2.5 0. 3.0 11.317 1.5 0. 1.5 1.0 0. 0. 0. -3.0 3.0 -3.0 ٥. 0. 0. 0. IMPROVED RESTRAINTS 0. 3.0 0. 0. 0. 0. 0. 0. 0. 0. -1.5 11.088 0. 0. 0. 0. 0. 0. 0. 0. INTEG AVIONICS AND DSPVS 0. 0. 0. ٥. 0. 0. 0. 2.0 -2.0 0., 0. ٥. 1.0 0. 0. 7.090 LOW/MEDM SPEED AIRFOILS 0. 0. э. ٥. 0. 0.5 0. 0.5 0. 0. -1.0 0. 0. 0. 0. 0\_ 1.0 4.608

and the second second

e.

٩,

FOWLER FLAPS	0.	0.	0. 0	. ð.	-0.5	٥.	0.5	0.	00.5	0. 0.	0.5	0.	0.	0.5	4.143
ENERGY ABSCRBING FLOOR	n.	0.5	0. 0	. 0,	. 0.	٥.	0.	0.	0. 0.	0. 0.	0.	0.	C.	Ο.	3.409
SGL LEVER THRUST/DRAG CT	0.	0.	0. 0	. 0.	. 0.	ο.	0.	0.	1.0 -0.5	ð. Ø.	0.	0.	0.	0.	3.092
AC ELECTRICAL SYSTEMS	0.	0.	0. 0	. 0.	. 0.5	٥.	0.	0.	0. 0.	0. 0.	0.	٥.	0.	0.	2.170
INTEG LOW-COST WG LVLER	0.	0.	01	.0 0.	-0.5	0.	0.	0.	2.0 -1.0	01.	5 0.	5.0	0.	0.	1.670
ANTI-ICING SFC COATINGS	0.	0.	01	.0 0.	0.5	٥.	0.	0.	01.5	00.	5 0.	3.0	0.	0.	0.990
ACTIVE STALL PREVENTION	0.	0.	01	.0 0.	-0.5	0.	0.	0.	1.0 -2.0	01.	0 0.	3.0	0.	٥.	-0.345
DATA LINKS	G.	0.	01	.0 0.	. 0.	0.	0.	0.	2.0 -1.0	01.	0 0.	1.0	0.	0.	-1.155
LOW-DRAG SFC COATINGS	0.	0.	0.5 0	. 0.	1.0	0.	0.5	0.	01.0	0.5 0.	0.	0.	0.	0.	-1,17!
MICROWAVE LANDING SYSTEM	0.	0.	0. 0	. 0.	. 0.	0.	0.	0.	-1.5 0.	0. 0.	0.	1.0	0.	0.	-1.602
IMPR STLL/SPNAERO TLRG	0.	0.	-1.0 -0	.5 0.	. o.	0.	-0.5	٥.	0. 0.	0. 0	0.	1.5	0.	0.	-1,821
BRST/TEAR RESIS FUEL TKS	0.	1.5	0. 0	. 0.	1.0	ο.	0.	0.	01.0	0. 0.	0.	0.	0.	0.	-2.628
LEADING EDGE DEVICES	0.5	٥.	1.0 0	. 0.	0.5	ο.	0.5	0.	01.5	0.5 -1.	5 1.0	0.	0.	0.	-3.753
NAVSTAR/GPS	0.	0.	00	.5 0.	. 0.	0.	0.	0.	0.5 -0.5	0. 0.	0.	0.	0.	0.	-4.536
CRT DISPLAYS	0.	0.	o. o	. 0.	. 0.	0.	0.	0.	1.0 -1.5	0. 0.	. Q.	0.	0.	0.	-5.423
DIRECT LIFT CONTROL	0.	0.	0. 0	. 0.	1.0	0.	0.	0.	1.0 -1.0	01.	0 0.	0.5	0.	0.5	-6.212
LORAN C	0.	0.	0. 0	. 0.	. 0.	0.	0.	0.	01.0	0. 0.	0.	0.	0.	Đ.	-8.515
ACT CTLS FOR RLX STBLTY	1.0	0.	1.0 0	. 0.	. 2.0	0.	1.5	0.	03.0	1.0 -3.	.0 0.	0.	0.	0.	-10.550
WINGLETS	0.5	0.	0. 0	.5 0.	2.0	0.	0.5	0.	02.0	0.5 0.	0.	Ô.	0.	0.	-11.319
LOW-LEVEL PRESSURIZATION	0.	0.	00	.5 0	1.0	0.	0.	2.0	01.0	01.		0.	0.	0.	-11.588
OMEGA	0.	0.	0. 0	. 0	. 0.	0.	0.	0.	01.5	0. 0.	0.	0.	0.	0.	+12.773
INTEGRATED YAW DAMPER	0.	0.	00	.5 0.	. 0.	0.	0.	0.	1.0 -2.5	01.	5 3.0	0.	0.	0.	-12.842
DUCTED PROPULSORS	0.	0.	-2.0 -1	.0 0	1.0	3.0	-2.0	3.0	01.0	-2.0 0.	0.	0.5	0.	1.5	-18.505
FLUIDIC AUTO FLT CTL SYS	0.	0.	01	.0 0	1.0	0.	0.	0.	0. 1.0	02.	0 0.	0.	0.	0.	-21,439
LASER GYROS	0.	0.	03	.0 0.	. 0.5	0.	0.	0.	03.3	0. 2.	0 0.	0.	0.	0.	-29.385
NICRO HUD	0.	0.	01	.0 0.	. 0.	0.	0.	0.	1.0 -2.0	02.	0 0.	0.5	0.	0.	-30.584
FINER OPTICS	0.	0.	01	<b>ο</b> υ.	0.5	٥.	0.	0.	02.0	00.	5 0.	0.	0.	0.	-31,533
ACT RIDE SM	0.	0.	01	.0 0	1.0	0.	0.	٥.	03.0	02.	0 3.0	0.	0.	0.	-37.169
MIR .	0.	0.	01	.5 0.	-1-5	0.	0.	0.	2.0 -3.0	02.	0 0.	1.0	٥.	0.	+37.502
NIRFET SIDE FORCE CTI	0.	0.	01	.0 0	-1.0	0.	-1.0	0.	-1.0 -1.0	01	0 0.	0.	0.	1.0	-38.247
wanter wave rente to the	~.														

Table B.9. Pessimistic Technology Evaluation (continued)

x

es.

1

1

÷ .

.

.

434

die alle

79

...

OPPLER NAVIGATION	0.	0.	٥.	-2.0	0.	-1.0	٥.	0.	0.	1.0	-3.0	0.	-2.0	0.	0.5	0.	0.	-51.345
NERTIAL NAVIGATION	0.	0.	٥.	-5.0	٥.	-1.0	0.	0.	0.	1.0	-3.0	0.	-2.0	0.	0.5	0.	0.	-51.345
CT GUST ALLV	0.	0.	0.	-1.5	0.	0.	0.	0.	0.	0.	-3-0	0.	-2.5	1.0	0.	0.	0.	-53.429
CT FLUTTER SUPRESSION	0.	0.	0.	-2.5	0.	1.0	0.	0.5	0.	0.	-3.0	0.	-3.0	0.	0.	0.	0.	-63,453
LY-BY-LIGHT	0.	۰.	ο.	-2.0	٥.	-1.0	0.	0.	0.	<b>C.</b>	-3.0	0.	-3.0	0.	0.	0.	0.	-72,259
LY-BY-WIRE	0.	0.	0.	-2.5	0.	-1.5	0.	0.	0.	0.	-3.0	σ.	-3.0	0.	0.	0.	0.	-78.382

-----

Table B.9. Pessimistic Technology Evaluation (continued)

- · · ·

- -

•

.

ł

Table B.9. Pessimistic Technology Evaluation (continued)

AIRPLANE B +++TABULAR RESULTS FOR ALL TECHNOLOGIES FOLLOW: FIG OF MERIT CRSH CRUS EMIS EMIY EXTR FUEL INTR PILT PACH RELI RIDE STAT TOLD CEIL WINY SPED DOC SION WONT NOIS EFFC NOIS WARL PACE RGE BLIY ALTY SFIY CMFT PERF CATEGORY WEIGHTS 5.45 7.33 7.48 9.45 2.16 5.66 5.41 8.61 7.45 7.00 7.76 7.18 9.55 7.24 9.54 6.71 7.41 56.784 KEVLAR COMPOSITES 2.0 0. 2.0 0. 0. -1.0 2.0 0. 0. 0. 0. 0. ٥. 0. 1.0 1.5 0. 50.969 QUIET PROP TECHNOLOGY ٥. 0. 0. 0. 0. 0.5 3.0 0. 3.0 c. 0. 0. 0.5 0. 0.5 0. 0. 3.0 0. 49.642 GRAPHITE COMPOSITES 0.5 0. 1.0 1.5 0. 2.0 0. 0. -3.0 5.0 0. ٥. 0. 0. 0. 44.864 GATE ENGINE 0. ٥. 0. 1.0 0. ٥. 0. 0. 0. ٥. 2.0 0. 1.0 0. 1.0 0. 0. 40.914 0.5 EFFICIENT PROP TCHNLGY ٥. 0. 0.5 0.5 0. 0.5 0. 0.5 0. 0. 0. 0.5 1.0 ٥. 0. 1.0 34.702 ٥. 0. FIBERGLASS COMPOSITES 0. 0. 1.0 0.5 0. 0.5 0. 1.0 ٥. ٥. 0.5 1.0 0. ٥. 0. 18.821 0.5 0. -0.5 0.5 0. 0.5 0. SPOILERS 0.5 0. 0.5 0.5 0. 0. 0. 0. ٥. 0. 16.542 0. ٥. 0. 0. 0. 0. 0. 1.0 0. ٥. 0. 0. 1.0 0. ٥. SEPARATE SEC TECHNOLOGY 0. 0. 0. 15.806 0. ٥. LOW/MEDM SPEED AIRFOILS ٥. 0. 0.5 0. 0.5 0. 1.0 C. ٥. -1.0 0. ٥. 1.0 0. -1.0 0.5 ٥. ٥. 0. 0. 0. 15.625 0.5 0.5 0. 0. 0. 1.0 0. 0. NAT LAM FLOW AIRFOILS 0.5 0. 0. 0. ٥. 0. ٥. 0. ٥. -1.0 0. 15.292 3.0 0. 0. 0. ο. 0. LOAD LIMITING SEATS 0. 0. 11.939 0. -1.5 3.0 ٥. 0. 0. 0. 0. 0. 0. 0. 0. 0. ٥. 0. 0. IMPROVED RESTRAINTS 0. 11.068 0. -1.5 0. 3.0 **C.** 0. -3.0 3.0 -3.0 0. 0. 0. 0. ACTIVE LAMINAR FLOW CTL 1.5 1.5 0.5 0. 0. 2.0 -1.0 0. 0. 0. 1.0 0. 11.053 -0.5 0. 0. 0. 0. 0. SYS STATUS DSP 0. 0. 0. 8.018 ٥. 0. 0. 2.0 -2.0 0. 0. ٥. 1.0 0. 0. 0. 0. 0. 0. 0. INTEG AVIONICS AND DSPYS 0. 7.333 0. 0. 0. 0. 0. 0. 0. ٥. ٥. 0. 0. 1.0 0. 0. 0. Ο. ENERGY ABSORBING FLOOR 0. 4.341 1.0 0. 0. 0. 0.5 0. ٥. 1.0 ٥. -2.O 0. 0. ANTI-ICING SFC COATINGS 0. 0. ٥. 0.5 0. 0. 1.0 -0.5 0. ٥. 0. 0. ٥. 3.117 0. 0. 0. 0. 0. Ο. ٥. 0. SGL LEVER THRUST/DRAG CT 0. 2.829 0. 0.5 0. 0. 0. 0. 0. 0. 0. 0. 0. 0. 0. AC ELECTRICAL SYSTEMS 0. 0. 0. 0. 2.142 1.0 -1.5 0. 0. ACTIVE STALL PREVENTION 0. -1.0 0. -0.5 0. 0. 0. -1.0 3.0 0. 0. 0. 0. 0.5 0. 0. -1.0 0.5 -1.5 1.0 0. 0. 0. 0.425 0. -0.5 0. LEADING EDGE DEVICES 0.5 0. 1.0 0.

. .

EV-

٠

, in the second se

Table B.9. Pessimistic Technology Evaluation (concluded)

INTEGRATED YAW DAMPER 0. ο. ٥. 0. 0. ٥. 0. 0. 0. 0. 0. 0. 0. ٥. 0. 0. 0. ٥. PICPOWAVE LANDING SYSTEM 0. 0. 0. ٥. ٥. ٥. 0. -1.5 0. -0.953 0. 0. 0. 0. 0\_ ٥. 1.0 0. INTEG LOW-COST WG LVLER 0. 0. -1.0 0. -0.5 0. 0. ٥. 2.0 -1.0 -1.277 ٥. 0. -1.5 0. 5.0 ٥. 0. LOW-DRAG SFC COATINGS 0. ٥. 0.5 0. 0. -1.0 0. 0.5 0. 0. -1.0 0.5 ٥. 0. 0. 0. 0. -1.785 IMPR STLL/SPN--XERO TLRG ٥. ٥. -1.0 -0.5 0. 0. ٥. -0.5 0. ٥. 0. ٥. 0. ٥. 1.5 0. 0. -2.193 DATA LINKS 0. ο. ٥. -1.0 Ο. ٥. Ο. ٥. 0. 2.0 -1.0 0. -1.0 ٥. 1.0 0. 0. -3.219 CRT DISPLAYS 0. ٥. 0. C. 0. ٥. 0. ٥. 0. 1.0 -1.5 0. 0. 0. ٥. 0. ٥. -4.644 NAVSTAR/GPS 0. ο. Ο. -0.5 0. 0. э. ٥. 0. 0.5 -0.5 ٥. ٥. ٥. 0. 0. 0. -5.107 DIRECT LIFT CONTROL ٥. Ο. ο. ٥. Ο. -1.0 Ο. ٥. 0. 1.0 -1.0 0. -1.0 ٥. 0.5 0. 0.5 -7.492 ACT CTLS FOR RLX STBLTY 1.0 0. 1.0 0. 0. 2.0 0. 1.5 ٥. ٥. -3.0 1.0 -3-0 ٥. 0. ٥. 0. -7.593 LORAN C 0. ٥. ο. с. Ο. 0. 0. ٥. ٥. ٥. -1.0 ٥. 0. ٥. 0. 0. 0. -7.761 BRST/TEAR RESIS FUEL TKS 0. -2.0 -1.0 -8-076 0. 1.5 0. 0. 0. ٥. 0. ٥. ٥. ٥. 0. 0. 0. 0. WINGLETS 0.5 ٥. 0. 0.5 0. э. 0.5 0. -2.0 -11.494 -2.0 0. 0.5 ٥. ٥. 0. 0. 0. OMEGA C. 0. 0. 0. ٥. ٥. ٥. -1.5 0. 0. 0. ٥. ٥. ٥. 0. ٥. 0. -11.641 LOW-LEVEL PRESSURIZATION 0. ٥. 0. -0.5 ٥. -1.0 0. 0. 2.0 -1.0 -12.787 0. 0. -1.0 0. 0. 0. 0. BUCTED PROPULSORS ٥. Ο. -2.0 -1.0 Ο. -1.0 3.0 -2.0 3.0 -1.0 ٥. -14.926 ٥. -2.0 ٥. 0.5 0. 1.5 FLUIDIC AUTO FLT CTL SYS 0. ٥. 0. -1.0 0. -1.0 0. ٥. 1.0 0. 0. -2.0 ٥. 0. 0. -26.441 0. 0. LASER GYPOS 0. 0. 0. - 3.0 0. 0.5 ٥. 0. -29.714 0. -3.0 0. 0. 0. 2.0 ٥. 0. 0. FIBER OPTICS 0. 0. 0. -1.0 0. 0. ٥. 0. 0. -2.0 -29.747 0... 0. -0.5 0. 0. 0. 0. MICRO HUD 0. 0. ٥. -1.0 0. ٥. Ο. ٥. 0. 1.0 -2.0 -32.297 0. -2.0 0.5 0. ٥. 0. ACT RIDE SM 0. 0. 0. -1.0 0. -1.0 0. ٥. -3.0 -35.771 0. 0. 0. -2.0 3.0 0. 0. 0. DIRECT SIDE FORCE CTL ٥. 0. -1.0 0. -1.0 ٥. ٥. -1.0 0. -1.0 -1.0 0. -1.0 0. ٥. 0. 1.0 -40.413 NUD 0. 0. ٥. -1.5 Ο. -1.5 0. 0. 0. 2.0 -3.0 -41\_499 0. -2.0 ٥. 1.0 0. 0. ACT GUST ALLV 0. 0. 0. -1.5 ٥. 0. 0. 0. 0. 0. -3.0 ٥. -2.5 1.0 0. 0. -56.089 ٥. BOPPLER NAVIGATION 0. σ. 0. -2.0 0. -1.0 ο. **c.** 0. 1.0 -3.0 0. -2.0 -55.144 ٥. 0.5 0. 0. INERTIAL NAVIGATION 0. ٥. 0. -2.0 0. -1.0 ٥. ٥. ٥. 1.0 -3.0 -55-146 0. -2.0 0. 0.5 0. 0. ACT FLUTTER SUPRESSION 0. 0. 0. -2.5 0. 1.0 0. 0.5 0. ٥. -3.0 ٥. -3.0 0. -45.592 ٥. 0. 0. FLY-BY-LIGHT 0. ο. 0. -2.0 0. -1.0 0. 0. 0. 0. -3.0 0. -3.0 0. 0. 0. 0. -75.483 FLY-BY-WIRE 0. 0. 0. -2.5 0. -1.5 0. 0. 0. 0. -3.0 0. -3.0 0. 0. 0. -84.037 D.

n i hen sin eine einer verste felse soften i frauten die besteren ige in die wertlichten under gegenzenten stereiter ige

• - -

FUNR QUALITY

1

1

.

5

,

î.

ł

î.

1

- N - 125

∢

Table B.10. Likely Technology Evaluation

\*\*\*TABULAR RESULTS FOR ALL TECHNOLOGIES FOLLOW:

· • ·

.

CRSH CRUS ENTS ENTY EXTR FUEL INTR PILT PRCH RELI BIBE STAT TOLD FIG OF CEIL WINY SPED DOC SION WONT NOIS EFFC NOIS WAKL PACE RGE BLTY BLTY SFTY CHFT PERF REBIT CATEGORY WEIGHTS 5.43 6.82 7.92 7.91 1.61 4.34 4.29 8.16 7.04 7.35 8.51 7.29 8.85 6.11 9.42 6.24 6.87

AIRPLANE A

GATE ENGINE	0.5	0.	2.0	2. O	1.0	3.0	٥.	2.0	0.	1.0	0.	1.0	2.0	1.0	2.0	0.	0.5	125.048
STRT CHG ROT COMB ENGINE	<b>0.</b>	0.	1.5	1.0	5.0	2.0	1.0	1.0	1.0	٥.	0.	1.0	1.0	0.	0.5	1.0	0.5	81.498
KEVLAR COMPOSITES	0.5	0.	1.0	2.0	0.	5.0	0.	2.0	0.	0.	0.	2.0	0.	0.	D.	0.	0.	66.023
GRAPHITE COMPOSITES	1.0	0.	1.5	2. Q	0.	3.0	0.	2.5	0.	0.	-2-0	2.0	0.	0.	0.	0.	0.	64.086
ADVANCED DIESEL ENGINE	0.	0.	0.	1.5	3.0	0.5	1.0	1.5	1.0	0.	-1.0	2.0	1.0	0.	0.5	0.	0.	42.037
SPOILERS	1.0	0.	2.0	1.0	0.	1.0	0.	1.0	0.	0.	٥.	1.0	0.	2.0	9.	0.	0.	61.186
STRAT CHE RECIP ENGINE	0.	0	0.5	1.5	3.0	0.5	0.	1.5	0.	ο.	0.	1.0	1.0	0.	0.5	0.	0.	55.898
HCRLB RECIP ENGINE	0.	0.	0.5	1.5	3.0	0.5	ο.	1.5	0.	0.	0.	1.0	1.0	0.	0.5	0.	0.	55.898
QUIET PROP TECHNOLOGY	0.	0.	0.	0.	0.	0.5	3.0	0.	3.0	0.	0.	0.	1.0	0.	0.5	0.	0.	49.707
EFFICIENT PROP TCHNLGY	0.	0.	0.5	0.5	0.	0.5	0.	1.0	0.	0.	0.	1.0	1.0	0.	0.5	0.	1.0	45.970
FIBERGLASS COMPOSITES	0.	0.	1.0	1.0	0.	1.0	٥.	1.0	0.	٥.	1.0	1.0	0.	0.	0.	0.	0.	44.130
IMPROVED TURBOCHARGING	3.0	0.	1.0	0.5	0.	0.	0.	1.0	0.	0.	-1.0	1.0	0.	0.	0.	0.	1.0	41.968
NAT LAR FLOW AIRFOILS	1.0	0.	1.0	1.5	0.	0.	٥.	1.5	0.	0.	-0.5	1.0	0.	0.	0.	0.	0.	40.478
LIQUID COOLING	0.	0.	0.5	1.0	2.0	-1.0	1.0	1.0	1.0	0.	0.	0.5	0.	0.	0.5	0.	0.	38.576
LEADING EDGE DEVICES	1.0	0.	1.0	0.5	0.	0.	0.	1.0	0.	0.	-1.0	1.0	-1.0	1.5	0.5	0.	0.	29.238
LOW/MEDK SPEED AIRFOILS	0.	0.	0.	0.5	0.	1.0	0.	1.0	0.	0.	-1.0	1.0	Q.	<b>0.</b>	0.	0.	2.0	28.973
FOWLER FLAPS	0.	0.	0.5	0.5	٥.	0.	0.	0.5	0.	٥.	0.	0.5	0.	1.0	0.	٥.	1.0	28.622
INTEG AVIONICS AND PSPYS	0.	0.	0.	0.5	0.	0.5	0.	0.	0.	2.0	-1.0	0.	٥.	0.	1.5	0.	0.	25.439
ACTIVE LAMINAR FLOW CTL	2.0	٥.	2.0	1.5	0.	-2.0	0.	3.0	0.	0.	-3.0	3.0	-3.0	C.	0.	0.	0.	24.115
LOAD LIMITING SEATS	0.	3.0	0.	0.	0.	0.5	0.	0.	0.	0.	0.	0.	0.	0.	0.	-0.5	0.	19.502
WINGLETS	1.0	0.	0.	1.0	0.	-1.0	0.	1.0	0.	D.	-1.0	1.0	0.	0.	0.	0.	0.5	17.344

11

438

و لاد و

17

#### 

SEPARATE SFC TECHNOLOGY	0.	0.	0.	0.	٥.	0.5	Ο.	0.	0.	1.0 0.	0.	0.	٥.	1.0	0.	0.	18.941
CRT DISPLAYS	0.	0.	0.	0.	0.	0.	ο.	0.	0.	1.5 -1.0	0.	1.0	٥.	0.5	0.	0.	16.073
SYS STATUS DSP	0.	0.	0.	0.	٥.	0.	٥.	0.	0.	2.0 -1.0	0.	0.	0.	1.0	0.	٥.	15.605
INPROVED RESTRAINTS	0.	3.0	0.	0.	٥.	0.	0.	0.	0.	0. 0.	0.	0.	0.	0.	-1.0	0.	14.210
LOW-DRAG SFC COATINGS	0.	0.	0.5	0.5	٥.	-0.5	ο.	1.0	0.	00.5	0.5	6.	٥.	0.	0.	0.	13.268
SGL LEVER THRUST/DRG CTL	C.	ο.	0.	0.	٥.	0.	٥.	0.	0.	1.0 0.	0.	0.	0.	0.5	0.	0.	12.060
NAVSTAR/GPS	0.	0.	٥.	0.	٥.	0.	٥.	0.	0.	1.0 0.	0.	0.	0.	0.5	0.	0.	12.060
INTEG LOW-COST WG LVLER	<b>c.</b>	0.	٥.	-0.5	٥.	-0.5	ο.	0.	0.	2.0 -1.0	0.	-1.0	0.	2.0	0.	0.	10.050
INTI-ICING SFC COATINGS	0.	0.	٥.	-0.5	٥.	-0.5	ο.	0.	0.	01.0	0.	-0.5	0.	3.0	0.	0.	9.201
C ELECTRICAL SYSTEMS	0.	ο.	0.	0.	٥.	1.0	ο.	0.	0.	0. 0.5	0.	0.	٥.	0.	0.	0.	8.598
DATA LINKS	0.	с.	0.	-0.5	٥.	0.	0.	0.	0.	2.0 -1.0	0.	-0.5	0.	1.0	0.	0.	7.225
MPR STLL/SPNAERO TLRG	0.	0.	-0.5	-0.5	0.	0.	Ο.	-0.5	0.	0. 0.	0.	0.	٥.	z.0	0.	0.	6.851
NERGY ABSORBING FLOOR	0,	1.0	0.	0.	ο.	0.	ο.	0.	0.	0. 0.	0.	0.	0.	0.	0.	0.	5.818
ICROWAVE LANDING SYSTEM	ο.	0.	0.	0.	0.	0.	٥.	0.5	0.	-1.0 0.	0.	0.	0.	1.0	0.	0.	6.15
IRECT LIFT CONTROL	0.	ο.	ο.	0.	с.	-0.5	Ο.	ο.	0.	1.0 -1.0	0.	-0.5	0.	1.0	0.	0.5	5.09
INTEGRATED YAW DAMPER	0.	0.	0.	0.	٥.	0	Ο.	0.5	0.	1.0 -2.0	0.	-1.0	3.0	0.	0.	0.	3.87
CTIVE STALL PREVENTION	0.	0.	٥.	-0.5	0.	-0.5	٥.	0.	0.	1.0 -2.0	0.	-1.0	0.	3.0	0.	0.	3.60
ACT CTLS FOR PLX STBLTY	1.0	٥.	1.0	1.0	ο.	2.0	0.	2.0	0.	03.0	1.0	-3.0	٥.	0.	0.	0.	1.43
BRST/TEAR RESIS FUEL TKS	0.	2.0	0.	0.	٥.	-1.0	0.	0.	0.	01.0	0.	0.	0.	0.	0.	0.	0.78
ORAN C	з.	0.	0.	0.	0.	0.	ο.	0.	0.	0.5 -0.5	0.	0.	٥.	0.	0.	<b>6.</b>	-0.58
DMEGA	0.	0.	٥.	0.	٥.	0.	0.	0.	0.	0.5 -1.0	0.	0.	0.	0.	0.	0.	-4.84
LOW-LEVEL PRESSURIZATION	0.	с.	0.	0.	٥.	-0.5	0.	0.	2.0	01.0	0.	-1.0	0.	0.	0.	0.	-5.46
FLUIDIC AUTO FLT CTL SYS	0.	0.	ο.	-0.5	0.	-0.5	٥.	0.	0.	0. 1.0	0.	-1.0	٥.	0.	0.	0.	-6.46
DUCTED PROPULSORS	0.	0.	-2.0	-1.0	0.	-1.0	3.0	-1.5	3.0	01.0	-1.5	0.	0.	0.5	0.	2.0	-7.34
TICRO HUD	0.	0.	0.	-0.5	٥.	0.	٥.	0.	0.	1.5 -2.0	0.	-1.0	0.	0.5	0.	0.	-14.10
ASER GYROS	0.	0.	0.	-2.0	٥.	1.0	ο.	0.	0.	03.0	0.	2.0	٥.	0.5	0.	0.	-14.59
FIBER OPTICS	0.	0.	0.	0.	0.	0.	0.	0.	0.	02.0	0.	0.	0.	0.	0.	0.	-17.03
NCT RIDE SM	0.	0.	0.	-0.5	0.	-0.5	0.	0.	0.	02.0	0.	-1.5	3.0	0.	0.	0.	-18.10
100	0.	0.	0.	-1.0	ο.	-1.0	0.	0.	0.	2.0 -3.0	0.	-1.0	٥.	1.0	0.	0.	-22.52
DIRECT SIDE FORCE CTL	0.	0.	0.	-1-0	0.	-1-0	0.	=1.0	٥.	-0.5 -1.0	٥.	-0.5	0.	0.	0.	1.0	-30-14

Table B.10. Likely Technology Evaluation (continued)

~

. ...

.

1.1

7

5

439

 $\mathbf{A} = \{\mathbf{x}_{i}, \dots, i\}$ 

÷

v

× 1

1, -

...

ACT GUST ALLV	0.	٥.	0.	-1.0	0.	0.5	٥.	0.	0.	0.	-3.0	0.	-2.0	1.0	0.	0.	0.	-42.879
OPPLER NAVIGATION	0.	0.	0	-5.0	٥.	-1.0	0.	<b>c.</b>	0.	1.0	-3.0	0.	-1.5	0.	0.5	0.	0.	-45.918
INERTIAL NAVIGATION	0.	0.	0.	-2.0	0.	-1.0	0.	0.	0.	1.0	-3.0	0.	-1.5	0.	0.5	0.	0.	-45.918
ACT FLUTTER SUPPRESSION	0.	0.	0.	-2.0	0.	1.0	0.	1.0	0.	<b>C.</b>	-3.0	0.	-2.5	0.	0.	0.	0.	-50.995
FLY-BY-LIGHT	0.	0.	0.	-1.5	٥.	-0.5	0.	0.	0.	0.	-3.0	0.	-5.0	0.	0.	0.	0.	-57,282
fly-by-wire	0.	0.	0.	- 2- 0	0.	-1_0	0.	0.	0.	0.	-3.0	0.	-2.0	0.	0.	0.	0.	-63.405

Table B.iO. Likely Technology Evaluation (continued)

1 10

. •

çî. E

· ·

-

\*

. -

.

Table B.10. Likely Technology Evaluation (continued)

1

1

·

\*

ter summer and

Annual Res

۲. ۱

•TABULAR RESULTS FOR AL	L TECK	SO LOS	IES F	0110														
	כנור	TIT S		20 0 0	N IS	ENTY UGNT	EXTR NOTS	FUEL EFFC	SICN	PILT VAKL		8 G E	5619	8196 GL 17	5577	STAT CHFT	7819	FIG OF
CATEGORY WEIGHTS	5.45 7	. 35 7	.48	. 45 2	-16	5.66	5.41	8.61	7.45	7-00	7.76	7.18	•.55	7.24	1.54	<b>6.</b> 71	7.41	
APHITE COMPOSITES	0*1		1.5	2.0	•	3.0	•	2.5	9	•	-2.0	2.0		6	5	` <b>o</b>	đ	72 . 895
VLAR COMPOSITES	0.5	<b>.</b>	0-1	2.0	•	2.0	•	2.0	•	•	•	2.0	å	-	•	•	•	766" 12
TET PROP TECHNOLOGY	•	•	•	•	•	0.5	3.0	•	3.0	•	•	•	1.0	•	0.5	d	9	55.742
FICLENT PROP TCHNLGY	•	•	0.5	0.5	•	0.5	•	1.0	6	•	•	1.0	1.6	•	0.5	•	1.0	48.807
GILERS	1.0	•	1.0	1.0	•	0.5	•	1.0	•	•	•	1.0		1.0	•	•	•	44.232
BERGLASS COMPOSITES	•	•	1.0	1.0	•	1.0	•	1.0	6	•	1.0		•		•	å	4	14.136
TE ENGINE	•	<b>.</b>	•	1.0	•	•	<b>.</b>	•	•	•	2.0		1.0	•	1.0	<b>;</b>	•	190-11
T LAM FLOW AIRFOILS	1.0	•	1.0	1.5	•	•	•	1.5	•		-0-5	1.6	<b>.</b>	•	0	5	•	43.333
W/MEDM SPEED ALAFOILS	•	•		1.0	•	1.0	•	1.5	ö	•	-1-0	1.0		•	•	•	2.0	42.251
ADING EDGE DEVICES	1.0	•	1.0	0.5	•	•	•	1.0	•	•	-0-2		-1-6	1.0	5.9	•	5	32,022
TEG AVIONICS AND DSPTS	••	•	••	0.5	•	0.5	•	•	•	2.6	-1.0	•	•	•		•	•	28.105
TIVE LAMINAR FLOW CTL	2.0	•	2.0	1.0	•	-1.0	•	3.0	•	•	-3.0	3.6	-3.6	9	ċ	•	5	25.085
TT-TCING SFC COATINGS	•	•	•	0.5	•	1.0	•	•	•	0.5	0.5	•	2.6	6	-1.5	•	5	22-546
AD LIMITING SEATS	<b>°</b>	3.0	•	•	•	0.5	•	•	5	•	•	•	•	å	•	ļ	6	21.474
N6LE75	1.0	•	•	1.0	•	-1.0	•	1.0	•	•	-1.0	1.0	ð	•	.0	•	0.5	114.05
PARATE SFC TECHNOLOGY	•	•	•	•		0.5	•	•	•	1.0	•	0	•		1.0	6		116.371
T DESPLAYS	•	•	•	•	•	•	•	•	•	:	1-1-0	•	1.6	•	0.5	•	ċ	17.055
S STATUS BSP	•	•		•	•	•	•	5	•	2.5	0-1-0	•	•		1.1	6	•	15.779
PROVED RESTRAINTS	•	3.0	•	<b>.</b>	•	•	•	•	•	•	•	•		9	6		•	15.242
W-BRAG SFC COATINGS	•	•	0.5	0.5	•	-0-5	•	1.0	6	6	-0-5	5	6	ġ	•	0	9	13.,951
	•	•	•	•	•	•	•	•	•		•	•	•	•	•	•	ł	

2.040.4

1

a public

441
Figure B.10. Likely Technology Evaluation (concluded)

~

----

;

ì

19. -

.

VSTAR/6PS	<b>.</b>	•	•	6	•	•		ò	•		•		<b>.</b>	•	•••	5	<b>.</b>	11.770
CROWAVE LANDENG SYSTEM	•		ġ	•		•	•	1.0	•	-1.0	•	•			1.0	5	•	11.152
ERGY ABSORDING FLOOR	•	1.5	•	•	•	•	•	•	5	•	•	•		.0	•	-		11.040
TIVE STALL PREVENTION	6	•	•	-0.5	9.	-0.5	•		•	1.0	-1.0		-1.0	•	3.0			10.746
ELECTRICAL SYSTEMS	•	•	•			1.0	•		•	•	0.5	•	•	•	ö	•	•	9.536
TEG LOW-COST NG LVLER	•	•	•	-0-5	•	-0-5	•	6	5	2.0	-1.0		-1.0	9	2.0	6	6	8.222
ITEGRATED VAN DAMPER	•	•	•	•	:	0.5	•	0.5	5			•	•		•	•	4	7.122
IPR STLL/SPMAERO TLRG	•		-0.5	-0.5	5	•		-0-5	6	•		•	•	•	2.0	•	9	\$12.4
ITA LINKS	ċ	•	••	-0.5	•	<b>.</b>	••	°.		2.0	-1.0	•	-0-5	ġ	1.0	•	ď	6.2.80
ET ETLS FOR RLX STBLTV	1.0	•	1.0	1.0	••	2.0		2.0		•	-3.0	1.0	-3.0		j	5		4.141
IRECT LIFT CONTROL	5	•		•	•	-0.5		•		1.0	-1.0	•	-0.5	ð	1.0	•	0.5	4.682
0878 C	•	•	•	•	•	••	•	•		0.5	-0.5	5			5	9	9	-0-342
RST/TEAR RESIS FUEL TKS	6	2-0		•	•	-1.5	<b>.</b>	•		•	-1.0	•			•		9	-1.581
UCTED PROPULSORS	•		-2.0	-1.0	•	-1.0	3.0	-1.5	3.0		-1.0	-1.5	•		0.5		2.0	-3.329
1664	•		•		•	•	•			0.5	-1-0				•	•		-4.242
DU-LEVEL PRESSURIZATION	6	•	•			-0.5			2.0	•	-1.0		-1.0		•	5	ö	-5.233
LUTBLE AUTO FLT CTL 575	5	•	•	-0-5	•	-0.5			•	•	1.0	<b>.</b>	-1.0	•	-	•	•	-1.360
ASER GYROS	•	5	•	-2.0	••	1.0	•		•	•	-3.0		2.0		0.5	0	ď	-12.662
TBER DFTICS	•	•	•	•		0.5		•	5	•	-2.0	•	•		ż	5		-12.493
1CR0 NUD	•	•	•	-0.5	•	•	•	•	•	1.5	-2.0		-1.0	ď	0.5	ö	9	-14.525
ct Ride Sn	ò	•	•	-0.5	•	-0-5			ċ		-2.0	•	-1.5	3.0	•	5		-15.483
97	6		•	-1.0		-1.0	•		•	2.0	-2.5	•	-1.0		1.0	5	6	-20.554
IRECT SIDE FORCE CTL	•	•	•	-1.0	•	-1.0	•	-1.0		-0.5	-1.0	ď	-0-5	9	•	6	1.0	-32.341
CT GUST ALLY	•		•	-1.0	•	0.5	<b>.</b>	.0		•	-3.0		-2.0	1.0	ď	•	9	-41.761
OPPLER MAVIGATION	•	•	•	-2.0		-1.0	•	•		1.0	-3.0	5	-1.5		0.5	•	ő	-50.392
NERTIAL RAVIGATION		r"	•	-2.0	••	-1.0	•	.0		1.0	-3.0		-1.5	•	0.5	5		-50, 392
CT FLUTTER SUPPRESSION	•	L	•	-2.0	•	1.9	•	1.0		•	-3.0	•	-2.5	ď	•	6		-51.789
1 - 8 1 - 4		3	•	-1.5		-0.5	<b>.</b>	•		•	-2.5	•	-2.0	•	•	•	•	-55.502
	¢	<	c	0.01	ç	-1-0	c	¢	c	ć	- 2 - 5	Ċ	-2-D	.0	ċ	Ğ	é	-45.854

ORIGINAL PAGE IS OF FOOR QUALITY

IBULAR RESULTS FOR ALL	1221	1	~ ~ ~ ~ ~		•													
5	EIL 6	I SH			NOI NOI	New T	EXTR FOIS		INTR NOIS			191			8 F T Y	STAT CAFT	1019	
CATEGORY WEIGHTS 5	• 5 •	-82	7-92 7.	5	5	. 36 .	•	8.14	7.04	7.35	8.51	7.29	8.85	11.4	27*6	1.24	4.87	
ENGLAE		:	2.0	2.0	•••	3.0	•	2.0		-	1-0		2.5	1.0	2.0	•	1.0	1 40.7
LERS	1.5	•	2.0	2.0	•	1.5	<b>.</b>	2.0	•	•	1.0	1.5	<b>.</b>	3.0			d	100.4
CHG ROT COMB ENGINE	ల	•	1.5	2-0	2.0	2.0	1.0	1.0	1.0	•	:	1.0	1.6		9	-	0.5	1-14
AR COMPOSITES	0.5	•	1.5	2.0	•	2.0	•	2.0	•	•	1.0	2.0	•	•		ė	6	78.4
NCED DIESEL ENGINE	ő	°.	••	2.0	3.0	1.0	1.0	2.0	1.0		-1.0	2.0			0.5	<b>.</b>	•	74.4
HITE COMPOSITES	1.5	5	2.0	2.0	•	3.0	•	3.0	•	•	-2.0	2.0	å	•	•	9	ð	76
T CMG RECIP ENGINE		•	1.0	1.5	3.0	1.0	•	2.0	•	•	•	1.0	::	<b>.</b>	<b>0</b> -2	9		70.5
B RECEP ENGINE	•	•	1.0	1.5	3.0	1.0	•	2.0	•	•	•	1.6		•	5	•	•	70.5
110 COOLING	•	•	0.5	1.5	2.0	6	1.0	1.0	1.0		0.5	1.6		ä	<b>.</b> .	5	9	59.1
LAM FLOW AIRFOILS	1.5	•	1.0	2.0	•	•	•	2.0	•	•	•	1	<b>.</b>	•	6	•	9	29.1
ING EDGE DEVICES	1.5	•	1.0	1.0	•	0.5	•	1.0	•	•	-0-	1.5				<b>.</b>	6	54.1
CIENT PROP TCHALGY	•	•	0.5	1.0	•	0.5	•	1.5	•		5	1.5	1.6		0-1	•	1.0	57.4
T PROP TECHNOLOGY	ö	•	•	•	•	1.0	3.0	•	3.0	•	•	•	-	•	<b>.</b> .	6	•	54.3
E AVIONICS AND DSPYS	•		•	1.0		0-5	•	•	•	3.6	; ;	•	5	•	2.6	。 。	9	55.3
RELASS COMPOSITES	•	•	1.0	1.5	•	1.5	•	1.0	•	•	<b></b>	1.6	6	ď	0	•	đ	5**5
OVED TURBOCHARGING	3.0	•	1.5	1.0	•	•	•	1.0	6	•	2	1.0	•	•	ď	9	1.0	2.1
4.675	۰.5	•	•	1.5	•	-0.5	•	2.0	5	•	5	2.6	•		6	ġ	-	B**6
ER FLAPS	0.5	•	1.0	1.0	•	0.5	•	1.0	•	•	0.5		•	1.1	•	å		
MEDN SPEED AIRFOILS	•		•	1.0	•			<b>.</b> -		•		1.6	•	•	ġ	•	<b>0</b> *2	12.5
WE LAMERAR FLOW CTL	2.5	•	2.5	2.0	•	-1.5	•	0 <b>.</b> 5	•	•	-3.0	3.6	-3.1	•	6	•	ő	34.4
APAC KET LATIMES	¢	•	•	•	•		•	•	•	•	•		•	•	•	•	•	

. . . . . . . . . .

Table B.11. Optimistic Technology Evaluation

1

1 111<sup>11</sup> to 3000

÷...

Ξ

Table B.11. Optimistic Technology Evaluation (continued)

· Ser Andrew Ser

, i -

٦

1981

-

f

. \*

-149 2.7

 $r \sim \eta$ 

24.7% -1.16 -10.057 23.985 9.769 3,072 -3.255 -9.717 -7.717 28.717 22.275 17.332 11.487 12.055 12.123 11.178 9,,523 8.301 -13.677 26.27 14.77 -2,035 21.111 11.430 17.639 112.01 13.904 15.22 -21.17 ÷, 5.5 2.9 å 4 4 å å ð å å ď å 2 -0.5 • ċ ċ ÷ ő ð <u>.</u> : °. 0.1 •••• ? **°**.2 : 3.0 1.0 ... 1.0 å 4 å đ ð .... ċ å ċ å ċ ċ d ċ å ò -0. \$ •.... ---. . <u>د</u>. -0.5 2 ... 0.2--1-0 -1.6 9 å j å å 5 å ð å ð å å ð 5 0. -1.0 -1.0 ; • ċ ő ċ ő ð ð ċ ð ċ -0-5 ••• 2.0 -2.0 •••• -3.0 -1-0 ..... 1.4 -1.0 1.0 -0.5 1.0 -1.0 1.0 0. -2.0 0. -3.0 -2.0 0.5 2.0 -1.0 1.0 -1.5 2.0 -1.0 2.0 -2.0 1.0 -1.0 5 ċ 6 ö • ċ å 2.0 . • 0.5 • . ċ . . . ċ . **5.0** 3.0 ċ 5 ð . ð ď ċ 5 ċ • ð • ð 6 ő • ċ ő • å ð å ... ... -1.0 -1.0 2.5 5.9 . . • ď . 6 • . ð • ð ċ • 6 • ő ð ċ ö • 5 • 0.0 . • . . • • • • • . . • . • å -1.0 -0-3 ••• **0**-2 ---1.5 -1.0 -1.0 ••• -0.5 -0.5 -0-5 ... -9.5 -0.5 • 5 5 5 • • ď . • 6 . • • • • . . . . ò . ÷ ċ ċ ċ • • -1.0 -0.5 -0-3 -1.0 ••• -2.0 0.5 2.0 -0.5 -0.5 -0.5 5 Ş 5 Ş . ð ö ď • 6 6 ċ ò 6 6 6 ò . 5 • • . . . • . . • ċ . . ċ . 0.5 3.0 <u>.</u> 3.0 . . • . å • • . . • • • • 6 . . . • . • • . 2 . . . ċ ċ . . ċ 6 6 . 6 ő • 6 IMPR STLL/SPM-AERO TLRG RICPONAVE LANDING SYSTEM BPST/TEAR RESES FUEL TKS FLUIDIC AUTO FLT CTL SYS LOW-LEVEL PRESSURIZATION SGL LEVER THRUSTIONG CTL SEPARATE SFC TECHNOLOGY INTEG LOW-COST NG LVLER ANTI-ICING SFC COATINGS ACT CTLS FOR MLE STOLTT ACTIVE STALL PREVENTION INTEGRATEA TAN BANPER ENERGY ABSORDING FLOOR AC ELECTRICAL SYSTERS binect side fonce cit IMPROVED RESTRAINTS LOAD LIMITING SEATS DIRECT LIFF CONTROL BUCTER PROPULSORS SYS STATUS BSP CRT DESPLAYS F1868 8PTICS HAVSTAR/6PS ACT RIDE SN LASER \$7805 BATA LINKS AICED NUD LONAN C ONEGA ŝ

! .

	_		_					0.00	_
7	t,	R	8	\$					
5.5	5.5	1	3	2					2
7	7	7	7	1	ſ				
	3	4			8		-		
	_	-	-	-	-				
ė	ė	d	ė	ė					
	5	0.5	5.0		;				
	0				-				
ġ	-	d	ġ	ġ					
.2.0	1	1.0	1.0	1.0					
					•				
0	-		•	-					
-3.0	-1.0	-3.0	-3.1						
		0	•	-	•				
-	-	•	-						
ö	9	ė	ö	ő	5				
					5				
	•	•	0	0					
•	-			•					
					•				
•	š	š	~	0	•				
	ę	÷	÷	÷					
•			•		•				
0	ø	Ö	Ö	Ö	i				
•	•	•	ċ	•	<b>;</b>				
_									
NOIS									
222		Z	10X						
-	~		GA T						
	ינו	9 I A E	IVA	5 M T	,				
111	51	ž	۲ ۲	÷.					
2	20	2		-					
ACT	L.	5	N	2	•				

Table B.11. Optimistic Technology Evaluation (continued)

Table B.11. Optimistic Technology Evaluation (continued)

yn 11 m. <sub>a</sub> befan. An an an an an an ar

a i

4

...

···· ABULAR RESULTS FOR AL	11 755	JONN	81E3 -		=													
	U II	C ass C ass C ass		v v	101	1111	TXTR 1015	1313		2111	115		111			E	22	114 W
CATEGORY NEIGHTS	3.45	7.33	1 27-2				11	17.8	7.45	7.00	7.76				- 35 - 4	2	1.41	
BRAPHETE COMPOSITES	1.5		2.0	2.0	•	3.0	•	3.0	5		-2.0	2.0	•	9	•	•	ġ	564.68
KEVLAR COMPOSITES	0.5	•	1.5	2.0	<b>.</b>	2.0	•	2.0			1.0	0.5	ö	•	6	•	•	547"53
54417543	1.5		1.5	1.5	•	1.0	•	1.5		9	1.0	1.5	•	::	d	ò		£1.523
BUIET PROP TECHNOLOGY	•	<b>°</b>		ő	•	1.0	0.0	•	3.0			5	1.5	o	0.5	ġ	ġ	63.344
HAT LAM FLOW AIRFOILS	1.5	5	1.0	2.0	0		•	2.0	•			1.5	ð	5	5	ċ	j	42.334
EFFICIENT PROP TONNLGY	•	•	0.5	1.0	•	0.5	•	1.5	•	•	•	1.5	1.0	•	0.5		1.0	61.426
LEADING EDGE DEVICES	1.5		1.0	1.0	•	0.5	°.	1.0	•	•	•	1.5	-0.5	1.0	1.0	•	•	59.314
LOW/REDR SPEED AIRFOILS	6	•	•	1.5	•	1.0	•	2.0	•	•	°.	1.0	ů	<b>.</b>	•	•	2.0	59.041
FIREAGLASS COMPOSITES			1.0	1.5	•	1.5	•	1.0	•	•	1.5	<b>1</b> .5		•	•	•	ò	57.5M
1 [ N G L E T S	2-5	•		1.5		-0.3	•	2.0	ð	-	-0.5	2.0		•	ė	•	1.0	57.343
INTEG AVIONICS AND BSPVS	•	•	•	1.0	•	0.5	•	•	•	3.0	•	•	0.5	å	2.0	ċ	•	52.133
647E E#614E	0	5	•	1.0	•	•	•	•	•	ö	2-0	•	5.0		1.0	•	ő	14.044
ACTIVE LAFIMAR FLOW CTL	2.5	•	2.5		•	-0.5	•	5.0	•	ő	-3.0	3.0	-3.0	•	6	•	ő	39.103
ARTI-ICING SFC COATINGS	<b>.</b>	•	•	0.5	•	1.0	•		•	1.0	0.5	0	2.0		-1.0	•	6	30.811
LOV-BRAG SFC COATINGS	•	•	1.0	1.0	•	-0.5	•	1.5	•	•	-0-5	1.0	ö	ö	ö		•	30.312
AET ETLS FOR RLK STOLTY	1.0	<b>.</b>	1.0	2.0	<b>.</b>	2.0	0	2.5			-3.0	1.0	-2.0		•	ć	6	297-62
CRT BISPLATS	5		•	•		•	•	•	•	2.0	-0.5	6	1.6		1.0	ė	i	7.20
LOAD LIMITING SEATS	0	3.6	.0.	•	•	1.0		•	6	•	•	•	ò	•	•	ċ	•	27.656
STATUS DSP		Ĵ	•	0.5	•	<b>.</b>	<b>*</b>	•	•	2.0	-1.0		0.5	ė	1.0	ò	•	25.27
SEPARATE SFC TECHNOLOGY	•	•	•	•	•	1.0		•	ċ	1.0	ů	ð	ö	•	1.0	•	6	22.199
2475748/6PS		5	•	0.5	•	••		0	•	1.0	0.5	9	ġ	9	0.5	ő	i	20.374

MICROWAVE LANDING SYSTEM	0.	0.	0.	0.	٥.	0.	0.	1.5	0.	-0.5	٥.	0.	٥.	0.	1.0	0.	0.	18. #54
IMPROVED RESTRAINTS	C.	3.0	0.	0.	0.	0.	0.	0.	0.	0.	0.	0.	0.	٥.	0.	-0.5	0.	18.646
INTEG LOW-COST NG LVLER	0.	0.	0.	с.	0.	-0.5	0.	0.	0.	2,0-	-1.0	0.	-0.5	0.	2.0	0.	0.	17.721
SGL LEVER THRUST/DRG CTL	0.	0.	0.	0.	0.	0.	٥.	0.	0.	.0	0.	0.	0.	0.	1.0	٥.	0.	16.542
AC ELECTRICAL SYSTEMS	ů.	8.	С.	0.	٥.	1.5	٥.	0.	0.	0.	1.0	0.	0.	0.	0.	0.	0.	16.247
ENERGY ABSOPBING FLOOR	0.	2.0	٥.	0.	٥.	0.	ο.	0.	0.	0.	Ũ.	0.	0.	٥.	0.	0.	0.	14.646
ACTIVE STALL PREVENTION	0.	0.	٥.	-0.5	0.	-0.5	٥.	0.	0.	1.0 -	-0.5	0.	-1.0	٥.	3.0	0.	0.	16.648
IMPR STLL/SPNAERO TLRG	٥.	0.	0.	-0.5	٥.	0.	٥.	-0.5	0.	0.5	0.	0.	9.	9.	2.0	٥.	0.	13,558
DUCTED PROPUL ORS	0.	٥.	-1.0	-1.0	ο.	-1.0	3.0	-1.0	3.0	0	-1.0	-1.0	0.	0.	0.5	0.	2.0	12.045
DATA LINKS	0.	e.	о.	0.	0.	0.	0.	0.	0.	2.0 .	-1.0	0.	-0.5	0.	1.0	0.	0.	11.005
INTEGRATED YAW DAMPER	0.	0.	0.	0.	0.	1.0	٥.	0.5	0.	0.	0.	0.	0.	0.	0.	0.	0.	9.960
DIRECT LIFT CONTROL	0.	0.	0.	0.	٥.	-0.5	٥.	0.	0.	1.0	-1.0	0.	0.	0.	1.0	0.	0.5	7.656
BRST/TEAR RESIS FUEL TKS	0.	3.0	0.	0.	0.	-1.0	0.	0.	0.	0.	-1.0	0.	0.	0.	D.	0.	0.	8.581
LORAN C	0.	c.	٥.	0.	ο.	0.	٥.	0.	0.	1.0	-0.5	0.	0.	0.	0.	0.	0.	3.117
FIBER OPTICS	0.	0.	0.	0.5	0.	1.0	0.	0.	٥.	0.	-2.0	0.	0.5	٥.	0.	0.	0.	-0.366
OMEGA	0.	0.	0.	0.	0.	0.	0.	0.	0.	1.0	-1.0	0.	0.	0.	0.	0.	0.	-0.763
LOW-LEVEL PRESSURIZATION	0.	0.	٥.	0.	٥.	0.	0.	0.	2.0	ð. ·	-1.0	0.	-1.0	0.	0.	0.	0.	-2.404
FLUIDIC AUTO FLI CTL SYS	0.	0.	<b>o</b> .	-0.5	٥.	-0.5	٥.	0.	0.	0.	1.0	0.	-0.5	٥.	0.	0.	0.	~4.567
LASER GYRDS	0.	0.	0.	-2.0	0.	1.5	ο.	0.	0.	0.	-3.0	0.	2.0	0.	1.0	0.	0.	-5.061
MICRO HUD	0.	0.	0.	-0.5	٥.	0.	0.	0.	0.	2.0	-ż.0	0.	-1.0	0.	1.0	0.	0.	-6.255
ACT RIDE SN	0.	0.	0.	-0.5	ç.	-0.5	0.	0.	0.	0.	-2.0	0.	-1.0	3.0	0.	0.	0.	-10.909
HUD	0.	0.	0.	-0.5	٥.	-1.0	0.	0.	0.	2.0	-2.0	0.	-1.0	e.	1.0	0.	0.	-11.912
ACT FLUTTER SUPPRESSION	0.	0.	0.	-1.0	٥.	2.0	0.	1.5	0.	0.	-3.0	0.	-2.0	0.	0.	0.	0.	-27.605
DIRECT SIDE FORCE CTL	0.	0.	0.	-1.0	0.	-1.0	0.	-1.0	0.	0.	-1.0	0.	-0.5	0.	0.	0.	1.0	-28.842
ACT GUST ALLY	0.	0.	0.	-0.5	0.	1.0	0.	0.	0.	0.	-3.0	0.	-1.5	1.0	0.	0.	0.	-29.434
FLY-BY-LIGHT	0.	0.	0.	-1.0	0.	0.	ο.	0.	0.	0.	-2-0	0.	-1.0	0.	0.	0.	0.	-34.520
FLY-BY-WIRE	0.	0.	0.	-1.0	0.	-0.5	0.	0.	0.	0.	-2.0	0.	-1.0	0.	0.	0.	0.	-37.349
DOPPLER NAVIGATION	c.	0.	٥.	-1.5	٥.	-1.0	٥.	0.	0.	1.0	-3.0	0.	-1.0	0.	0.5	0.	D.	-40.894
INERTIAL NAVIGATION	0.	0.	0.	-1.5	0.	-1.0	٥.	0.	0.	1.0	-3.0	0.	-1+0	0.	0.5	0.	0.	-40.894

Table B.11. Optimistic Technology Evaluation (concluded)

an an the second se

· . . .

---

.

----1

fill the trainer of

11 BA 61 BA HAVE

• •

1. E. E.

Fig. B.3. Airplane A PLOE Figure of Merit Comparisons

GATE Engine Strt Chg Rot Comb Eng Kevlar Composites Graphite Composites Adv Diesel Engine Spoilers Strt Chg Recip Engine HCRLB Recip Engine Quiet Prop Tech Efficient Prop Tech Fiberglass Composites I



449

¥.

Improved Turbocharging Nat Laminar Flow Airfoils Liquid Cooling Leading Edge Devices Low/Med Speed Airfoils Fowler Flaps Integ Avionics and Displays Active Laminar Flow Control Load Limiting Seats Winglets Separate Surface Technology

.



and a second second

Leven and the part of the leven of the second se

1. . . . . .

S. S. S. S. Synamous

CRT Displays Systems Status Display Improved Restraints Low-Drag Sfc Coatings Sgi Lever Thrust/Drag Ctri Navstar/GPS Integ Low-Cost Wing Leveler Anti-Icing Sfc Coatings AC Electrical System Data Links Impr Stall/Spin-Aero Tailoring



Issimistic

Jara (\* 1949) de

**Energy Abs Floor** Microwave Land Sys **Direct Lift Control** Integ Yaw Damper Active Stall Prevention Act Ctrls for Rixd Stability Brst/Tear Resis Fuel Tanks Loran C Omega Low-Level Pressurization Fluidic AFCS **Ducted Propulsors** 



Fig. в.з. Airplane A PLOE Figure of Merit Comparisons (concluded)

Micro HUD Laser Gyros **Fiber Optics** Act Ride Smoothing HUD Direct Side-Force Ctrl Act Gust Alleviation Doppler Navigation Inertial Navigation Act Flutter Suppression Fly-by-Light Fly-by-Light



ť

Graphite Composites Keviar Composites Quiet Prop Technology Efficient Prop Technology **Spoilers** Fiberglass Composites **GATE Engine** Nat Laminar Flow Airfoils Low/Med Speed Airfoils Leading Edge Devices Integ Avionics and Displays

والمرجع والمرجع



Fig. B.4. Airplane B PLOE Figure of Merit Comparisons (continued)

Act Laminar Flow Control Anti-Icing Sfc Coatings Load Limiting Seats Winglets Separate Surface Tech CRT Displays Systems Status Displays Improved Restraints Low-Drag Sfc Coatings Sgi Lever Thrust/Drag Ctri Navstar/GPS



IRPLANE

œ

455

Microwave Landing System Energy Absorbing Floor Active Stall Prevention AC Electrical System Integ Low-Cost Wing Leveler Integ Yaw Damper Impr Stall/Spin-Aero Tailoring Data Links Act Ctris for Rixd Stability **Direct Lift Control** Loran C



AIRPLANE B

- timistic

- xpected

- 'essimistic

Burst/Tear Resis Fuel Tanks **Ducted Propulsors** Omega Low-Level Pressurization Fluidic AFCS Laser Gyros Fiber Optics Micro HUD Act Ride Smoothing HJD **Direct Side-Force Control** 

wa in e



AIRPLANE B

- Optimistic Likely Expected Pessimistic

8

456

Fig.

B.4.

Airplane

8

PLOE Figure

of

Merit Comparisons (continued)

Ŋ



ų

· 17.\*\*\* · 17.\*

4

an Brith

Fig. B.4. Airplane B PLOE Figure of Merit Comparisons (concluded)

## B.5 <u>TABULAR RESULTS OF THE TECHNOLOGY EVALUATION WITH AN</u> EXAMINATION OF THE EFFECT OF EMPTY WEIGHT

This section contains computer output from Program TCHLST and represents three different sets of category rating data for Airplane A and Airplane B. In all cases, the matrix of relative benefits for both airplanes was kept constant.

These data examine the effects of perturbations to the Empty Weight category because this category was awarded unusually low weightings by the respondents to Survey 2. This was cause for concern to the present research effort because empty weight is directly related to payload and, for a given takeoff gross weight, technologies which reduce empty weight will ultimately increase payload if fuel capacity is maintained. As a result, Payload was specifically eliminated as a category during Survey 1 because it overlapped almost completely with Empty Weight. At the conclusion of Survey 2, however, Empty Weight had an average weighting of 4.340 for Airplane A and was ranked 15th of 17 categories. Airplane B likewise had an average weighting of 5.657 and here the category was ranked 14th of 17 categories. Evaluation ratings varied from 0.0 to 9.5 for both airplanes. However, although the category showed some convergence to its mean for Airplane A, there was no apparent agreement on the weighting for Airplane B, and scores appeared to be distributed uniformly as shown in the histograms of Appendix B.2. Consequently, results from average category weightings as well as those resulting when Empty Weight was rated 9.5 and 0.0 are included.

458

## B.5.1 Average Category Weights

Four tables are presented where the average category weights as generated by Survey 2 are utilized. Table B.12 shows relative benefits and technology rankings by figures of merit within technology groups for Airplane A. These figures of merit are those shown in the tables of Section 4.5. Table B.13 is a duplicate of Table 4.10 for Airplane A and shows overall technology rankings for this airplane. Table B.14 shows relative benefits and technology rankings by figures of merit within technology groups for Airplane B. These data were also reflected in the tables of Section 4.5 Finally, Table B.15 is a duplicate of Table 4.11 for Airplane B and shows the overall technology rankings. These four figures are included here to illustrate data output by technology group with relative benefits shown (Tables B.12 and B.14) and to form a baseline against which to compare the overall technology rankings of the next two sections.

## B.5.2 Maximum Category Weight For Empty Weight

Here, the maximum Empty Weight rating (9.5) is used for both airplanes as shown in Tables B.16 and B.17. When Table B.16 is compared with B.13 for Airplane A, it can be seen that the effect of increasing the significance of Empty Weight is to increase the range of figures of merit, while the relative rankings remain almost constant. Likewise, the same trend is noted for Airplane B when Tables B.17 and B.15 are compared. The significant points to be noted in these comparisons are that:

- (1) those technologies which improve empty weight also improve several other measurements when integrated into an airplane and hence are already ranked at the top of the overall technology list for both airplanes (first 8 for Airplane A and first 3 for Airplane B).
- (2) the effect of increasing the Empty Weight rating serves to make these technologies even more attractive when compared to other technologies.
- (3) those technologies which are ranked low do not impact Empty Weight significantly and consequently retain their relative unattractiveness with respect to a figure of merit of zero.
- B.5.3 Minimum Category Weight For Empty Weight

Here, the minimum Empty Weight rating (0.0) is used for both airplanes as shown in Tables B.18 and B.19. When these tables are compared with Tables B.13 and B.10, results inferred from the previous section are noted.

- Relative technology rankings exhibit insignificant changes for both airplanes.
- (2) Those technologies which are ranked high retain their relative dominance over other technologies but do not display advantages as distinct as originally reflected.
- (3) Unattractive technologies remain so and do not reflect changes in figures of merit with respect to zero.

								<b>A</b> 1	IRPLAJ	IE A									
	+++TABULAR RESULTS BY CA	TEGORI	FOL	LOW:															
		CEIL	C RSH W TH Y	CRUS SPED	9 C C	EM15 SION	EMTY WGNT	EXTR NOIS	FUEL	INTR GDIS	PILT WRKL	PRCH PRCE	RGE	BLTY B	LTY	SFTY	8 TAT CHFT	TOLD PERF	116 <b>97</b> MEL 17
	CATEGORY WEIGHTS	5.43	6.82	7.92	7. 91	1.61	4.34	4.29	8.16	7.04	7,35	8.51	7.29	8.85 6	5,11	9.42	6.24	6.87	
	+AERODYNAMICS:																		
	NAT LAM FLOW AIRFOILS	1.0	0.	2.0	2.0	0.	1.0	0.	1,1	` O.	0.	-0.5	2.0	-1.0	0.	0.	0.	1.0	57.924
	SPOILERS	-0.5	0.	2.0	1.0	0.	1.0	) 0.	1.0	) 0.	0.5	0.	1.0	0.	2.0	0.	0.	0.	54.717
	FOWLER FLAPS	-0.5	; O.	z.0	1.0	0.	1.0	0.	1.0	0.	0.	-0.5	1.0	0.	2.0	) 0.	0.	0.	48.785
	LOW/MEDM SPEED AIRFOILS	0.	0.	1.0	0.5	0.	0.5	i 0.	0.5	i 0.	0.	-0.5	0.5	0.	0.	0.9	5 9.	1.0	29.09(
1	LEADING EDGE DEVICES	1.0	) 0.	0.5	0.5	0.	0.	0.	0.9	5 0.	٥.	-1.0	0.5	i 0.	1.(	) 0.	0.	1_0	25.530
	IMPR STLL/SPNAERO TLRG	0.	0.	0.	0.	0.	0.	0.	0.	0.	0.	-0.5	0.	0.	0.	3.(	0.	D.	24.00
	ACTIVE LAMINAR FLOW CTL	2.0	0.	2.0	-2.0	0.	0.	٥.	2.0	) 0.	0.	-3.0	3.0	-3.0	٥.	0.	0.	1.0	3.84
	WINGLETS	0.5	i 0.	0.	0.5	0.	-2.0	) 0.	0.9	5 0.	0.	-1.0	0.5	· 0.	0.	0.	0.	0.	-2.80
	LOW-DRAG SFC COATINGS	0.	0.	0.	-0.5	0.	-0.5	i 0.	0.9	5 0.	0.	-1.0	U.	0.	0.	0.	0.	-0.5	-13.99
	++AIRCRAFT SYSTEMS:																		
	ICEPHOBIC SET COATINGS	-0.5	0.	0.	-0.5	0.	-1.0	0.	0.	0.	0.	-1.0	-0.5	0.	٥.,	3.0	0.	0.	5.04
	AC ELECTRICAL SYSTEMS	0.	0.	0.	0.	0.	0.	0.	0.	0.	٥.	0.	0.	0.	0.	0.	0.	0.	0.
	++CRASHWORTHINESS:																		
1	LOAD LIMITING SEATS	0.	3.	0 0.	0.	0.	0.	0.	0.	0.	0.	0.	0.	0.	0.	0.	-0.5	i 0.	17.32(
	IMPROVED RESTRAINTS	0.	3.	0 0.	0.	0.	0.	0.	0.	0.	0.	-0.5	0.	0.	0.	0.	0.	0.	16.19
	ENERGY ABSORBING FLODR	0.	3.	0 0.	0.	0.	-0.5	i 0.	0.	0.	0.	-0.5	0.	0.	٥.	C.	0.	0.	14,.02(
I	BRST/TEAR RESIS FUEL TKS	0.	3.	0 0.	0.	0.	-1.0	0.	0.	0.	٥.	-1.0	0.	0.	0.	0.	0.	0.	7.59
	++FLIGHT CONTROL SYST	EMS 1																	
1	INTEG LOW-COST WG LVLER	0.	0.	0.	0.	٥.	-0.5	5 D.	0.	0.	3.0	) -0.5	0.	-0.5	٥.	3.	0 0.	0.	39.45
Í	SEPARATE SFC TECHNOLOGY	0.	0.	0.	-0.5	, o.	-0.5	5 0.	0.	0.	3.0	) -2.O	0.	-0-5	3.0	0 1-0	0 0-	0.	22.21

Table B.12. Technology Ranking By Technology Group For Airplane A

.

ar district supervision of the state of the supervision of the supervi

۰.

.

The construction of the co

•

Table B.12. Technology Ranking By Technology Group For Airplane A (continued)

ACTIVE STALL PREVENTION	0.	0.	0.	0.	٥.	-0.5	0.	0.	0.	٥.	-1.0	0.	-0.5	0.	3.0	0.	0.	13,154	
DIRECT LIFT CONTROL	0.	0.	Ο.	0.	٥.	-0.5	0.	e.	0.	1.0	-1.0	0.	-0.5	1.0	1.0	0.	0.5	11.206	
INTEGRATED YAW DAMPER	0.	0.	0.	0.	0.	0.	٥.	0.	٥.	1.0	-2.0	٥.	0.	2.0	0.	0.	D.	2.539	
ACT CTLS FOR RLX STBLTY	0.5	0.	0,5	1.0	0.	1.0	0.	1.0	٥.	0.	-3.0	1.0	-1_0	0.	0.	0.	0.	-3.030	
SGL LEVER THRUST/DRG CT	L 0.	0.	0.	0.	ο.	0.	0.	0.	0.	2.0	-1.0	0.	-1.0	0.	0.	٥.	0.	-2.671	
FEUIDIC AUTO FET CTE SY	s -0.5	0.	D.	0.	0.	-0.5	Ο.	-0.5	Ο.	0.	-1.0	-0.5	1.0	٥.	0.	0.	0.	-12.269	
ACT RIDE SMOOTHING	0.	0.	Ο.	-0.5	0.	-0.5	0.	0.	0.	0.5	-3.0	0.	-0.5	3.0	0.	0.	0.	-14,090	
DIRECT SIDE FORCE CTL	C.	ο.	0.	-0.5	٥.	-0.5	ο.	٥.	0.	0.5	-3.0	0.	-0.5	3.0	0.	0.	D.	-14.090	
ACT FLUTTER SUPPRESSION	1.	Ο.	Ο.	-0.5	Ο.	0.	٥.	0.	0.	0.	-3.0	0.	-1.0	0.	ο.	0.	0.	-38.352	
ACT GUST ALLV	0.	э.	0.	-1.0	0.	0.	ο.	0.	0.	0.	-3.0	0.	-1.0	0.	ũ.	0.	0.	-42.305	
FLY-BY-WIRE	0.	0.	Ο.	-1.0	0.	-1.0	0.	0.	0.	0.	-3.0	0.	-1.0	٥.	٥.	0.	0.	-45.645	
FLY-BY-LIGHT	٥.	٥.	٥.	-1.0	٥.	-1.0	٥.	0.	٥.	٥.	-3.0	٥.	-1.0	٥.	٥.	0.	0.	-45.645	
**TNEORMATION SYSTEM	¢.																		
DIGITAL DATA LINKS	0.	n .	٥.	Ω.	٥.	٥.	۵.	0.	٥.	3.0	-2.0	0.	0.	0.	3.0	0.	0.	33, 283	
INTEG AUTONICS AND DSPY	s 0.	η.	D.	1.0	0.	0.	0.	0.	0.	3.0	-2.0	0.	0.	٥.	2.0	0.	0.	31.767	
SYS STATHS DISPLAYS	0.	р.	o.	0.	0.	0.	0.	0.	0.	2.0	-1.0	0.	0.	0.	2.0	0.	n.	25.027	
COT DISPLAYS	0.	0.	0.	0.	0.	0.	0.	с. С.	0.	2.0	-2-0	0.	1.0	0.	0-5	0.	0.	11.233	
	0.	n.	0.	-0.5	0.	-0.5	0.	n.	0.	2.0	+2.0	0.	-1.0	a.	2.0	0_	0.	1.515	
JACER GYPOS	0.	0.	0.	-0.5	0.	0.	0.	0.	0.	0.	-3-0	0.	1.0	0.	0.5	0.	0.	-15.933	
HEADS-HP DISPLAY	0.	0.	0.	-1-0	0.	-1.0	0.	0.	0_	2.0	-3.0	0.	+1.0	0.	1-0	0.	0.	-22.525	
FIBER OPTICS (DATA TRNS	· · · ·	σ.	ο.	0.	0.	0.	ο.	0.	0.	0.	-3.0	0.	ο.	D.	D.	0.	0.	-25,545	
**NAVIGATION CONCEPT	5:																		
MICROWAVE LANDING SYSTE	r 0.	ο.	0.	0.5	0.	0.	1.0	0.	0.	0.	-1.0	¢,•	0.	٥.	1.0	0.	0.	9.147	
NAVSTAR/GPS	0.	0.	Ο.	0.	Ο.	0.	٥.	0.	0.	1.0	-1.0	0.	0.	0.	1.0	0.	0.	8.256	
LORAN C	0.	0.	0.	-0.5	0.	-0.5	0.	0.	0.	0.	-1.0	0.	-0.5	0.	-0.5	0.	0.	-23.776	
OMEGA	0.	0.	0.	-0.5	0.	0.	D.	0.	٥.	0.	-1,0	0.	-1.0	0.	-0.5	0.	0.	-26.033	
DOPPLER NAVIGATION	0.	0.	0.	-2.0	٥.	~0.5	٥.	0.	0.	1.0	-2.0	0.	-1.0	0.	0.	0.	0.	-36.517	
INERTIAL NAVIGATION	0.	0.	0.	- 3. 0	0.	-0.5	0.	0.	٥.	1.0	-3.0	0.	-1.0	٥.	٥.	٥.	<b>0.</b>	-52.938	
INERTIAL NAVIGATION	U.	υ.	0.	-3.0	υ.	-0.5	0.	U.	υ.	1.0	-3.0	U.	-1.0	<b>v.</b>	u.	0.	u.	- >2 . ¥ 56	

. ..

•

.

.

ч

Table B.12. Technology Ranking By Technology Group For Airplane A (concluded)

۰.

.

...N015E: 23.901 0.5 0, 0.5 2.0 0. 2.0 0. -1.0 0.5 0. ٥. QUIET EFFICIENT PROPS 0. 0. 0. U. ٥. ٥. -0.5 0. -1.0 -1.0 0. -1.0 3.0 -1.0 3.0 0. -1.0 -2.0 0. ٥. 0.5 0. 2.0 -1.703 DUCTED PROPULSORS LOW-LEVEL PRESSURIZATION 0. 0. 0. -0.5 0. -0.5 0. -6.5 2.0 0. -1.0 -0.5 -1.0 0. 0. 0. -17.141 ٥. ++PROPULSION: 85.017 1.0 1.0 1.0 3.0 0. 2.0 1.0 1.0 -3.0 1.0 1.0 1.0 1.0 0. 1.0 GATE ENLINE 2.0 0. 1.0 2.0 2.0 1.0 0.5 2.0 1.0 1.0 -2.0 1.0 1.0 1.0 84.205 1.0 0. ٥. STRT CHE ROT COMB ENGINE 1.0 0. 2.0 0. 36.205 0. 0.5 3.0 0.5 1.0 0. ٥. 0. 0. 2 STR ADV DIESEL ENGINE 0. 0. 0. 0. -2.0 0. 31.110 1.0 0. 1.0 0.5 0. 0. Ο. 1.0 0. 0. -1.0 1.0 0. 0. ٥. 0. 1.0 IMPROVED TURBOC ARGING 12.677 1.0 1.0 0. 0. 2.0 0. -1.0 1.0 0. 0. -1.0 0. -1.0 0. 0. LIQUID COOLING 0. ο. 1.0 2.0 -0.5 0. 1.0 0. 0. -1.0 0.5 0. 0. 0. 12.235 0. 0. STRAT CHE RECIP ENGINE 0. 0. 0. Ο. 0.5 0. 0. -0.5 0.5 0. 10.631 0.5 2.0 0. 0. 0. ٥. 0. HCRLB RECIP ENGINE 0. 0. 0. ++STRUCTURAL MATERIALS: 55-375 1.0 0.5 0. 2.0 0. 1.0 0. 0. 1.0 1.0 0. 0. 0. 2.0 0. 0. 0. FIBERGLASS COMPOSITES 3.0 0. 51.2.30 2.0 0. 1.0 0.5 0. 1.0 ٥. ٥. 0. 1.0 Ú. ٥. ٥. 0. KEVLAR COMPOSITES 0. 1.0 0.5 0. 3.0 0. 1.0 0. 42.485 2.0 0. 0. -1.0 1.0 0. 0. 0. GRAPHITE COMPOSITES 0. ٥.

•

۰, I

++TABULAR RESULTS FOR AL	L TZC	HNOL	GIES	FOLLO														
	* * * *	CRSH	CRUS	100	EMIS	EMTY	EXTR	FUEL	INTR	PILT	PRCH	9 G F	RELI	RIDE	8.F.T.Y	STAT	TOLO	FIG OF MERIT
CATEGORY WEIGHTS	5.43	6.82	7.92	7.91	1.61	4.34	4.29	8.16	7.04	7.35	8.51	7.29	8.85	6.11	9.42	6.24	6.87	
SATE ENGINE	2,0	٥.	1.0	1.0	1.0	3.0	0.	2.0	) 1.0	) 1.	0 -3.0	1.0	0 1.0	1.0	1.0	0.	1.0	85.017
STRT CHG ROT COMB ENGINE	1.0	с.	1.0	2.0	2.0	1.L	0.5	s 516	1.0	o 1.	D -2.0	) 1.(	0 1.0	1.0	1.0	0.	0.	84.205
NAT LAM FLOW AIRFOILS	1.0	0.	2.0	2.1	n 0.	1.0	9.	1.0	) 0.	0.	-0.9	i 2.(	0 -1.0	0.	0.	٥.	1.0	57.924
SPOILERS	-0.5	٥.	2.0	1.0	0.	1.0	е.	1.0	) 0.	0.	5 0.	1.	D 0.	2.0	0.	0.	0.	56.717
IBERGLASS COMPOSITES	2.0	0.	1.0	0.5	i 0.	2.0	0.	1.0	0.	ο.	1.0	1.0	0.0	٥.	0.	0.	0.	55.375
CEVLAR COMPOSITES	2.0	0.	1.0	0.5	5 0.	3.0	ΰ.	1.0	) O.	٥.	0.	1.0	<b>0</b> 0.	0.	0.	0.	0.	51.200
FOWLER FLAPS	-0.5	е.	2.0	1.0	) 0.	1.0	0.	1.0	0.	Ο.	-0.5	5 1.1	0.0	2.0	0.	0.	0.	48.785
FAPHITE COMPOSITES	2.0	9.	1.0	0.5	5 0.	3.0	ο.	1.0	0.	0.	-1.0	1.0	0.	0.	٥.	0.	с.	42.685
INTEG LON-COST WC LVLCR	0.	0.	С.	Ο.	0.	-0.5	Ο.	0.	0.	3.	0 -0.9	. O.	-0.5	0.	3.0	0.	0.	39.458
STR ADV DIESEL ENGINE	0.	ο.	٥.	2.0	) ) <b>.</b>	0.	0.9	5 3.0	0.0	5 0.	-2+0	1.	0.	0.	0.	0.	0.	36.205
DIGITAL DATA LINKS	0.	٥.	0.	0.	٥.	ο.	0.	٥.	0.	3.	o -2.0	) 0.	0.	0.	3.0	) 0.	0.	33.283
INTEG AVICNICS AND DSPYS	0.	٥.	С.	1.0	0.	0.	Ο.	0.	0.	3.0	o -2.0	0.	٥.	0.	z.0	0.	0.	31.767
MPROVES TURBOCHARGING	1.0	0.	1.0	0.5	i D.	0.	0.	1.0	0.	с.	-1.0	1.(	) û.	0,	0.	••	1.0	31.110
LOW/MEDM SPEED AIRFOILS	0.	0.	1.0	0.5	i 0.	0.5	٥.	0.5	5 O.	0.	-0.5	5 0. S	5 0.	0.	0.5	. O.	1.0	29.096
EADING EDGE DEVICES	1.0	٥.	0.5	0.5	i 0.	0.	0.	0.5	; O.	0.	-1.0	0.	5 0.	1.0	6.	0.	1.0	25.536
SYS STATUS DISPLAYS	0.	ο.	0.	0.	0.	٥.	٥.	0.	0.	2.0	0 -1.0	) 0.	0.	0.	2.0	0.	0.	25.027
IMPR STLL/SPNAERO TLRG	0.	0.	٥.	0.	0.	0.	ο.	0.	0.	0.	-0.5	i 0.	0.	0.	3.0	0.	0.	24.006
QUIET EFFICIENT PROPS	0.	0.	0.	0.5	5 0.	0.5	2.0	) 0.	2.0	0.	-1.(	0.	5 0.	0.	0.	0.	0.	23.901
SEPARATE SFC TECHNOLOGY	0.	0.	о.	+0.5	5 0.	-0.5	٥.	0.	0.	3.1	0 -2.(	) 0.	-0.5	3.0	1.0	) 0.	0.	22.219
LOAD LIMITING SEATS	0.	3.0	o .	0.	٥.	٥.	Ο.	٥.	0.	0.	<b>c.</b>	0.	0.	٥.	0.	-0.	5 0.	17.326
IMPROVED RESTRAINTS	0.	3.(	) O.	о.	0.	٥.	٥.	٥.	٥.	0.	-0.5	i 0.	0.	٥.	0.	0.	0.	16.191
NERGY ABSORBING FLOOR	٥.	3.0	) <b>0.</b>	0.	٥.	-0.5	0.	0.	0_	0.	-0-5	5 O-	0.	0.	0.	0_	0.	14.020
CTIVE STALL PREVENTION	ο.	0.	0.	0.	0.	-0.5	0.	0.	0-	0.	-1-1	N 0-	+0.5	0.	3.0	0.	0.	13,144
TRUTH COOLING	0.	0	<u> </u>	2,0	n n	-1.0			· · ·	<b>.</b> .		, we	_ 1 0				<u> </u>	17 477

1. N. F. F.

Table B.13. Overall Technology Ranking For Airplane A

,

. .

Ň

...

STHAT CHG RECIP ENGINE	••	0.	••	1.0	2.0	-0.5	••	1.0	••	•	-1.0	0.5	•	•	••	•	•	12.235
CR: D/SPLAYS	••	•	•	<b>.</b>	••		•		•	2.0	-2.0	•	-	9	0.5	•	•	11.233
DIRECT LIFT CONTROL	••	•		5	•	-0.5	•	<b>.</b>	•	1.0	-1.0	•	-0-5	-	1.0	•	0.5	11.204
HCRLP RECIP ENGINE	•	•	•	0.5	2.0		•	0.5		•	-0.5	0.5	•	•		<b>°</b>	•	12.631
ALCROWAVE LANDING SYSTEM	•	•	•	0.5	ċ	•	1.0	•	•	•	-1.0	•	•	5	1.0	9	<b>.</b>	9.147
NAVSTAR/GPS	••	•	•	•	•	•	<b>.</b>		•	1.0	-1-0	•		•	1.0	6	•	8.256
BRST/TEAR RESIS FUEL TKS	°.	3.0	•		•	-1.0	••	•		•	-1-0	5	•	•	<b>.</b>		•	7.593
ICEPHOBIC SFC COATINGS	-0-5	•		-0.5		-1.0	•	•		•	-1.0	-0-5	5		3.0	•		8+0-5
ACTIVE LAMINAR FLOW CTL	2.0	•	2 <b>.0</b>	-2.0	•	<b>.</b>	•	2.0	•	•	-3.0	3.0	-3.0	•		5	1.0	3.841
INTEGRATED YAN DAMPER	••	•	•	•		•	•	•	•	1.0	-2.0	•	•	2.6	•	•	•	2.539
HICRO HUD	•	•	•	-0.5	•	-0.5		•	ö	2.0	-2.0	•	-1-	•	2.0	•	•	1.535
AC ELECTRICAL SYSTEMS	•	•	•	•				5			•	•	6	•	•			•
ACT CTLS FOR RLK STOLTY	0.5	•	0.5	1.0	••	1.0	••	1.0	•	•	-3.0	1.0	-1.0	9	ä	•	•	-0.030
BUCTED PROPULSORS	-0.5	•	-1.0	-1.0	5	-1.0	3.0	-1.0	3.0	•	-1-0	-2.0	•	ď	0.5	•	2.0	-1.703
SGL LEVER THRUST/DRG CTL	•	•	•	°,	•		э.	<b>.</b>	•	2.0	-1.0	•	-1-0	5	•		•	-2.671
WINGLETS	0.5	۰.	•	0.5	•	-2.0	••	0.5	•	5	-1.0	0.5	4	•		9	•	-2.804
FLUIDIC AUTO FLT CTL SYS	-0.5	•	•	•	•	-0.5	•	-0.5	•	•	-1.0	-0-5		ð	•	0	6	-12.269
LOW-BRAG SFC COATINGS	••	<b>.</b>	•	-0.5		-0.5		0.5	•		-1-0	•	•	•	•	•	-0-5	-13.997
DIRECT SIDE FORCE CTL	•	•	•	-0.5	•	-0.5	•	•	•	0.5	-3.0	•	-0-3	2.0		•	•	-14.040
ACT REDE SMOOTHING	•	•	•	-0.5	•	-0.5	•	•		0.5	-3.0	•	-	T.	•	•	9	-14.040
LASER GYROS	••	•	•	-0-5	•		•	•		•	-3.0	•		•	0.5	•	6	-15.933
LOW-LEVEL PRESSURIZATION	•	•	•	-0.5	•	-0.5		-0.5	2.0	•	-1-0	-0.5		•	•	•	0	-17.141
HEADS-UP DISPLAT	•	•	•	-1.0	•	-1.0	•	•	•	2.0	• 1.•	•			1.0	•	9	-22.525
LORAN C	•	•	•	-0.5	••	-0.5	•	•	•	•	-1-0	•	-0-	•	-9-5	•	•	-23.776
FIBER DPTICS (BATA TRNS)	•	•	•		•	•	•	•	•	•	-3.0	ċ	•	•	•	•	•	-25.545
ONEGA	••	•	•	-0.5	•		<b>.</b>	•	•	•	-1-0	•		•	-0-2	•	•	-24.033
BOPPLER KAVIGATION	•	•	•	- 2.0	•	-0.5	•			•••	- 2 - 0	•		•	•	••	•	-34.517
ACT FLUTTER SUPPRESSION	•	ċ	•	-0.5	•		•	•	•	•	-3.0	•	-1-0	•	•	0		-33.352
ACT GUST ALLV	•	•	•	-1.0	•		•	•		•	-3.0	•	-1.0	•	•	•	•	-42.305
F. Y-8Y-13RE	•	•	•	-1-0		-1.0		•		•	-3.0			•	•		•	-46.445
fLY-8Y-LIGHT	•	Ľ	•	-1.0		-1.0	•	•			-3.0	•		ď	•	•	•	-45.645
INERTIAL NAVIGATION	•	•		-3.0	•	-0.5	•	•	•		-3.0	•	-1.0		•	•		-52.936

.

A set and a s

, . ,

;

Table B.13. Overall Technology Ranking For Airplane A (concluded)

-----

A .....

.

Table B.14. Technology Ranking By Technology Group For Airplane B

***TABULAR RESULTS BY CA	EGORI	FOLI	LO¥:															
	CEIL	C RSH WTHY	CRUS SPED	DOC	EMIS SION	ENTY WGHT	EXTR NOIS	FUEL EFFC	INTR NUIS	PILT WRKL	PRCH PRCE	RGE	RELI BLTT	RIDE	SFTV	S TA T CHFT	TOLD PERF	FIG O VERI
CATEGORY WEIGHTS	5.45	7.33	7.48	9.45	2.16	5.66	5.41	8.61	7.45	7.00	7.76	7,18	9.55	7.24	9.54	6.71	7.41	
++AEPCDYNAMICS:																		
SPOILERS	0.	0.	1.0	1.0	0.	0.5	0.	0.5	٥.	0.5	0.	0.5	0.	1.0	) 0.	٥.	0.	38.39
LOW/PEDM SPEED AIPFOILS	C.	с.	1.0	0.5	· 0.	0.5	٥.	0.5	٥.	0.	-0.5	0.5	٥.	0.	0.5	i 0.	1.0	31.22
LEADING EDGE DEVICES	1.0	e.	0.5	0.5	i 0.	σ.	0.	0.5	0.	٥.	-0.5	0.5	0.	0.9	i 0.	0.	1.0	28.95
NAT LAM FLOW AIRFOILS	٤.5	5 O.	1.0	1.0	) O.	1.0	ο.	1.0	C.	0.	-0.5	1.0	-2.0	0.	0.	0.	1.0	25.53
FOULER FLAPS	0.	С.	0.5	0.5	i 0.	0.	0.	0.5	٥.	0.	0.	0.5	0.	1.0	0.	0.	٥.	23.59
INPR STEL/SPNAERO TERG	Ο.	0.	٥.	٥.	٥.	٥.	Ο.	0.	٥.	0.	-0.5	0.	0.	0.	2.0	<b>.</b>	0.	15.20
WINGLETS	0.5	0.	0.	0.5	0.	-2-0	0.	0.5	٥.	0.	-1.0	0.5	0.	٥.	<b>G.</b>	٥.	0.	-3.73
LOW-DRAG SEC COATINGS	٥.	٥.	٥.	-0.5	0.	-0.5	٥.	٥.	0.	0.	-0.5	٥.	٥.	٥.	0.	0.	-0.5	-15.13
ACTIVE LAMINAR FLOS CTL	2.0	0.	٥.	- 3. (	0.	٥.	٥.	1.0	0.	0.	-3.0	1.0	-3.0	0.	0.	0.	1.0	-45.18
++AIRCRAFT SYSTEMS:																		
ICEPHOBIC SFC COATINGS	1.0	0.	0.	C.	c.	1.0	0.	0.5	0.	0.	1.0	0.	0.5	i 0.	0.9	5 0.	0.	32.71
AC ELECTRICAL SYSTEMS	0.5	i 0.	0.	٥.	0.	0.5	٥.	0.	٥.	0.	1.0	0.5	0.	0.	0.	0.	0.	14.90
++CRASHWORTHINESS:																		
LOAD LIMITING SEATS	0.	3.0	D 0.	0.	0.	0.	0.	0.	0.	0.	0.	0.	Ũ.	٥.	0.	-0.5	0.	18.64
IMPROVED RESTRAINTS	0.	3.0	<b>.</b>	0.	٥.	0.	٥.	0.	0.	0.	-0.5	0.	0.	0.	0.	0.	Đ.	18.11
ENERGY ABSORBING FLOOR	٥.	3.0	) O.	0.	٥.	-0.5	٥.	0.	٥.	0.	-0.5	0.	0.	٥.	n.	0.	0.	15.29
BRST/TEAR RESIS FUEL TKS	0.	3.0	) <b>0.</b>	0.	0.	-1.0	0.	с.	0.	0.	-1-0	0.	0.	٥.	0.	0.	0.	8.58

Table B.14. Technology Ranking By Technology Group For Airplane B (continued)

en a casa Mar

٠

محادية والمراجع

		•	~				•	•	•			•	•			•	A	38 7-4
SEPARATE SFC TECHNOLOGY	0.	0.	0.	0.	0.	0.	0.	0.	0.	1.0	0.	0.	0.	1.0	1.0	9.	0.	<i>23.78</i> 0
ACTIVE STALL PREVENTION	0.	0.	0.	0.	0.	0.	0.	U.	0.	0.	-1.0	0.	-0.5	0.	3.0	U.	U.	10.07/
DINECT LIFT CONTROL	0.	0.	0.	0.	0.	-0.5	0.	0.	0.	1.0	-1.0	0.	-0.5	1.0	1.0	0.	0.3	12.120
INTEG LOW-COST WE LVLER	0.	0.	0.	0.	0.	0.	0.	0.	0.	0.	1.0	0.	0.	0.	0.	g.	0.	/./01
INTEGRATED YAW DAMPER	0.	0.	0.	0.	0.	0.	0.	0.	0.	1.0	-2.0	0.	0.	2.0	0.	0.	0.	3.736
ACT CTLS FOR RLX STBLTY	0.5	0.	0.5	1.0	0.	1.0	0.	1.0	0.	0.	-3.3	1.0	-1.0	0.	0.	0.	0.	4.321
ACT RIDE SMOOTHING	0.	0.	0.	-0.5	0.	0.	0.	0.	0.	1.0	-2.0	0.	-0.5	3.0	0.	0.	0.	3.671
FLUIDIC AUTO FLT CTL SYS	-0.5	0.	0.	0.	0.	0.	0.	-0.5	0.	0.	0.	-0.5	1.0	0.	0.	0.	0.	-1.070
SGL LEVER THRUST/DRG CTL	0.	0.	0.	0.	0.	0.	0.	0.	0.	2.0	-1.0	0.	-1.0	0.	0.	0.	0.	-3.312
DERECT SIDE FORCE CTL	0.	0.	0.	-0.5	0.	0.	0.	0.	0.	0.	-3.0	э.	-0.5	3.0	9.	0.	0.	-11.065
FLY-BY-LIGHT	0.	0.	0.	-1.0	0.	0.5	0.	0.	0.	з.	-3.0	J.	r,	0.	0.	0.	0.	-29.905
FLY-BY-WIRE	0.	0.	0.	-1.0	0.	0.	0.	0.	0.	0.	-3.0	0.	0.	0.	0.	0.	0.	-32.734
ACT GUST ALLEVIATION	0.	0.	٥.	-1.0	0.	0.	0.	0.	0.	0.	-3.0	0.	-1.0	1.0	0.	0.	0.	-35.043
ACT FLUTTER SUPPRESSION	0.	0.	0.	-0.5	0.	0.	0.	0.	0.	0.	-3.0	0.	-1.0	0.	0.	0,	0.	-37,556
INFORMATION SYSTEMSI																		
INTEG AVIONICS AND DSPYS	0.	ο.	0.	1.0	0.	0.	٥.	0.	0.	3.0	-1.0	0.	0.	0.	2.0	0.	0.	41.772
DIGITAL DATA LINKS	0.	0.	0.	0.	0.	0.	0.	0.	0.	3.0	-2.0	0.	0.	0.	3.0	0.	0.	34.194
SYS STATUS DISPLAYS	0.	0.	٥.	0.	0.	0.	٥.	0.	0.	2.0	-1.0	0.	0.	0.	5.O	0.	0.	25.321
CRT DESPLAYS	0.	0.	0.	0.	0.	0.	Ο.	0.	0.	2.0	-1.0	0.	1.0	0.	0.5	0.	0.	20.554
ATCHO NAD	0.	0.	0.	-0.5	0.	<b>C.</b>	0.	0.	0.	2.0	-2.0	0.	-1.0	0.	2.0	0.	0.	3.281
LASER GYROS	0.	C.	0.	-0.5	٥.	0.	Ο.	0.	0.	0.	-3.0	0.	1.0	0.	0.5	0.	0.	-13.681
FIBER OPTICS (DATA TRNS)	0.	0.	c.	0.	0.	0.	0.	0.	0.	0.	-3.0	0.	0.	0.	0.	¢.,	0.	-23.283
HEADS-UP DISPLAY	0.	0.	0.	-1.0	٥.	-1.0	0.	0.	0.	2.0	-3.0	0.	-1.0	0.	1.0	0.	0.	-24.39
NAVIGATION CONCEPTS:																		
MICROWAVE LANDING SYSTEM	0.	0.	0.	1.0	٥.	٥.	1.0	0.	0.	0.	-0.5	0.	0.	0.	1.0	0.	0.	20.527
NAVSTAR/GPS	0.	0.	0.	0.	0.	٥.	٥.	0.	0.	1.0	-1.0	0.	0.	0.	1.0	0.	0.	8.781
LORAN C	0.	0.	0.	-0.5	٥.	-0.5	0.	<b>0</b> .	0.	0.	-1-0	0.	-0-5	٥.	-1-0	٥.	0.	-29.43

1 i k

F

4 k i

₹

. . . . .

÷.

Table B.14. Technology Ranking By Technology Group For Airplane B (concluded)

0. 0. 0. -0.5 0. 0. 0. 0. 0. -1.0 0. -1.0 0. -1.0 0. OMEGA -31.577 ٥. SCPPLER NAVIGATION C. D. D. -2.0 D. -0.5 D. ٥. 0. 1.0 -2.3 0. -1.0 0. 0. 0. ٥. - 39.802 0. 0. -3.0 C. -0.5 D. 0. 0. 1.0 -3.0 0. -1.0 0. U. 0. INERTIAL NAVIGATION ٥. -57.013 \*\*NOISE: 0.5 0. 0.5 2.0 0. 2.0 0. -1.0 0.5 0. QUIET EFFICIENT PROPS C. D. D. ٥. 0. ٥. 0. 29.114 DUCTED PROPULSORS -0.5 0. -1.0 -0.5 0. -1.0 3.0 -0.5 3.0 0. -1.0 -1.0 0. ٥. 0.5 0. 18.350 2.0 LOW-LEVEL PRESSURIZATION C. 0. 0. -0.5 C. -0.5 0. -0.5 2.0 G. -1.0 -0.5 -1.0 D. 0. 0. -17.452 0. ++PROPULSION: GATE ENGINE 0. 0. 0. 1.0 0. 0. 0. 1.0 0. ο. 0. 2.0 0. 32.419 ٥. 0. 0. 0. STRT CHG RCT COMB ENGINE 0. 0. 0. 1.0 1.0 -1.0 0. 0. 0. 0. 1.0 1.0 0. 0. 0. ٥. ٥. 20.893 ++STRUCTURAL MATERIALS: FIGERGLASS COMPOSITES 2.0 0. 1.0 0.5 0. 2.0 0. 1.0 0. 0. 1.0 1.0 0. 57.962 ٥. 0. ٥. 0. KEVLAR COMPOSITES 2.0 0. 1.0 0.5 0. 3.0 9. 1.0 0. 0. ٥. 1.0 0. 55.858 0. 0. 0. 0. GRAPHITE COMPOSITES 2.0 0. 1.0 0.5 0. 3.0 0. 1.0 0. 0. -1.0 1.0 0. 0. 0. 0. 48.097 0.

c

s to all allocated to prove shake to come shake

.

. ..

. \*

۰.

.

Table B.15. Overall Technology Ranking For Airplane B

ALAPLANE

ì

£

28- 95 FIC OF 57.942 55.454 38.390 31.226 23.700 23.597 20.493 20.527 11.644 11.350 41.772 32.419 29.114 11.118 31.10 32.714 25.323 20.554 48.097 25.531 : 2.5 •••• 2.5 **.** ď å å ð CEIL WIWY SPED DOC SION WENT MOIS EFFC NOIS WALL PACH REL RIDE STAT TOLD 6 5 CATEGORT WEIGHTS 5.45 7.33 7.48 9.45 2.16 5.66 5.41 8.61 7.45 7.00 7.74 7.18 9.55 7.24 9.54 6.71 7.41 5 ē.5 • 5 ð • 6 5 5 o.s 2.0 0.1 5.9 2.0 5.9 2.2 . 4 . . . . å . å • ... 0.5 . • . • 5 ð å 5 . • ď -2.0 1.0 :. ď . 5 5 9 ď ð 5 . 5 • • 5 9 ď ð 0. ... ... 2.0 0.5 -1.0 ... 0.5 : 5.0 0.5 5 • • • 6 • • ö • 5 .. -1-0 -0.5 -0.5 5.0 -1.0 -2.0 : -1.0 -0.5 -1.0 2 2.0 -1.0 • 9 Ş ò . • . • 0.2 2 • ? . • . • • 3.0 . • . . • ď ċ . ċ • 5 å å • 3.0 -0.5 ... ? 2.1 1.0 ... 0.5 5.0 ... 0.5 5 . 5 6 . 5 . . . •••• . ; . • • • • • . . **.** . ò . • -1.0 3.0 ••• 0.5 <u>د</u> 3.0 0.5 •• ? 3 . . . • . • . • • • . ; • . . . . . • ; 5 • . • • . 5 . ó \*\*\*TABULAR RESULTS FOR ALL TECHMOLOGIES FOLLOW: ••• : 1.0 °. o. 5 0.5 °. : -1.0 -0.5 1.0 e.s 0.5 <u>°</u> :0 . . 6 5 5 . ? • ? ••• ... ••• 0.5 <u>.</u> ; . . • . . . . . ő . 3.0 3.0 **.** • . • • . 0 . . . . • . . . • 2.0 -0.5 °.° **2.**0 0.5 ••• 2 . 5 . . • . • . . • . • • 9 INTEG AVIONICS AND DSPYS STRT CHG ROT COMB ENGINE ALCROWAVE LANDING SYSTEM LOW/REAM SPEED AIRFOILS SEPARATE SFC TECHNOLOGY ICEPHOBIC SFC COATINGS QUIET EFFICIENT PROPS FIBERGLASS COMPOSITES NAT LAP FLOW AIRFOILS LEADING EDGE DEVICES STS STATUS BESPLAYS GRAPHITE COMPOSITES LOAD LEFITING SEATS IMPROVED RESTRAINTS DIGITAL DATA LINKS KEVLAR COMPOSITES BUCTED PROPULSORS FOULER FLAPS CAT DISPLATS GATE ENGINE SPOILERS

Table	B.15.	Overall	Technology	Ranking	For	Airplane	B	(concluded)
-------	-------	---------	------------	---------	-----	----------	---	-------------

AC ELECTRICAL SYSTEMS	0.5	0.	0.	٥.	0.	0.5	٥.	٥.	0.	0.	1.0	L.S 0.	0.	0.	0.	0.	16.904
ACTIVE STALL PREVENTION	٥.	0.	٥.	0.	Ο.	0.	٥.	0.	0.	٥.	-1.0	00.5	0.	3.0	0.	0.	16.097
ENERGY ABSORBING FLOOR	0.	5.0	٥.	0.	0.	-0.5	٥.	٥.	0.	٥.	-0.5	0. 0.	0.	٥.	0.	0.	15.290
IMPR STLL/SPNAERO TLRG	0.	е.	С.	0.	0.	0.	Ο.	0.	0.	٥.	-0.5	0. 0.	٥.	2.0	٥.	0.	15.207
DIRECT LIFT CONTROL	0.	٥.	٥.	0.	0.	-0.5	Ο.	0.	0.	1.0	-1.0	00.5	1.0	1.0	0.	0.5	12.120
NAVSTARJGPS	0.	0.	0.	Û.	0.	0.	٥.	٥.	0.	1.0	-1,0	0. 0.	٥.	1.0	٥.	0.	8.781
BR\$7/TEAR RESIS FUEL TKS	0.	3.0	0.	с.	0.	-1.0	ο.	0.	0.	٥.	-1.0	0. 0.	0.	Ø.	٥.	0.	8.581
INTEG LOW-COST WG LVLER	0.	0.	٥.	0.	٥.	0.	٥.	٥.	٥.	0.	1.0	0. 0.	0.	0.	0.	0.	7.761
INTEGRATED YAW DAMPER	0.	0.	٥.	0.	Ö.	0.	٥.	0.	0.	1.0	-2.0	0. 0.	2.0	0.	0.	0.	5.952
ACT CTLS FOR REX STBLTY	0.5	0.	0.5	1.0	0.	1.0	٥.	1.0	0.	٥.	-3.0	1.0 -1.0	0.	٥.	0.	0.	4.521
ACT RIDE SHOOTHING	0.	α.	0.	-0.5	0.	٥.	G.	٥.	0.	1.0	-2.0	00.5	3.0	0.	0.,	0.	3.691
WICRO HUD	٥.	c.	0.	-0.5	0.	0.	θ.	٥.	0.	2.0	-2.0	01.0	٥.	2.0	0.	0.	3.289
FLUIDIC AUTO FLT CTL SYS	-0.5	٢.	٥.	0.	٥.	0.	٥.	-0.5	0.	0.	0.	-0.5 1.0	3.	0.	0.	0.	-1.070
SGL LEVER THRUST/DRG CTL	0.	J.	ο.	e.	0.	٥.	э.	٥.	0.	2.0	-1.0	01.0	0.	0.	٥.	0.	-3.312
#INGLETS	0.5	0.	٥.	0.5	с.	-2.0	σ.	0.5	с.	٥.	-1.0	0.5 0.	0.	0.	0.	0.	-3.73
DIRECT SIDE FORCE CTL	0.	ο.	с.	-0.5	٥.	0.	٥.	٥.	٥.	٥.	-3.0	00.5	3.0	0.	0.	0.	-11.06
LASER GYROS	c.	0.	٥.	-0.5	0.	0.	٥.	0.	0.	٥.	-3.0	0. 1.0	٥.	0.5	0.	0.	-13.68
LOW-DRAG SFC COATINGS	0.	٥.	с.	-0.5	0.	-0.5	٥.	0.	0.	0.	-0.5	0. 0.	٥.	0.	0.	-0.5	-15.13
LOW-LEVEL PRESSURIZATION	<b>6.</b>	0.	0.	-0.5	0.	-0.5	٥.	-0.5	2.0	0.	-1.0	-0.5 -1.0	0.	¢.	0.	0.	-17.85
FIBER OPTICS (DATA TRNS)	0.	0.	0.	0.	٥.	0.	ο.	0.	0.	0.	-3.0	0. D.	0.	0.	0.	0.	-23.28
HEADS-UP DISPLAY	C.	0.	α.	-1.0	0.	-1.0	0.	0.	0.	2.0	-3.0	01.0	٥.	1.0	0.	0.	-76.39
LORAN C	٥.	σ.	ο.	-0.5	0.	-0.5	ο.	0.	0.	0.	-1.0	00.5	0.	-1.0	0.	0.	-29.63
FLY-BY-LIGHT	0.	٥.	Ο.	-1.0	σ.	0.5	٥.	0.	0.	٥.	-3.0	0. 0.	0.	0.	0.	0.	-29.90
OMEGA	0.	0.	ο.	-0.5	ο.	0.	٥.	0.	0.	0.	-1.0	01.0	D.	-1.0	0.	э.	-31.57
FLY-BY-WIRE	0.	σ.	ο.	-1.0	0.	0.	٥.	0.	0.	0.	-3.0	0. 0.	0.	0.	0.	0.	-32.73
ACT GUST ALLEVIATION	0.	0.	0.	-1.0	٥.	0.	0.	0.	0.	0.	-3.0	01.0	1.0	0.	0.	0.	-35.04
ACT FLUTTER SUPPRESSION	0.	0.	0.	-0.5	٥.	0.	ο.	0.	0.	<b>c.</b>	-3.0	01.0	0.	0.	0.	0.	-37.55
DOPPLER NAVIGATION	0.	0.	ο.	-2.0	ο.	-0.5	٥.	0.	0.	1.0	-2.0	01.0	0.	0.	0.	0.	-39, 802
ACTIVE LAWINAR FLOW CTL	2.0	0.	ο.	-3.0	0.	0.	0.	1.0	0.	0.	-3.0	1.0 -3.0	٥.	0.	0.	1.0	-65.18
INERTIAL NAVIGATION	•	0	~	-1.0	~		•										

. \*

Table B.16. Overall Technology Ranking With Empty Weight = 9.5 For Airplane A

.

-\*<sup>r</sup>---<del>-</del>2

٠

,

AIRPLANE A

+++TABULAR RESULTS FOR ALL TECHMOLOGIES FOLLOW:

716 OF CEIL UTHY SPED DOC SHOT ENTY EXTR FUE IMTR PILT PACH RELI RIDE STAT TOLD CEIL UTHY SPED DOC SION WGHT HOIS EFFC HOIS WAKL PACE AGE BLTY GLTT SFTY CAFF PERF CATEGORY WEIGHTS 5.43 6.82 7.92 7.91 1.61 9.50 4.29 8.16 7.04 7.35 8.51 7.29 8.85 6.11 9.42 6.24 6.87

17E E4618E	2.0	•	1.0	1.0	-	3.0	•	2.0	1.0	1.0	-3.0	1.0	1.0	<b>.</b> .	1.0		1.0	100.497
TRT CHG ROT COMB ENGINE	1.0	•	1.0	2.0	2.0	1.0	0.5	2.0	1.0	1.0	-2.0		1.0	1.0	1.0			89.345
EVLAR COMPOSITES	2.0	•	1.0	0.5	•	3.0		1.0	•			1.0			•			66.480
BERGLASS COMPOSITES	2.0	•	1.0	0.5	•	2.0	••	1.0	•	•	1.0	1.0			•	•	•	65.495
AF LAM FLOW AIRFOILS	1.0	•	2.0	2.0		1.0		1.0			-0.5	2.0 -	-1.0	<b>.</b>	°.	•	1.0	43.005
DILERS	-0-5	•	2.0	1.0	•	1.0		1.0	•	0.5	.0	1.0	ð	2.0		ď	°.	61.877
RAPHITE COMPOSITES	2.0	•	1.0	0.5	•	3.0	•	0.1			-1.0	1.0	•	<b>.</b>	•			58.165
DULER FLAPS	-3.5	•	2.0	1.0		1.0	:	1.0			-0.5	1.0		2.0	•	ď	•	53.945
NTEG LON-COST NG LVLER	•	•	•	•		-0.5				3.0	-0.5		-0.5		3.0	e e	6	35.47
STR ADV DIESEL ENGINE	5	•	•	2.0	•		0.5	3.0	0.5		-2.0	1.0	ď	÷		ő	6	34.205
IGETAL BATA LINKS			•		•				•	3.0	-2.0				3.0			33, 283
UTEG AVIONECS AND DSPTS	•	ċ	•	1.0	•	•	•	•	•	3.0	-2.0				2.0			31.767
JW/MEBN SPEED AIRFOILS	•		1.0	0.5		0.5		0.5		•	-0.5	0.5			0.5		1.0	31.676
PROVED TURBOCHARGING	1.0	•	1.0	0.5		•	•	1.0	5		-1.0	1.0	•				1.0	31.110
ULET EFFLCIENT PROPS	•	•		0.5		0.5	2.0		2.0		-1-0	0.5	•					24.487
EADING EDGE DEVICES	1.0	•	0.5	0.5	•	•		0.5		•	-1-0	0.5					0.1	25.536
TS STATUS BESPLATS	•	•	•	ö	•	•				2.0	-1-0				2.0			25-027
RPR STLL/SPNAERO TLRG	•	•	•	•	•	•		•		5	-0.5			•	9.0	•	ď	24.00
EPARATE SIC TECHNOLOGY	•	•	•	-0.5	•	-0.5	•	•	•	3.0	-2.0		·0.5	3.0	0.1	ő		19.439
DAD LIMITING SEATS	•	3.0	•	•		•	•		•	•	•	6				-0.5		17.326
PPOVED RESTRAINTS	•	3.0	•	•	•	•		•			-0.5	5		.0	ď	Ğ	•	

Table B.16. Overall Technology Ranking With Empty Weight = 9.5 For Airplane A (continued)

01.0
•••
~ ~
0
; ;
c
c
-
:
5
5 0
;;
VE LANSMAN FLOW CTL

•

9

e

-

1

. l

DOPPLER RAVIGATION ACT GUST ALLV FLV-DV-WIRE	J. 0. 0.	0. 0. 0.	0. 0. 0.	-2.0 -1.0 -1.0	0. 0. 0.	-0.5 0. -1.0	D. 0. 0.	0. 0. 0.	0. 0. 0.	1.0 0. 0.	-2.0 -3.0 -3.0	0. 0. 0.	-1.0 -1.0 -1.0	0. 0. 0.	0. 0. 0.	0. 0. 0.	0. 0. 0.	-39.09 -42.30 -51.80
FLT-GT-LIGHT INERTIAL NAVIGATION	0. C.	0. 0.	0. 0.	-1.0 -3.0	0. 0.	-1.0 -0.5	¢. Q.	0. 0.	0. 0.	0. 1.0	-3.0 -3.0	0. 0.	-1.0 -1.0	0. 0.	0. 0.	0. 0.	0. 0.	-51.80 -55.51

.

e . .

 $(\cdot)$ 

Table B.16. Overall Technology Ranking With Empty Weight = 9.5 For Airplane A (concluded)

н н<sub>а</sub>

.

.

\*

-

Å,

1

,

į

يراقب الرائد المراجع المراجع المراجع المراجع المراجع المراجع

Table B.17. Overall Technology Ranking With Empty Weight = 9.5 For Airplane B

. ....

3 1 F 18 19

where a maximum of all patients where a provide applied

We have not an advantaged when here and the set

AIRPLANE B +++TABULAR RESULTS FOR ALL TECHNOLOGIES FOLLOW: CRSH CRUS EMIS ENTY EXTR FUEL INTR PILT PACH RELI RIDE STAT TOLD CEIL WIWY SPED DOC SION WONT HOIS EFFC NOIS WAKL PACE RGE BLTY GLTY SFTY CHFT PERF FIG OF CATEGORY WEIGHTS 5.45 7.33 7.48 9.45 2.16 9.56 5.41 8.61 7.45 7.00 7.76 7.18 9.55 7.24 9.54 6.71 7.47 KEVLAR COMPOSITES 67.387 2.0 0. 1.0 0.5 0. 3.0 0. 1.0 0. 0. 0. 1.0 0. 0. ٥. ٥. ٥. FIBERGLASS COMPOSITES 2.0 0. 1.0 0.5 0. 5\*0 0. 1.0 ٥. ٥. 1.0 1.0 0. ٥. ٥. 0. 65.648 ۵. GRAPHITE COMPOSITES 2.0 0. 1.0 0.5 0. 3.0 0. 1.0 ٥. ٥. -1.0 1.0 ٥. 0. 0. 0. 59.626 0. INTEG AVIONICS AND DSPYS ٥. 0. 0. 1.0 0. ٥. 0. 0. 0. 3.0 -1.0 0. 0. ΰ. 2.0 0. 0. 41.772 SPOILERS 0. 0. 1.2 1.0 0. 0.5 0. 0.5 0. 0.5 0. 0.5 ٥. 1.0 ٥. ٥. 0. 40.312 ICEPHOBIC SFC COATINGS 1.0 ο. Ο. 0. 0. 1.0 0. 0.5 0. 0. 1.0 0. 0.5 0. 0.5 0. 0. 36.556 DIGITAL DATA LINKS Ο. 0. **c.** 34.104 0. ٥. ٥. 0. ٥. 0. 3.0 -2.0 0. ٥. 0. 3.0 0. 0. LOW/MEDM SPEED AIRFOILS 0. 1.0 0.5 Ο, 0.5 0. 0.5 0. D. -0.5 33.148 ٥. 0.5 0. 0. 0.5 0. 1.0 0. ο. 0. 1.0 ٥. 2.0 32.419 GATE ENGINE 0. 0. 0. 1.0 ٥. 0. 0. ۵. ٥. 0. 0. QUIET EFFICIENT PROPS ٥. 0.5 0. 0.5 2.0 0. 2.0 0. -1.0 31.035 ٥. 0. 0.5 0. 0. 0. 0. 0. NAT LAH FLOW AIRFOILS 0.5 C. 1.0 1.0 0. 1.0 0. 1.0 0. 0. -0.5 1.0 -2.0 0. 6. 0. 1.0 29.374 0. LEADING EDGE DEVICES 1.0 0. 0.5 0.5 0. 0. ٥. 0.5 0. -0.5 0.5 0. 0.5 ٥. ٥. 1.0 28.952 SYS STATUS DISPLAYS 25.323 0. 0. 0. 0. 0. 0. 2.0 -1.0 0. ٥. С. 0. 0. 0. 5.0 0. 0. 0. SEPARATE SFC TECHNOLOGY 0. ٥. 0. ٥. 0. 0. 0. 0. 1.0 0. 0. 0. 1.0 1.0 23.780 0. 0. FOWLER FLAPS 23.597 ٥. 0. 0.5 0.5 0. 0. ٥. 0.5 0. 0. 0. 0.5 0. 1.0 ٥. 0. 0. CRT DISPLAYS 0. 0. 0. ٥. 0. 0. Ο. 0. 0. 2.0 -1.0 0. 1.0 ٥. 0.5 0. 0. 20.554 MICROWAVE LANDING SYSTEM 0. ٥. 1.0 0. 1.0 ٥. ٥. D. -0.5 20.527 ٥. 0. 0. ٥. ٥. 1.0 0. 0. AC ELECTRICAL SYSTEMS 0.5 0. 0. 0. 0. 0.5 ٥. 0. ō. 0. 1.0 18.825 0.5 0. 0. ٥. 0. 0. LOAD LIMITING SEATS 0. 0. 3.0 ο. 0. 0. 0. 0. ٥. 0. 0. 18.646 0. a. ٥. 0. -0.5 0. IMPROVED RESTRAINTS ۰.0 0. 0. 0. 0. 0. 0. 0. 0. 0. 18.118 -0.5 ú. 0. ٥. 0. 0. 0. STRT CHG ROT COMB ENGINE 0. ٥. 1.0 1.0 -1.0 0. ٥. ٥. ٥. 0. 1.0 1.0 0. ٥. 0. 17.050 0. 0.

. . .

1.1

and a grant again the second second

5

Table B.17. Overall Technology Ranking With Empty Weight = 9.5 For Airplane B (concluded)

ومستهور والمعارية بالمتعامية والمتعامين والمتعارية

• •

٠

ACTIVE STALL PREVENTION	0.	0.	0.	0.	0.	ο.	٥.	6.	<b>6.</b>	01.0	00.5	0.	۰.0	0.	0.	16.097	
IMPR STLL/SPNAERO TLRG	0.	0.	0.	<b>^.</b>	٥.	٥.	D.	0.	0.	00.5	0. 0.	0.	2.0	0.	0.	15-207	
BUCTED PROPULSORS	-0.5	0.	-1.0	-0.5	с.	-1.0	3.0	-0.5	3.0	01.0	-1.0 0.	٥.	0.5	0.	2.0	14.507	
ENEIGT ABSORBING FLOOR	ο.	3.0	٥.	0.	0.	-0.5	0.	0.	0.	00.5	0. 0.	٥.	٥.	ø.	0.	13.349	
DIRECT LIFT CONTROL	0.	0.	0.	0.	0.	-0.5	0.	0.	0.	1.0 -1.0	00.5	1.0	1.0	0.	0.5	10.199	
NAVSTAR/GPS	0.	0.	0.	C.	Ο.	0.	٥.	0.	0.	1.0 -1.0	0. 0.	0.	1.0	0.	0.	8.781	
ACT CTUS FOR RUX STBUTY	0.5	٥.	0.5	1.0	0.	1.0	0.	1.0	0.	03.0	1.0 -1.0	<b>G.</b>	0.	0.	0.	8.372	
INTEG LOW-COST NG LVLER	0.	0.	0.	0.	c.	0.	0.	0.	0.	0. 1.0	0. 0.	<b>G.</b>	0.	0.	0.	7.761	
INTEGRATED YAW DAMPER	0.	0.	ο.	0.	٥.	0.	٥.	0.	0.	1.0 -2.0	0. 0.	2.0	0.	0.	0.	5.952	
BRST/TEAR RESIS FUEL TES	٥.	3.0	0.	0.	٥.	-1.0	٥.	0.	0.	01.0	a. o.	0.	0.	0.	0.	4.738	
ACT RIDE SMODTHING	0.	0.	0.	-0.5	0.	0.	٥.	٥.	0.	1.0 -2.0	00.5	3.0	0.	٥.	0.	3.691	
MICRO HUD	٥.	0.	٥.	-0.5	0.	٥.	0.	0.	0.	2.0 -2.0	01.0	9.	2.0	0.	0.	3.289	
FLUIDIC AUTO FLT CTL SYS	-0.5	0.	٥.	đ.	ú.	٥.	٥.	-0.5	0.	0. 0.	-0.5 1.0	0.	٥.	0.	0.	-1.070	
SGL LEVER THRUST/DRG CTL	0.	0.	0.	0.	0.	0.	0.	0.	0.	2.0 -1.0	01.0	0.	0.	0.	0.	-3.312	
DIRECT SIDE FORCE CTL	0.	0.	0.	-0.5	0.	0.	0.	0.	0.	03.0	00.5	3.0	G.	0.	0.	-11.068	
UINGLETS	0.5	0.	٥.	5	0.	-2.0	0.	0.5	0.	01.0	0.5 0.	٥.	9.	9.	0.	-11_418	
LASER GTROS	0.	0.	0.	-0.5	0.	0.	٥.	0.	0.	03.0	0. 1.0	0.	0.5	0.	0.	-13.689	
LOW-LRAG SFC COATINGS	0.	C.	0.	-0.5	٥.	-0.5	٥.	0.	0.	00.5	0. 0.	٥.	0.	0.	-0.5	-17.059	
LLW-LEVEL PRESSURIZATION	0.	0.	٥.	-0.5	0.	-0.5	0.	-0.5	2.0	01.0	-0.5 -1.0	0,	0.	0.	٥.	-19.773	
FIBER OPTICS (BATA TRNS)	0.	0.	0.	<b>6.</b>	с.	0.	С.	0.	0.	03.0	0. 0.	0.	0.	0.	٥.	-23.283	
FLI-BV-LIGHT	0.	0.	0.	-1.0	0.	0.5	٥.	0.	0.	03.0	0. 0.	0.	0.	0.	٦.	-27.984	
HEADS-UP DISPLAT	đ.	0.	0.	-1.0	0.	-1.0	ο.	0.	0.	2.0 -3.0	01.0	0.	1.0	0.	0.	-28.241	
LORAN C	٥.	0.	0.	-0.5	0.	-0.5	٥.	0.	0.	01.0	00.5	0.	-1.9	0.	9.	-31.554	
OMEGA	0.	0.	0.	-0.5	0.	0.	0.	0.	0.	01.0	01.0	0.	-1.0	0.	0.	-31.577	
FLT-BY-WIRE	0.	e.	0.	-1.0	0.	D.	0.	0.	0.	03.0	0. 0.	0.	0.	0.	9.	-32.734	
ACT GUST ALLEVIATION	0.	0.	0.	-1.0	٥.	0.	0.	0.	0.	03.0	01.0	1.0	<b>0.</b>	0.	0.	-35.043	
ACT FLUTTER SUPPRESSION	0.	0.	0.	-0.5	0.	0.	0.	0.	0.	03.0	01.0	0.	0.	0.	0.	-37.556	
BOPPLER NAVIGATION	0.	0.	0.	-2.0	0.	-0.5	٥.	0.	<b>c.</b>	1.0 -2.0	01.0	0.	0.	0.	0.	-41.723	
ACTIVE LAMINAR FLOW CTL	2.0	0.	0.	-3.0	٥.	0.	٥.	1.0	0.	03.0	1.0 -3.0	0.	0.	0.	1.0	-45.189	
INERTIAL NAVIGATION	0.	0.	0.	-3.0	٥.	-0.5	0.	0.	0.	1.0 -3.0	01.0	٥.	0.	9.	đ.	-58.935	

A PLE INTERNET COMPANY

.

Table B.18. Overall Technology Ranking With Empty Weight = 0.0 For Airplane A

。 1999 电子号子 医子宫 1994 - 1994 - 1987 - 198

AIRPLANE A \*\*\*TABULAR RESULTS FOR ALL TECHNOLOGIES FOLLOW: CASH CAUS EMIS ENTY EXTR FUEL INTR PILT PRCH RELI RIDE STAT TOLD CEIL WINY SPED DOC SION WENT NOIS EFFC NOIS WAKL PRCE RGE DLTY OLTY SFTY CHFT PERF FIG OF CATEGORY WEIGHTS 5.43 6.82 7.92 7.91 1.61 9.50 4.29 8.16 7.04 7.33 8.51 7.29 8.85 6.11 9.42 6.24 6.87 +\*SEE 'NOTES' BELOW: (1) 79.845 1.0 2.0 2.0 1.0 0.5 2.0 1.0 1.0 -2.0 1.0 1.0 1.0 1.0 ۵. 0. STRT CHG ROT COMB ENGINE 1.0 0. 71.997 1.0 -3.0 1.0 1.0 0. 1.0 1.0 1.0 3.0 0. 2.O 1.0 1.0 1.0 SAVE ENGINE 2.0 0. 1.0 2.0 -1.0 ٥. 0\_ 1.0 53.584 2.0 1.0 0. 1.0 0. 0- -0-5 0. NAT LAN FLOW AIRFOILS 1.0 С. 2.0 0. 52.377 0.5 ٥. 1.0 ٥. 2.0 0. 0. 0. 1.0 0. 1.0 ٥. SPOILERS -0.5 0. 2.0 1.0 0. 44.495 2.0 0. 1.0 1.0 0. ٥. 0. 0. 0. 2.0 ο. 1.0 0.5 0. 1.0 0. 0. FIBERGLASS COMPOSITES 44.445 1.0 0. -0.5 ٥. 0. 0. 0. -0.5 2.0 1.0 0. 1.0 0. ٥. 1.0 2.0 FOULER FLAPS 0. 41.628 -0.5 0. 0. 3.0 -0.5 0. -0.5 0. 3.0 ٥. 0\_ 0. 0. ٥. С. 0. INTEG LOW-COST WE LVLEP C. 38.780 0. 0. 0. 2.0 0. 0.5 0. 3.0 ٥. 1.0 0. ٥. 1.0 ٥. 0. 0. KEVLAR COMPOSITES 1.0 36.205 3.0 0. 0. 2 STR ADV DIESEL ENGINE 0. 0. 2.0 0. 0. 0.5 0.5 0. -2.0 1.0 0. ٥. 0. ٥. 33.283 3.0 0. 0. ٥. 0. 9. 0. ٥. 0. 0. 3.0 -2.0 0. 0. ٥. DEGITAL DATA LINKS 0. ٥. 31.767 ٥. 2.0 0. INTEG AVIONICS AND DSPYS 0. ٥. ٥. 1.0 6. 0. 0. 0. 0. 3.0 -2-0 0. 0. 0. 0. 31.110 0. -1.0 1.0 0. ٥. 0. 1.0 IMPROVED TURBOCHARGING 1.0 ٥. 1.0 0.5 0. 0. ٥. 1.0 0. 29.665 -1.0 ٥. 0. 0. GRAPHITE COMPOSITES 2.0 0. 1.0 0.5 0. 3.0 0. 1.0 0. 0. 1.0 0. ٥. 26.926 0.5 0. 0.5 0. 0.5 0. 0. -0.5 0.5 0. ٥. 1.0 0. 1.0 0.5 0. LOW/HEDM SPEED AIRFOILS 0. 0. 25.536 э. 0. 0.5 0. -1.0 0.5 0. 1.0 0. 1.0 LEADING EDGE DEVICES 1.0 0. 0.5 0.5 0. 0. 25.027 SYS STATUS DISPLAYS 0. 0. ο. 0. 0. 0. 0. 0. 2.0 -1.0 J. 0. 0. 2.0 0. 0. D. 24.389 0. 0. -0.5 0. -0.5 0. 0. 0. 3.0 -2.0 0. -0.5 3.0 1.0 0. 0. SEPARATE SEC TECHNOLOGY 0. 26.003 1"PR STLL/SPN--AERO TLRG 0. 0. 0. 0. 0. 0. 0. ٥. 0. -0.5 0. ٥. ٥. 3.0 0. 0. 0. 21.731 0. 0. QUIET EFFICIENT PROPS 0. 0. 0. 0.5 0. 0.5 2.0 0. 2.0 0. +1.0 0.5 0. ٥. 0. 17.326 3.0 0. Ο. ٥. 0. LOAD LIMITING SEATS 0. 0. 0. 0. 0. 0. ٥. 0. 0. -0.5 ٥.

5 . . . .

LIQUID COOLING	0.	0.	0.	2.0	0.	-1.0	1.0	1.0	1.0	0.	0.	0.	-1.0	0.	-1.0	0.	G.	17.017
IMPROVED RESTRAINTS	0.	3.0	0.	0.	0.	0.	0.	0.	0.	0.	-0.5	0.	0.	ũ.	0.	0.	0.	16.191
ENERGY ABSORBING FLOOR	0.	3.0	0.	0.	0.	-0.5	0.	0.	0.	0.	-0.5	0.	0.	0.	0.	0.	0.	16.191
ACTIVE STALL PREVENTION	<b>c.</b>	0.	ο.	0.	0.	-0.5	0.	0.	0.	0.	-1.0	0.	-0.5	0.	3.0	0.	0.	15.324
STRAT CHG RECIP ENGINE	0.	<b>e.</b>	٥.	1.0	2.0	-0.5	0.	1.0	0.	0.	-1.0	0.5	0.	٥.	0.	0.	0.	14.405
BIRECT LIFT CONTROL	0.	0.	0.	0.	0.	-0.5	٥.	0.	G.	1.0	-1.0	0.	-0.5	1.0	1.0	0.	0.5	15.376
BRST/TEAR RESIS FUEL TKS	0.	3.0	ο.	0.	٥.	-1.0	0.	0.	0.	0.	-1.0	0.	G.	٥.	0.	0.	0.	11,933
CAT DISPLAYS	0.	ο.	0.	0.	0.	0.	ο.	0.	0.	2.0	-2.0	0.	1.0	0.	0.5	0.	8.	11.233
HCRLB RECIP ENGINE	0.	٥.	0.	0.5	2.0	0.	٥.	0.5	0.	0.	-0.5	0.5	0.	0.	0.	0.	0.	10.631
ICEPHOBIC SFC COATINGS	-0.5	0.	0.	-0.5	0.	-1.0	٥.	0.	0.	0.	-1.0	-0.5	0.	0.	3.0	0.	0.	9.438
HICROWAVE LANDING SYSTER	<b>6.</b>	0.	0.	0.5	0.	0.	1.0	0.	0.	0.	-1.0	0.	0.	0.	1.0	0.	D.	9.147
NAVSTAR/GPS	0.	0.	0.	0.	0.	0.	٥.	0.	D.	1.0	-1.0	0.	9.	0.	1.0	0.	0.	8.256
JINGLETS	0.5	٥.	٥.	0.5	0.	-2.0	0.	0.5	0.	٥.	-1.0	0.5	0.	0.	٥.	0.	9.	5.876
ACTIVE LAMINAR FLOW CTL	2.0	0.	2.0	-2.0	0.	0.	٥.	2.0	0.	0.	-3.0	3.0	-3.0	0.	0.	0.	1.0	3.841
HICRO HUD	0.	0.	C.	-0.5	0.	-0.5	0.	0.	0.	2.0	-2.0	0.	-1.0	٥.	2.0	0.	0.	3.705
BUCTED PROPULSORS	-0.5	0.	-1.0	-1.0	0.	-1.0	3.0	-1.0	3.0	0.	-1.0	-2.0	0.	0.	0.5	Q.	2.0	2.637
INTEGRATED YAW DAMPER	O.	0.	0.	0.	٥.	0.	٥.	0.	<b>C.</b>	1.0	-2.0	0.	0.	2.0	0.	0.	0.	2.539
AC FLECTRICAL SYSTERS	٥.	С.	0.	0.	0.	0.	0.	0.	0.	0.	0.	0.	0.	0.	<b>C.</b>	0.	0.	0.
SGL LEVER THRUST/ORG CTL	0.	0.	٥.	0.	0.	0.	٥.	0.	0.	2.0	-1-0	0.	-1.0	0.	0.	0.	0.	-2.671
ACT CTLS FOR RLX STBLTY	0.5	0.	0.5	1.0	٥.	1.0	٥.	1.0	0.	0.	-3.0	1.0	-1.0	0.	0.	0.	0.	-4.370
FLUIDIC AUTO FLT CTL SYS	-0.5	0.	0.	0.	0.	-0.5	٥.	-0.5	0.	٥.	-1.0	-0.5	1.0	0.	0.	0.	9.	-10.099
LOW-BRAG SEC COATINGS	0.	٥.	0.	-0.5	0.	-0.5	٥.	0.5	0.	0.	-1.0	ð.	٥.	0.	0.	9.	-0.5	-11.827
DIRECT SIDE FORCE CTL	0.	ο.	0.	-0.5	0.	-0.5	0.	0.	0.	0.5	-3.0	0.	-0.5	3.0	0.	٥.	٥.	-11.920
ACT RIBE SMOOTHING	0.	0.	0.	-0.5	0.	-0.5	0.	0.	0.	0.5	-3.0	0.	-0.5	3.0	0.	G.	0.	-11.920
LOW-LEVEL PRESSUREENTION	٥.	0.	0.	-0.5	0.	-0.5	0.	-0.5	2.0	0.	-1.0	-0.5	-1.0	<b>Q.</b>	ø.	0.	0.	-14.971
LASER GYROS	0.	٥.	0.	-0.5	0.	0.	0.	0.	0.	٥.	-3.9	0.	1.0	0.	0.5	0.	0.	-15.933
HEADS-UP DISPLAY	0.	0.	0.	-1.0	0.	-1.0	٥.	0.	0.	2.0	-3.0	0.	-1.0	0.	1.0	0.	O.	-18.185
LORAN C	0.	0.	0.	-0.5	0.	-0.5	0.	0,	۹.	0.	-1.0	0.	-0.5	0.	-0.5	0.	0.	-21.606
FIBER OPTICS (BATA TRNS)	0.	0.	0.	0.	0.	0.	٥.	0.	C.	0.	-3.0	٥.	٥.	Q.	0.	0.	0.	-25.545
OMEGA	0.	0.	0.	-0.5	٥.	0.	0.	0.	0.	0.	-1.0	0.	-1.0	<b>Q.</b>	-0.5	0.	0.	-26.033

Table B.18. Overall Technology Ranking With Empty Weight = 0.0 For Airplane A (continued)

ة مستحد مست

**.** .

.

۰.

.

i

477

1

14 yr 14

3

The states in the second second states
Table B.18. O	verall I	echnology	Ranking	With	Empty	Weight	<b>\$</b> 1	0.0	For	Airplane	A	(concluded)
---------------	----------	-----------	---------	------	-------	--------	-------------	-----	-----	----------	---	-------------

0. -34.347 0. DOPPLER NAVIGATION 0. 0. 0. -2.0 0. -0.5 0. **c.** 0. 1.0 -2.0 0. -1.0 0. 0. ٥. -38.352 0. -3.0 0. ٥. 0. 0. -0.5 0. ο. 0. 0. -1.0 0. ACT FLUTTER SUPPRESSION 0. 0. ĉ., -42.305 0. -1.0 0. -1.0 0. 0. 0. 0. -3.0 0. -1.0 0. 0. ٥. 0. 0. 0. FLY-BY-LIGHT -42.305 с. 0. -3.0 0. -1.0 0. 0. 0. 0. 0. -1.0 0. 0. 0. 0. ACT GUST ALLV 0. 0. 0. 0. -42.305 0. -1.0 0. -1.0 0. 0. 0. 0. -3.0 0. -1.0 0. 0. FLY-BY-WIRE 0. ο. 0. 1.0 -3.0 0. -1.0 0. 0. ٥. 0. -50.748 0. -3.0 0. -0.5 0. 0. INERTIAL NAVIGATION 0. 0. • • NO 7 E S : ( 1) EMPTY WEIGHT IS OMITTED.

And the second state of th

Inter Contraction in the second statement of all the mention of the second statement of the second sta

. \*

. .

s a mer tri un all millionen av stantäännäären ja sääsitti anaankonninka on a Ponsahilliks

;} 2<sup>4</sup>111

3

.

.

Table B.19. Overall Technology Ranking With Empty Weight = 0.0 For Airplane B

1. . . . . . . . .

. . . .

ATEPLANE B +++TABULAR RESULTS FOR ALL TECHNOLOGIES FOLLOW: CPSH CRUS EMIS ENTY EXTR FUEL INTR PILT PACH RELI RIDE STAT TOLD CEIL WITHY SPED DGC SION WENT NOIS EFFC NOIS WAKL PACE RGE OLTY GLTY SFTY CAFT PERF FIG OF CATEGORY WEIGHTS 5.45 7.33 7.48 9.45 2.16 9.50 5.41 8.61 7.45 7.00 7.76 7.18 9.55 7.24 9.54 6.71 7.4/ \*\*SEE 'NOTES' BELOW: (7) 46.648 1.0 0. 0. 1.0 Ο. ۵. ٥. 0. FIBERGLASS COMPOSITES 2.0 0. 1.0 0.5 0. 2.0 0. 1.0 0. ٥. 41.772 ٥. 1.0 0. 0. ٥. 0. 3.0 -1.0 0. 0. ۵. 2.0 0. 0. INTEG AVIONICS AND DSPYS 0. ٥. 38.887 0.5 0. 3.0 0. 1.0 ٥. 0. 0. 1.0 0. ٥. 0. ô. 0. KEVLAR COMPOSITES 2.0 0. 1.0 35-542 0. 0.5 0. 0.5 0. 0.5 0. 0.5 0. 1.0 ٥. ٥. ۸. 0. 0. 1.0 1.0 SPOILERS 34-104 **o**. 0. 0. 0. 0. ٥. 3.0 -2-0 ٥. 0\_ ٥. 3.0 0. D. DIGITAL DATA LINKS C. 0. ٥. 0. ٥. ٥. 0. 0. 32.419 0. 1.0 0. 0. 1.0 0. 2.0 0. ٥. GATE ENGINE 0. 0. 0. 31-126 0. 0. 3.0 0. 1.0 0. 0. -1-0 1.0 ٥. ٥. 0. 0. ٥. GRAPHITE COMPOSITES 2.0 1.0 0.5 28.952 1-0 0. 0.5 0.5 0. ۵. ٥. 0.5 0. 0. -0-5 0.5 0. 0.5 0. 0. 1.0 LEADING EDGE DEVICES 0. 0.5 0. 0.5 0. 0. -0.5 0.5 ٥. 0. 0.5 0. 1-0 28.398 1.0 0.5 LOW/MEDH SPEED AIRFOILS 0. 0. 0. 1.0 0. 0.5 0. 0. 1.0 0. 0.5 0. 0.5 0. 0. 27.057 ICEPHOBIC SEC COATINGS 1.0 0. 0. ٥. -1.0 0. 0. 0. 1.0 1.0 0. 0. 26.550 STRT CHG ROT COMB ENGINE 0. ٥. 0. 1.0 1.0 0. 0\_ ٥. 0. 0. 0.5 0. 0.5 2.0 0. 2.0 0. -1-0 0.5 0. ٥. 0. 0. 0. 26.285 QUIET EFFICIENT PROPS 0. 0. SYS STATUS DISPLAYS 0. 0. ٥. 0. 0. 0. 0. 0. 2.0 -1.0 0. 0. 0. 2.0 0. 0. 25-323 0. 0. -1.0 -0.5 0. -1.0 3.0 -0.5 3.0 0. -1.0 -1.0 0. D. 0.5 0. 2.0 24-007 DUCTED PROPULSORS -0.5 ٥. 0. 0. ٥. 0. 0. 1.0 0. 0. 0. 1.0 1.0 0. 0. 23.788 SEPARATE SFC TECHNOLOGY 0. 0. 0. FOULER FLAPS 0. 0. 0.5 0.5 0. 0. 0. 0.5 0. ٥. ٥. 0.5 0. 1.0 ۵. 0. 0. 23.597 0. 0. 0. 0. ٥. 0. 0. 2.0 -1.0 0. 1.0 0. 0.5 0. 0. 20.554 CRT DISPLAYS 0. 0. MICROWAVE LANDING SYSTEM ·0. **C**. 0. 1.0 ٥. 0. 1.0 0. 0. 0. -0.5 0. ٥. 0. 1.0 0. 0. 20.527 NAT LAR FLOW AIRFOILS 0.5 0. 1.0 1.0 0. 1.0 0. 1.0 0. 0. -0.5 1.0 -2.0 0. 0. 0. 1.0 19.874 LOAD LIMITING SEATS 0. 0. 0. 0. ٥. -0.5 0. 18.646 ٥. 3.0 a. 0. ٥. 0. 0\_ 0. ٥. 0.

Table B.19. Overall Technology Ranking With Empty Weight = 0.0 For Airplane B (continued)

۷.,

and the second of the stand of the second seco

45.1

15.118 0. 0. ENERGY ABSORBING FLOOR 3.0 ٥. -0.5 ٥. ٥. ٥. ٥. -0.5 ٥. ٥. ٥. ٥. 0. 0. ٥. 18.118 0. 0. 0. IMPROVED RESTRAINTS 0. 3.0 0. 0. 0. ٥. ٥. 0. ٥. ٥. -0.5 Ω. ٥. ٥. 16.097 ACTIVE STALL PREVENTION ٥. -1.0 0. -0.5 ٥. 3.0 0. 0. 0. 0. 0. 0. 0. 0. 0. 0. 0. 15.207 -0.5 ٥. 2.0 0. 0. IMPR STLL/SPN--AERO TLRG 0. 0. ٥. 0. 0. 0. ٥. 0. ٥. ٥. ٥. ٥. 16.949 0. 0.5 DIRECT LIFT CONTROL 0. 0. 0. 0. ٥. -0.5 ٥. ٥. ٥. 1.0 -1.0 0. -0.5 1.0 1.0 14.238 0. đ. -1.0 ٥. 0. -1.0 ٥. ٥. ٥. 0. BRST/TEAR RESTS FUEL TKS 0. 3.0 0. 0. Ο. 0. ٥. 14.075 ٥. 0. AC ELECTRICAL SYSTEMS 0.5 0. ٥. 0. 0. 0.5 Ο. 0. 0. 0. 1.0 0.5 0. ٥. 0. 8.781 ٥. ٥. 1.0 0. 0. NAVSTAR/GPS 0. 0. 0. 0. 0. 0. 0. 1.0 -1.0 0. ٥. 0. 7.761 ٥. 0. 0. 0. Ũ. 0. INTEG LOW-COST WG LVLER 0. 0. G. 0. ο. 0. 0. 1.0 ٥. 0. ٥. 7.582 ٥. 0. -2.0 ٥. 0.5 0. -1-0 0.5 ۵. ٥. 0. WINGLETS 0.5 0. 0. 0.5 0. ρ. 5.952 0. 0. 0. 0. 0. INTEGRATED YAW DAMPER 0. 0. ٥. 0. ٥. Ο. 0. 1.0 -2.0 0. 0. 2.0 3.691 0. ٥. -2.0 0. -0.5 3.0 ٥. 0. 0. ACT RIDE SMOOTHING Ο. -0.5 0. ٥. 0. 1.0 ٥. 0. 3.289 ο. 0. 2.0 -2.0 -1.0 ٥. 2.0 0. 0. 0. ٥. -0.5 Ο. 0. 0. 0. MICRO HUD 0. -1.070 0. ٥. -0.5 ٥. -0.5 ٥. 0. 0. 0. FLUIDIC AUTO FLT CTL SYS -0.5 0. 0. ٥. ٥. 1.0 0. 0. -1.128 1.0 0. 1.0 -3.0 -1.0 0. 0. 0. 0.5 0. 0.5 0. 1.0 ٥. ٥. 1.0 ٥. ACT CTLS FOR RLX STBLTY -3.312 0. ٥. 0. ٥. 0. ٥. 2.0 -1.0 0. 0. 0. C. 0. SGL LEVER THRUST/DRG CTL ٥. 0. е. -1-0 -11.048 0. 0. ٥. ٥. -3.0 -0.5 3.0 0. 0. 0. DIRECT SIDE FORCE CTL 0. 0. -0.5 ٥. ٥. ٥. ٥. -12.309 LOW-DRAG SEC COATINGS 0. 0. -0.5 0. -0.5 0. 0. 0. 0. -0.5 0. 0. 0. ٥. 0. -0.5 ٥. -13.689 LASER GYROS Ο. Ο. -0.5 0. 0. 0. 0. ٥. -3.0 0. 0. 0.5 0. 0. ٥. ٥. 1.0 -15.023 -0.5 -1.0 LOW-LEVEL PRESSURIZATION 0. 0. 0. -0.5 0. 0. -0.5 5.0 ٥. -0.5 -1.0 0. 0. 0. 0. -18,741 -1.0 HEADS-UP DISPLAY 0. 0. 0. -1.0 0. ٥. 0. 0. 2.0 -3.0 0. -1.0 0. 1.0 0. 0. -23.283 FIBER OPTICS (DATA TRNS) 0. 0. 0. ٥. 0. 0. Ο. 0. 0. 0. -3.0 0. 0. ٥. 0. 0. 0. -0.5 -26.804 0. LORAN C 0. ٥. 0. -0.5 0. 0. 0. 0. -1.0 0. -0.5 0. -1\_0 0. 0. -31.577 OMEGA 0. 0. 0. -0.5 0. 0. 0. 0. 0. 0. -1-0 0. -1.0 0. -1.0 0. 0. -32.734 0.5 0. 0. 0. 0. FLY-BY-LIGHT 0. 0. 0. -1.0 0. 0. -3.0 ū. 0. ٥. 0. 0. -32.734 0. 0. ٥. 0. 0. 0. 0. FLY-BY-WIRE 0. 0. -1.0 ο. 0. -3.0 ٥. 0. 0. 0. -35-043 ACT GUST ALLEVIATION 0. -1.0 0. 0. 0. 0. 0. 0. -3.0 0. -1.0 1.0 0. 0. ٥. 0. 0.

pi,

Ł

"e" , i al matisaera

. . .

Table B.19. Overall Technology Ranking With Empty Weight = 0.0 For Airplane B (concluded)

-34.973 0. 1.0 -2.0 0. -1.0 ٥. ٥. ٥. BOPPLER NAVIGATION с. 0. 0. -2.0 0. -0.5 0. 0. ٥. -37.556 0. 0. 0. -0.5 0. 0. 0. 0. 0. 0. -3.0 0. -1.0 0. 0. 0. 9. ACT FLUTTER SUPPRESSION ACTIVE LAMINAR FLOW CTL 2.0 0. 0. -3.0 0. 0. 0. 1.0 0. 1.0 -46.189 0. -3.0 1.0 -3.0 0. 0. 0. INERTIAL NAVIGATION D. 0. 0. -3.0 D. -0.5 D. 0. 0. 1.0 -3.0 D. -1.0 D. 0. 0. 0. 0. -54.185 ...NOTES: ( 1) EMPTY WEIGHT IS OMITTED.

í

۰.

a.

481

.

16<sup>1</sup>

· · · · · · · ·

.

### APPENDIX C

### 6-PASSENGER AND COMMUTER DESIGN STUDIES

This appendix details basic equations used in the configuration trades for both the 6-passenger and commuter aircraft. Weight and drag breakdowns are shown for both baseline and advanced conventional configurations for both airplanes. Also, the Range Factor derivation is shown.

#### C.1 6-PASSENGER DESIGNS

Three airplanes were designed: a current technology airplane, an advanced technology conventionally configured airplane, and an advanced technology canard configured cirplane. Specifications and technologies incorporated are detailed in Chapter 5 as is the cabin layout which is common to all three designs.

The following paragraphs discuss, in order, the parametric trade studies, the current technology airplane, the conventional configuration advanced technology airplane, and the canard configuration advanced technology airplane.

#### C.1.1 Trade Studies

e

Three trade studies were done covering cruise, takeoff, and landing performance. Derivations are as follows (the resulting graphs may be found in Chapter 5).

Cruise trades were formulated to show effective power loading required to cruise at 250 kt. The relationship is given in Equation C.1.

$$(n_{p} P/W) = \frac{\rho V^{3}}{4882} \left\{ \frac{23.95 C_{D_{o}}}{(W/s)} + \frac{(W/s)}{\pi (Ae) \rho^{2} V^{4}} \right\}$$
for  $\rho$  in kg sec<sup>2</sup>/m<sup>4</sup> W/S in N/m<sup>2</sup>  
V in m/s  $(n_{p} P/W)$  in kw/N

Note: (n P/W) is that required at cruise and is not the sea level maximum.

÷.

L/D can also be included in the plots because it is related to the effective power loading as given by Equation C.2.

$$L/D = (n_p P/W)^{-1} (V/)$$
 (C.2)

for V in m/s (n<sub>p</sub> P/W) in kw/N

The takeoff trade study was formulated to show wing loading required for a 2000 ft takeoff over a 50 ft obstacle. The approximations of Reference 214 were used to model the takeoff maneuver. Equation C.3 presents the resulting expression.

$$\frac{W}{S} = \frac{15.7 \sqrt{A}}{16.505 (\frac{P}{W}) \sqrt{A} - .3}$$

$$\frac{.1198}{18.332 (P/W)C_{L} - .02 C_{L} - .72 C_{D} + \frac{.1694}{C_{L}}}$$
(C.3)

Computations showed that the required W/S was a very weak function of  $C_{D_o}$  and A so the plots were generated with W/S = f(P/W,  $C_{L_o}$ ) only. max<sub>TO</sub>

The landing trade study was formulated to show W/S and C<sub>L</sub> required to land in 610 over a 15.2 m obstacle. The max<sub>LDG</sub> approximations of Reference 214 were used to model the landing maneuver. The relation proved to be linear as shown in Equation C.4.

$$C_{L} = .0011 (W/S) (C.4)$$

### C.1.2 Current Technology Design - 6PAXBL

Preliminary sizing of the 6PAXBL was accomplished using a standard weight breakdown and the Breguet range equation as given by Equations C.5 and C.6.

 $W_G = W_E + W_p + W_f$  (C.5) where  $W_G = \text{gross weight}$   $W_p = \text{payload weight}$  $W_E = \text{empty weight}$   $W_f = \text{fuel weight}$ 

$$R = 198.13 (\eta_p / sfc) (L/D) \ln (W_G/(W_G - W_f))$$
(C.6)

for R in nm sfc in kg/kw/hr

Use of Equations C.5 and C.6 resulted in the following preliminary weight breakdowns:

W <sub>E</sub> =	1086 kg	W = 545 kg
W <sub>f</sub> =	213 kg	W <sub>G</sub> = 1844 kg

Next, the wing was sized for  $W/S = 1053 \text{ N/m}^2$  and the vertical and horizontal tails were sized with volume coefficients. With initial sizing completed, detailed component weight estimations were done primarily using the methods of References 214 and 154. Wing location and horizontal tail area were then determined from static margin and takeoff rotation requirements using a program which is briefly described tollowing the design discussions in Section C.1.5. The final step in the manual conceptual design process was the calculation of the zero lift drag for which the methods of References 179, 214, 90, and 137 were used.

The final sizing began by calibrating the GASP weight and drag routines against the previous hand calculations by varying already-provided inputs. GASP then sized the engine and airframe to meet the required specifications. Table C.1 presents the final weight breakdown and Table C.2 presents the final drag breakdown for 6PAXBL.

Table C.1. 61	AXBL We	lght -	Bree	kdowa
---------------	---------	--------	------	-------

Group	Iten	Weight (kg)/(lb)
Propulsion	Engine (dry) Engine Installation Fuel System Propeller	281.7 / 621 31.8 / 70 10.4 / 23 33.6 / 74
Structures & Miscellaneous	Wing Horizontal Teil Vertical Tail Fuselage Landing Gear Fuel Tanks Battery & Wiring	177.4 / 391 27.7 / 61 11.3 / 25 219.1 / 483 62.1 / 137 18.1 / 40 22.7 / 50
Flight Controls & Fixed Equipment	Surface Controls Avionics & Instrumentation Seats & Furnishings	50.8 / 112 33.1 / 73 71.7 / 158
Empty Weight = 1 Payload Weight = Fuel weight = Gross Weight = 1		051.4 kg (2318 1b) 544.3 kg (1200 1b) 264.0 kg (582 1b) 859.8 kg (4100 1b)

Iten	Flatplate Area (m <sup>2</sup> )	ູດີ	Wetted Area (m <sup>2</sup> )	
Wing	.1315	.00759	28.938	
Fuselage	.0854	.00493	29.381	
Vert. Tail	.0123	.00071	3.185	
Horiz. Tail	.0277	.00160	6.202	
Increment	.0710	.00410	0.	
Totels	.3279	.01 <b>89</b> 4	67.706	
* Accounts for roughness, protuberances, and cooling drag.				
Breakdown is for cruise, gear and flaps up.				
Altitude = 7620	m, Speed =	250 kt,	Mach = .415	
Reynolds number per meter = 4.59 x 10 <sup>6</sup> .				
Cruise Drag Polar : $C_D = .0189 + .0511 C_L^2$				

### Table C.2. 6PAXBL Drag Breakdown

7

C.1.3 Advanced Technology Conventional Design - 6PAXAD

Preliminary sizing is analogous to that of 6PAXBL (see Eq. C.5 and C.6). The methods of Reference 154 were used to account for composite structures as outlined below:

- For preliminary sizing, assume that composites result in a 16% empty weight reduction.
- (2) For component weight estimation, fuselage, wing, and empennage weights are reduced by 25% through use of composite materials.

- ---- --

Applying (1) above and Equations C.5 and C.6 resulted in the following preliminary weight breakdown:

We	= 648	kg	Wp	-	545 kg	
W <sub>f</sub>	= 101	kg	W <sub>G</sub>	-	1294 kg	

GASP was used for final sizing as discussed previously for 6PAXBL. Tables G.3 and C.4 present the final weight and drag breakdowns for 6PAXAD.

C.1.4 Advanced Technology Canard Design - 6PAXC

Due to problems discussed in Chapter 5, the design of 6PAXC did not proceed beyond the preliminary sizing stage. Based on limited analysis, the following conclusions were made:

- (1) Wetted area appears to be less than that of 6PAXAD. Hence,  $C_D$  may be expected to be slightly smaller.
- (2) Structural weights are expected to be similar or slightly less than those of 6PAXAD.
- (3) Feasibility (from a trim standpoint) of full span flaps on the main wing is unknown.
- (4) In view of items 1 and 2, and assuming full span flaps are feasible, the performance and efficiency characteristics of 6PAXC are expected to be similar to or better than those of 6PAXAD.

### C.1.5 Tail Sizing Computer Program

A short program was written to aid in horizontal tail sizing. The program requires aerodynamic, geometric, and weight inputs to compute horizontal tail area and weight (for conventional or

Table C.3. 6PA	AD Weig	ht Brea	kdown
----------------	---------	---------	-------

Group	Item	Weight (kg) / (lb)
Propulsion	Engine (dry) Engine Installation Fuel System Propeller	92.1 / 203 25.9 / 57 7.7 / 17 26.3 / 58
Structures & Miscellaneous	Wing Horizontal Tail Vertical Tail Fuselage Landing Gear Battery & Wiring	62.1 / 137 12.7 / 28 6.8 / 15 127.0 / 280 43.5 / 96 22.7 / 50
Flight Controls & Fixed Equipment	Surface Controls Avionics & Instrumentation Seats & Furnishings	50.8 / 112 33.1 / 73 71.7 / 158
Empty Weight = Payload Weight = Fuel Weight = Gross Weight = 1		532.0 kg (1283 lb) 544.3 kg (1200 lb) 132.5 kg (292 lb) 258.7 kg (2775 lb)

Item	Flatplate Area (m <sup>2</sup> )	°0°	Wetted Area (m <sup>2</sup> )		
Wing	,0494	.00862	8.883		
Fuselage	.0801	.01397	25.842		
Vert. Tail	.0099	.00173	3.347		
Horiz. Tail	.0156	.00273	3,223		
* Increment	.0086	.00150	0.		
Totals	.1636	.02856	40.367		
* Accounts for a assumed neglig	* Accounts for roughness and protuberances. Cooling Drag is assumed negligible for liquid cooled engine.				
Breakdown is for cruise, gear and flaps up.					
Altitude = 9140 m Speed = 250 kt, Mach = .424					
Reynolds number per meter = 3.94 x 10 <sup>6</sup>					
Cruise Drag Polar: $C_{\rm D}$ = .0286 + .0408 $C_{\rm T}^2$					

ł

### Table C.4. 6PAXAD Drag Breakdown

**v**3

composite materials) to meet static margin and/or rotation requirements.

Static margin and rotation requirements are based on the methods of References 174 and 180.

In running the program, the operator has the following options:

Fix main gear location or locate main gear 15<sup>0</sup> (from the vertical) aft of the computed center of gravity.

- (2) No tail sising; compute input characteristics.
- (3) Size tail for static margin requirement only.
- (4) Size tail for rotation requirement only.
- (5) Size tall for the critical requirement.

Generally, a static margin of at least .10  $(dC_L \le -.10)$ and rotation at or below 120% of the stall speed  $(V_R \le 1.2 V_s)$ were required.

#### C.2 COMMUTER DESIGN

Two aircraft were designed: an advanced technology conventional configuration and an advanced technology canard configuration. Specifications, technologies incorporated into the aircraft, and cabin layout (common to both aircraft) are detailed in Chapter 5. In addition, a baseline, current technology aircraft was analyzed for comparison with the advanced technology aircrait.

The following paragraphs discuss the parametric trade studies and the three commuter aircraft that were synthesized and/or analyzed.

#### C.2.1 Trade Studies

Two trade studies were performed covering cruise and balanced field length performance. The cruise trade studies are not covered in this section and the reader is referred to Section C.1 for their derivation. The Balanced Field Length (BFL) trades were developed to illustrate the effects of effective power loading ( $n_p$  P/W), maximum lift in takeoff configuration (C<sub>L</sub>), zero lift drag max<sub>TO</sub>

coefficient (C<sub>D</sub>), effective aspect ratio (Ae), and wing loading (W/S) on BFL. The relationship for these factors is presented in Equation C.7 and was derived from the approximations of Reference 214.



$$x \left\{ 2.7 + \left( \frac{1}{\frac{n_{p}^{P}}{W} - \frac{459.9}{(\frac{W}{S} \times \frac{1}{C_{L_{max_{TO}}}})^{\frac{1}{2}}} - (.02 + .72 - \frac{C_{D_{O}}}{C_{L_{max_{TO}}}})^{\frac{1}{2}} \right\}$$

sea level standard conditions (ISA)
concrete runway (µ = .02)
W/S in N/m<sup>2</sup>.
n<sub>p</sub> P/W in kw/kg.
BFL in m.

Calculations indicated that variations in Ae and  $C_{D_o}$  had little effect on BFL. The final plots were constructed with Ae and  $C_{D_o}$  held constant.

#### C.2.2 Current Technology Configuration - COMBL

The drag and weight routines of GASP were calibrated to known or predicted characteristics of COMBL, an existing commuter aircraft. This was done by iterating over a set of input values until the desired outcome was obtained. The existing propulsion characteristics of the aircraft were also input into the program. Table C.5 presents the final weight breakdown for COMBL, while Table C.6 presents the final drag breakdown for the aircraft.

#### C.2.3 Advanced Technology Conventional Configuration - ADCOM

Development of ADCOM proceeded as outlined for 6PAXAD in Section C.1.3, with the exception that only engine sizing was allowed in the GASP analysis. The final weight breakdown and the final drag breakdown appear in Tables C.7 and C.8 respectively.

#### C.2.4 Advanced Technology Canard Configuration - ADCOMCN

Because of the canard analysis problems discussed in Chapter 5, developmental work on ADCOMCN was halted after preliminary sizing. The following conclusions can be made based on limited

for

Group	Iten	Weight (kg) / (1b)
Propulsion	Engine (dry)	325 / 716
	Engine Installation	116 / 256
	Fuel System	24 / 53
	Propellers	137 / 303
Structures &	Wing	620 /1367
MISCELLANCOUS	Horizontal Tail	89 / 196
	Vertical Tail	70 / 154
	Fuselage	685 /1443
	Landing Gear	283 / 623
	Engine Section	186 / 411
Flight Controls	Cockpit Controls	14 / 31
Equipment	Fixed Wing Controls	65 / 144
	Fixed Equipment	796 /1753
Empty	vWeight = 3	3363 kg (7450 1b)
Operational Items (includes crew)		311 kg (686 lb)
Max Payload Weight		1994 kg (4395 1b)
Max Fuel Weight		1970 kg (4342 1b)
Gross	weight (Design) = 5	5670 kg (12500 lb)
Gross mi	Weight (multi- lssion) = (	5350 kg (14000 lb)

# Table C.5. COMBL Weight Breakdown

•1

1

4

2

ŧ

.

**1** 

Ì

Iten	Flatplate Area (m <sup>2</sup> )	с <sub>р</sub>	Wetted Area (m <sup>2</sup> )		
Wing	.1971	.00764	43.114		
Fuselage	.2539	.00985	80.530		
Vert. Tail	.0325	.00126	10.025		
Horiz. Tail	.0465	.00181	12.843		
Engine Nacelles	.0768	.00298	18.729		
Increment*	.0410	.00159	0.		
Totals	.6478	.02513	165.241		
* Accounts for roughness and protuberances.					
Breakdown is for	Breakdown is for cruise, gear and flaps up.				
Altitude = 3050 m, Speed = 250 kt, Mach = .4					
Reynolds number per meter = $7.034 \times 10^6$					
Cruise Drag Polar: $C_D = .0251 + .0511 C_L^2$					

### Table C.6. COMBL Drag Breakdown

preliminary analysis of the configuration:

- Wetted area is approximately the same as ADCOM, and therefore the drag characteristics are anticipated to be approximately the same.
- (2) Both configurations are characterized by essentially identical weights. Wing sweep and landing gear location cause the basic structural weight differences.

Group	Item	Weight (kg) / (lb)
Propulsion	Engines (dry) Engine Installations <sup>1</sup> Fuel System Propellers	278 / 613 63 / 139 143 / 315 189 / 417
Structures & Miscellaneous	Wing Horizontal Tail Vertical Tail Fuselage Landing Gear Systems <sup>2</sup>	284 / 627 36 / 80 16 / 35 854 / 1883 308 / 679 210 / 463
Flight Controls & Fixed Equipment	Flight Controls Cockpit Accomodations Fixed Equipment <sup>3</sup>	135 / 298 136 / 300 405 / 893
Empty Opera (: Max 1 Max 1 Gross	y Weight = 3058 ational Items includes crew) = 273 Payload Weight = 1723 Fuel Weight = 908 s Weight = 5706	kg (6742 1b) kg (601 1b) kg (3800 1b) kg (2002 1b) kg (12580 1b)

## Table C.7. ADCOM Weight Breakdown

1. includes engine mount and accessories.

٠.

2. includes hydraulic, electrical, and airconditioning systems.

3. includes cabin furnishings and emergency equipment.

Iten	Flatplate Area (m <sup>2</sup> )	с <sub>л</sub>	Wetted Area (m <sup>2</sup> )
Wing	,1525	.00887	28.726
Fuselage	.2075	.01207	91.856
Vert. Tail	.0314	.00183	9.415
Horiz. Tail	.0172	.00154	6.488
Increment*	.0431	.00251	0.
Total	.4610	.02682	136.485
* Accounts for roughness and protuberances.			
Breakdown is for cruise, gear and flaps up.			
Altitude = 3050 m, Speed = 250 kt, Mach = .4			
Reynolds number per meter = 7.034 x 10 <sup>6</sup> .			
Cruise Drag Polar: $C_D = .0268 + .0376 C_L^2$			

### Table C.8. ADCOM Drag Breakdown

- - - ----

٩

(3) Trim problems with full span flap deflection may exist. This is an area that merits more research.

### C.3 RANGE FACTOR PARAMETRIC ANALYSIS

The Range Factor (RF) gives a useful measure of an airplane's <u>cruise</u> efficiency and utility in terms of kilogram (payload) kilometers per liter.

Equations C.5 and C.6 from Section C.1 can be combined to give an expression for fuel weight required as given by Equation C.8.

$$W_{f} = W_{p} \left\{ \frac{e^{x} - 1}{1 - \left(\frac{W_{E}}{W_{C}}\right) e^{x}} \right\}$$
(C.8)  
where  $x = \left\{ \frac{R \left(\frac{sfc}{\eta_{p}}\right)}{198.10 \left(\frac{L}{D}\right)} \right\}$ 

1

. .

. . . ·

Next, an expression for L/D is obtained as given by Equation C.9:

$$\frac{L}{D} = \frac{.0085 \pi \rho V^2 (Ae) (\frac{W}{S})}{.0418 \pi \rho^2 V^4 (Ae) C_D + .0017 (\frac{W}{S})^2}$$
(C.9)

for 
$$ho$$
 in kg sec<sup>2</sup>/m<sup>4</sup>  
V in m/s  
W/S in N/m<sup>2</sup>

RF is defined by Equation C.10 and is expressed in terms of airplane and mission parameters by combining Equations C.8 and C.9 as given by Equation C.11.

$$RF = W_{p} R \left(\frac{.705}{W_{f}}\right)$$
(C.10)

L

for RF in kg-nm/liter

R in ma  
W<sub>p</sub> in kg  
W<sub>f</sub> in kg  
RF = .705 R 
$$\left\{ \frac{1 - \left(\frac{W_E}{W_G}\right) e^X}{e^X - 1} \right\}$$
 (C.11)

where 
$$x = \frac{\frac{sfc}{\eta_p} .0418 \left\{ \pi \rho^2 V^4 (Ae) C_{D_0} + .0017 \left(\frac{W}{S}\right)^2 \right\}^{\frac{1}{2}}}{1.6934 \pi \rho V^2 (Ae) \left(\frac{W}{S}\right)}$$

for RF in kg-nm/liter
 V in m/s
 R in nm
 W/S in N/m<sup>2</sup>
 ρ in kg sec<sup>2</sup>/m<sup>4</sup>
 sfc in kg/kw/hr

Note: (1) valid for cruise flight only

(2) R is usually expressed as a fixed range (R<sub>f</sub>) plus 45 minutes reserve:  $R = R_f + .75 V$